Michael W. Stewart

Diabetic Retinopathy

Current Pharmacologic Treatment and Emerging Strategies



Diabetic Retinopathy

Michael W. Stewart, MD

Diabetic Retinopathy

Current Pharmacologic Treatment and Emerging Strategies



Michael W. Stewart, MD Mayo Clinic Jacksonville Florida USA

ISBN 978-981-10-3508-1 DOI 10.1007/978-981-10-3509-8

ISBN 978-981-10-3509-8 (eBook)

Library of Congress Control Number: 2017933287

© Mayo Foundation for Medical Education and Research 2017

This work is subject to copyright. All rights are reserved by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

The publisher, the authors and the editors are safe to assume that the advice and information in this book are believed to be true and accurate at the date of publication. Neither the publisher nor the authors or the editors give a warranty, express or implied, with respect to the material contained herein or for any errors or omissions that may have been made. The publisher remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Printed on acid-free paper

This Adis imprint is published by Springer Nature

The registered company is Springer Nature Singapore Pte Ltd.

The registered company address is: 152 Beach Road, #21-01/04 Gateway East, Singapore 189721, Singapore

Introduction

Diabetes mellitus (DM) currently affects more than 8% of the US population and ranks seventh as a cause of death. Approximately 90% of affected patients have type 2 DM and because of the growing obesity epidemic, which predisposes patients to the development of type 2 DM, the number of affected patients is increasing rapidly. Despite the large number of patients in industrialized nations, the majority of diabetics live in developing countries with the greatest number found in India. The abandonment of traditional foods in favor of westernized diets that are rich in processed carbohydrates and saturated fats has increased the incidence of metabolic syndrome in many of these countries. The number of patients with DM challenges healthcare systems throughout the world with both industrialized and developing countries struggling to manage the global diabetes epidemic.

Diabetic retinopathy (DR) is the most common neurodegenerative and microvascular complication affecting patients with DM. Comorbidities such as systemic arterial hypertension and hyperlipidemia predispose to the development of DR, but the most important risk factors are average serum glucose levels and duration of disease. Internists and endocrinologists have an increasing number of drugs at their disposal to treat DM, but several factors including national and regional drug availability, drug costs, healthcare insurance coverage, access to medical care, patient education, and patient compliance (among others) limit the effectiveness of treatment and influence the development of retinopathy. Chapter 1 of this book discusses the incidence and prevalence of DM and DR from both global and regional perspectives.

The pathophysiology of DR is extraordinarily complex and remains incompletely understood. Because 75% of diabetes-related vision loss results from diabetic macular edema (DME) and most of the remaining cases stem from complications of proliferative diabetic retinopathy, research and drug development has focused on the retinal vascular endothelial cell. Several chemokines and inflammatory cytokines, including vascular endothelial growth factor (VEGF), have been associated with the development of DR. Unfortunately, the pertinent biochemical pathways are rarely linear, with numerous crosstalk associations and feedback loops that make it difficult to fully understand the differences between causative molecules and disease markers. Chapter 2 delves into the complex biochemical mix found in eyes with DR.

Only during the past 3 decades have proven treatments for DR (laser photocoagulation and vitrectomy surgery) emerged. Chapter 3 discusses these alternative treatments and highlights their effectiveness in the treatment of DR.

The relatively recent (1989) discovery of VEGF has fueled most of the recent advances in the treatment of DR. VEGF inhibitory drugs have been available to ophthalmologists since 2004, but their use in patients with DME has advanced more slowly than in patients with neovascular age-related macular degeneration. Nevertheless, anti-VEGF treatment has become the preferred treatment for DME, and their use in PDR is increasing. Chapter 4 describes the available anti-VEGF agents and details the important clinical studies that have shaped our current approaches to the treatment of DR.

Inflammation plays a key role in the development of DR, and its discovery quickly led to the use of intravitreal triamcinolone acetonide injections, and, more recently, dexamethasone and fluocinolone sustained release inserts. Chapter 5 discusses both the off-label use of corticosteroids and the pivotal phase III trials that have led to regulatory approval of sustained release inserts.

With data from prospective trials that have evaluated several drugs, treatment choices for DME and DR abound. Chapter 6 constitutes the author's attempt to present evidence-based treatment guidelines after a synthesis of the literature. Several important questions regarding treatment choices remain unanswered, and these will be addressed in accordance with the best available data.

Multiple lines of evidence implicate the vitreoretinal interface in the development of DR. Several drugs and preparations that induce posterior vitreous detachment are being evaluated in patients with DR. Chapter 7 discusses the off-label use of both approved and investigational drugs that promote vitreous detachments.

Despite our available treatments, many patients still respond poorly to current therapies. Development of new drugs, novel formulations of existing medications, and extended duration drug-release devices are proceeding on several fronts. Chapter 8 discusses off-label use of approved medications as well as the current status of investigational products.

Drug safety is an important outcome during the development and approval of all pharmaceuticals, and it becomes particularly important in diabetic patients, many of whom have advanced vascular disease. The pivotal trials found acceptable safety profiles for each of the approved drugs, but important ocular and systemic risks must be considered when treating patients. Chapter 9 discusses the risks associated with pharmacotherapy.

Pharmacotherapy for DME comes at a significant cost to patients, medical care insurance companies, and national healthcare programs. The cost-effectiveness of available treatment strategies has been calculated in several economic analyses. Chapter 10 discusses some of the economic impacts of pharmacotherapy.

The diabetic retinopathy body of literature has become vast, and published reports vary from case discussions through multicenter, randomized, phase III trials. With tens of thousands of manuscripts published in hundreds of journals, this book

is not and cannot be a complete review of the literature. Contradictions in the literature are frequent and recommendations for nearly every contingency have been advanced. The best data come from carefully planned, multicenter, randomized, prospective, double-blind trials. This book attempts to discuss these studies in the greatest detail as they provide the most reliable information. Other studies that lack the same rigorous methodology have been included, but these are fewer in number, and their conclusions must be viewed with greater skepticism. Despite the focus on the pivotal trials, some important studies have undoubtedly been omitted.

This book comes after years of writing about drug pharmacokinetics, pharmacodynamics, and clinical pharmacotherapy. My interest in academic writing has been nurtured and supported by several excellent physicians including Mayo Clinic colleagues James Bolling, MD; Tom Liesegang, MD (past editor-in-chief of the *American Journal of Ophthalmology*); and George Bartley, MD (editor-in-chief of *Ophthalmology*). Others who have significantly supported my academic development include Phil Rosenfeld, MD; David Browning, MD; Kurt Gitter, MD; son-inlaw Gregory P. Forlenza, MD; and son Michael Llort Stewart. I am forever indebted to Maurice B. Landers, III, MD, who has served as my mentor and valued colleague for 30 years.

I dedicate this work to the ladies in my life: my daughter-in-law Gwen Hochman Stewart, JD; daughter Tania Llort Stewart, MBA; and granddaughters Olympia Stewart Forlenza, Julia Claire Stewart, and Genevieve Stewart Forlenza. Most importantly, I thank my wife Enid Llort, MBA, from whom I have received neverending support and encouragement. She is the smartest, most intuitive woman I have ever met, and without her pushing and pulling me for the past 37 years, I would never have been able to compose this volume.

Contents

1	Diał	oetes ar	nd Diabetic Retinopathy: Overview of a Worldwide	
	Epic	lemic .		1
	1.1	Introd	uction	1
	1.2	Incide	nce of Diabetes	4
	1.3	Incide	nce of Diabetic Retinopathy	6
	1.4	Como	rbidities for Diabetes and Diabetic Retinopathy	13
	1.5	Concl	usions	18
	Refe	erences.		18
2	The	Diabet	ic Retina: Anatomy and Pathophysiology	29
	2.1	Introd	uction	29
	2.2	The R	etina	30
		2.2.1	History	30
		2.2.2	Anatomy	30
		2.2.3	Microanatomy of the Retina Neurons	34
		2.2.4	Intercellular Spaces	36
		2.2.5	Internal Limiting Membrane	36
		2.2.6	Circulation	37
		2.2.7	Arteries.	37
		2.2.8	Veins	38
		2.2.9	Capillaries	38
	2.3	Hemo	dynamics, Macular Edema, and Starling's Law	39
	2.4	Bioch	emical Basis for Diabetic Retinopathy	41
		2.4.1	Increased Polyol Pathway Flux	44
		2.4.2	Advanced Glycation End Products (AGEs)	44
		2.4.3	Activation of Protein Kinase C (PKC)	45
		2.4.4	Increased Hexosamine Pathway Flux	46
	2.5	Early	Pathophysiology of Diabetic Retinopathy	46
	2.6	Macul	lar Edema	47
		2.6.1	Blood-Retinal Barrier.	49
		2.6.2	Biochemical Abnormalities Responsible for Diabetic	
			Retinopathy	50

		2.6.3 Mechanism of Blood-Retinal Barrier Breakdown	53
		2.6.4 Renin/Angiotensin System	57
	2.7	Development of Proliferative Diabetic Retinopathy	57
	2.8	Clinical Findings of Diabetic Retinopathy	58
	2.9	Conclusions	63
	Refe	erences	63
3	Trea	atment of Diabetic Retinopathy: A Historical	
	Pers	spective	73
	3.1	Introduction	73
	3.2	Pituitary Ablation	74
	3.3	Interferon	75
	3.4	Laser Photocoagulation	76
		3.4.1 Techniques for Performing Laser Photocoagulation	77
	3.5	Treatment of Background Diabetic Retinopathy	83
	3.6	Treatment of Proliferative Diabetic Retinopathy (PDR)	85
	3.7	New Laser Technologies	86
		3.7.1 Micropulse Laser	87
		3.7.2 Navigated Laser	88
	3.8	Pars Plana Vitrectomy	89
	3.9	Conclusion	91
	Refe	erences	91
4	Targ	geting Vascular Endothelial Growth Factor	99
	4.1	Introduction	99
	4.2	Vascular Endothelial Growth Factor	100
	4.3	Anti-VEGF Drugs	106
		4.3.1 Pegaptanib	107
		4.3.2 Bevacizumab	110
		4.3.3 Ranibizumab	114
		4.3.4 Aflibercept	123
	4.4	Other Studies	128
	4.5	Comparison Trials	129
	4.6	Conclusions and Outstanding Questions	130
	Refe	erences	130
5	Cor	ticosteroids: Targeting Multiple Cytokines	
	and	Chemokines	141
	5.1	Introduction	141
	5.2	Characteristics of Corticosteroids	141
	5.3	Corticosteroid Delivery to the Eye.	144
	5.4	Triamcinolone	146
	5.5	Dexamethasone	149
	5.6	Fluocinolone	155
	5.7	Conclusions	158
	Refe	erences	158

6	Cur	rent Treatment Recommendations	163
	6.1	Introduction	163
	6.2	General Medical Care	164
		6.2.1 Diabetes Mellitus	164
		6.2.2 Systemic Arterial Hypertension	166
		6.2.3 Hyperlipidemia	167
		6.2.4 Sleep Apnea	167
	6.3	Screening Guidelines	167
		6.3.1 Screening Methods.	168
	6.4	Imaging	169
	6.5	Treatment of Diabetic Macular Edema	171
		6.5.1 Non-Center-Involving Diabetic Macular Edema	171
		6.5.2 Center-Involving Diabetic Macular Edema	172
	6.6	Treatment Failures	175
	6.7	Cataract Surgery and DME	179
	6.8	Proliferative Diabetic Retinopathy.	180
	6.9	Pregnancy	181
	6.10	Conclusions	182
	Refe	erences	182
7	Vitr	eolysis: Targeting the Vitreoretinal Interface	187
'	7 1	Introduction	187
	7.2	Vitreous Anatomy	188
	73	Posterior Vitreous Detachment	188
	74	Anomalous Posterior Vitreous Detachment	189
	7.5	Association Between Posterior Hyaloid and Diabetic Retinopathy	191
	7.6	Treatment of Vitreomacular Traction	192
		7.6.1 Streptokinase	195
		7.6.2 Hyaluronidase	196
		7.6.3 Nattokinase	196
		7.6.4 Chondroitinase	196
		7.6.5 Plasmin.	196
		7.6.6 Tissue Plasminogen Activator (tPA)	197
		7.6.7 Autologous Plasmin Enzyme (APE)	197
		7.6.8 Vitreosolve	198
		7.6.9 Ocriplasmin	198
	7.7	Conclusions	199
	Refe	erences.	200
8	Invo	ostigational Modications	205
0	8 1	Corticosteroid-Related Molecules	205 206
	0.1	8 1 1 Danazol	207
		8.1.2 Devamethasone-Cyclodextrin Microparticle Drops	207
		8.1.3 Difluprednate Ophthalmic Emulsion	208
		8.1.4 FGP-437	200 208
		8 1 5 Lotenrednol Etabonate	200 200
			209

	8.2	Vascul	ar Endothelial Growth Factor Inhibitors	209
		8.2.1	Abicipar Pegol	211
		8.2.2	Conbercept	211
		8.2.3	Encapsulated Cell Technology	212
		8.2.4	Gene Therapy	212
		8.2.5	Implantable Drug Delivery Pump (PMP)	213
		8.2.6	PAN-90806.	213
		8.2.7	Ranibizumab Sustained Release Reservoir	213
		8.2.8	RTH258	214
		8.2.9	Ziv-Aflibercept	214
	8.3	Tumor	· Necrosis Factor-α Inhibitors	215
		8.3.1	Adalimumab	216
		8.3.2	Etanercept	216
		8.3.3	Infliximab	216
		8.3.4	Pegsunercept	216
	8.4	Nonste	eroidal Anti-inflammatories	216
		8.4.1	Aspirin	217
		8.4.2	Diclofenac	217
	8.5	Other		217
	0.0	8.5.1	Adenosine Kinase Inhibitor	217
		8.5.2	Ang2 Inhibition	221
		8.5.3	Antioxidants	221
		8.5.4	ASP8232	222
		8.5.5	Darapladib	222
		8.5.6	Enalrestat	222
		857	Fasudil	223
		858	iCo-007	223
		859	Luminate (ALG-1001)	224
		8 5 10	Mecamylamine	224
		8 5 11	Microspheres	225
		8 5 12	Minocycline	225
		8 5 13	PF-04523655	225
		8 5 14	Plasma Kallikrein Inhibitor	220
		8 5 1 5	Rubovistaurin	227
		8 5 16	Sirolimus	227
		8 5 17	Saualamine	220
		8518	Tie? Λ gonist (Λ KB 0778)	229
		0.J.10 8 5 10	Tocilizumah	229
		8 5 20	Tenrotumumah	230
	06	0.J.20	Therepios	230
	0.0 Dofe	Future		230
	Rele	iences.		231
9	Safe	ty Cons	siderations of Pharmacotherapy	239
	9.1	Introdu	uction	239
		9.1.1	Intravitreal Injections	240
	9.2	Ocular	Complications	241

	9.3	Elevated Intraocular Pressure (IOP) and Glaucoma	242
	9.4	Cataracts	244
	9.5	Endophthalmitis	245
	9.6	Counterfeit Drugs.	247
	9.7	Other Complications	248
	9.8	Systemic Complications.	249
	9.9	Conclusion	251
	Refe	rences	252
10	Soci	Deconomic Cost of Diabetic Retinopathy and Therapy	257
	10.1	Introduction	257
	10.2	National Costs	258
	10.2 10.3	National Costs Cost-Effectiveness of Treating Diabetic Macular	258
	10.2 10.3	National Costs Cost-Effectiveness of Treating Diabetic Macular Edema	258 259
	10.2 10.3 10.4	National Costs Cost-Effectiveness of Treating Diabetic Macular Edema Comparative Cost-Effectiveness Models	258 259 263
	10.2 10.3 10.4 10.5	National Costs Cost-Effectiveness of Treating Diabetic Macular Edema Comparative Cost-Effectiveness Models Future Considerations Comparative Cost-Effectiveness Models	258 259 263 265
	10.2 10.3 10.4 10.5 Refe	National Costs Cost-Effectiveness of Treating Diabetic Macular Edema Comparative Cost-Effectiveness Models Future Considerations rences	258 259 263 265 265

Chapter 1 Diabetes and Diabetic Retinopathy: Overview of a Worldwide Epidemic

1.1 Introduction

By 2013, the world's population reached 7.2 billion with 138.8 million annual births and 54.9 million annual deaths [145]. The global population is projected to continue to increase, reaching 8.3 billion people by 2013, with 13% of these people falling into the fastest-growing age group – those over the age of 65 [119]. The world's changing demographic profile results from a remarkable transition in the causes of death. No longer are people dying primarily from communicable diseases. Rather, most deaths today are the result of noncommunicable diseases, a shift that has dominated the statistical patterns of industrialized countries for decades and is now also the trend in low- and middle-income countries.

Between 1970 and 2010, global life expectancy increased from 56.4 to 67.5 years for males and from 61.2 to 73.3 years for females [162]. During this same period, age-specific death rates were declining. Longer life expectancy prolongs exposure to various environmental and age-related risk factors and increases the incidence of deaths from noncommunicable diseases. In 2013, more than two-thirds of global deaths were due to noncommunicable diseases, with 17.3 million deaths due to cardiovascular and circulatory diseases, 8 million deaths due to cancer, and anywhere from 1.5 to 5.1 million deaths due to diabetes mellitus, making diabetes the eighth leading cause of death worldwide [30, 154]. Among older adults, diabetes is most heavily concentrated in Oceania, Latin America, North Africa, and the Middle East where a high body mass index is an important risk factor. High blood pressure is an important risk factor in Central and Eastern Europe, whereas high body mass index is important in Latin America, Oceania, North Africa, and the Middle East [98]. Of the 671 million obese individuals worldwide, 62% live in developing countries [122, 123]. The incidence rates of diabetes are likely to continue rising in association with the rapidly increasing obesity epidemic.

Diabetes mellitus (DM) represents a leading cause of morbidity and mortality throughout the world. Approximately 347 million persons worldwide (approximately 8.3% of the adult population) have diabetes mellitus [30, 154], and this has



† Prevalence in 2013 (millions) †† Projected prevalence in 2035 (millions) Projected increase from 2013 to 2035 (%)

International Diabetes Federation.. IDF Diabetes Atlas: 6th Edition. 2013

Fig. 1.1 This drawing shows projected increases in the prevalence of diabetes mellitus by continent from 2013 to 2035. The total number of diabetic patients throughout the world is projected to increase by 55% to 592 million. Increases are particularly significant in developing areas of Africa, the Middle East, and India

been projected to grow to 592 million by 2035 (Fig. 1.1) [12], numbers that have been revised upward several times in recent years. Diabetes ranks among the top ten causes of disability, primarily due to loss of ambulation from lower limb amputations and blindness due to diabetic retinopathy (DR). Because of its effects on pregnancy, DM adversely affects more than 21 million live births annually.

Diabetes disproportionately affects poor societies, as low- and middle-income countries account for 80% of diabetes-related mortality [72]. Diabetes-attributed mortality outstrips the combined annual deaths from malaria (600,000), tuberculosis (1.5 million), and AIDS (1.6 million) [169–171]. When combined with cardio-vascular disease, cancer, and chronic respiratory diseases, diabetes accounts for an annual global death toll of 35 million – over 60% of the world's mortality.

Worldwide concern over the increasing importance of DM even led to the passage of a 2006 United Nations resolution designating November 14 as "World Diabetes Day." It was not the mortality rate, however, but rather the macroeconomic impact that prompted global action on diabetes and other noncommunicable diseases (NCDs). It has been estimated that for every 10% rise in the incidence of NCDs, a country loses 0.5% of its gross domestic product [152]. Noncommunicable diseases ranked among the top global risks to business due to the trillions of dollars incurred annually through lost productivity [168]. Diabetes incurs an annual healthcare cost of US\$548 billion for 20–79-year-olds [30]. In low- and middle-income countries, blindness due to diabetes often disables breadwinners, burdens caregivers, and perpetuates a cycle of poverty.

Type 1 diabetes mellitus (T1DM) results from unopposed autoimmune destruction of the islet β -cells of the pancreas, which leads to insufficient production of insulin. Only 5-10% of worldwide cases of diabetes are classified as type 1 with the balance being type 2. Genetic factors appear to play important roles in the development of T1DM [112, 129, 153] but efforts to identify the responsible genes and their variants have met with limited success. More than half of the significant association signals for T1DM have been identified within chromosome 6p21 [15, 125, 129, 147], which maps to the human leukocyte antigen (HLA) region. This constitutes the most important region in the vertebrate genome regarding infection and autoimmunity and is crucial in regulating adaptive and innate immunity. A recent study of 4075 patients with T1DM identified 452 associated genes. Seven genes including four nonhuman leukocyte antigen (HLA) genes (RASIP1, STRN4, BCAR1, and MYL2) were replicated in at least one independent population and were differentially expressed in peripheral blood mononuclear cells or monocytes [136]. This emphasizes the fact that the genetic susceptibility pattern for T1DM appears complex [73] and may involve both susceptible and protective haplotypes. Despite our improved understanding of genetic associations, the triggering mechanism for the development of T1DM remains unknown, and in genetically susceptible individuals, it may result from exposure to as yet unidentified environmental factors.

The relative contributions of these haplotypes and their interactions with environmental factors and other genetic loci might partially explain the ethnic and racial variations in the frequency of T1DM [71]. Type 1 diabetes predominantly affects individuals of European ancestry, with the highest rates in Finland and Sardinia [11]. Asian and sub-Saharan African countries generally report low frequencies of T1DM, but Kuwait and China recently have reported higher rates [113, 180]. The SEARCH Study for Diabetes in Youth (2009) determined that T1DM remains a Caucasian-dominated disease with prevalence rates among 0–19-year-olds of 2.00 per thousand in non-Hispanic white patients, 1.31 for African-Americans, 0.99 for Hispanics, 0.94 for Navajos, and 0.52 for Asians and Pacific Islanders [11, 103].

Remarkable economic growth in Asia during the past 30 years has greatly improved the region's standard of living and extended life expectancy. Growth is transforming the region from a predominantly low-technology, agrarian society toward an industrialized society with greater urbanization. This has increased food availability and caloric consumption and created a more sedentary lifestyle with a declining rate of communicable diseases. Obesity rates have increased as have the prevalences of T2DM, further straining the already challenged healthcare systems in the developing countries of the region.

Type 2 diabetes is characterized by peripheral insulin resistance, impaired regulation of hepatic glucose production, and declining β -cell function that eventually leads to β -cell failure and dependence on exogenous insulin [154]. With rapid growth in the prevalence of T2DM in the last two decades, there has been a surge in the reports of T2DM-related diabetic retinopathy (DR), especially from Asia. In 2030, the largest numbers of patients with diabetes will be in India and China. About 150 million people in China currently show early symptoms of diabetes [27]. By the year 2033, 80 million people in India will suffer from T2D [154] and 1 million people with diabetes will die every year [37, 56].

1.2 Incidence of Diabetes

Substantial data showing the prevalence rates of DM among specific ethnic and racial populations and within several countries has been published. There appears to be uniform agreement that the prevalence rates among most of these groups will significantly increase in the foreseeable future. Some authors have characterized the rate of increase in the global prevalence of T2DM, particularly in some areas of the globe, as alarming [157, 164]. It is estimated that the prevalence of diabetes among people over the age of 16 years will rise by 28.3% between 2010 and 2030, with 54.5% of this increase being attributed to increased obesity [2]. The incidence of T1DM is also rising but not to the same degree [11].

In 2008, there were an estimated 2.4 million Canadians (out of a population of approximately 30 million) with DM, and this is expected to increase to 3.7 million by 2018 [66]. Five and one-half percent of the adult population of France suffers from T2DM. Among affected patients, the average age is 65.9 years, 55% are men, the prevalence of obesity is 43%, and 18% take insulin [46]. The number of people with DM in the Middle East is expected to triple from the year 2000 to 2030 to approximately 60 million [148].

The prevalence of T2DM in South African adults rose from 5.5% in 2000 to 9.0% in 2009. Two million South Africans currently live with T2DM with 115,000 new cases estimated to occur each year [12]. Eye screening programs have been ineffective because an estimated 55% of adults with diabetes remain undiagnosed [65].

According to the International Diabetes Federation, Brazil's population of patients with DM is the fourth largest in the world (11.9 million in 2013) with an estimated prevalence of 10.3% [8].

The annual age-adjusted prevalence and incidence rates of diabetes in the United States did not change significantly during the 1980s, but each rate increased yearly from 1990 through 2008 before leveling off until 2012. The prevalence per 100 population was 3.5 in 1990, 7.9 in 2008, and 8.3 in 2012 (Fig. 1.2). The incidence per 1000 persons was 3.2 in 1990, 8.8 in 2008, and 7.1 in 2012. Trends in many subpopulations were similar to these overall trends, but the incidence rates among non-Hispanic blacks (P = .03) and Hispanic adults (P = .01) continued to increase at rates significantly greater than those for non-Hispanic white adults. The prevalence rate increased faster for adults with a high school education or less, than for those with more than a high school education [57]. Based upon population and obesity trends, it was recently predicted that for a child born in the United States in 2000, the lifetime probability of being diagnosed with diabetes is 33% for males and 39% for females [117].

For predisposed individuals who do not yet meet criteria for the diagnosis of DM, early changes in lifestyle can prevent the onset of T2DM. The Diabetes Prevention



Diabetes Prevalence Among Adults, Age 20+: United States 2004-2010



Centers for Disease Control and Prevention. County Level Date. 2010

Program was a multicenter, randomized, controlled clinical trial that enrolled overweight individuals with elevated blood glucose levels but without definitive criteria for the diagnosis of diabetes. The consistent adoption of lifestyle changes including a low-fat diet, weight loss, and increased physical activity reduced the development of T2DM by 58% compared with placebo. Metformin (850 mg twice daily) lowered the incidence of diabetes by 31% compared with placebo. The study also demonstrated that approximately 8% of the patients who were prediabetic by commonly used clinical criteria already had diabetic retinopathy [7]. This suggests that the current criteria necessary for the diagnosis of DM exclude some patients who already show evidence of diabetes-related damage. Therefore, an argument can be made for changing the criteria required to diagnose T2DM.

Good control of blood glucose concentrations is the best way to prevent the development and progression of diabetes-related complications, but many patients with DM fail to achieve or maintain adequate metabolic control. Unfortunately, patients who manage to keep their HbA1c low have an increased risk of symptomatic hypoglycemia. The incidence of severe hypoglycemia in the Diabetes Control and Complications Trial (DCCT) was three times higher in the intensive treatment group compared with the conventional treatment group. Therefore, it appears unrealistic for patients with T1DM to pursue perfect glucose control because of the elevated risk of hypoglycemic episodes. Intensive glycemic control in the DCCT was also associated with a weight gain of 4.6 kg more than those in the conventional treatment group [51, 52]. It is difficult for patients with T2DM to achieve optimal metabolic control as patients in the intensive treatment group in the UKPDS were also plagued by increased hypoglycemic episodes and weight gain [51]. Metabolic control deteriorated over time in the UKPDS, possibly because of progressive loss of islet beta cell function [156]. The Action to Control Cardiovascular Risk in Diabetes (ACCORD) study was stopped because of increased all-cause mortality in people whose glucose was extremely tightly controlled with insulin and multiple oral agents [1].

Obesity is a major risk factor for the development of T2DM [44, 55, 68, 97, 160], and increased prevalences of both diabetes and obesity are closely correlated [110]. The increases in obesity prevalence rates may have recently plateaued [50, 128] because of declines in overall food purchases and caloric intake, which could mean that the growth rate of T2DM may stabilize [53, 122]. Bariatric (weight loss) surgery is an effective and increasingly commonly used treatment for obese patients with T2DM [179].

Physical inactivity as part of a sedentary lifestyle is considered to be an important modifiable risk factor for T2DM and cardiovascular disease. Increased physical activity may slow the onset and progression of insulin resistance and improve glycemic control, blood pressure, and the lipid profile [69, 76].

1.3 Incidence of Diabetic Retinopathy

Visual impairment is a major public health problem that significantly diminishes the quality of life of affected individuals. Visually impaired patients report having difficulty reading, driving a car, and preparing meals [58, 105, 106, 161]. Vision impairment leads to higher incidences of social isolation, poor overall health, and falls. Not only does vision impairment impose a major burden on the affected individuals but also on their families, caregivers, and society.

Diabetes is the leading cause of vision loss in patients between the ages of 20 and 74 years in industrialized nations [33, 140]. Diabetes affects all parts of the eye, ocular adnexae, neurosensory pathway, and ocular motility system, but most DM-related vision loss stems from retinopathy due to microvascular complications. Early diagnosis and treatment of DR is important because substantial and permanent vision loss may occur if DR is left untreated for 1 year or longer [3, 47]. An estimated 50,000 new cases of retinal neovascularization and diabetic macular edema (DME) occur yearly [120, 130], and as many as half of the patients who would benefit from treatment remain untreated [87].

Preventable causes of blindness such as cataracts, glaucoma, and infectious diseases are viewed as the most important ophthalmic public health problems in many parts of the world, making screening for diabetic retinopathy and subsequent treatment lesser priorities (Fig. 1.3). Fortunately, the incidence rate of nonproliferative diabetic retinopathy appears to be declining in the United States, thus supporting the contention that more aggressive management of DM and its comorbidities limits the development of DR-related vision loss. In response to the results of randomized, controlled trials, the initiation of the VISION 2020 program in 1999 [54, 172] and DR surveillance programs [141] has intensified control of risk factors [38, 143, 156], and continuing improvements in healthcare systems have contributed to the decreasing rates of DR [81]. Unfortunately, studies from less developed areas of the world have been more limited in scope and do not show similar trends [148].

The number of people worldwide with DR will increase from 126.6 million in 2010 to 191.0 million by 2030, and the number with vision-threatening diabetic retinopathy (VTDR) is estimated to increase from 37.3 million to 56.3 million



[183]. Among patients aged 20–79 years with diabetes, the worldwide prevalence of any grade of DR has been estimated to be 35% and that of PDR to be 7% [80]. Among patients with DM for at least 25 years, 50% of those with T1DM and 15% of those with T2DM will have proliferative retinopathy [6].

A pooled analysis using individual participant data (22,896 individuals) from 35 population-based studies (1980–2008) from around the world was performed to determine the prevalence of DR [177]. The overall prevalence of DR was 34.6% (95% CI 34.5–34.8%) for any DR, 6.96% (CI 6.87–7.04%) for proliferative DR, 6.81% (CI 6.74–6.89%) for diabetic macular edema, and 10.2% (10.1–10.3%) for VTDR. All DR prevalence endpoints increased with diabetes duration, hemoglobin A1c levels, and blood pressure measurements, and were higher in people with T1DM compared with T2DM. The prevalence estimates of any DR and VTDR were similar in men and women and were highest in African-Americans and lowest in Asians [146, 176]. Higher total serum cholesterol was associated with a higher prevalence of DME, bringing clarity to previously conflicting reports about this risk factor [167]. The authors concluded that there are approximately 93 million people with DR, 17 million with proliferative DR, 21 million with diabetic macular edema, and 28 million with VTDR worldwide.

An analysis of pooled data from several population-based studies estimated that approximately 40% of patients with diabetes who are over the age of 40 years have some retinopathy and 8.2% have vision-threatening retinopathy [81].

Many have attempted to define the contribution of hereditary factors to the development of DR [144], but no firm data suggests that DR has a genetic component. This differs from diabetic nephropathy where important genetic associations have been described recently [43]. Candidate gene studies and GWAS (genome-wide association studies) may ultimately find genetic linkage to retinopathy phenotypes.

The WHO (World Health Organization) has estimated that diabetic retinopathy accounts for approximately 15-17% of total blindness in Europe and the United States [134]. The widespread use of drugs that inhibit the actions of vascular endothelial growth factor (VEGF) has significantly decreased the incidence of blindness due to age-related macular degeneration (AMD), prompting some authors to speculate that diabetes has become the overall leading cause of blindness [28, 45]. Further complicating the care of many of these patients is the fact that diabetes is also a leading cause of renal failure through its effects on the microvasculature [29, 77, 126]. Since prevalence rates of systemic arterial hypertension – a well-established risk factor for DR – are also increasing, the importance of DR to vision loss in industrialized nations is unlikely to diminish [70, 131].

Diabetic retinopathy is responsible for up to 17% of all blindness in parts of the Americas, Europe, and the Western Pacific [163]. The numbers of patients with both DM and DR appear to be increasing, and by 2050 as many as 50 million or more individuals in the United States will have DM with half having some form of retinopathy [10, 18, 19, 75, 81, 116]. In 2004, the Eye Diseases Prevalence Research Group estimated the prevalence of diabetic retinopathy from the compilation of eight separate population-based studies from the United States and elsewhere that had been conducted in the late 1980s or early 1990s [81]. Their report recommended that more recent estimates of diabetic retinopathy prevalence be obtained from the nationally representative sample of the National Health and Nutrition Examination Survey (NHANES) [181].

The 2005–2008 National Health and Nutrition Examination Survey evaluated 5222 Americans aged 40 years and over for general visual impairment (distance visual acuity worse than 20/40 in the better-seeing eye) and visual impairment not due to refractive error (distance visual acuity worse than 20/40 after refraction). The overall prevalences of visual impairment and of visual impairment not due to refractive error were 7.5% (95% CI, 6.9%, 8.1%) and 2.0% (95% CI, 1.7%, 2.3%), respectively. Not surprisingly, the prevalence of visual impairment not due to refractive error was significantly higher among people with DR (3.5%) compared to those without DR (1.2%) [28]. The prevalences of DR and vision-threatening DR were found to be 28.5% (95% CI, 24.9-32.5%) and 4.4% (95% CI, 3.5-5.7%). A slightly greater prevalence of DR was found among men (31.6%, 95% CI, 26.8–36.8%) than women (25.7%, 95% CI, 21.7–30.1%, P = 0.04). Non-Hispanic blacks, compared to non-Hispanic whites, had higher incidences of DR (38.8%, 95% CI, 31.9-46.1% vs. 26.4%, 95% CI, 21.4–32.2%, P = 0.01) and vision-threatening retinopathy (9.3%, 95% CI, 5.9–14.4% vs. 3.2%, 95% CI, 2.0–5.1%, P = 0.01). Male gender was independently associated with the presence of DR (OR 2.07, 95% CI 1.39-3.10), higher HbA1c (OR, 1.45, 95% CI, 1.20-1.75), longer duration of diabetes (OR, 1.06 per year duration, 95% CI, 1.03-1.10), insulin use (OR, 3.23, 95% CI, 1.03-1.10), and higher systolic blood pressure (OR, 1.03 per mmHg, 95% CI, 1.02-1.03).

A previous analysis of NHANES III data suggests that the prevalence of diabetic retinopathy is 46% higher in non-Hispanic black individuals and 84% higher in Mexican Americans than in non-Hispanic whites [61, 63]. The prevalence of vision-threatening diabetic retinopathy was 190% higher in non-Hispanic blacks and 130% higher in Mexican Americans than in non-Hispanic whites, probably because non-Hispanic blacks and Mexican Americans have poorer diabetes control and are less likely to be screened and treated in a timely manner [127]. Non-Hispanic black individuals and Hispanics are less likely to use eye care services [182]. These findings highlight the need to reduce disparities in care among racial, ethnic, and socio-economic groups [182].

Latino Americans have one of the highest rates of visual impairment but the prevalence and risk of undetected eye disease cannot be accurately quantified. In one study, the 4-year incidences of best corrected visual impairment and blindness from DR in Latinos were 1.2 and 0.3%, respectively [159]. Access to healthcare by this population is inconsistent and unequal compared to other populations, and this may influence disease statistics. In contrast, no association was found between socioeconomic status and DR in cohorts of Mexican Americans and Caucasian patients with T2DM in Texas [60]. Approximately 13% of African-Americans have T2D, with the prevalence and incidence of T2DM being at least twice as high as that among white Americans [18, 62].

There are more than two million Native Americans on the North American continent, comprising more than 500 tribal organizations, and a comprehensive review of complications of T2DM in this indigenous population reveals high prevalence rates of DR for all populations studied. High prevalence rates of DR have been observed among the Alberta First Nations of Canada (40%) and the Pima Indians in Arizona (37.8%). In the Southern Alberta Study of Diabetic Retinopathy, DR in nonnatives tended to be more advanced, but the prevalences of DR in native and nonnative Canadians were identical (40%).

The Eye Disease Prevalence Research Group estimated that 4.1 million Americans had diabetic retinopathy in 2010 and projected that this would rise to 7.2 million by 2020 [118]. It has been estimated that one in every 12 Americans with diabetes over the age of 39 has vision-threatening retinopathy. A recently published study of the prevalence of DR and vision-threatening DR in a nationally representative sample of US adults aged 40 years or older showed that approximately 1.5% (95% CI, 1.1–2.2%) of adults with diabetes had proliferative DR and 2.7% (95% CI, 1.8–4.0%) had CSME [181]. Over a 10-year period, nonclinically significant DME and CSME will develop in 14% and 10%, respectively, of Americans with known diabetes [89]. Approximately half of all patients with DME will lose two lines or more of vision within 2 years [47, 48].

A cross-sectional, subgroup analysis of fundus photographs from 1038 participants with diabetes aged 40 years or older showed that 55 had DME, for an overall weighted prevalence of 3.8% (95% CI, 2.7–4.9%); this translates to 746,000 persons with DME in the US 2010 population. There were no differences in prevalence rates due to age or gender, but the chances of having DME were higher for non-Hispanic blacks than for non-Hispanic whites (odds ratio [OR], 2.64). Elevated

hemoglobin A1c levels (OR, 1.47; P < .001) and longer duration of diabetes (OR, 8.51; P < .001) were also associated with DME [13].

Diabetic retinopathy appears to be strongly associated with all-cause mortality. An analysis of 20 studies with data from 19,234 patients showed that any degree of DR increased the risk of all-cause mortality by 2.34-fold (95% CI, 1.96–2.80) among type 2 diabetics and 2.41-fold (95% CI, 1.87–3.10) among type 1 diabetics [95]. Fortunately, however, rates of non-ocular diabetic complications have declined during recent decades. The hospitalization rate for lower extremity amputations among individuals with diabetes began decreasing in 1997 [57], and the prevalence of diabetes-related end-stage renal disease decreased between 1996 and 2006 [16]. It is reasonable to assume from these favorable trends in incidence data that improved diabetes care, such as effective management of blood glucose levels, blood pressure, and serum lipid levels, may also be reducing the incidence of diabetic retinopathy [16, 57].

Population-based studies have shown that nearly all patients with T1DM and more than 60% of those with T2DM develop DR during the first two decades of the disease [51]. Recent population-based studies have reported decreases in the prevalence and incidence of severe DR [67, 92, 106]. These findings, however, were limited to regional populations, and broad application of this data to national populations cannot be done with certainty [181]. The prevalence of DR has been changing because of improvements in management of blood glucose, serum lipids, and blood pressure [4]. In the 8 years between the beginning of the Wisconsin Epidemiologic Study of Diabetic Retinopathy (WESDR) and the beginning of the Beaver Dam Eye Study (BDES), the prevalence of any DR in persons with T2DM fell by 30% (from 50% in the WESDR in 1980–1982 to 35% in the BDES in 1988–1990), and the prevalence of vision-threatening DR fell by 70% (from 10% in the WESDR in 1980–1982 to 3% in the BDES in 1988–1990) [85, 86, 88].

The Wisconsin Epidemiologic Study of Diabetic Retinopathy reported that the 10-year incidence of DME in the United States was 20.1% among type 1 diabetics, 25.4% among insulin-dependent type 2 diabetics, and 13.9% for non-insulin-dependent type 2 diabetics. The prevalence of DME was 3% (diabetes duration, 0–14 years) or 6% (>15 years) in mild NPDR, 37% (0–14 years) or 63% (>15 years) in moderate/severe NPDR, and 73% (0–14 years) or 74% (>15 years) in PDR [82]. Nearly half of the patients with DME will lose two or more lines of visual acuity within 2 years.

National DR screening programs have been introduced to all four countries of the United Kingdom during the last decade resulting in significant reorganization of care for these patients [17, 124, 135, 141]. Analysis of a large UK database found that 15.8–18.1% of diabetics had center-involving macula edema [80]. Of these, 14% of eyes had mild NPDR, and 24% of eyes with moderate NPDR, 31% of eyes with severe NPDR, and 22% of eyes with PDR also had CSME (whether or not center-involving). Prevalence figures for DR are approximately three times higher in the hospital databases than whole population estimates, reflecting the tendency that patients with more advanced grades of DR are managed in the hospital systems. The prevalence of any grade of DR was around 80% in this study, compared with

20–40% in most population studies of patients with diabetes. PDR prevalence was around 20%, and CSME prevalence was around 18% in the hospital database, compared with 1–8% and 6%, respectively, in the general populations [14, 107, 111, 133, 150, 178].

From the English National Health System database, the number of people with diabetes in England in 2010 was estimated at 2,342,951 of which 2,334,550 were over the age of 12 years. An estimated 166,325 (7.12%) had DME in one or both eyes, and of these, 64,725 individuals had DME that reduced the visual acuity to poorer than 6/6 in at least one eye. The overall health and social care costs in 2010, from screening through treatment, rehabilitation, and care in the home, were estimated at $\pounds116,296,038$.

The community-based National Diabetic Retinopathy Screening Service for Wales performed a cross-sectional analysis of 91,393 persons with diabetes (5003 with T1DM and 86,390 with T2DM) at their first screening from 2005 to 2009. The prevalence of any DR and sight-threatening DR in patients with T1DM was 56.0% and 11.2%, respectively, and in patients with T2DM was 30.3% and 2.9%, respectively. The presence of DR was strongly associated with increasing duration of diabetes for patients with either T1DM or T2DM and was also associated with insulin therapy in patients with T2DM.

A UK database study evaluated the incidence of DR in newly diagnosed diabetics [100]. By 9 years after the diagnosis of diabetes, 28% of T2DM and 24% of T1DM patients had developed DR (7899 incident DR cases). During the first 2 years with diabetes, the incidence rate was almost two times higher in patients diagnosed with diabetes in 2006–2007 than among those diagnosed in 2000–2001. Among patients with retinopathy at baseline, the study found a cumulative incidence of DME in 12.1% of T2DM patients and 18.8% of T1DM patients within 9 years.

The World Health Organization estimates that over 40 million Chinese will have diabetes by 2030 [164]. Similar to the studies in South Asia, the Beijing Eye Study reported that DR is a relatively minor cause of blindness (7.7%) in the Chinese population [78]. Nonetheless, with continued economic progress, this is becoming a threat to public health in many areas. In China, the prevalence of diabetic retinopathy among people with diabetes is predicted to reach 43% [177], with up to 9.2 million people in rural areas having diabetic retinopathy and 1.3 million having sight-threatening diabetic retinopathy. Reflecting the fact that DM and DR are related to lifestyle, DR is more prevalent in South Asians living outside the Indian subcontinent [138] and in urban cities [137] compared to those who reside in rural areas within India [115]. Projections suggest that for India alone, 0.7 million people will have proliferative diabetic retinopathy (PDR) in 2030, and 1.8 million will have clinically significant macular edema [139].

A Japanese study found a significant correlation between the severity of DME and the DR grade [175]. In this report, 28% of patients with mild/moderate NPDR, 67% with severe NPDR, and 51% with PDR also had DME.

Studies from Australia that date back more than three decades support the impact of health education and better glycemic control on the prevalence and incidence of DR [35, 104, 114]. In the past three decades, the prevalence and incidence of

DR among patients with T1DM have declined in the United States, Australia, and other developed countries [183]. The earliest clinic-based study of DR in Australia, the Newcastle Diabetic Retinopathy Study (1977–1988), reported a 35% prevalence of DR, but subsequent population-based studies have reported lower rates [109] with the Australian Diabetes, Obesity, and Lifestyle study (AusDiab) reporting a DR prevalence of only 21.9%.

Comparing the prevalence rates for DR among studies is difficult because of the changing classification of diabetes over time, the different grading protocols employed, and differences in the characteristics of "similar" populations [177, 178, 181]. Findings from these studies, however, can provide policy makers with important information to plan eye care services, with the understanding that the prevalence of sight-threatening DR may be underestimated. The strong association between duration of diabetes and risk of retinopathy underscores the importance of early detection via office-based examinations or screening programs. An effectively structured screening program may reduce the incidence of blindness by 40% within 4 years [121].

Improving glycemic control by lowering the level of glycosylated hemoglobin (HbA1c) is the most effective way to slow the progression of DR. Keys to the discovery that optimal metabolic control could reduce the incidence and progression of DR were the DCCT and the UKPDS trials. Not only did intensive glycemic control inhibit the development of DR, these effects persisted well beyond the course of treatment [39, 40].

The Diabetes Control and Complications Trial (DCCT) (1982–1993) was a multicenter, controlled clinical trial that compared the effects of intensive blood glucose control (INT) with standard control (CON) on the onset and progression of DR. The Epidemiology of Diabetes Interventions and Complications (EDIC) study (1994– present) is an observational follow-up of the DCCT cohort. Of the 1441 DCCT subjects, 726 had no DR (primary prevention cohort) and 715 had mild DR (secondary intervention cohort) at baseline. Subjects were followed for a mean of 6.5 years. The median HbA1c was 7% in the INT group compared with a median of 9% in the CON group. INT reduced the adjusted mean risk for the development of DR by 76% and slowed progression of DR by 54% compared with CON.

Following DCCT, the HbA1c levels in the original INT and CON groups converged (year 8, INT 7.98%, CON 8.07%), but the INT group continued to enjoy a durable effect with significantly lower incidence of further DR progression (hazard reduction, 53–56%). Serious retinal outcomes and the need for procedures to treat them were reduced by 50% in the original INT group.

DCCT demonstrated that intensive glucose-lowering therapy for a mean of 6.5 years reduces the risk of development and progression of retinopathy by as much as 76% compared with conventional therapy. Much of the original effect persisted for over 18 years of follow-up in EDIC – the so-called metabolic memory – as the cumulative incidence of each retinal outcome continued to be lower in the former INT group. Interestingly, the year-to-year incidence of these outcomes is now similar in the two groups because of a reduction in risk in the former CON group [40].

1.4 Comorbidities for Diabetes and Diabetic Retinopathy

Higher levels of hemoglobin A1c, longer duration of diabetes, insulin use, and higher systolic blood pressure have been independently found to be associated with diabetic retinopathy [85, 86, 151, 165, 166, 182]. The early randomized controlled clinical trials showed that modifying risk factors such as hyperglycemia and blood pressure could reduce the burden of diabetic retinopathy and prevent vision loss [38, 155, 156] (Table 1.1). Efforts to prevent the development and progression of

Table 1.1 This table lists some of the important trials that have studied the effects of the major systemic contributing factors (hyperglycemia, systemic arterial hypertension, hyperlipidemia) on the development and progression of diabetic retinopathy

Important trials associating diabetic retinopathy with major systemic risk factors			
Hyperglycemia			
Diabetes Complications and Control Trials (DCCT) Type 1 diabetes mellitus	 Intensive glucose control: 1. Reduced the adjusted mean risk for the development of DR by 76% 2. Slowed progression of DR by 54% 		
UK Prospective Diabetes Study (UKPDS) Type 2 diabetes mellitus	Intensive glucose control: 1. Reduced progression by 35% per A1C point 2. Reduced moderate vision loss by 47%		
Systemic arterial hypertensio	n		
Wisconsin Epidemiologic Study of Diabetic Retinopathy	Progression of retinopathy was associated with:1. Higher diastolic BP at baseline2. Increase in diastolic BP over a 4-year period		
UK Prospective Diabetes Study (UKPDS) Type 2 diabetes mellitus	Found that systolic BP <150 mmHg:1. Decreased the progression of DR2. Decreased the need for macular laser photocoagulation for DME		
ACCORD-Eye trial	Found that intensive BP control did not decrease the progression of DR		
EUCLID study	Found that lisinopril decreased the progression of DR in normotensive type 1 diabetics		
Diabetic Retinopathy Candesartan Trials (DIRECT)	 In type 1 diabetics, 5 years of treatment: 1. Decreased the incidence of DR 2. Had no effect on progression of established DR In type 2 diabetics, there was a 34% regression of DR (<i>P</i> = 0.009) Less severe retinopathy in types 1 and 2 (<i>P</i> = 0.03) 		
RASS trial	Evaluated 285 normotensive patients treated with enalapril, losartan, or placebo for 5 yrs Progression of retinopathy by two steps in: Placebo (38%) Enalapril (25%, $P = 0.02$) Losartan (21%, $P = 0.008$) Enalapril and losartan increased the likelihood of less DR progression by 65% and 70% independent of blood pressure lowering		

(continued)

Hyperlipidemia			
Fenofibrate Intervention and Event Lowering in Diabetes Study (FIELD)	 Found that fenofibrate: 1. Decreased the requirement for the first laser and the development of DME 2. Decreased the need for laser treatment compared to the control group (3.4% vs. 4.9%, <i>P</i> = 0.0002) 3. Appeared to have protective effects independent of blood glucose, blood pressure, and baseline lipid values 		
ACCORD-Eye study	The addition of fenofibrate to basal statin therapy resulted in: 1. A decrease in the progression of DR, in a similar manner to that observed with intensifying blood glucose control, but with a good safety profile without increasing the risk of hypoglycemia 2. Questions regarding fenofibrate's mechanism of action and the pathogenesis of DR/DME		

Important trials associating diabetic retinonathy with major systemic risk factors

Table 1.1 (continued)

DR diabetic retinopathy, BP blood pressure, DME diabetic macular edema

DR should target individuals with the highest risk of developing retinopathy, those with longer duration of diabetes, and those using insulin. Improved availability of care for vulnerable sections of the population should be able to reduce the risk of blindness in diabetes. Identifying and protecting individuals at risk of T2DM, by reducing body weight and increasing physical activity, may also help delay the onset of T2DM and reduce complications of diabetic retinopathy.

Diabetic retinopathy is associated with increased cardiovascular (CV) and allcause mortality in patients with T2DM and T1DM [41, 84, 149, 158]. Diabetic retinopathy also predicts all-cause mortality, more than just CV events. Several mortality studies showed that CV disease was the cause of death in fewer than 35% of diabetic patients [61, 84], suggesting that an alternative mechanism increases the death rate in many DR patients. An autonomic neuropathy could link DR and CV events since it was recently demonstrated that autonomic deregulation can lead to alterations in blood pressure and cardiac rhythm and the development of DR [5, 94].

In a recent systematic review of 17 studies of 14,896 people with T2DM (mean age 58 years and mean follow-up 9 years), people with any retinopathy were more than twice as likely to die or suffer a fatal or nonfatal CV event than people without retinopathy, and a fourfold higher risk was noted for people with advanced retinopathy. The same review also included four studies of 4438 people with T1DM (mean age 33 years and mean follow-up 12 years) and reported a 3.5–4-fold higher risk of death as well as CV events in the presence of any retinopathy and a sevenfold higher risk with advanced retinopathy.

Several explanations may account for the relationship between retinopathy and CV outcomes. First, both retinopathy and incident CV outcomes are recognized consequences of diabetes. Second, the degree of retinopathy is progressively related to the degree of several independent risk factors of CV outcomes, including hyper-glycemia, elevated blood pressure, albuminuria, renal insufficiency, hyperlipidemia,

and other abnormalities. People with these risk factors would therefore have both more severe retinopathy and a higher incidence of CV outcomes. Third, the microvascular abnormalities present in the retina may also be occurring in many other vascular beds, and CV outcomes may be due in part to accumulated microvascular abnormalities in the myocardial microcirculation, arterial wall, and elsewhere. If this is true, changes in retinal pathology may closely reflect changes in microvascular pathology in these other vascular beds, and people with the most rapid progression in retinal pathology may be the ones most likely to suffer adverse CV events.

Recognizing and classifying diabetic retinopathy can improve the accuracy of a diabetic patient's CV risk stratification. A dilated fundus examination is inexpensive and should be routinely performed annually to screen for retinopathy, making this examination highly cost-effective. Previously mentioned findings support the view that changes in the retina may reflect changes in an individual's CV risk and may therefore identify those individuals whose CV risk is rising and who may benefit from particularly aggressive CV risk reduction therapies. Such a possibility should be tested in future clinical trials. Moreover, a retinal benefit of some therapy after only a few years (despite no CV effect during that period) may predict a CV benefit over a much longer period of time. Indeed, the findings of both the Diabetes Control and Complications Trial (in patients with T1DM) and the UK Prospective Diabetes Study (in patients with T2DM) showed that therapies that reduce the progression of retinal disease in the short-term subsequently reduce CV outcomes after long-term follow-up. If subsequent studies, including ongoing follow-up of the ACCORD participants, support this finding, retinal assessments may become integral to the regulatory assessment of the CV risk of drugs to treat diabetes. Furthermore, whether retinopathy progresses or not could be a gauge of whether a particular patient is reducing his or her risk of CV outcomes in response to the cardioprotective therapies that have been prescribed and provide an empirical basis to change therapies that are not working.

Hypertension is a major risk factor for the development and progression of both DR and DME. The WESDR found that progression of retinopathy was associated with higher diastolic blood pressure at baseline and an increase in diastolic blood pressure over a 4-year follow-up period [90]. The UKPDS demonstrated that control of blood pressure (systolic blood pressure <150 mmHg) led to a reduction in the progression of diabetic retinopathy and reduced need for laser treatment in the tight blood pressure control group compared with the less tight control group [156]. In the United Kingdom, this finding has contributed to the establishment of a target blood pressure measurement in diabetic patients of 140/80 mmHg. The American Diabetes Association and the National Institutes of Health recommend a target blood pressure of less than 130/80 mmHg for diabetics. The ACCORD study, however, produced contrasting findings to those of UKPDS in terms of the effects of intensive blood pressure control did not reduce progression of DR.

The SisHyperDia national registry for diabetes and hypertension found that 4.3% of diabetic patients had foot disorders, 2.2% had a previous amputation, 7.8% had renal disease, 7.8% had a previous myocardial infarction, and 8% had a previous

stroke [142]. Registry data also suggested that age- and gender-adjusted mortality figures showed a 57% higher risk of death among diabetes.

A meta-analysis that included 40 trials (100,354 participants) evaluated the effects of HTN on morbidity and mortality in diabetics. It concluded that each 10 mmHg lowering of systolic blood pressure reduced mortality rates (relative risk [RR], cardiovascular events (RR, 0.89 [95% CI, 0.83–0.95]), coronary artery disease (RR, 0.88 [95% CI, 0.80–0.98]), stroke (RR, 0.73 [95% CI, 0.64–0.83]), albuminuria (RR, 0.83 [95% CI, 0.79–0.87]), and retinopathy (RR, 0.87 [95% CI, 0.76–0.99]) [45].

Attempts have been made to determine whether blocking the renin-angiotensin system (RAS) can reduce the incidence or progression of DR. The EUCLID study group investigated the effect of lisinopril, an angiotensin-converting enzyme (ACE) inhibitor, on progression of retinopathy in normotensive subjects with T1DM [21, 22]. The investigators found that lisinopril can decrease retinopathy progression in non-hypertensive patients with T1DM and little or no nephropathy. It is not known if this effect is due to RAS blockade or incremental lowering of blood pressure in normotensive subjects. Other studies, however, have not shown that ACE inhibitors decrease retinopathy development in normotensive patients with T1DM [20, 96].

The Diabetic Retinopathy Candesartan Trials (DIRECT) were a group of large, randomized trials designed to determine if blockade of the RAS with candesartan reduces the incidence or progression of DR in normotensive patients without albuminuria [22]. Almost 5 years of candesartan treatment in patients with T1DM reduced the incidence of retinopathy by two or more steps (ETDRS) in severity by 18% (P = 0.0508) and, in a post hoc analysis, reduced the incidence of retinopathy by three-step progression by 35% (P = 0.034). In patients with T1DM, there was no effect on progression of established retinopathy, but 5 years of candesartan treatment to patients with T2DM resulted in a 34% regression of retinopathy (P = 0.009). A significant overall trend toward less severe retinopathy was noted in patients with both T1DM and T2DM (P = 0.03). Though the analysis was corrected for blood pressure values obtained throughout the trial, it could not be proven conclusively that the effects on retinopathy were due to RAS blockade and not to blood pressure lowering [173]. The favorable effect of blocking the RAS was confirmed by the RASS study [102], a multicenter, randomized, double-blind, placebo-controlled, investigator-initiated trial conducted on 285 normotensive patients treated with enalapril 20 mg/day, losartan 100 mg/day, or placebo for 5 years. The primary functional endpoint was a two-step progression of retinopathy severity, which was seen in 38% of patients receiving placebo, but only 25% of those receiving enalapril (P = 0.02) and 21% of those receiving losartan (P = 0.008). Enalapril and losartan increased the likelihood of slowing the progression of retinopathy by 65% and 70%, respectively, independent of changes in blood pressure.

In 1991, Gordon et al. [59] found that lipid-lowering therapy reduced hard exudates and microaneurysms in DR and may decrease the risk of vision loss in patients with DR [26]. Several large studies have subsequently studied this question by randomizing patients to statin and or fenofibrate treatment.

The Action to Control Cardiovascular Risk in Diabetes (ACCORD) trial was a large, randomized, multicenter, controlled trial that evaluated the effects of

intensive versus standard glucose lowering, intensive versus standard systolic blood pressure (SBP) lowering, and the addition of a fibrate versus placebo to a statin on the 5-year incidence of serious CV outcomes in people with type 2 diabetes and other CV risk factors. After a mean of 3.5 years, patients receiving intensive glucose control were switched to standard glucose control because they had a higher risk of death from all causes (5% vs. 4%) [1]. Tighter glucose control lowered the risk of retinopathy progression, but among patients without DR at baseline, no differences in the incidence of developing retinopathy were seen. By 4 years, the risk of having a CV event increased to 38% (hazard ratio [HR] 1.38; 95% confidence interval [CI], 1.10, 1.74) for each category change in retinopathy severity. Compared with no retinopathy, the adjusted hazard ratios (95% CI) for the development of cardiovascular outcomes rose from 1.49 (1.12–1.97) for mild retinopathy to 2.35 (1.47–3.76) for severe retinopathy. Although the relationship was insignificant after adjusting for baseline and follow-up A1C levels, systolic blood pressure, and lipids, the authors concluded that the retina may reflect the effect of metabolic and hemodynamic factors on future CV outcomes and that these two complications of diabetes may have shared pathophysiology. ACCORD found that fenofibrate combined with simvastatin reduced the progression of DR at 2 years, but the ACCORDION extension trial showed no effect at 8 years [25].

Both the FIELD (Fenofibrate Intervention and Event Lowering in Diabetes) and ACCORD trials suggested an important role of fenofibrate for the treatment of DR, and results were aggregated for analysis. These trials included 11,388 patients with T2DM, of which 5701 were treated with fenofibrate (+/-statin) for up to 5 years. In the FIELD study, retinopathy progression was defined as laser treatment for PDR or macular edema or an increase by ≥2 steps on the Early Treatment Diabetic Retinopathy Study (ETDRS) severity scale. Disease progression in the ACCORD-Eve study was defined as an increase of ≥ 3 steps on the ETDRS severity scale or proliferative disease requiring laser or vitrectomy treatment. In FIELD, fenofibrate (200 mg/day) reduced the requirements for laser therapy and was shown to arrest disease progression in patients with preexisting diabetic retinopathy. In ACCORD-Eye, fenofibrate (160 mg/day) taken with simvastatin yielded a 40% reduction in the odds of retinopathy progression when compared with simvastatin alone over 4 years. Fenofibrate reduced the need for first laser treatment by 31% (*P* = 0.0002) and progression of diabetic retinopathy with absolute reductions of 5.0% over 5 years (P = 0.022, FIELD) and 3.7% over 4 years (P = 0.006, ACCORD-Eye) [24, 79, 167, 174].

A meta-analysis of 35 articles from 1980–2008 found that higher total serum cholesterol was associated with a higher prevalence of DME, thereby clarifying previously conflicting reports regarding this risk factor [167, 177]. This was particularly relevant to trials suggesting that fenofibrate, a lipid-altering agent, may slow the development and progression of DR [79]. Fenofibrate acts mostly on triglycerides, and its effects on retinopathy in those trials were independent of lipid levels achieved. Statins, however, did not affect DR severity in the few studies in which cholesterol was evaluated, although this was not a primary outcome [31, 32].

High caloric and sodium intakes are significant and independent risk factors for the progression of DR in African-Americans with T1DM [140]. Other contributing factors in all ethnic groups include high body mass index (BMI) [36, 93], lack of physical activity, dyslipidemia in CSME [108], microalbuminuria [23], smoking, and socioeconomic factors [132]. Elevated hemoglobin levels predict the incidence of PDR in T1DM [34], whereas moderate alcohol consumption reduces the risk of PDR [9]. Advancing age at onset significantly decreases the long-term risk of PDR with the highest risk in the 5–14 years group and the lowest risk in the 15–40 years group [64]. Hormonal changes induced by puberty, including increased growth hormone and insulin-like growth factor, and the effect of the duration of prepubertal diabetes on the development of DR remain controversial [42, 49].

Increased ocular axial length, as is seen in high myopia, may be protective against the development of DR [99]. The relationship between smoking and the development of DR is unclear as some studies have discovered a link [91], whereas others have not [101].

1.5 Conclusions

The rapidly increasing number of patients with diabetes mellitus throughout the world threatens to stress healthcare systems. Epidemiological studies show that there have been remarkable improvements in the care and management of diabetes over the last 30 years that have resulted in significant decreases in the prevalence and incidence of DR in patients with T1DM [83]. Unfortunately, only limited long-term epidemiological data are available to determine whether similar trends exist for patients with T2DM [83]. Since optimal treatment of the key risk factors for DR and DME requires coordinated medical treatment, ophthalmologists caring for patients with diabetic eye disease should discuss management options with patients and primary care physicians.

References

- Action to Control Cardiovascular Risk in Diabetes Study Group. Effects of intensive glucose lowering in type 2 diabetes. N Engl J Med 2008;358:2545 e59.
- Alwakeel JS, Sulimani R, Al-Asaad H, et al. Diabetes complications in 1952 type 2 diabetes mellitus patients managed in a single institution in Saudi Arabia. Ann Saudi Med. 2008;28:260–6.
- American Academy of Ophthalmology Retina/Vitreous Panel. Preferred practice pattern[®] guidelines: diabetic retinopathy. San Francisco: American Academy of Ophthalmology; 2008. http://www.aao.org/ppp. Accessed April 30, 2013.
- 4. Antonetti DA, Klein R, Gardner TW. Diabetic retinopathy. N Engl J Med. 2012;366:1227-39.
- Ayad F, Belhadj M, Pariés J, Attali JR, Valensi P. Association between cardiac autonomic neuropathy and hypertension and its potential influence on diabetic complications. Diabet Med. 2010;27:804–11.

- Backlund LB, Algvere PV, Rosenqvist U. New blindness in diabetes reduced by more than one-third in Stockholm County. Diabet Med. 1997;14:732–40.
- Barrett JC, Clayton DG, Concannon P, et al. Genome-wide association study and metaanalysis find that over 40 loci affect risk of type 1 diabetes. Nat Genet. 2009;41:703–7.
- 8. Bertoldi AD, Kanavos P, Franca GV, et al. Epidemiology, management, complications and costs associated with type 2 diabetes in Brazil: a comprehensive literature review. Global Health. 2013;9:62.
- Beulens JW, Kruidhof JS, Grobbee DE, et al. Alcohol consumption and risk of microvascular complications in type 1 diabetes patients: the EURODIAB Prospective Complications Study. Diabetologia. 2008;51:1631–8.
- 10. Blumenkranz MS. Optimal current and future treatments for diabetic macular oedema. Eye. 2010;24(3):428–34.
- 11. Borchers AT, Uibo R, Gershwin ME. The geoepidemiology of type 1 diabetes. Autoimmun Rev. 2010;9:A355–65.
- 12. Bradshaw D, Norman R, Pieterse D, Levitt NS. Estimating the burden of disease attributable to diabetes in South Africa in 2000. S Afr Med J. 2007;97(8):700–6.
- Bressler NM, Varma R, Doan QV, Gleeson M, Danese M, Bower JK, Selvin E, Dolan C, Fine J, Colman S, Turpcu A. Underuse of the health care system by persons with diabetes mellitus and diabetic macular edema in the United States. JAMA Ophthalmol. 2014;132(2):168–73.
- Broadbent DM, Scott JA, Vora JP, Harding SP. Prevalence of diabetic eye disease in an inner city population: the Liverpool Diabetic Eye Study. Eye. 1999;13:160–5.
- Bugawan TL, Klitz W, Alejandrino M, et al. The association of specific HLA class I and II alleles with type 1 diabetes among Filipinos. Tissue Antigens. 2002;596:452–69.
- Burrows NR, Li Y, Geiss LS. Incidence of treatment for end-stage renal disease among individuals with diabetes in the US continues to decline. Diabetes Care. 2010;33(1):73–7.
- 17. Cardiff and Vale University Health Board. Diabetic Retinopathy Screening Service for Wales (DRSSW). http://www.cardiffandvaleuhb.wales.nhs.uk/drssw.
- 18. Centers for Disease Control and Prevention (CDC). National Diabetes Fact Sheet: general information and National estimates on diabetes in the United States, 2005. Atlanta, GA:US Department of Health and Human Services; 2010.
- 19. Centers for Disease Control and Prevention. National Diabetes Fact Sheet: National Estimates and General Information on Diabetes and pre Diabetes in the United States, 2011. Department of health and human services, centers for disease control and prevention. Atlanta; 2011.
- Chase HP, Garg SK, Harris S, et al. Angiotensin converting enzyme inhibitor treatment for young normotensive diabetic subjects: a two-year trial. Ann Ophthalmol. 1993;25:284–9.
- Chaturvedi N, Sjolie AK, Stephenson JM, et al. Effect of lisinopril on progression of retinopathy in normotensive people with type 1 diabetes. The EUCLID Study Group EURODIAB controlled trial of lisinopril in insulin-dependent diabetes mellitus. Lancet. 1998;351:28–31.
- 22. Chaturvedi N, Sjolie AK, Svensson A, DIRECT Programme Study Group. The Diabetic Retinopathy Candesartan Trials (DIRECT) Programme, rationale and study design. J Renin Angiotensin Aldosterone Syst. 2002;3:255–61.
- Chetthakul T, Likitmaskul S, Plengvidhya N, et al. Thailand diabetes registry project: prevalence of diabetic retinopathy and associated factors in type 1 diabetes mellitus. J Med Assoc Thai. 2006;89(Suppl 1):S17–26.
- Chew EY, Ambrosius WT, Davis MD, et al. Effects of medical therapies on retinopathy progression in type 2 diabetes. N Engl J Med. 2010;363(3):233–44.
- 25. Chew EY, Davis MD, Danis RP, Lovato JF, Perdue LH, Greven C, Genuth S, Goff DC, Leiter LA, Ismail-Beigi F, Ambrosius WT, for the Action to Control Cardiovascular Risk in Diabetes Eye Study research group. The effects of medical management on the progression of diabetic retinopathy in persons with type 2 diabetes. The Action to Control Cardiovascular Risk in Diabetes (ACCORD) Eye Study. Ophthalmology. 2014;121:2443–51.
- Chew EY, Klein ML, Ferris III FL, et al. Association of elevated serum lipid levels with retinal hard exudate in diabetic retinopathy. Early Treatment Diabetic Retinopathy Study (ETDRS) Report 22. Arch Ophthalmol. 1996;114:1079–84.

- 27. China faces 'diabetes epidemic', research suggests. BBC, 25 March 2010. Retrieved 8 June 2012.
- Chou C-F, Cotch MF, Vitale S, Zhang X, Klein R, Friedman DS, Klein BEK, Saaddine JB. Age-related eye diseases and visual impairment among U.S. adults. Am J Prev Med. 2013;45(1):29–35.
- 29. Ciulla TA, Amador AG, Zinman B. Diabetic retinopathy and diabetic macular edema. Diabetes Care. 2003;26(9):2653–64.
- 30. Colagiuri R, Dain K, Moylan J. The global response to diabetes: action or apathy? Med J Aust. 2014;201(10):581–3.
- Colhoun HM, Betteridge DJ, Durrington PN, et al. CARDS investigators. Primary prevention of cardiovascular disease with atorvastatin in type 2 diabetes in the Collaborative Atorvastatin Diabetes Study (CARDS): multicentre randomized placebo controlled trial. Lancet. 2004;364:685–96.
- 32. Collins R, Armitage J, Parish S, Sleigh P, Peto R, Heart Protection Study Collaborative Group. MRC/BHF Heart Protection Study of cholesterol-lowering with simvastatin in 5963 people with diabetes: a randomised placebo-controlled trial. Lancet. 2003;361:2005–16.
- 33. Congdon N, O'Colmain B, Klaver CC. et al.; Eye Diseases Prevalence Research Group. Causes and prevalence of visual impairment among adults in the United States. Arch Ophthalmol. 2004;122(4):477–85.
- Conway BN, Miller RG, Orchard TJ. Are hemoglobin levels elevated in type 1 diabetes? Diabetes Care. 2010;33:341–3.
- 35. Cugati S, Cikamatana L, Wang JJ, et al. Five-year incidence and progression of vascular retinopathy in persons without diabetes: the Blue Mountains Eye Study. Eye. 2006;20:1239–45.
- De Block CE, De Leeuw IH, Van Gaal LF. Impact of overweight on chronic microvascular complications in type 1 diabetic patients. Diabetes Care. 2005;28:1649–55.
- 37. Diabetes can be controlled in 80 percent of Cases in India. IANS. www.news.biharprabha. com Retrieved 6 Feb 2014.
- The Diabetes Control and Complications Trial Research Group. The effect of intensive treatment of diabetes on the development and progression of long-term complications in insulindependent diabetes mellitus. N Engl J Med. 1993;329:977–86.
- 39. Diabetes Control and Complications Trial/Epidemiology of Diabetes Interventions and Complications Research Group. Retinopathy and nephropathy in patients with type 1 diabetes four years after a trial of intensive therapy. N Engl J Med. 2000;342:381–9.
- 40. The Diabetes Control and Complications Trial (DCCT)/Epidemiology of Diabetes Interventions and Complications (EDIC) Research Group. Effect of intensive diabetes therapy on the progression of diabetic retinopathy in patients with type 1 diabetes: 18 years of follow-up in the DCCT/EDIC. Diabetes. 2015;64:631–42.
- Dinneen SF, Gerstein HC. The association of microalbuminuria and mortality in non-insulindependent diabetes mellitus. A systematic overview of the literature. Arch Intern Med. 1997;157:1413–8.
- 42. Donaghue KC, Fairchild JM, Craig ME, et al. Do all prepubertal years of diabetes duration contribute equally to diabetes complications? Diabetes Care. 2003;26:1224–9.
- 43. Doria A. Genetics of diabetes complications. Curr Diab Rep. 2010;10:467-75.
- Edelstein SL, Knowler WC, Bain RP, et al. Predictors of progression from impaired glucose tolerance to NIDDM: an analysis of 6 prospective studies. Diabetes. 1997;46(4):701–10.
- 45. Emdin CA, Rahimi K, Neal B, Callender T, Perkovic V, Patel A. Blood pressure lowering in type 2 diabetes. A systematic review and meta-analysis. JAMA. 2015;313(6):603–15.
- 46. Eschwege E, Basdevant A, Crine A, Moisan C, Charles M-A. Type 2 diabetes mellitus in France in 2012: results from the ObEpi survey. Diabetes Metab. 2015;41:55–61.
- 47. Ferris III FL. How effective are treatments for diabetic retinopathy? JAMA. 1993;269(10):1290-1.
- Ferris III FL, Patz A. Macular edema: a complication of diabetic retinopathy. Surv Ophthalmol. 1984;28(Suppl):452–61.

- 49. Flack A, Kaar ML, Laatikainen L. A prospective, longitudinal study examining the development of retinopathy in children with diabetes. Acta Paediatr. 1996;85:313–9.
- Flegal KM, Carroll MD, Kit BK, Ogden CL. Prevalence of obesity and trends in the distribution of body mass index among US adults, 1999-2010. JAMA. 2012;307(5):491–7.
- 51. Fong DS, Aiello LP, Ferris 3rd FL, et al. Diabetic retinopathy. Diabetes Care. 2004;27:2540–53.
- Fong DS, Aiello L, Gardner TW, et al. American Diabetes Association Retinopathy in diabetes. Diabetes Care 2004;27(suppl 1):S84–7.
- Ford ES, Dietz WH. Trends in energy intake among adults in the United States: findings from NHANES. Am J Clin Nutr. 2013;97(4):848–53.
- 54. Foster A, Resnikoff S. The impact of Vision 2020 on global blindness. Eye (Lond). 2005;19:1133–5.
- 55. Fox CS, Pencina MJ, Meigs JB, Vasan RS, Levitzky YS, D'Agostino Sr RB. Trends in the incidence of type 2 diabetes mellitus from the 1970s to the 1990s: the Framingham Heart Study. Circulation. 2006;113(25):2914–8.
- Gale J. (2010). India's diabetes epidemic cuts down millions who escape poverty. Bloomberg. Retrieved 8 June 2012.
- Geiss LS, Wang J, Cheng YJ, Thompson TJ, Barker L, Li Y, Albright AN, Gregg E. Prevalence and incidence trends for diagnosed diabetes among adults aged 20 to 79 years, United States, 1980-2012. JAMA. 2014;312(12):1218–26.
- Goldzweig CL, Rowe S, Wenger NS, et al. Preventing and managing visual disability in primary care: clinical applications. JAMA. 2004;291(12):1497–502.
- Gordon B, Chang S, Kavanagh M, et al. The effects of lipid lowering on diabetic retinopathy. Am J Ophthalmol. 1991;112:385–91.
- 60. Haffner SM, Hazuda HP, Stern MP, et al. Effects of socioeconomic status on hyperglycemia and retinopathy levels in Mexican Americans with NIDDM. Diabetes Care. 1989;12:128–34.
- Hanis CL, Chu HH, Lawson K, et al. Mortality of Mexican Americans with NIDDM. Retinopathy and other predictors in Starr County, Texas. Diabetes Care. 1993;16:82–9.
- 62. Harjo TC, Perez A, Lopez V, et al. Prevalence of diabetes and cardiovascular risk factors among California Native American adults compared to other ethnicities: the 2005 California Health Interview Survey. Metab Syndr Relat Disord. 2011;9:49–54.
- 63. Harris MI, Klein R, Cowie CC, Rowland M, Byrd-Holt DD. Is the risk of diabetic retinopathy greater in non-Hispanic blacks and Mexican Americans than in non-Hispanic whites with type 2 diabetes? a US population study. Diabetes Care. 1998;21(8):1230–5.
- 64. Hietala K, Harjutsalo V, Forsblom C, et al. Age at onset and the risk of proliferative retinopathy in type 1 diabetes. Diabetes Care. 2010;33:1315–9.
- Hofman KJ, Cook C, Levitt N. Preventing diabetic blindness: a priority for South Africa. S Afr Med J. 2014;104(10):661–2.
- 66. Hooper P, Boucher MC, Colleaux K, Cruess A, Greve M, Lam WC, Shortt S, Tourville E. Contemporary management of diabetic retinopathy in Canada: from guidelines to algorithm guidance. Ophthalmologica. 2014;231(1):2–15.
- Hovind P, Tarnow L, Rossing K, et al. Decreasing incidence of severe diabetic microangiopathy in type 1 diabetes. Diabetes Care. 2003;26(4):1258–64.
- Hu FB, Manson JE, Stampfer MJ, et al. Diet, lifestyle, and the risk of type 2 diabetes mellitus in women. N Engl J Med. 2001;345(11):790–7.
- 69. Hu FB, Sigal RJ, Rich-Edwards JW, et al. Walking compared with vigorous physical activity and risk of type 2 diabetes in women: a prospective study. JAMA. 1999;282(15):1433–9.
- Hypertension in Diabetes Study (HDS). Prevalence of hypertension in newly presenting type 2 diabetic patients and the association with risk factors for cardiovascular and diabetic complications. J Hypertens. 1993;11(3):309–17.
- Ikegami H, Noso S, Babaya N, et al. Genetic basis of type 1 diabetes: similarities and differences between east and west. Rev Diabet Stud. 2008;5:64–72.

- 72. International Diabetes Federation. IDF diabetes atlas. 6th ed. Brussels: IDF; 2013. http:// www.idf.org/sites/default/files/EN_6E_Atlas_Full_0.pdf. Accessed Aug 2014.
- Irvine KM, Gallego P, An X, et al. Peripheral blood monocyte gene expression profile clinically stratifies patients with recent-onset type 1 diabetes. Diabetes. 2012;615:1281–90.
- 74. Ismail-Beigi F, Craven T, Banerji MA, et al. Effect of intensive treatment of hyperglycaemia on microvascular outcomes in type 2 diabetes: an analysis of the ACCORD randomised trial. Lancet. 2010;376:419–30.
- Jain A, Sarraf D, Fong D. Preventing diabetic retinopathy through control of systemic factors. Curr Opin Ophthalmol. 2003;14(6):389–94.
- Jeon CY, Lokken RP, Hu FB, van Dam RM. Physical activity of moderate intensity and risk of type 2 diabetes: a systematic review. Diabetes Care. 2007;30(3):744–52.
- Jha V, Garcia-Garcia G, Iseki K, et al. Chronic kidney disease: global dimension and perspectives. Lancet. 2013;382(9888):260–72.
- 78. Jonas JB, Xu L, Wang YX. The Beijing eye study. Acta Ophthalmol. 2009;87:247-61.
- Keech AC, Mitchell P, Summanen PA, et al. FIELD study investigators. Effect of fenofibrate on the need for laser treatment for diabetic retinopathy (FIELD study): a randomised controlled trial. Lancet. 2007;370:1687–97.
- Keenan TDL, Johnston RL, Donachie PHJ, Sparrow JM, Stratton IM, Scanlon P. United Kingdom National Ophthalmology Database Study: diabetic retinopathy; Report 1: prevalence of centre-involving diabetic macular oedema and other grades of maculopathy and retinopathy in hospital eye services. Eye. 2013;27:1397–404.
- Kempen JH, O'Colmain BJ, Leske MC, et al. The prevalence of diabetic retinopathy among adults in the United States. Arch Ophthalmol. 2004;122(4):552–63.
- 82. Klein R. Epidemiology of diabetic retinopathy. In: Duh E, editor. Diabetic retinopathy. Totowa: Humana Press; 2008.
- Klein R, Klein BE. Are individuals with diabetes seeing better?: a long-term epidemiological perspective. Diabetes. 2010;59:1853–60.
- Klein R, Klein BE, Moss SE, Cruickshanks KJ. Association of ocular disease and mortality in a diabetic population. Arch Ophthalmol. 1999;117:1487–95.
- Klein R, Klein BE, Moss SE, Davis MD, DeMets DL. The Wisconsin epidemiologic study of diabetic retinopathy II: prevalence and risk of diabetic retinopathy when age at diagnosis is less than 30 years. Arch Ophthalmol. 1984;102(4):520–6.
- Klein R, Klein BE, Moss SE, Davis MD, DeMets DL. The Wisconsin epidemiologic study of diabetic retinopathy. III. Prevalence and risk of diabetic retinopathy when age at diagnosis is 30 or more years. Arch Ophthalmol. 1984;102:527–32.
- Klein R, Klein BE, Moss SE, et al. The Wisconsin epidemiologic study of diabetic retinopathy VI. Retinal photocoagulation. Ophthalmology. 1987;94(7):747.
- Klein R, Klein BE, Moss SE, Linton KL. The Beaver Dam Eye Study. Retinopathy in adults with newly discovered and previously diagnosed diabetes mellitus. Ophthalmology. 1992;99:58–62.
- Klein R, Klein BEK, Moss SE, et al. The Wisconsin epidemiologic study of diabetic retinopathy, XV: the long term incidence of macular oedema. Ophthalmology. 1995;102:7–16.
- Klein R, Klein BE, Moss SE, et al. The Wisconsin epidemiologic study of diabetic retinopathy. XVII. The 14-year incidence and progression of diabetic retinopathy and associated risk factors in type 1 diabetes. Ophthalmology. 1998;105:1801–15.
- Klein R, Lee KE, Gangnon RE, Klein BE. Relation of smoking, drinking, and physical activity to changes in vision over a 20-year period: the Beaver Dam Eye Study. Ophthalmology. 2014;121(6):1220–8.
- 92. Klein R, Lee KE, Knudtson MD, Gangnon RE, Klein BE. Changes in visual impairment prevalence by period of diagnosis of diabetes: the Wisconsin Epidemiologic Study of Diabetic Retinopathy. Ophthalmology. 2009;116(10):1937–42.
- Ko GT, Chan JC, Lau M, et al. Diabetic microangiopathic complications in young Chinese diabetic patients: a clinic based cross-sectional study. J Diabetes Complications. 1999;13:300–6.

- 94. Kramer CK, Leitão CB, Canani LH, et al. Late afternoon blood pressure increase is associated with diabetic retinopathy in normotensive type 2 diabetes mellitus patients. Diabetes Res Clin Pract. 2009;84:e12–4.
- Kramer CK, Rodrigues TC, Canani LH, Gross JL, Azevedo MJ. Diabetic retinopathy predicts all-cause mortality and cardiovascular events in both type 1 and 2 diabetes. Diabetes Care. 2011;34:1238–44.
- 96. Larsen M, Hommel E, Parving HH, et al. Protective effect of captopril on the blood-retina barrier in normotensive insulin dependent diabetic patients with nephropathy and background retinopathy. Graefes Arch Clin Exp Ophthalmol. 1990;228:505–9.
- 97. Lee ET, Welty TK, Cowan LD, et al. Incidence of diabetes in American Indians of 3 geographic areas: the Strong Heart Study. Diabetes Care. 2002;25(1):49–54.
- 98. Lim SS, Vos T, Flaxman AD, Danaei G, Shibuya K, Adair-Rohani H, Amann M, Anderson HR, Andrews KG, Aryee M, Atkinson C, Bacchus LJ, Bahalim AN, Balakrishnan K, Balmes J, Barker-Collo S, Baxter A, Bell ML, Blore JD, Blyth F, Bonner C, Borges G, Bourne R, Boussinesq M, Brauer M, Brooks P, Bruce NG, Brunekreef B, Bryan-Hancock C, Bucello C, Buchbinder R, Bull F, Burnett RT, Byers TE, Calabria B, Carapetis J, Carnahan E, Chafe Z, Charlson F, Chen H, Chen JS, Cheng AT, Child JC, Cohen A, Colson KE, Cowie BC, Darby S, Darling S, Davis A, Degenhardt L, Dentener F, Des Jarlais DC, Devries K, Dherani M, Ding EL, Dorsey ER, Driscoll T, Edmond K, Ali SE, Engell RE, Erwin PJ, Fahimi S, Falder G, Farzadfar F, Ferrari A, Finucane MM, Flaxman S, Fowkes FG, Freedman G, Freeman MK, Gakidou E, Ghosh S, Giovannucci E, Gmel G, Graham K, Grainger R, Grant B, Gunnell D, Gutierrez HR, Hall W, Hoek HW, Hogan A, Hosgood HD 3rd, Hoy D, Hu H, Hubbell BJ, Hutchings SJ, Ibeanusi SE, Jacklyn GL, Jasrasaria R, Jonas JB, Kan H, Kanis JA, Kassebaum N, Kawakami N, Khang YH, Khatibzadeh S, Khoo JP, Kok C, Laden F, Lalloo R, Lan Q, Lathlean T, Leasher JL, Leigh J, Li Y, Lin JK, Lipshultz SE, London S, Lozano R, Lu Y, Mak J, Malekzadeh R, Mallinger L, Marcenes W, March L, Marks R, Martin R, McGale P, McGrath J, Mehta S, Mensah GA, Merriman TR, Micha R, Michaud C, Mishra V, Mohd Hanafiah K, Mokdad AA, Morawska L, Mozaffarian D, Murphy T, Naghavi M, Neal B, Nelson PK, Nolla JM, Norman R, Olives C, Omer SB, Orchard J, Osborne R, Ostro B, Page A, Pandey KD, Parry CD, Passmore E, Patra J, Pearce N, Pelizzari PM, Petzold M, Phillips MR, Pope D, Pope CA 3rd, Powles J, Rao M, Razavi H, Rehfuess EA, Rehm JT, Ritz B, Rivara FP, Roberts T, Robinson C, Rodriguez-Portales JA, Romieu I, Room R, Rosenfeld LC, Roy A, Rushton L, Salomon JA, Sampson U, Sanchez-Riera L, Sanman E, Sapkota A, Seedat S, Shi P, Shield K, Shivakoti R, Singh GM, Sleet DA, Smith E, Smith KR, Stapelberg NJ, Steenland K, Stöckl H, Stovner LJ, Straif K, Straney L, Thurston GD, Tran JH, Van Dingenen R, van Donkelaar A, Veerman JL, Vijayakumar L, Weintraub R, Weissman MM, White RA, Whiteford H, Wiersma ST, Wilkinson JD, Williams HC, Williams W, Wilson N, Woolf AD, Yip P, Zielinski JM, Lopez AD, Murray CJ, Ezzati M, AlMazroa MA, Memish ZA. A comparative risk assessment of burden of disease and injury attributable to 67 risk factors and risk factor clusters in 21 regions, 1990-2010: a systematic analysis for the Global Burden of Disease Study 2010. Lancet. 2012;380(9859):2224-60.
- 99. Man RE, Lamoureux EL, Taouk Y, Xie J, Sasongko MB, Best WJ, Noonan JE, Kawasaki R, Wang JJ, Luu CD. Axial length, retinal function, and oxygen consumption: a potential mechanism for a lower risk of diabetic retinopathy in longer eyes. Invest Ophthalmol Vis Sci. 2013;54(12):7691–8.
- Martin-Merino E, Fortuny J, Rivero-Ferrer E, Garcia-Rodriguez LA. Incidence of retinal complications in a cohort of newly diagnosed diabetic patients. PLoS One. 2014;9(6):e100283.
- 101. Martín-Merino E, Fortuny J, Rivero-Ferrer E, Lind M, Garcia-Rodriguez LA. Risk factors for diabetic retinopathy in people with Type 2 diabetes: A case-control study in a UK primary care setting. Prim Care Diabetes. 2016;10(4):300–8.
- 102. Mauer M, Zinman B, Gardiner R, et al. Renal and retinal effects of enalapril and losartan in type 1 diabetes. N Engl J Med. 2009;361:40–51.
- 103. McCance DR, Hadden DR, Atkinson AB, et al. Long-term glycaemic control and diabetic retinopathy. Lancet. 1989;2:824–8.

- 104. McCarty CA, Taylor KI, McKay R, et al. Diabetic retinopathy: effects of national guidelines on the referral, examination and treatment practices of ophthalmologists and optometrists. Clin Experiment Ophthalmol. 2001;29:52–8.
- 105. McKean-Cowdin R, Varma R, Wu J, Hays RD, Azen SP. Los Angeles Latino Eye Study Group. Severity of visual field loss and health-related quality of life. Am J Ophthalmol. 2007;143(6):1013–23.
- 106. McKean-Cowdin R, Wang Y, Wu J, Azen SP, Varma R. Los Angeles Latino eye study group. Impact of visual field loss on health-related quality of life in glaucoma: the Los Angeles Latino Eye Study. Ophthalmology. 2008;115(6):941–8.
- 107. McLeod BK, Thompson JR, Rosenthal AR. The prevalence of retinopathy in the insulinrequiring diabetic patients of an English town. Eye. 1988;2:424–30.
- 108. Mendes AB, Fittipaldi JA, Neves RC, et al. Prevalence and correlates of inadequate glycaemic control: results from a nationwide survey in 6,671 adults with diabetes in Brazil. Acta Diabetol. 2010;47:137–45.
- 109. Mitchell P. Development and progression of diabetic eye disease in Newcastle (1977-1984): rates and risk factors. Aust NZ J Ophthalmol. 1985;13:39–44.
- 110. Mokdad AH, Bowman BA, Ford ES, Vinicor F, Marks JS, Koplan JP. The continuing epidemics of obesity and diabetes in the United States. JAMA. 2001;286(10):1195–200.
- 111. Morgan CL, Currie CJ, Stott NC, Smithers M, Butler CC, Peters JR. The prevalence of multiple diabetes-related complications. Diabet Med. 2000;17:146–51.
- 112. Mosaad YM, Auf FA, Metwally SS, et al. HLA-DQB1* alleles and genetic susceptibility to type 1 diabetes mellitus. World J Diabetes. 2012;38:149–55.
- 113. Moussa MA, Alsaeid M, Abdella N, et al. Prevalence of type 1 diabetes among 6- to 18-yearold Kuwaiti children. Med Princ Pract. 2005;14:87–91.
- 114. Muller A, Keeffe JE, Taylor HR. Changes in eye care utilization following an eye health promotion campaign. Clin Experiment Ophthalmol. 2007;35:305–9.
- Namperumalsamy P, Kim R, Vignesh TP, et al. Prevalence and risk factors for diabetic retinopathy: a population based assessment from Theni District, south India. Br J Ophthalmol. 2009;93:429–34.
- Narayan KMV, Boyle JP, Geiss LS, Saaddine JB, Thompson TJ. Impact of recent increase in incidence on future diabetes burden: U.S., 2005–2050. Diabetes Care. 2006;29(9):2114–6.
- Narayan KM, Boyle JP, Thompson TJ, Sorensen SW, Williamson DF. Lifetime risk for diabetes mellitus in the United States. JAMA. 2003;290(14):1884–90.
- 118. National Eye Institute. Vision loss from eye diseases will increase as Americans age. Available at http://www.nei.nih.gov/news/pressreleases/041204.asp. Accessed 8 Nov 2012.
- 119. National Institute on Aging, "Why population aging matters: a global perspective" (National Institutes of Health and U.S. Department of State, publication no. 07–6134, 2007).
- 120. National Society to Prevent Blindness. Operational research department. Vision problems in the US: a statistical analysis. New York: National Society to Prevent Blindness; 1980. p. 146.
- 121. Newsom R, Moate B, Casswell T. Screening for diabetic retinopathy using digital colour photography and oral fluorescein angiography. Eye. 2000;14:579–82.
- 122. Ng M, Fleming T, Robinson M, Thomson B, Graetz N, Margono C, Mullany EC, Biryukov S, Abbafati C, Abera SF, Abraham JP, Abu-Rmeileh NM, Achoki T, AlBuhairan FS, Alemu ZA, Alfonso R, Ali MK, Ali R, Guzman NA, Ammar W, Anwari P, Banerjee A, Barquera S, Basu S, Bennett DA, Bhutta Z, Blore J, Cabral N, Nonato IC, Chang JC, Chowdhury R, Courville KJ, Criqui MH, Cundiff DK, Dabhadkar KC, Dandona L, Davis A, Dayama A, Dharmaratne SD, Ding EL, Durrani AM, Esteghamati A, Farzadfar F, Fay DF, Feigin VL, Flaxman A, Forouzanfar MH, Goto A, Green MA, Gupta R, Hafezi-Nejad N, Hankey GJ, Harewood HC, Havmoeller R, Hay S, Hernandez L, Husseini A, Idrisov BT, Ikeda N, Islami F, Jahangir E, Jassal SK, Jee SH, Jeffreys M, Jonas JB, Kabagambe EK, Khalifa SE, Kengne AP, Khader YS, Khang YH, Kim D, Kimokoti RW, Kinge JM, Kokubo Y, Kosen S, Kwan G, Lai T, Leinsalu M, Li Y, Liang X, Liu S, Logroscino G, Lotufo PA, Lu Y, Ma J, Mainoo NK, Mensah GA, Merriman TR, Mokdad AH, Moschandreas J, Naghavi M, Naheed A, Nand D,
Narayan KM, Nelson EL, Neuhouser ML, Nisar MI, Ohkubo T, Oti SO, Pedroza A, Prabhakaran D, Roy N, Sampson U, Seo H, Sepanlou SG, Shibuya K, Shiri R, Shiue I, Singh GM, Singh JA, Skirbekk V, Stapelberg NJ, Sturua L, Sykes BL, Tobias M, Tran BX, Trasande L, Toyoshima H, van de Vijver S, Vasankari TJ, Veerman JL, Velasquez-Melendez G, Vlassov VV, Vollset SE, Vos T, Wang C, Wang X, Weiderpass E, Werdecker A, Wright JL, Yang YC, Yatsuya H, Yoon J, Yoon SJ, Zhao Y, Zhou M, Zhu S, Lopez AD, Murray CJ, Gakidou E. Global, regional, and national prevalence of overweight and obesity in children and adults during 1980-2013: a systematic analysis for the Global Burden of Disease Study 2013. Lancet. 2014;384(9945):766–81.

- 123. Ng SW, Slining MM, Popkin BM. Turning point for US diets? Recessionary effects or behavioral shifts in foods purchased and consumed. Am J Clin Nutr. 2014;99(3):609–16.
- 124. NHS Scotland. NHS Scotland National Diabetes Retinopathy Screening. http://www.ndrs. scot.nhs.uk.
- 125. Noble JA, Valdes AM. Genetics of the HLA region in the prediction of type 1 diabetes. Curr Diab Rep. 2011;116:533–42.
- 126. Nordwall M, Bojestig M, Arnqvist HJ, Ludvigsson J. Linköping Diabetes Complications Study. Declining incidence of severe retinopathy and persisting decrease of nephropathy in an unselected population of type 1 diabetes-the Linköping Diabetes Complications Study. Diabetologia. 2004;47(7):1266–72.
- 127. Nsiah-Kumi P, Ortmeier SR, Brown AE. Disparities in diabetic retinopathy screening and disease for racial and ethnic minority populations—a literature review. J Natl Med Assoc. 2009;101(5):430–7.
- 128. Ogden CL, Carroll MD, Kit BK, Flegal KM. Prevalence of obesity among adults: United States, 2011-2012. NCHS Data Brief. 2013;131:1–8.
- 129. Park Y, Eisenbarth GS. Genetic susceptibility factors of Type 1 diabetes in Asians. Diabetes Metab Res Rev. 2001;171:2–11.
- 130. Patz A, Smith RE. The ETDRS and diabetes 2000. Ophthalmology. 1991;98:739-40.
- 131. Pechere-Bertschi A, Greminger P, Hess L, Philippe J, Ferrari P. Swiss Hypertension and Risk Factor Program (SHARP): cardiovascular risk factors management in patients with type 2 diabetes in Switzerland. Blood Press. 2005;14(6):337–44.
- 132. Pedro RA, Ramon SA, Marc BB, et al. Prevalence and relationship between diabetic retinopathy and nephropathy, and its risk factors in the North-East of Spain, a population based study. Ophthalmic Epidemiol. 2010;17:251–65.
- 133. Prasad S, Kamath GG, Jones K, Clearkin LG, Phillips RP. Prevalence of blindness and visual impairment in a population of people with diabetes. Eye. 2001;15:640–3.
- 134. Prevent Blindness America. The economic impact of vision problems. www.preventblindness.org/sites/default/files/national/documents/EI_introduction.pdf.
- 135. Public Health Agency, Belfast: Northern Ireland Diabetic Retinopathy Screening Programme. http://www.publichealth.hscni.net/publications/northernireland-diabetic-retinopathy-screening-programme.
- 136. Qiu Y-H, Deng F-Y, Li M-J, Lei S-F. Identification of novel risk genes associated with type 1 diabetes mellitus using a genome-wide gene-based association analysis. J Diabetes Invest. 2014;5:649–56.
- 137. Raman R, Rani PK, Reddi RS, et al. Prevalence of diabetic retinopathy in India: Sankara Nethralaya diabetic retinopathy epidemiology and molecular genetics study report 2. Ophthalmology. 2009;116:311–8.
- Raymond NT, Varadhan L, Reynold DR, et al. Higher prevalence of retinopathy in diabetic patients of South Asian ethnicity compared with white Europeans in the community: a crosssectional study. Diabetes Care. 2009;32:410–5.
- 139. Rema M, Premkumar S, Anitha B, et al. Prevalence of diabetic retinopathy in urban India: the Chennai Urban Rural Epidemiology Study (CURES) eye 3. Invest Ophthalmol Vis Sci. 2005;46:2328–33.
- 140. Roy MS, Janal MN. High caloric and sodium intakes as risk factors for progression of retinopathy in type 1 diabetes mellitus. Arch Ophthalmol. 2010;128:33–9.

- 141. Scanlon PH. The English national screening programme for sight-threatening diabetic retinopathy. J Med Screen. 2008;15:1–4.
- 142. Schmidt MI, Duncan BB, Azevedo e Silva G, et al. Chronic non-communicable diseases in Brazil: burden and current challenges. Lancet. 2011;377(9781):1949–61.
- 143. Schrier RW, Estacio RO, Mehler PS, et al. Appropriate blood pressure control in hypertensive and normotensive type 2 diabetes mellitus: a summary of the ABCD trial. Nat Clin Pract Nephrol. 2007;3:428–38.
- 144. Schwartz JA, Lantis II JC, Gendics C, Fuller AM, Payne W, Ochs D. A prospective, noncomparative, multicenter study to investigate the effect of cadexomer iodine on bioburden load and other wound characteristics in diabetic foot ulcers. Int Wound J. 2012; doi:10.1111/j.1742-481X.2012.01109.x.
- 145. Sepulveda J, Murray C. The state of global health in 2014. Science. 2014;345(6202): 1275-8.
- 146. Shaw JE, Sicree RA, Zimmet PZ. Global estimates of the prevalence of diabetes for 2010 and 2030. Diabetes Res Clin Pract. 2010;87:4–14.
- 147. Sheehy MJ, Scharf SJ, Rowe JR, et al. A diabetes-susceptible HLA haplotype is best defined by a combination of HLADR and -DQ alleles. J Clin Invest. 1989;833:830–5.
- 148. Sivaprasad S, Gupta B, Crosby-Nwaobi R, Evans J. Prevalence of diabetic retinopathy in various ethnic groups: a worldwide perspective. Surv Ophthalmol. 2012;57(4):347–70.
- 149. Soedamah-Muthu SS, Chaturvedi N, Witte DR, Stevens LK, Porta M, Fuller JH, EURODIAB Prospective Complications Study Group. Relationship between risk factors and mortality in type 1 diabetic patients in Europe: the EURODIAB Prospective Complications Study (PCS). Diabetes Care. 2008;31:1360–6.
- 150. Sparrow JM, McLeod BK, Smith TD, Birch MK, Rosenthal AR. The prevalence of retinopathy and maculopathy and their risk factors in the non-insulin treated diabetic patients of an English town. Eye. 1993;7:158–63.
- 151. Stratton IM, Kohner EM, Aldington SJ, et al. UK-PDS 50: risk factors for incidence and progression of retinopathy in type II diabetes over 6 years from diagnosis. Diabetologia. 2001;44(2):156–63.
- 152. Stuckler D, Basu S, McKee M. Drivers of inequality in Millennium Development Goal progress: a statistical analysis. PLoS Med. 2010;7:e1000241.
- 153. Sugihara S. Genetic susceptibility of childhood type 1 diabetes mellitus in Japan. Pediatr Endocrinol Rev. 2012;10(Suppl. 1):62–71.
- 154. Tao Z, Shi A. Zhao. Epidemiological perspectives of diabetes. Cell Biochem Biophys. 2015;73(1):181–5.
- 155. Prospective UK. Diabetes Study Group. Prospective UK diabetes study 16. Overview of 6 years' therapy of type II diabetes: a progressive disease. Diabetes. 1995;44:1249–58.
- 156. UK Prospective Diabetes Study (UKPDS) Group. Tight blood pressure control and risk of macrovascular and microvascular complications in type 2 diabetes: UKPDS 38. BMJ. 1998;317:703–13.
- 157. Van DS, Beulens JW, van der Schouw YT, et al. The global burden of diabetes and its complications: an emerging pandemic. Eur J Cardiovasc Prev Rehabil. 2010;17(Suppl 1):S3–8.
- 158. van Hecke MV, Dekker JM, Stehouwer CD, et al. EURODIAB prospective complications study. Diabetic retinopathy is associated with mortality and cardiovascular disease incidence: the EURODIAB prospective complications study. Diabetes Care. 2005;28:1383–9.
- 159. Varma R, Chung J, Foong AW, et al. Four-year incidence and progression of visual impairment in Latinos: the Los Angeles Latino Eye Study. Am J Ophthalmol. 2010;149:713–27.
- 160. Vazquez G, Duval S, Jacobs Jr DR, Silventoinen K. Comparison of body mass index, waist circumference, and waist/hip ratio in predicting incident diabetes: a meta-analysis. Epidemiol Rev. 2007;29:115–28.
- 161. Vu HT, Keeffe JE, McCarty CA, Taylor HR. Impact of unilateral and bilateral vision loss on quality of life. Br J Ophthalmol. 2005;89(3):360–3.
- 162. Wang H, Dwyer-Lindgren L, Lofgren KT, Rajaratnam JK, Marcus JR, Levin-Rector A, Levitz CE, Lopez AD, Murray CJ. Age-specific and sex-specific mortality in 187 countries, 1970-2010: a systematic analysis for the Global Burden of Disease Study 2010. Lancet. 2012;380(9859):2071–94.

- 163. Wessel MM, Nair N, Aaker GD, Ehrlich JR, D'Amico DJ, Kiss S. Peripheral retinal ischaemia, as evaluated by ultra-widefield fluorescein angiography, is associated with diabetic macular oedema. Br J Ophthalmol. 2012;96:694–8.
- 164. Wild S, Roglic G, Green A, et al. Global prevalence of diabetes: estimates for the year 2000 and projections for 2030. Diabetes Care. 2004;27:1047–53.
- 165. Wong TY, Cheung N, Tay WT, et al. Prevalence and risk factors for diabetic retinopathy: the Singapore Malay Eye Study. Ophthalmology. 2008;115:1869–75.
- 166. Wong TY, Klein R, Islam FM, et al. Diabetic retinopathy in a multi-ethnic cohort in the United States. Am J Ophthalmol. 2006;141(3):446–55.
- 167. Wong TY, Simo R, Mitchell P. Fenofibrate—a potential systemic treatment for diabetic retinopathy? Am J Ophthalmol. 2012;154(1):6–12.
- World Economic Forum. Global risks 2010: a Global Risk Network report. Geneva: WEF. 2010. http://www3.weforum.org/docs/WEF_GlobalRisks Report_2010.pdf. Accessed Aug 2014.
- 169. World Health Organization. Fact sheet No. 94. Malaria. Geneva: WHO, 2014. http://www. who.int/mediacentre/factsheets/fs094/en. Accessed Aug 2014.
- 170. World Health Organization. Fact sheet No. 104. Tuberculosis. Geneva: WHO, 2014. http:// www.who.int/mediacentre/factsheets/fs104/en. Accessed Aug 2014.
- 171. World Health Organization. Number of deaths due to HIV/AIDS. Geneva: WHO, 2012. http://www.who.int/gho/hiv/epidemic_status/deaths_text/en. Accessed Aug 2014.
- 172. World Health Organization. WHO launches Vision 2020 to combat avoidable blindness. Public Health Rep. 1999;114:210.
- 173. Wright AD, Dodson PM. Diabetic retinopathy and blockade of the renin-angiotensin system: new data from the DIRECT study programme. Eye (Lond). 2010;24:1–6.
- 174. Wright AD, Dodson PM. Medical management of diabetic retinopathy: fenofibrate and ACCORD Eye studies. Eye (Lond). 2011;25(7):843–9.
- 175. Yamamoto T, Iimuro S, Ohashi Y, Sone H, Yamashita H, Ito H. Japanese Elderly Intervention Trial Research Group. Prevalence and risk factors for diabetic maculopathy, and its relationship to diabetic retinopathy in elderly Japanese patients with type 2 diabetes mellitus. Geriatr Gerontol Int. 2012;12:134–40.
- 176. Yang W, Lu J, Weng J, et al. China National Diabetes and Metabolic Disorders Study Group. Prevalence of diabetes among men and women in China. N Engl J Med. 2010;362: 1090–101.
- 177. Yau JWY, Rogers SL, Kawsaki R, Lamoureux EL, Kowalski JW, Bek T, Chen S-J, Dekker JM, Fletcher A, Grauslund J, Haffner S, Hamman RF, Ikram MK, Kayama T, Klein BEK, Klein R, Krishnaiah S, Mayurasakorn K, O'Hare JP, Orchard TJ, Porta M, Rema M, Roy M, Sharma T, Shaw J, Taylor H, Tielsch JM, Varma R, Wang JJ, Wang N, West S, Xu L, Yasuda M, Zhang X, Mitchell P, Wong TY, for the Meta-Analysis for Eye Disease (META-EYE Study Group. Global prevalence and major risk factors of diabetic retinopathy. Diabetes Care. 2012;35:556–64.
- 178. Younis N, Broadbent DM, Harding SP, Vora JR. Prevalence of diabetic eye disease in patients entering a systematic primary care-based eye screening programme. Diabet Med. 2002;19:1014–21.
- 179. Yuen A, Sugeng Y, Weiland TJ, Jelinek GA. Lifestyle and medication interventions for the prevention or delay of type 2 diabetes mellitus in prediabetes: a systematic review of randomised controlled trials. Aust N Z J Public Health. 2010;34:172–8.
- 180. Zhang H, Xia W, Yu Q, et al. Increasing incidence of type 1 diabetes in children aged 0–14 years in Harbin, China (1990–2000). Prim Care Diabetes. 2008;2:121–6.
- 181. Zhang X, Saaddine JB, Chou C-F, Cotch MF, Cheng YJ, Geiss LS, Gregg EW, Albright AL, Klein BEK, Klein R. Prevalence of diabetic retinopathy in the United States, 2005-2008. JAMA. 2010;304(6):649–56.
- 182. Zhang X, Saaddine JB, Lee PP, et al. Eye care in the United States: do we deliver to high-risk people who can benefit most from it? Arch Ophthalmol. 2007;125(3):411–8.
- 183. Zheng Y, He M, Congdon N. The worldwide epidemic of diabetic retinopathy. Indian J Ophthalmol. 2012;60(5):428–31.

Chapter 2 The Diabetic Retina: Anatomy and Pathophysiology

2.1 Introduction

Throughout the world diabetic retinopathy (DR) has emerged as a major cause of permanent loss of vision among people over the age of 20 years. Retinopathy has generally been considered a vasculopathy that results from breakdown of the blood-retinal barrier and closure of retinal capillaries. Recent evidence, however, suggests that DR begins as a neuro-retinopathy with vascular changes occurring later during the disease.

Knowledge of retinal anatomy and pathophysiology are important to surgeons so that diagnoses and disease characterizations can be made accurately. Additionally, a detailed understanding of vitreoretinal anatomy enables surgeons to correctly deliver intravitreal depot injections of pharmaceutical agents and perform complex vitreoretinal surgery.

The biochemical abnormalities responsible for the development of diabetic retinopathy may not be important during the day-to-day practice of ophthalmology, but these abnormalities form the base of knowledge from which drugs have been developed to treat DR. Biochemical pathways that are pertinent to the development of diabetic retinopathy are complex and highly interrelated. Recognizing pivotal molecules such as vascular endothelial growth factor (VEGF) from within the biochemical milieu is an extraordinarily difficult process but one that is critical to drug development.

This chapter will cover many of the important highlights of retinal biochemistry in the setting diabetes, though a deep analysis of all affected pathways is beyond the scope of a single book chapter. The chapter begins with a discussion of retinal anatomy, then transitions to the microscopic composition of the blood-retinal barrier, and the biochemical abnormalities found in eyes affected by diabetes mellitus (DM). Finally, a brief discussion of the funduscopic characteristics of diabetic retinopathy will be provided.

2.2 The Retina

2.2.1 History

The retina was first described by Herophilus of Chalcedon (c. 300 B.C.) and it was eventually named by Rufus of Ephesus (c. 110 A.D.). To early anatomists the retina appeared as a net that surrounded and supported the vitreous. Galen noted structural similarities between the retina and the brain, but he did not understand the retina's function. Kepler was the first to suggest that the retina served as the eye's primary photoreceptor tissue. Treviranus (1835) performed the first detailed microscopic analysis of retinal structure by employing alcohol fixation. Only with the use of electron microscopy, trypsin digest, clinical fluorescein angiography, and optical coherence tomography were scientists able to understand the retina's cellular connections, ultrastructure, and retinal vasculature, and ultimately correlate anatomic and clinical findings [169].

2.2.2 Anatomy

The retina is a translucent tissue that lines the posterior two-thirds of the eye from the macula posteriorly to the ora serrata anteriorly [87] (Fig. 2.1). The retina is contiguous with the optic disc since axons of the retinal nerve fiber form the optic nerve as they leave the eye. Anteriorly, the retina attaches to the nonpigmented epithelium of the pars plana at the ora serrata. Interdigitations between the photoreceptor outer segments and the cells of the retina pigment epithelium weakly attach the outer retina to the wall of the eye. The retina is internally attached to the vitreous at the optic nerve, macula, retinal vessels, and vitreous base.

The central 5–6 mm of the retina, referred to as the retina centralis, has more than one layer of ganglion cells (Fig. 2.2). The central 1.5 mm of the retina is bounded by the termination of the retinal capillary circulation and is called the macula lutea because of the high concentration of xanthophyll in the ganglion and bipolar cell layers [218]. The 0.35 mm depression within the central macula, called the fovea by clinicians and the foveal by anatomists, has only cones in the photoreceptor layer. The center of the foveal depression, the clivus, is situated 3.4 mm temporal and 0.8 mm inferior to the center of the optic disc.

Beyond the macula, the retina extends past the vortex veins until terminating at the ora serrata. The equatorial diameter of the human eye averages 24.06–24.08 mm, from which the retinal area is calculated to be 1206 mm² [169, 218]. The ora serrata lies between 5.73 mm (nasally) and 6.52 mm (temporally) posterior to Schwalbe's line and is characterized by protruding extensions of the retina (teeth) that are separated by indentations of the ciliary epithelium (ora bays) from the pars plana. The strong adhesion between the vitreous and the peripheral retina and pars plana, the



Fig. 2.1 This drawing details the important anatomic and circulatory structures of the eye and supporting tissues. The retina lines the inner, posterior two-thirds of the globe from the optic nerve to the ora serrata (just beyond the termination of the retinal arterioles). The vitreous is the most voluminous (4 ml) tissue within the eye

so-called vitreous base, extends from 1.8 to 3 mm posterior to the ora bays to 1-2 mm anterior to the ora serrata [150].

A cross-sectional view through the retina just outside the retina centralis shows nine layers (from internal to external): nerve fiber layer, ganglion cell layer, inner plexiform layer, inner nuclear layer, outer plexiform layer, outer nuclear layer, external limiting membrane (ELM), rod and cone inner and outer segments, and retinal pigment epithelium (RPE) (Figs. 2.3 and 2.4). The retina is thickest around the optic disc because of the densely packed nerve fiber and ganglion cell layers, but thickness tapers to 0.18 mm at the equator and 0.11 mm at the ora serrata, since the density of all neural elements decreases in the peripheral retina.

The layered pattern of cells and connecting neural fibers within the retina enables photic energy to be converted into neuronal signals. Rhodopsin within the outer segments of rods and cones detects incoming photons and converts their energy into an electrical potential by depolarizing adjacent cellular membranes. Photoreceptor cells synapse with dendrites of bipolar cells (first-order neurons) and the processes of horizontal cells (integrating neuronal cells) in the outer plexiform layer. In the inner plexiform layer, bipolar cells synapse with dendrites of ganglion cells (secondorder neurons) and amacrine cells (which provide cross wiring). Ganglion cell axons comprise the nerve fiber layer, which aggregate to form the optic nerves until they synapse with third-order neurons in the lateral geniculate body.



Fig. 2.2 This drawing shows the anatomic and clinical areas of the macula. Despite being referenced more frequently by anatomists, the terms parafovea and perifovea are often used in clinical practice to describe the locations of macular pathology

The central 0.4 mm of the macula is capillary-free with its cells obtaining their nutrients from the choriocapillaris [54]. The fovea contains long and slender red and green (but not blue) cones that are aligned perpendicular to the RPE, which maximally sensitizes them to incoming photons. Because the fovea lacks the inner nuclear layer, inner plexiform layer, ganglion cell layer, and nerve fiber layer, light scatter is minimized. The external plexiform layer has an unusual configuration as its axons make a right angle turn to assume a course parallel with the retina surface. After coursing a short distance parallel to the retinal surface (while forming Henle's layer), the axons turn perpendicular to the RPE to synapse with dendrites of overlying bipolar cells outside the central 0.2 mm diameter of the fovea.

Xanthophyll pigments within bipolar and ganglion cells give the retina its characteristic yellow color and function to decrease chromatic aberration, absorb potentially toxic blue light, and scavenge free radicals. Lutein is scattered throughout the posterior pole, whereas zeaxanthin is concentrated in the foveal region [81].

The parafovea and perifovea are frequently referenced regions in clinical practice. The parafovea is a 0.5 mm ring of retina that surrounds the fovea and is characterized by an accumulation of ganglion and inner nuclear cells with a thickened Henle's layer. The density of cones in the parafovea is lower than within the fovea and rods are beginning to be found. The perifovea is the outermost ring of the retina centralis, between 1.25 and 2.75 mm from the foveal center. The perifovea



Fig. 2.3 This cross-sectional drawing of the retina near the optic nerve shows the layers of the retina and the locations of important cell types. The pumping mechanism of the retinal pigment epithelium (RPE), which removes water from the subretinal space to keep the photoreceptors in contact with the RPE, is illustrated

begins where the ganglion cell layer is four nuclei thick and ends where it thins to a single layer.

The ora serrata is the anterior boundary of the retina where it forms the junction between the multilayered neurosensory retina and the monolayered, nonpigmented epithelium of the ciliary body. The tissue in this region is thin and avascular and enjoys a close relationship with the vitreous base and zonular fibers. Collagen fibrils



Internal limiting membrane Nerve fiber layer Ganglion cell layer Inner plexiform layer Outer plexiform layer Outer plexiform layer Outer nuclear layer External limiting membrane Ellipsoid zone Cone outer segment line Retinal pigment epithelium

Fig. 2.4 The labeling on this spectral domain optical coherence tomography (OCT) scan through the papillomacular bundle highlights the tissue layers of the retina and important barrier structures. Structures that are commonly identified on the OCT that are often not referred to by anatomists as retinal layers – internal limiting membrane and external limiting membrane – have been included

insert into the internal limiting membrane (ILM) of the retina at the vitreous base, which normally extends from 2 to 4 mm posterior to the ora. The vitreoretinal adhesion is particularly strong along the posterior margin of the vitreous base, which makes this a common site for the development of retinal tears. A gradual loss of the nerve fiber layer, ganglion cell layer, and plexiform layers occurs near the ora as these layers are replaced with neuroglia and Müller cells. Both the ILM, into which vitreous base inserts, and the external limiting membrane (ELM), which continues anteriorly between the pigmented and nonpigmented layers of pars plana, are thick-ened in this region.

2.2.3 Microanatomy of the Retina Neurons

Cells within the retina fall into one of the three histologic groups: neuronal, glial, and vascular. The intricate interactions between the three types of neural cells, photoreceptors, interneurons, and ganglion cells, give the retina its primary function – converting light energy into electrical impulses. The densely packed rod and cone photoreceptor cells act as the primary neurons in the visual pathway by detecting individual photons to accurately construct visual images. Any change in the axial arrangement of the photoreceptors alters visual perception by producing micropsia if the cells are abnormally separated or metamorphopsia if the alignment is irregular.

Cone photoreceptor cells are comprised of four distinct anatomic portions: inner segments, outer segments that contain the visual pigment, a perikaryal region that contains the cell nucleus, and a synaptic terminal. The three different types of cones each contain only one light-sensitive pigment, which results in three different spectral sensitivities. The blue, green, and yellow cone pigments maximally absorb incident light with wavelengths of 450 nm, 530 nm, and 565 nm, respectively. The visual pigment in rods, rhodopsin, is composed of the light-sensitive chromophore retinol attached to the protein opsin [197] and is most sensitive to light with a wavelength of 500 nm.

The photoreceptor outer segments have two important anatomic connections, to the inner segments (cell bodies) of the photoreceptor cells and to the extracellular matrix that separate the outer segments from the RPE. The acid mucopolysaccharides of the matrix are probably synthesized by the photoreceptor inner segments and form the "glue" that keeps the outer segments attached to the RPE [5]. The nuclei of the cones are situated $3-4 \mu m$ internal to the ELM and comprise the outer nuclear layer. The photoreceptors form synaptic junctions with both interneurons and Müller cells. The plasma membrane of each photoreceptor and Müller cell is differentiated into a dense band known as the external limiting membrane. Rods and cones do not come into direct contact as they are insulated from each other by the Müller cells. The Müller cells contact each other in zonula adherens, which are thought to form a relative diffusion barrier between the intercellular space of the inner retina and the extracellular matrix between the photoreceptor outer segments and the RPE.

The outer plexiform layer lies between the inner and outer nuclear layers. The synaptic zone is composed of numerous intercellular junctions and synapses between neural and glial processes and creates the middle limiting membrane. This resembles the ELM and may also act as a partial barrier to diffusion of fluid and larger molecules as it may prevent exudates, hemorrhages, and cysts from spreading throughout the entire retina.

The inner plexiform layer is located between the ganglion and inner nuclear cell layers and contains Müller cell branches, retinal blood vessels, and synaptic processes of the bipolar, ganglion, and amacrine cells. Synaptic units called dyads each consist of a bipolar cell contacting two processes, one from a ganglion cell. Dyad density is approximately 2.9 million/mm² [47].

The ganglion cell bodies comprise a distinct layer between the inner plexiform and nerve fiber layers. Throughout much of the retina, there is one ganglion cell for every 100 rods and every 4–6 cones, but the ganglion cell to photoreceptor ratio is higher in the macula. This creates a smaller receptor field for each ganglion cell and greater image resolution. There are no ganglion cells at the center of fovea, but the ganglion cells are so densely packed within the extrafoveal macula that there may be two or more for each cone [42]. Ganglion cells consist of two types: midget and diffuse. Midget ganglion cells cover small areas (<10 μ m²) and synapse with only one midget bipolar cell, though each midget bipolar cell may synapse with several ganglion cells. The large, polysynaptic, diffuse ganglion cells synapse with retinal bipolar and amacrine cells via dendrites and cells in the lateral geniculate body via axons.

Ganglion cell axons course through the nerve fiber layer of the inner retina until entering the optic nerve. The axons remain unmyelinated until they reach the lamina cribrosa at which point they acquire myelin sheaths. Axons directly contact each other without interposed glial cells, except for interdigitating Muller cell processes. Most axons assume a generally radial course toward the optic nerve except for those immediately temporal to the disc, which form the papillomacular bundle. Since these axons are the first to develop, they form the center of the optic nerve. The nerve fiber layer thickens as axons converge at the optic nerves.

Axons cannot survive if they are detached from their respective cell bodies [152] as both proximal and distal axonal degeneration are seen after acute retinal or optic nerve ischemia. Cotton wool spots or optic disc edema are usually seen on funduscopic examination. Though cotton wool spots were long believed to represent focal infarctions of the retinal nerve fiber layer, they may actually be boundary sentinels of inner retinal ischemia [135]. After axonal degeneration, residual defects in the nerve fiber layer can often be seen on optical coherence tomography (OCT) scanning or funduscopic examination.

Müller cells form tight junctions with both other Müller cells and with neural cells. A continuous row of Müller cell zonula adherens forms the external limiting membrane, which acts as a relative barrier to metabolite movement into and out of the retina [142]. Müller cells comprise the majority of retinal glial cells but astrocytes are more widely distributed between blood vessels and neurons.

2.2.4 Intercellular Spaces

Neural cells are $10-20 \,\mu\text{m}$ apart within the retina, with spacing similar to that found in the brain. Intercellular spaces are filled with low density matrix material that permits diffusion of even large proteins [173] until they reach the relatively impervious external limiting membrane. The potential subretinal space is filled with an inter-photoreceptor matrix that is comprised of glycosaminoglycans, glycoproteins, and filamentous structures [85]. The most common inter-receptor matrix protein is interstitial retinal binding protein (IRBP), which is synthesized and secreted by rod photoreceptor cells and binds all-trans retinal and 11-cis retinal. Little is known about other matrix proteins.

2.2.5 Internal Limiting Membrane

The ILM is composed of laminin, basement membrane (BM) proteoglycans, fibronectin, and collagen [96] and represents the retina's only true basement membrane. The outer portion consists of the basement membrane of the Müller cells, whereas the inner portion is formed by vitreous fibrils and mucopolysaccharides. The ILM is 2000 nm thick over the macula but thins to only 20 nm over the fovea where the density of Müller cells decreases [209]. Müller cell processes form a continuous but uneven border of attachment with the ILM, though the exact nature of the vitreous attachment to the ILM is not known (see Chap. 7).

2.2.6 Circulation

The retina's demand for oxygen is higher than that of any tissue in the body, and to meet this high metabolic demand, the retina relies on two distinct circulations: the inner two-thirds of the retina relies on the retinal vasculature, whereas the outer one-third relies on the choroidal circulation. Flow through the choroidal circulation is high but variable with easy molecular transfer with the surrounding tissues. Flow through the retinal circulation is low but constant with a high rate of oxygen extraction [11].

2.2.7 Arteries

The central retinal artery (CRA) penetrates the optic nerve about 10 mm posterior to the globe, leaves the nerve at the optic disc, and rapidly branches into progressively smaller arterioles. The central retinal artery supplies the entire inner two-thirds of the retina, except in the 20% of eyes with cilioretinal arteries. The histologic structure of the CRA resembles that of other comparable-sized arteries with a luminal diameter of 200 μ m, a wall thickness of 35 μ m, a single layer of endothelial cells, a subendothelial elastica, an internal elastic lamina, a medium of smooth muscle, and an external elastic lamina that merges with the adventitia. When the artery enters the eye, the elastic lamina is lost but the muscularis is unusually prominent. Degenerative diseases that affect muscular arteries, such as atherosclerosis and giant cell arteritis, also affect the intraneural retinal artery. Arteries within the retina are spared from giant cell arteritis because they lack an internal elastic lamina, whereas atherosclerosis, with its subendothelial plaque formation and hyperplasia of the intimal and endothelial layers, can affect any portion of the retinal artery.

The retinal artery divides into superior and inferior branches immediately upon entering the eye, then to smaller branches according to either dichotomous (equalsized bifurcation) or side-arm branching. The arteries are situated in the nerve fiber layer or ganglion cell layers, and only the smaller arterioles descend into the inner plexiform layer to supply capillaries [88]. Strong connections exist between the arteries and cortical collagen in the ILM. Vitreous traction on the ILM can elevate the retinal arteries without affecting deeper retinal layers. The arteries are usually superficial to the veins but may dive to the inner nuclear layer at arteriovenous crossing sites.

Within smaller branches of the artery, the muscular layer thins from seven cell layers at the disc to two layers at the equator, and the luminal diameter thins from 120 μ m at the disc to 8–15 μ m at the equator. The vascular endothelial cells contain tight junctions that prevent the passage of large molecules into or out of the vascular lumens [167], which limits transfer of materials to diffusion and endothelial pinocytosis. Müller and astrocyte cell processes surround the vessels and insulate them from surrounding retinal neural tissue.

Tissue oxygen tensions, metabolic by-products, and intraocular and systemic blood pressures autoregulate the retinal circulation [52]. It is not known if sympathetic or parasympathetic nerves innervate the retinal arteries, but studies suggest that adrenergic binding sites are present on cells and that retinal blood flow can be altered by adrenergic agonists and antagonists [58, 168].

2.2.8 Veins

The central retinal vein consists of a layer of endothelial cells, subendothelial connective tissue, a medium of mostly elastic fibers, a few smooth muscle cells, and a thin connective tissue adventitia. The vein is separated from the surrounding neural tissue by Müller cells and astrocyte processes. The diameter of the vascular lumen decreases from 150 μ m at the disc to 20 μ m at the equator. In the peripheral retina, smooth muscle cells in the medium are lost and pericytes take their place. Muscle cells enable the venous diameter to change according to the pressure differential across the lumen. In patients with diabetes or carotid artery disease, sluggish flow relaxes the smooth muscles and the veins become sausage-shaped. The central retinal vein is the only outlet for the retinal circulation, but potential anastomoses between the retinal and choroidal circulations at the disc may become manifest in cases of central retinal vein occlusion or compressive lesions of the optic nerve.

2.2.9 Capillaries

Capillaries are found throughout the retina except in the foveal avascular zone, the retina adjacent to major arteries and veins, and the far periphery. The capillary network originates from the arterioles in the ganglion cell layer and spreads through the inner nuclear cell layer. No vessels are found in the outer plexiform and outer nuclear layers. Capillaries are distributed in two layers with a superficial network in the ganglion and nerve fiber cell layers and a deeper network in the inner nuclear layer. The diameters of these vessels range from 5 to 10 μ m. The volume of the outer vascular network is relatively constant throughout the retina, whereas the volume of the inner network varies with the thickness of the nerve fiber layer. Though only one capillary layer is found in the perifoveal region, up to four different capillary layers are found in the peripapillary region.

Peripapillary capillaries drain directly into venules on the optic nerve [84]. Within 2 disc diameters of the nerve, these capillaries have long, straight, or slightly curved paths with minimal anastomoses. This unique anatomy makes the capillaries susceptible to elevated IOP and changes in retinal perfusion pressure, which has been used to explain the development of arcuate scotomas in glaucoma, peripapillary flame-shaped hemorrhages in papilledema and hypertension, and cotton wool spots in disorders causing retinal ischemia.

The capillary wall consists of endothelial cells, pericytes, and a basement membrane. The narrow vascular lumen $(3.5-6 \ \mu m)$ and thin endothelial cell bodies cause nuclei to bulge inward, which requires erythrocytes to distort and mold when passing.

The capillary endothelial cells are connected by tight junctions that comprise the inner blood-retinal barrier (BRB) [167]. Pinocytotic vesicles transfer metabolites from the circulation to the retina. Diseases such as diabetes that damage the endothelium and disrupt the BRB allow protein and lipid to leak into the retina. The leakage is reversible through endothelial cell mitosis and the formation of new tight junctions [206].

Pericytes are situated within the endothelial cell basement membranes. Contraction of mammalian pericytes has not been demonstrated in vivo, but pericytes contain contractile proteins and contract in vitro when exposed to endothelin [29], thromboxane A_2 [46], and angiotensin II [132]. Loss of pericytes, due to ischemic retinopathies such as diabetes mellitus, weakens capillary walls and results in the formation of microaneurysms [35].

2.3 Hemodynamics, Macular Edema, and Starling's Law

Movement of fluid both into and out of capillaries depends upon hydrostatic and oncotic pressures. Formation and resorption of macular edema can be described by Starling's law (Fig. 2.5).

The four primary Starling's forces are as follows:

- 1. Hydrostatic pressure within the capillary lumen (P_c)
- 2. Hydrostatic pressure within the retinal interstitium (P_i)



Fig. 2.5 Starling's law states that movement of fluid across capillary walls is determined by the net hydrostatic pressure less the net oncotic pressure. In eyes with diabetic retinopathy, greater blood flow increases the net oncotic pressure, and breakdown of the blood-retinal barrier enables protein to accumulate in the interstitium and increase the net oncotic pressure. The end result is net fluid movement out of the capillaries into the interstitium leading to macular edema

- 3. Capillary oncotic pressure (Q_c)
- 4. Interstitial oncotic pressure (Q_i)

Capillary hydrostatic pressure is primarily determined by systemic blood pressure, whereas tissue hydrostatic pressure is approximately the same as intraocular pressure. Most of the capillary oncotic pressure is created by albumin, whereas, with healthy vascular endothelium, tissue oncotic pressure is determined by interstitial proteins. The net force pushing fluid out of capillaries is the difference between hydrostatic pressures and oncotic pressures and can be represented by the following equation:

 $F = (P_{\rm c} - P_{\rm i}) - (Q_{\rm c} - Q_{\rm i})$

where F is the resultant force that determines fluid movement. If F is positive, fluid moves out of the capillary into the interstitium and forms tissue edema. If F is negative, then the net movement of fluid is out of the tissue and into the capillary. At equilibrium:

 $F = 0 = \Delta P - \Delta Q$

where there is no net movement of fluid across the capillary walls.

Edema can be defined as the abnormal swelling of soft tissues, which in the case of diabetic retinopathy represents the retinal interstitium. Edema can be cytotoxic, where the fluid accumulates within cells, or vasogenic, where fluid accumulates within the interstitial spaces. Cytotoxic edema occurs with severe ischemic conditions such as central retinal artery occlusions. Starling's law applies to the formation of vasogenic edema, which is the most common form of edema in retinal vasculopathies such as diabetic macular edema and retinal vein occlusions.

Retinal edema occurs when the net hydrostatic force (forcing fluid into the interstitium) exceeds the net oncotic force (drawing fluid into the capillary lumen) across capillary walls. This usually occurs because of an increase in transluminal hydrostatic pressure, as with systemic hypertension or ocular hypotony, or because of a decrease in transluminal oncotic pressure, as occurs with increased interstitial proteins due to breakdown of the BRB or with a decrease in plasma proteins as with liver disease or protein wasting nephropathies.

Intraluminal pressure falls through the length of the arteriole system, but hydrostatic pressure in the capillaries and venules still depends upon the arterial blood pressure. Systemic arterial hypertension increases capillary hydrostatic pressure and aggravates the severity of diabetic macular edema. Patients with diabetic macular edema should undergo a thorough blood pressure assessment, and if systemic hypertension is discovered, appropriate pressure lowering treatment should be initiated. Reduction of systemic blood pressure may be sufficient to improve diabetic macular edema in some cases [134, 186].

Poiseuille's law describes flow within a tube, where the resistance to flow is inversely related to the radius of the lumen raised to the fourth power. Retinal arterioles dilate under hypoxic conditions such as diabetic retinopathy, which decreases resistance and increases hydrostatic pressure [178–180]. Retinal blood vessels have been observed to progressively dilate as diabetic macular edema forms [118].

Macular edema also forms in eyes with ocular hypotony. Since tissue hydrostatic pressure is equal to intraocular pressure, hypotonous eyes develop macular edema

due to a widened hydrostatic pressure gradient. Edema may improve in these eyes if intraocular pressure rises [115]. When the retina is subjected to vitreous traction, interstitial hydrostatic pressure decreases and macular edema or subretinal fluid accumulates [122].

Capillary oncotic pressure decreases in patients with hypoalbuminemia due to the nephrotic syndrome, protein-deficiency malnutrition, or severe liver disease. On the other hand, breakdown of the BRB allows albumin to leak into the interstitial spaces and increase tissue oncotic pressure. Endothelial damage, like that seen with diabetic retinopathy and vein occlusions, compromises the intercellular tight junctions and breaks down the blood-retinal barrier. Increased vascular porosity results in oncotic pressure shifts and the development of interstitial macular edema. There is a strong correlation between increased vascular endothelial growth factor (VEGF) levels, breakdown of the BRB, and the formation of macular edema [148].

2.4 Biochemical Basis for Diabetic Retinopathy

Patients with DM develop similar microvascular abnormalities in the retinal vasculature, renal glomeruli, and vasa vasorum of the peripheral nerves. Alterations in retinal blood flow and increases in vascular permeability are seen soon after the development of DM. Decreased activity of vasodilators such as nitric oxide together with increased activity of vasoconstrictors such as angiotensin II and endothelin-1 can be measured. The resultant qualitative and quantitative abnormalities in the extracellular matrix contribute to increases in vascular permeability. Microvascular cell loss results from programmed cell death. Cell loss together with the overproduction and deposition of extracellular matrix proteins and periodic acid-Schiffpositive proteins induced by growth factors such as TGF- β leads to progressive capillary occlusion. Chronic hyperglycemia decreases the production of endothelial and neuronal cell trophic factors, which leads to edema, ischemia, and hypoxiadriven neovascularization [25]. In nondiabetic patients, atherosclerosis begins with endothelial dysfunction [131], whereas in diabetic patients initiation seems to involve insulin resistance [90].

Four hypotheses have been advanced to explain the mechanism of hyperglycemiainduced microvascular damage (Fig. 2.6). These are:

- 1. Increased polyol pathway flux
- 2. Accumulation of advanced glycation end products (AGEs)
- 3. Activation of protein kinase C (PKC)
- 4. Increased hexosamine pathway flux

Inhibitors specific to aldose reductase, AGE formation, PKC activation, and the hexosamine pathway each prevent various diabetes-induced abnormalities. No common element to these pathways had been noted until the recent discovery that each causes overproduction of superoxide by the mitochondrial electron transport chain (Fig. 2.7) [25]. Consistent with this observation is that both diabetes and hyperglycemia increase oxidative stress [79].



Fig. 2.6 Hyperglycemia dysregulates four biochemical pathways: protein kinase C pathway, advanced glycation end product (AGE) pathway, hexosamine pathway, and polyol pathway. Each of the pathways has been associated with the development of diabetic retinopathy

To understand how hyperglycemia increases reactive oxygen species (ROS), one must examine changes in the electron transport within the mitochondria. Hyperglycemia causes overproduction of the electron donors (NADH and FADH₂) by the tricarboxylic acid (TCA) cycle [49], which increases the proton gradient across the inner mitochondrial membrane. This sufficiently prolongs the lifespan of electron transport intermediates, such as ubisemiquinone, to generate superoxides. Experimental evidence shows that two regulatory enzymes can be exploited to uncouple hyperglycemia-induced production of ROS. Upregulation of manganese superoxide dismutase (MnSOD) eliminates the production of reactive oxygen species, and an excess of uncoupling protein-1 (UCP-1) eliminates the protein electrochemical gradient [73]. Furthermore, overexpression of either MnSOD or UCP-1 prevents PKC activation of the hexosamine pathway, AGE formation, and an increase in polyol pathway flux. This evidence strongly supports the contention that excessive superoxide is pivotal to the unified theory of diabetic retinopathy.

Other experimental evidence links hyperglycemia, ROS, and the four previously mentioned biochemical pathways. A hyperglycemia-induced increase in ROS decreases glyceraldehyde 3-phosphate dehydrogenase (GAPDH) activity, which increases the concentrations of upstream glycolytic metabolites. These increase flux through the polyol pathway. Elevated triose phosphate levels increase the formation of methylglyoxal-derived AGE, the most common AGE that results from hyperglycemia.



Fig. 2.7 In the presence of O₂, electron transfer through the mitochondrial cytochromes regenerates adenosine triphosphate (ATP) from adenosine diphosphate. Dysregulation of each of the previously mentioned biochemical pathways disrupts electron transfer and shunts the electrons toward the synthesis of superoxide (O₂⁻)

Triose phosphate levels rise due to GAPDH inhibition by ROS [50]. ROSinduced decreases in GAPDH activity increase the concentration of fructose-6phosphate, the primary substrate for the hexosamine pathway. Inhibition of GAPDH increases concentrations of dihydroxyacetone phosphate, which increases DAG concentrations and activates PKC.

Several experimental models have shown that elevated MnSOD or UCP-1 activity prevents hyperglycemia-induced complications. Overexpression of either protein prevents monocyte adhesion to aortic endothelial cells [25], hyperglycemia-induced decrease in eNOS activity [49], and collagen-induced platelet aggregation and activation [210]. Increased MnSOD activity prevents increase collagen synthesis [37] and decreases programmed cell death induced by hyperglycemia.

Continued oxidative stress leads to chronic inflammation, which is a strong mediator of most chronic diseases including diabetes. Oxidative stress upregulates several transcription factors including AP-1, HIF-1 α , β -catenin/Wnt, NF- κ B, PPAR- γ , p53, and Nrf2. Activation of these factors leads to the expression of over 500 genes, including those for growth factors, inflammatory cytokines, chemokines, cell regulatory molecules, and anti-inflammatory molecules [156].

Inflammatory cells also produce soluble mediators, such as metabolites of arachidonic acid, cytokines, and chemokines that recruit additional inflammatory cells to the site of damage and produce more reactive species. These key mediators activate signal transduction cascades and induce changes in transcription factors that mediate immediate cellular stress responses. Induction of cyclooxygenase-2 (COX-2), inducible nitric oxide synthase (iNOS), aberrant expression of inflammatory cytokines (tumor necrosis factor (TNF), interleukin-1 (IL-1), and IL-6) and chemokines (IL-8, CXC chemokine receptor 4 (CXCR4)), and alterations in the expression of specific microRNAs have also been reported to play a role in oxidative stress-induced inflammation [92].

Our current understanding of retinal biochemistry identifies a series of pathways, all of which share substrates and enzymes and are regulated by positive and negative feedback loops. This complicated biochemical mosaic allows for considerable and complicated crossover reactions among pathways, which suggests that successful inhibition of any single pathway will probably produce an incomplete clinical response. If the unified theory of diabetic retinopathy – hyperglycemic production of reactive oxygen species – is correct, then limiting the formation of ROS may ultimately yield the most effective therapies.

A considerable amount of clinical research continues to focus on decreasing diabetic complications by minimizing changes in the four hyperglycemia-driven pathways. Further discussion of these pathways, therefore, appears warranted.

2.4.1 Increased Polyol Pathway Flux

Aldose reductase (the first enzyme in the polyol pathway) has a low affinity for glucose at normal concentrations, but during hyperglycemia glucose is increasingly converted to sorbitol with associated decreases in NADH. Sorbitol is subsequently oxidized to fructose with reconstitution of NADH. Sorbitol oxygenation increases the NADH/NAD+ ratio in the cytosol, which inhibits glyceraldehyde-3-aldehyde dehydrogenase (GAPDH) activity. This increases concentrations of triose phosphate [205], which increases formation of methylglyoxal (a precursor of AGEs) and diacylglycerol (DAG) (which activates PKC). Reduction of glucose to sorbitol consumes NADPH, and since NADPH is also required to reduce glutathione, this could worsen oxidative stress. Unfortunately, attempts to inhibit the polyol pathway *in vivo* have yielded only mixed results. A 5-year study of polyol pathway inhibition in diabetic dogs prevented diabetic neuropathy but failed to prevent retinopathy [51]. Zenarestat, an aldose reductase inhibitor, had a positive effect on diabetic neuropathy in humans [77].

2.4.2 Advanced Glycation End Products (AGEs)

Intracellular hyperglycemia promotes the formation of AGEs, which are found in elevated concentrations in diabetic retinal blood vessels [182, 184, 185] and glomeruli [89]. The intracellular auto-oxidation of glucose to glyoxal, the decomposition

of the Amadori product (glucose-derived 1 amino-1-deoxyfructose lysine adducts) to 3-deoxyglucosone, and the fragmentation of glyceraldehyde-3-phosphate and dihydroxyacetone phosphate to methylglyoxal, each reacts with amino groups of intracellular and extracellular proteins to form AGEs [44, 192, 202]. The AGE inhibitor aminoguanidine partially prevents microvascular damage in animal models [176] and lowers urinary protein concentrations and may slow progression of retinopathy in humans [23].

The intracellular production of AGE precursors damages target cells by modifying critical proteins and altering their functions. These change extracellular matrix components and integrins and modify plasma proteins that bind to AGE receptors. The end result is receptor-mediated production of reactive oxygen species.

The formation of AGEs alters the properties of several extracellular matrix proteins. Advanced glycation end product-induced cross-linking expands the molecular packing of type I collagen and changes the composition of type IV collagen in basement membranes, thereby altering the function of blood vessels [91]. Advanced glycation end product formation on laminin decreases the self-assembly of polymers, decreases binding to type IV collagen, and decreases binding to heparin sulfate proteoglycan [30]. The formation of AGEs on the extracellular matrix interferes with matrix-cell interactions. Modification of type IV collagen-binding domains in the matrix decreases endothelial cell adhesion.

Several cell-associated proteins that bind AGEs have been identified. These include OST-48, 80 K-H, galectin-3, macrophage scavenger receptor type II, and RAGE. These proteins mediate the long-term effects of AGEs on macrophages, glomerular mesangial cells, and vascular endothelial cells through the expression of cytokines and growth factors (interleukin-1, insulin-like growth factor-I, tumor necrosis factor- α , TGF- β , macrophage colony-stimulating factor, granulocyte-macrophage colony-stimulating factor, and platelet-derived growth factor) by macrophages and mesangial cells and the expression of pro-coagulatory and pro-inflammatory molecules (thrombomodulin, tissue factor, and VCAM-1) by endothelial cells. Many of these ligands bind to endothelial AGE receptors and mediate VEGF-induced capillary wall hyperpermeability [129].

2.4.3 Activation of Protein Kinase C (PKC)

The protein kinase C family is made up of at least 11 isoforms, 9 of which are activated by diacylglycerol (DAG). Intracellular hyperglycemia increases DAG synthesis from dihydroxyacetone phosphate in both the retina and renal glomeruli [117]. Hyperglycemia also activates PKC isoforms indirectly by ligating AGE receptors [153] and increasing activity of the polyol pathway [103]. Activation of PKC- β isoforms depresses nitric oxide production and increases endothelin-1 activity to mediate blood flow abnormalities in the retina and kidney [93].

Activated PKC increases the accumulation of matrix protein by upregulating the expression of TGF- β 1, fibronectin, and type IV collagen, which is mediated by

downregulating nitric oxide [38]. Protein kinase C overexpresses the fibrinolytic inhibitor PAI-1 and activates NF-kB in endothelial and vascular smooth muscle cells [151, 211] and regulates and activates various membrane-associated NADP(H)-dependent oxidases.

A PKC- β -specific inhibitor reduces activity in the retina and renal glomeruli of diabetic animals, reverses diabetes-induced increases in retinal mean circulation time, normalizes glomerular filtration rate, and corrects urinary albumin excretion [116].

2.4.4 Increased Hexosamine Pathway Flux

Excess flux through the hexosamine pathway activates genes that lead to vascular endothelial dysfunction and other changes that are seen in diabetic retinopathy. Excess intracellular glucose in the form of fructose-6-phosphate is diverted from glycolysis in order to provide substrates for reactions requiring UDP-Nacetylglucosamine. End products of these reactions include proteoglycans and O-linked glycoproteins. The mechanism by which increased hexosamine pathway flux mediates hyperglycemia-induced increases in gene transcription is not known, but covalent modification of the transcription factor Sp1 by N-acetylglucosamine (G1cNAc) might explain the link between the hexosamine pathway and hyperglycemia-induced changes in transcription of the gene for PAI-1. The glycosylated form of Sp1 upregulates transcription more than the deglycosylated form. Glycosylated Sp1 is inversely associated with phosphorylated Sp1, which suggests that the two molecules compete for the same binding site and may represent a more generalized mechanism for regulating glucose-responsive gene transcription [82].

2.5 Early Pathophysiology of Diabetic Retinopathy

One of the earliest histopathological changes observed in diabetic patients and diabetic models is thickening of the capillary BM (basement membrane) [66, 67]. Thickening is thought to result from increased synthesis of collagen IV, fibronectin, and laminin and/or reduced degradation of the BM by catabolic enzymes [128, 158, 182, 183]. It remains uncertain whether BM thickening is of primary or secondary importance in the development of DR, but it has been speculated that these matrix modifications may contribute to impaired endothelial-pericyte communication, defects in capillary autoregulation, or inappropriate cell interaction with constituent BM proteins [20, 146, 159, 184].

Mitochondria of retinal pericytes exhibit fragmentation, metabolic dysfunction, and reduced extracellular acidification when exposed to high glucose environments. This may be responsible for the early apoptotic death of both pericytes and smooth muscle cells during DM [67, 138]. The ratio of pericytes to endothelial cells in

patients with diabetes decreases from 1:1 to 1:4 [19, 80]. Vascular endothelial cells become dysfunctional at the same time, but over the short term, they can sufficiently compensate for the developing cellular deficit. Because of prolonged exposure to the diabetes-induced biochemical milieu, the replicative ability of the endothelial cells ultimately becomes critically compromised [83, 126].

It has been proposed that the primary cellular defect in diabetic retinopathy lies with the vascular endothelium and that retinopathy could be considered an endotheliopathy [104]. The endothelium is critical to normal function of other cells in the capillary complex [66], and though the precise basis of endothelial, pericyte, and smooth muscle dysfunction in the diabetic retinal microvasculature remains obscure, it is most likely related to an array of cumulative biochemical insults coupled with impaired ability of the cells to repair and renew themselves. Retinal pericytes and smooth muscle cells rely on key growth/survival factors, such as PDGF (platelet-derived growth factor)-B [61, 86], which is selectively depleted during diabetes [36] and is closely related to the formation of acellular capillary casts [80].

2.6 Macular Edema

Retinal nutrition is supplied by two distinct circulations: the choriocapillaris nourishes the outer third of the retina and the two vascular layers (superficial plexus in the axon and ganglion cell layers, deep plexus in the inner nuclear layer) of the retinal capillary circulation. Two distinct, though often coexisting, types of diabetic maculopathy are seen: ischemic maculopathy results from capillary dropout, and macular edema results from breakdown of the blood-retinal barrier in the retinal capillaries and/or RPE.

The exact location of macular edema – extracellular or intracellular – has not been well established as previous studies have implicated both locations. Histologic studies of enucleated eyes with diabetic retinopathy have identified swelling and degeneration of Müller, bipolar, ganglion, and photoreceptor cells [59] and extracellular cystic spaces [69]. Though angiographic ischemia and macular edema may coexist, their effects on visual acuity appear to be independent. Arend [12] found no correlation between blood flow and macular edema, thereby concluding that inner retinal ischemia does not cause macular edema. Retinal thickening appears to correlate with blood-retinal barrier breakdown, but ischemia was not correlated with macular thickening, thereby suggesting that the major cause of retinal thickening was extracellular expansion [174]. These studies conclude that ischemia may lead to a small amount of intracellular swelling, but extracellular fluid accumulation is the primary contributor to macular edema [8].

Even before the development of obvious retinopathy, diabetes can either increase or decrease retinal blood flow. Once retinopathy becomes established, however, retinal blood flow usually increases [171]. The autoregulatory function of retinal blood vessels is lost, and both arterioles and venules are noted to dilate and elongate. This leads to decreases in arteriole resistance and loss of pressure across the length of the



Fig. 2.8 This optical coherence tomography scan demonstrates cystoid macular edema, most which can be seen in the inner and outer nuclear layers

arteriole. Effective intraluminal pressure in the retinal capillaries increases, which, according to Starling's law, increases fluid movement into the extracellular space. The concomitant release of vasoactive cytokines (such as VEGF) from endothelial cells, neuroglia, and activated leukocytes breaks down the blood-retinal barrier via several mechanisms and leads to albumin movement into the interstitium.

Total flow through the choroidal circulation decreases in patients with diabetes [120]. Bulk flow to the choroid is relatively unchanged in diabetics, but indocyanine green angiography shows focal areas of choroidal non-perfusion in eyes with diabetic pigment epitheliopathy. Significant leakage across the pigment epithelium sometimes occurs in eyes that have only minimal retinopathy [200]. Both neovascularization within the choroidal and occlusion of the choriocapillaris occur more commonly in diabetics than in individuals without diabetes. These processes probably result from the same inciting events – basement membrane degeneration and angiogenesis – that lead to the development of diabetic retinopathy [63].

Breakdown of the blood-retinal barrier can be detected by vitreous fluorophotometry before edema accumulates within the retina. This probably results from early protein loss from the gap junctions that allows molecules smaller than albumin to pass out of the capillary lumen. The first manifestation of diabetic retinopathy in other patients, however, may be a retinal pigment epitheliopathy [200]. Retinal pigment epithelial dysfunction is probably underappreciated in patients with DR since overlying retinal vascular disease masks RPE dysfunction on fluorescein angiography.

Histopathologic studies have shown that cystoid macular edema develops in the inner nuclear layer and the outer plexiform layer (Fig. 2.8) [194], but large volumes of fluid can "spill over" to the outer nuclear layer (Fig. 2.9). Both histologic studies and OCT scans suggest the presence of multiple cystoid spaces, but further analysis shows that fluid exists in large cavities that are often spanned by Müller cells.

Fluid accumulation due to leaking capillaries of the inner retina is usually limited by the inner and outer plexiform layers, causing fluid to accumulate within the inner nuclear layer. This fluid displaces tissue in the plexiform layers, increases tissue density, and forms a relative barrier to further spread of the edema. Breakdown of the outer blood-retinal barrier often causes fluid to accumulate between the outer Fig. 2.9 This optical coherence tomography scan through the macula shows significant edema in the outer nuclear layer. Note the prominent epiretinal membrane and the intact external limiting membrane and ellipsoid zone



plexiform layer and the external limiting membrane. As demonstrated in horseradish peroxidase studies in diabetic rats [193], eyes with intact external limiting membranes will develop limited exudative retinal detachments with fluid accumulation between the retinal pigment epithelium and photoreceptor outer segments.

2.6.1 Blood-Retinal Barrier

The retina is protected by two distinct blood-retinal barriers: the inner BRB consists of the retinal capillary endothelial cells and their tight junctions, and the outer BRB consists of the retinal pigment epithelium and their tight junctions. The cell membranes of the epithelial and endothelial cells are bilayers with hydrophilic glycerols bounding hydrophobic long-chain fatty acid moieties. Under physiologic conditions, the cell membranes allow small hydrophobic molecules, water, and small uncharged polar molecules to pass, but form an impenetrable barrier to ions and larger uncharged polar molecules.

Adjacent retinal vascular endothelial cells and retinal pigment epithelial cells are connected by tight junctions that are composed of several intercellular proteins including occludin, claudin (23 isoforms), 7H6, cingulin, zonula occludens (ZO)-1,2,3, junction adhesion molecule (JAM), membrane-associated guanylate kinases with inverted domain structures (MAGI)-1,2,3, partition defective genes (PAR)3/6, and multi-pdz protein-1 (MUPP1) (Fig. 2.10) [133]. Some of the proteins have been extensively studied and well characterized, with occludin and the claudins controlling much of the barrier function [106]. Loss of occludin, which is specific to vascular endothelial cells, [10] in diabetic rats correlates with increased blood-retinal permeability to albumin (molecular weight of 66 kD) but not rhodamine-dextran (molecular weight of 10 kD), which may be due to changes in hydrophobicity but not pore size. Claudin-5 prevents passage of small (<0.8 kD) molecules across the blood-brain barrier [141] and probably serves the same purpose in the retina [55]. In retinal capillary endothelial cell tight junctions, ZO-1 is believed to coordinate the intracellular assembly of the junctional complex [53] and is a reliable indicator of



overall tight junction function [68]. The junctional adhesion molecules (JAM) facilitate interactions between endothelial cells and leukocytes and assist with tight junction assembly. Junctional proteins may assist with cellular structure since intracellular, circumferential actin bundles aggregate at sites of occludin-positive cell contact, but not at sites that lack occludin.

Transport of ions and molecules across the BRB may be either transcellular or paracellular (Fig. 2.11) [181]. Transcellular passage occurs because of changes in the barrier state or pumping capacity of endothelial cells, whereas paracellular passage occurs because of loss of integrity of the tight junctions. In rat and monkey models of diabetes and in humans, increased fluorescein levels in the vitreous can be detected with vitreous fluorophotometry before fluorescein angiograms show evidence of diabetic retinopathy [41, 97]. These observations suggest that early breakdown of the BRB allows passage of small molecules such as fluorescein but not larger proteins such as albumin, which are necessary for the formation of macular edema.

2.6.2 Biochemical Abnormalities Responsible for Diabetic Retinopathy

During recent years there has been considerable focus on the role of inflammation in the pathophysiology of diabetic retinopathy (Table 2.1) [191]. Increased expression of pro-inflammatory cytokines within the neural retina and upregulation of Fig. 2.11 Fluid movement across the blood-retinal barrier (BRB) may be paracellular (through the intercellular tight junctions) or transcellular (across the cell). Vascular endothelial growth factor-mediated breakdown of the BRB promotes fluid movement via both pathways



Table 2.1This table listsseveral of the chemokines(pro-inflammatory molecules)and cytokines (growthfactors) that have beenassociated with thedevelopment of diabeticretinopathy

Chemokines
Chemokine (C-C) motif ligand 2 (CCL2)
Endothelin-1
Intercellular adhesion molecule-1 (ICAM-1)
Interleukin 1α (IL- 1α)
Interleukin 1 β (IL-1 β)
Interleukin 6 (IL-6)
Interleukin 8 (IL-8)
CXCL 10/Interferon induced protein 10 (IP-10)
Monocyte chemotactic protein 1 (MCP-1)
P-selectin
Vascular cell adhesion molecule-1 (VCAM-1)
Cytokines
Angiopoietin – 2 (Ang 2)
Platelet-derived growth factor AA (PDGF-AA)
Transforming growth factor β (TGF- β)
Tumor necrosis factor-α (TNF-α)
Vascular endothelial growth factor (VEGF)

vascular adhesion molecules that promote leukostasis have been linked to both neurovascular dysfunction and formation of acellular capillaries [98]. Evidence suggests that leukocytes may actively damage the retinal vascular endothelium [123, 189]. Global mRNA expression profiling has detected altered expression of proinflammatory cytokines not only in the retinal vessels [72] but also in the neuroglia [26]. Within the diabetic retina, a complex milieu of dysregulated pro-inflammatory factors, including interleukin (IL)-1 α , IL-1 β , IL-6, and tumor necrosis factor $(TNF)-\alpha$, develops [191]. Though microglia and infiltrating monocytes are central to inflammatory-mediated central nervous system pathology, the role of these cells in diabetic retinopathy is considerably less well understood. A number of in vitro studies and in vivo investigations of animal models and human postmortem specimens indicate that activation of retinal microglia could play an important regulatory role in diabetes-mediated retinal inflammation [213] by modulating cytokine expression [27] and other pathologic responses [119]. Monocytes and microglia play important roles in retinal homoeostasis, but they are also central to neuroinflammation, and these have been shown to increase in diabetes, in both humans and animal models [15, 160, 214].

Very soon after the onset of experimental diabetes in animals, a variety of molecules are upregulated in the retinas and physiological changes are observed. Retinal oxygen tensions decrease before the loss of retinal capillaries in rodent models of diabetic retinopathy [39, 125]. Experimental diabetes in animals results in an increase in retinal VEGF mRNA within the first week, and some authors have concluded that VEGF upregulation is caused by oxidative stress [48, 147]. Upregulation of VEGF correlates with increased expression of intercellular adhesion molecules (ICAM-1), leukostasis, and breakdown of the blood-retinal barrier (BRB) [3]. Pro-inflammatory cytokines such as TNF- α [99] and complement factors [215] are upregulated, and activation of the kallikrein-kinin [121] and renin-angiotensin systems [32] occurs. Each of these changes has been implicated in the pathogenesis of DR, and since some of these pathways are not VEGF dependent, they provide further evidence that inflammation and hypoxia may be independently responsible for DR [4, 43, 45]. Vitreous concentrations of pro-inflammatory cytokines (TNF- α , IL-8, IL-6, VEGF), chemokines (monocyte chemotactic protein-1 (MCP-1)), and other proteins (endothelin-1, sE-selectin, ICAM-1, CXCL10/IP-10) are higher in patients with proliferative diabetic retinopathy or diabetic macular edema than in controls [109].

Evidence that links leukostasis to oxidative stress and other downstream mediators comes from the observation that alpha-lipoic acid decreases leukocyte adhesion. Mechanisms linked to PKC pathways are responsible for hemodynamic alterations that occur concomitantly with leukostasis [1].

Advanced glycation end products bind to specific membrane receptors to upregulate vascular cell adhesion molecule-1 (VCAM-1), which promotes leukostasis and may accelerate diabetic vasculopathy [2, 164].

A few hours of hypoxia increases passage of retinal RNA into the general circulation in humans, and when oxygen tensions are normalized, the levels return to normal within 24 h [100, 207]. Hypoxia stimulates the release of hypoxia-regulated vasoproliferative factors, such as VEGF, but VEGF has been found to be increased in retinas of diabetic animals before capillary closure, indicating that other factors upregulate it early during the course of diabetes. VEGF has been shown to break down the BRB by both promoting molecular movement across cells and by damaging tight junctions. VEGF injections induce membrane fenestrations in rat and frog endothelial cells but not monkey cells. Transcellular gaps occur with increased frequency and lead to increased hydraulic conductivity. Within 24 h after vascular endothelial cells are perfused with VEGF, vesiculovacuolar organelles form a continuous, transcellular chain, separated only by fenestrations [16–18].

When exogenous VEGF is administered to eyes affected by diabetes, three changes in the tight junctions have been observed: (1) tight junction proteins are phosphorylated; (2) existing junctions are reorganized; (3) junctional protein levels are decreased [55].

2.6.3 Mechanism of Blood-Retinal Barrier Breakdown

The exact mechanism by which breakdown of the blood-retinal barrier occurs is not fully understood. Current evidence suggests that it may be due to either damage of the junctional complexes between capillary endothelial cells and RPE cells or changes in the cells' membrane state or pumping capacity. Proposed mechanisms that cause leakage have included the development of fenestrations across the endothelial cell cytoplasm, increased transcellular transport via vesicles, and increased infolding of the RPE that promotes choroidal to subretinal space transudation [195]. High glucose concentrations have been shown to reduce electrical resistance of cultured capillary endothelial cells and induce insulin-mediated breakdown of the blood-retinal barrier [76]. The discovery of vascular endothelial growth factor (VEGF) and other inflammatory cytokines, which exert considerable effects on junctional and interstitial proteins, has focused attention on the tight junctions. Cultured astrocytes simultaneously increase both intercellular barrier function and ZO-1 synthesis, suggesting a strong link between the two processes [68].

Diabetes changes the concentrations of junctional and matrix proteins. Occludin is decreased in the retinas of diabetic rats, and occludin and ZO-1 levels are decreased in cultured brain endothelial cells treated with VEGF [199]. Levels of MMP-2, MMP-9, and MMP-14 mRNA are increased in the retinas of diabetic animals. Cells treated with MMP-2 and MMP-9 develop altered gap junctions through increased transepithelial electrical resistance and degradation of occludin [74].

Though the precise biochemical pathway that leads to breakdown of the bloodretinal barrier is unknown, a large volume of research has focused on vascular endothelial growth factor. Much has been learned about VEGF since its discovery in 1989, and understanding its effects has been critical to developing therapy for clinically significant diabetic retinopathy. Detailed VEGF chemistry will be covered in Chap. 4, but general actions of VEGF on the blood-retinal barrier and development of neovascularization will be discussed in this chapter.



Fig. 2.12 Vascular endothelial growth factor-A binds to transmembrane receptors and activates several downstream pathways. These promote cellular survival, proliferation, and migration, and they increase blood flow through vasodilation and angiogenesis

Most ocular actions of VEGF result from the binding of VEGF-A isoforms to the transmembrane receptor VEGFR2. Vascular endothelial growth factor-mediated dimerization and phosphorylation of VEGFR-2 increase vascular permeability and stimulate angiogenic and mitogenic changes in vascular endothelial cells [56, 57].

Vascular endothelial growth factor increases vascular permeability through several mechanisms (Fig. 2.12). First, it stimulates inositol triphosphate (IP₃), which releases intracellular calcium. High calcium concentrations increase nitrous oxide and cyclic GMP levels and relax vascular smooth muscle. Vascular permeability, angiogenesis, vessel diameter, and blood flow are proportional to nitrous oxide synthetase levels [62]. Secondly, VEGF stimulates DAG production, which increases cellular permeability directly through DAG-sensitive Ca^{2+} channels. Thirdly, increased synthesis of DAG activates PKC.

In mouse models of experimental neovascularization, exogenous administration of a VEGF-trap decreases neovascularization by 66% [162], and transfer of the VEGFR-1 genes decreases neovascularization by 53–86% [13, 71]. These data suggest that VEGF is critically important to the development of neovascularization. VEGF is sufficient to cause leakage from normal blood vessels as shown in animal studies following implantation of sustained VEGF-release devices [145] or intravit-real injections [162]. Though hundreds of papers have been published regarding the

effect of VEGF on vascular permeability, there have been very few *in vitro* studies. Notable findings include the ability of VEGF to increase hydraulic conductivity [17] and diffusive permeability to albumin [208], but VEGF has no effect on the oncotic reflection coefficient (the probability that a molecule will bounce off a pore rather than go through it) to albumin [16]. In fact, hydraulic conductivity (a measure of the ease by which fluid moves through a lumen) and compliance (the inverse of stiffness) may be stimulated separately, suggesting that permeability and angiogenesis may be stimulated or inhibited separately.

In addition to its effects on the retinal vasculature, VEGF protects against apoptosis by acting as a survival factor for endothelial cells both *in vitro* and *in vivo*. Vascular endothelial growth factor prevents starvation-induced endothelial apoptosis by blocking the phosphatidylinositol 3-kinase/V-akt murine thymoma viral oncogene homolog (PI3 kinase/Akt) pathway, possibly by inducing the antiapoptotic proteins Bcl-2, A1, XIAP, and survivin.

Vascular endothelial growth factor induces inflammation by upregulating ICAM-1 (intercellular adhesion molecule) synthesis, which causes leukocyte adherence to vascular walls [130]. These activated leukocytes synthesize VEGF, thereby amplifying VEGF production [70]. Adherent leukocytes narrow capillary lumens and decrease downstream perfusion, which exacerbates localized retinal ischemia, further amplifies VEGF production, and compromises the blood-retinal barrier [22].

Intravitreal VEGF levels correlate with the severity of diabetic retinopathy as higher concentrations are generally seen in eyes with PDR and lower concentrations in eyes with BDR [24]. Even within the subset of patients with DME, aqueous VEGF levels closely correlate with the severity of DME [65].

Vascular endothelial growth factor is upregulated by several growth factors including TNF- α , TGF- β , epidermal growth factor, insulin-like growth factor-1, fibroblastic growth factor, platelet-derived growth factor, and also by inflammatory cytokines including IL-1 α and IL-8. One of the major stimulants of VEGF mRNA expression is low tissue O₂ tension. Similarities exist between the hypoxic regulation of VEGF and erythropoietin. A 28-base sequence in the promoter of the human VEGF gene, which has similar binding characteristics to hypoxia-inducible factor (HIF)-1 α , a key mediator of hypoxic responses, has been identified [166]. Hypoxia-inducible factor has been termed the "cell's oxygen sensor" since it serves as the key transcriptional regulator of the hypoxic response.

In the presence of oxygen, free HIF-1 α is hydroxylated by prolyl hydroxylase. Hydroxylated HIF-1 α binds to the von Hippel-Lindau factor (pVHL), and the pVHL/HIF-1 α complex is subsequently degraded by intracellular proteasomes [95]. Low tissue oxygen tension prevents hydroxylation of HIF-1 α and allows it to dimerize with HIF-1 β [161]. The stabilized dimer moves to the nucleus where it binds to the promoter region of the VEGF gene and stimulates transcription. Hypoxia increases levels of HIF-1 α subunits both by stabilizing the protein [198] and allowing HIF-1 α mRNA accumulation [201]. Stabilization of HIF-1 α during hypoxia requires an intact mitochondrial electron transport chain to generate reactive oxygen species (ROS), and it is the concentration of ROS that may constitute the cell's oxygen sensor [165]. Low intracellular oxygen tension is indirectly represented by high levels of ROS, which limits a cell's ability to hydroxylate HIF-1 α .

In addition to ROS, several other molecules including IGF-1 and 2 and AGEs affect the stability of HIF-1 α . Insulin-induced stabilization of HIF-1 α with subsequent upregulation of VEGF may explain why very tight glucose control worsens retinopathy [154]. Several biochemical opportunities exist for HIF-1 α modulation, including phosphatidylinositol 3-kinase inhibitors, mitogen-activated protein kinase inhibitors (MAPK), prolyl-4-hydroxylase domain activators, microtubule disrupting agents, cyclooxygenase-2 (COX-2) inhibitors, heat shock protein inhibitors, and antisense therapy. Breakthroughs in treating ischemic retinal disease may follow new cancer therapies such as topotecan and camptothecin analogs [149].

The protein kinase C pathway is upregulated by the presence of diacylglycerol. Activated PKC increases vascular permeability via two mechanisms: endothelial cell contraction through phosphorylation of the cytoskeletal proteins caldesmon and vimentin [177] and the activation of serine/threonine phosphatases or the inactivation of kinases, both of which dephosphorylate the tight junction proteins occludin and the claudins [34]. Protein kinase C increases matrix protein deposition – a hallmark of diabetic retinopathy – by inhibiting NO synthesis, which leads to increased TGF- β 1, fibronectin, and type IV collagen [187].

Several studies have shown how activated PKC may contribute to BRB breakdown. Protein kinase C inhibitors block VEGF-mediated BRB breakdown in nondiabetic rats [7]. The intravitreal injection of PKC can induce BRB breakdown in nondiabetics. Protein kinase C inhibitors prevent hyperglycemia-induced VEGF production [204], and in advanced diabetic retinopathy, PKC can mediate the action of VEGF, thereby creating a reinforcing cycle. Hyperglycemia may worsen existing retinopathy by activating PKC and also by upregulating p42/p44 MAPK. Both of these pathways depend on HIF-1\alpha stabilization [154].

Maintenance of the normal retinal vasculature requires a precise balance of angiogenic factors (such as VEGF) with inhibitors (such as angiostatin and pigment epithelium-derived factor (PEDF)) [136]. Pigment epithelium-derived factor, which is secreted by Müller and endothelial cells, has anti-inflammatory and anti-permeability properties [127]. Vitreous levels of PEDF are low in eyes with BDR, and levels rise to normal after pan-retinal photocoagulation for proliferative diabetic retinopathy. Pigment epithelium-derived factor modulates the vasculopathic effects of VEGF by decreasing HIF-1 α levels via MAPK-mediated activation, which stimulates endogenous PEDF production [217], and by competitive binding with VEGF for VEGFR-2.

Patients with rheumatoid arthritis appear to have less severe diabetic retinopathy, which has prompted speculation that the anti-inflammatory medications they take antagonize the effects of vasoactive cytokines. Though these drugs may antagonize the effects of VEGF, they do not decrease VEGF levels. Aspirin and TNF- α inhibitors decrease ICAM-1 levels and leukocyte adhesion by decreasing nitrous oxide synthetase expression. Aspirin decreases the expression of integrins that bind to ICAM-1 [LFA-1 (CD11a/CD18) and Mac-1 (CD11b/CD18)].

2.6.4 Renin/Angiotensin System

The renin/angiotensin system (RAS) modulates a diverse group of biological functions [105]. The system's major active product, angiotensin II, is converted from angiotensin I by angiotensin-converting enzyme (ACE). Angiotensin II modulates vasoconstriction, electrolyte homeostasis, drinking behavior, and pituitary hormone release [6, 40, 94]. As a growth factor, angiotensin II promotes cellular differentiation, apoptosis, and the deposition of extracellular matrix [101, 143, 144, 188]. Angiotensin receptors are found in endothelial cells, glia, and neurons [139, 140], which suggests that angiotensin II may regulate functions within these cells.

Angiotensin exerts significant effects on vascular smooth muscle cells including growth, proliferation, and the deposition of extracellular matrix proteins [190]. These are mediated by factors such as TGF- β 1, PDGF, VEGF, insulin-like growth factor, and connective tissue growth factor [203]. Though the effects of angiotensin II on pericytes are less well studied, they appear to include pericyte uncoupling and migration [102].

The pro-angiogenic effect of angiotensin II on mammalian retinas with oxygeninduced retinopathy is mediated by VEGF [56]. Pharmacologic blockade of the RAS decreases angiogenesis by downregulating VEGF and VEGFR2.

The ACE inhibitor captopril prevents degeneration of retinal capillaries, and the angiotensin receptor antagonist losartan inhibits leukostasis in streptozotocininduced rats and vascular cell adhesion molecule-1 (VCAM) expression in human vascular endothelial cells [216]. Inhibition of angiotensin-converting enzyme prevents the retinal overexpression of VEGF in experimental diabetes [75] and the progression of diabetic retinopathy in normotensive patients with type I diabetes [31].

Elevated angiotensin II levels correlate with vitreous VEGF concentrations in patients with diabetic macular edema [64]. Experimental evidence suggests that this may be mediated through the AT1-R/NF- κ B pathway [139], which creates new target sites for the prevention of diabetic retinopathy.

2.7 Development of Proliferative Diabetic Retinopathy

Apoptosis of retinal vascular endothelial cells together with the formation of pericyte ghosts results in acellular capillaries, hypoxia due to retinal non-perfusion, and with sufficient upregulation of cellular growth factors, proliferative retinopathy [21, 138]. Death of the vascular cells results from the upregulation of several cytotoxic factors including TNF- α . The forkhead transcription factor FOXO1 regulates cell death after being upregulated by TNF- α . Inhibition of FOXO1 by RNAi reduces cell apoptosis and microvascular cell death both *in vitro* and *in vivo* in type 1 and type 2 diabetic rats.

These factors significantly increase the activities of genes that modulate endothelial cell behavior, including angiogenesis and vascular remodeling. Occlusion of retinal capillaries leads to patchy non-perfusion, with the inner two-thirds of the retina suffering from hypoxia. Hypoxia-induced stabilization of HIF-1 α upregulates VEGF production, which is temporally related to the development of neovas-cularization [4].

In addition to VEGF, other growth factors such as insulin-like growth factor-1, hepatocyte growth factor (HGF), basic fibroblast growth factor (b-FGF), plateletderived growth factor, and pro-inflammatory cytokines and angiopoietins are involved in the pathogenesis of PDR. Also found within the retina and vitreous are anti-angiogenic factors such as pigment epithelium-derived factor (PEDF), TGF- β , thrombospondin, (TSP), and somatostatin. It is widely believed that neovascularization results from an imbalance of angiogenic and anti-angiogenic factors [170].

2.8 Clinical Findings of Diabetic Retinopathy

The clinicopathology of the diabetic retina has been extensively studied, but the precise cellular and molecular defects that lead to retinal vascular, neural, and glial cell dysfunction remain somewhat elusive. Though vascular dysfunction and decreased perfusion remain the hallmarks of diabetic retinopathy, a growing body of evidence suggests that neuroretinal function is compromised, probably before overt vascular changes are seen [9]. Electrophysiological studies of patients with diabetes suggest that alterations in the neural retina, including loss of color vision [157] and contrast sensitivity [175], and abnormalities in the electroretinogram (ERG) occur soon after the onset of diabetes [212]. Glial abnormalities develop during hyperglycemia, with Müller cells overexpressing glial fibrillary acidic protein [137], with concomitant synthesis of glutamate as a function of disruption of the glutamate transporter [155]. This may contribute to excitotoxicity and eventual depletion of retinal neurons, thus making them an integral part of diabetic retinopathy [124]. Taken together, the retinopathy literature suggests that neurodegeneration constitutes a significant pathophysiological defect in diabetes.

The earliest functional changes in nonproliferative diabetic retinopathy (NPDR) cannot be visualized with photography, but they include changes in retinal blood flow and loss of autoregulatory mechanisms that adjust retinal capillary perfusion to local metabolic demand [114]. As early as the 1930s, it was suggested that retinal blood flow was markedly altered in diabetic patients [196], and as technology evolved there were indications that retinal vessel caliber consistently increased during diabetes [14, 172]. Since the 1970s it has been recognized that hemodynamic changes in the retinal vasculature play a role in the pathophysiology of diabetic retinopathy and possibly serve as an early indicator of diabetes -related retinal dysfunction [110, 112]. Since then various patient-based studies have shown that retinal hemodynamic abnormalities occur before the onset of clinically observed diabetic retinopathy [28]. In diabetic patients without retinopathy, retinal arteriolar vasoconstriction and decreased total retinal blood flow have been reported [28, 33, 107]. Decreased retinal blood flow early in the course of diabetes may reduce oxygen and

nutrient delivery to the neuro-retina and contribute to the initial neuroglial abnormalities that have been observed in the diabetic retina. As the diabetes progresses, retinal arterioles begin to dilate and bulk retinal blood flow increases in proportion to the severity of retinopathy [163]. Enhanced blood flow may hasten the progression of diabetic retinopathy by causing shear stress-induced endothelial cell damage and death [114]. A recent trial has suggested that these changes could be regarded as robust risk factors for the subsequent development of retinopathy [108]. Patients with increased retinal blood flow who fail to demonstrate improved hemodynamic indices after normalization of blood glucose experience a more rapid progression of the disease [78].

The earliest ophthalmoscopically visible signs of diabetic retinopathy are usually retinal capillary microaneurysms. These represent saccular dilations of the capillaries and appear as dark red or white fundus spots. Fluorescein angiography reveals perfused microaneurysms as discrete hyperfluorescent spots with early pooling of dye followed by late leakage. Many microaneurysms are fully or partially perfused during fluorescein angiography, whereas others are sclerosed and non-perfused [111, 113].

Retinal hemorrhages can be either "flame-shaped," which generally occur within the nerve fiber layer and arise from the superficial capillary plexus, or "dot and blot," which occur within the spaces between the vertical-oriented axons of the inner plexiform layer and arise from the deep capillary plexus. Hemorrhages usually indicate a more serious form of DR than do microaneurysms.

Exudation from microaneurysms and leaking areas of the capillary beds lead to diabetic macular edema (DME) (Fig. 2.13). After resorption of interstitial fluid, lipids frequently precipitate and appear as yellow spots in the inner retina [Fig. 2.14]. These are often found at the junction of edematous and non-edematous retina, surrounding areas of microaneurysms.

Fig. 2.13 This macular photograph shows significant diabetic retinopathy with flameshaped and dot-and-blot hemorrhages. Scattered hard exudates are seen primarily in the superior macula, and macular edema obscures the underlying choroidal detail



Fig. 2.14 This posterior pole photograph shows nonproliferative diabetic retinopathy with diabetic macular edema and significant hard exudates



Fig. 2.15 This photograph of the posterior pole shows prominent cotton wool spots (soft exudates) in each quadrant. This indicates advancing retinal non-profusion and progressive ischemia

Cotton wool spots (CWS) and intraretinal microvascular abnormalities (IRMA) are usually associated with early retinal ischemia (Fig. 2.15). Cotton wool spots represent stasis of flow within the axons, whereas IRMA are vessels that shunt blood past occluded capillary beds.

Advanced retinal ischemia leads to the development of retinal neovascularization. New vessels sprout from pre-existing retinal (neovascularization elsewhere or NVE) or disc (neovascularization of the disc or NVD) vessels and grow along the scaffold provided by the posterior hyaloid surface (Fig. 2.16). The amount of blood vessel growth generally correlates with intraocular VEGF levels, with vessels growing toward higher VEGF concentrations. New vessel growth defines proliferative diabetic retinopathy (PDR) though patients may also display all the characteristics of nonproliferative diabetic retinopathy, including diabetic macular edema.





Most diabetes-related vision loss results from DME due to nonproliferative diabetic retinopathy. Severe vision loss (worse than 20/200) more commonly results from complications of PDR: vitreous hemorrhage, traction and traction/rhegmatogenous retinal detachments, neovascular glaucoma, and widespread retinal non-perfusion [60].

Though diabetic retinopathy is broadly categorized as nonproliferative and proliferative based on the presence of neovascularization, the severity of each condition varies from patient to patient. Nonproliferative DR can be subdivided into mild, moderate, severe, and very severe, depending on the extent of hemorrhages, retinal venous beading, and intraretinal microvascular abnormalities (IRMA) (Table 2.2). Though each of these sub-diagnoses carries a different prognosis, and the frequency of follow-up examinations is usually based on the sub-diagnosis, they are not usually used to make treatment decisions.

Over the past 30 years, macular edema has been classified according to two definitions:

- 1. Clinically significant macular edema (CSME). This was first used in the Early Treatment of Diabetic Retinopathy Study and is defined as any of the following:
 - (a) Thickening of the retina within 500 μ m of the fovea
 - (b) Hard exudates within 500 μ m of the fovea with adjacent retinal thickening
 - (c) Thickening of the retina at least 1 disc area in size that extends to within 1 disc diameter (DD) of the fovea

Thickening is determined by contact lens examination of the macula.

2. Diabetic macular edema (DME). This definition, rather than CSME, is used in the era of optical coherence tomography (OCT) and retinal pharmacotherapy. This is defined as a 2 DD area of macular thickening, any part of which is within 1 DD of the fovea. Thickening is determined by OCT scanning.

Historically, color fundus photography has been the "gold standard" for detecting diabetic retinopathy and assessing its severity. In clinical practice, treatment decisions were usually based on the results of biomicroscopic evaluation of the
Category Characteristics Nonproliferative diabetic retinopathy (NPDR)		
Nonproliferative diabetic retinopathy (NPDR)		
Mild NPDR Microaneurysms only		
ModerateMore than just microaneurysms (hemorrhages and exudates) but less severeNPDRthan severe NPDR		
Severe NPDR Any of the following but with no signs of proliferative		
"4:2:1 rule" retinopathy:		
Severe intraretinal hemorrhages and microaneurysms in		
each of the four retinal quadrants		
Venous beading in two or more quadrants		
Moderate intraretinal microvascular abnormalities (IRMA) in		
one or more quadrant		
Very severe Two or more criteria for severe NPDR but without		
NPDR neovascularization		
Proliferative diabetic retinopathy (PDR)		
PDR Neovascularization of disc or retina (elsewhere)		
Vitreous or pre-retinal hemorrhage		
Severe PDR At least three of the following characteristics:		
Neovascularization		
Neovascularization of the disc		
Neovascularization at least one-half disc diameter in size		
Vitreous or pre-retinal hemorrhage		

Table 2.2 This table classifies diabetic retinopathy according to severity

This popular classification system is used by many prospective and retrospective studies

macula, indirect ophthalmoscopic evaluation of the retina, or $30^{\circ}-50^{\circ}$ photographs of the fundus. Fluorescein angiography was usually used to identify areas of leaking in order to guide laser photocoagulation of the macula. Diabetic treatment trials usually required 7-field photographs, which provided views of the retina that extended to the mid-periphery.

Recent advances in technology enable physicians to take widefield and ultrawidefield laser-generated photographs. The Heidelberg Spectralis system with the Ocular Staurenghi 230 SLO Retina Lens (Ocular Instruments, Bellevue, Washington) enables photographers to take color and angiographic images that span 120°. The Optos C200MA noncontact SLO (Optos PLC, Dunfermline, UK) enables the photographer to take ultra-widefield photographs and fluorescein angiography (Fig. 2.17) images that span 200°. The system enables physicians to identify retinal non-perfusion anterior to the equator and may be superior to indirect ophthalmoscopy for the detection of peripheral neovascularization. The extent of peripheral retinal ischemia has been correlated with the severity of treatment-naïve DME, and peripheral ischemia may be identified in eyes where the DME responds poorly to pharmacotherapy. A retrospective study of eyes that responded poorly to anti-VEGF therapy showed that 80% had untreated peripheral non-perfusion and the poorest responders had the largest areas of non-perfusion. Many physicians perform targeted peripheral laser photocoagulation to eyes responding poorly to therapy, but its ability to resolve DME has not been established.

Fig. 2.17 This fluorescein angiogram image taken with an ultra-widefield system shows areas of peripheral capillary non-perfusion and retinal neovascularization in all quadrants. Pre-retinal hemorrhage is noted just beneath the inferotemporal arcade, and the macula is partially obscured by vitreous hemorrhage



2.9 Conclusions

Diabetic retinopathy has historically been considered a retinal vasculopathy, but recent data suggests that it affects the entire neurovascular unit. Diabetes results in dysfunction and eventual death of several key cells that maintain the BRB: pericytes, vascular endothelial cells, Müller cells, and neurons. Complex biochemical pathways upregulate inflammatory chemokines and cytokines that lead to the development of diabetic retinopathy, diabetic macular edema, and retinal neovascularization. Recent advances in our understanding of several important biochemical pathways have helped us identify potential therapeutic targets.

References

- Abiko T, Abiko A, Clermont AC, et al. Characterization of retinal leukostasis and hemodynamics in insulin resistance and diabetes: role of oxidants and protein kinase-C activation. Diabetes. 2003;52(3):829–37.
- Adamiec-Mroczek J, Oficjalska-Młyńczak J. Assessment of selected adhesion molecule and proinflammatory cytokine levels in the vitreous body of patients with type 2 diabetes – role of the inflammatory-immune process in the pathogenesis of proliferative diabetic retinopathy. Graefes Arch Clin Exp Ophthalmol. 2008;246:1665–70.
- Adamis AP, Berman AJ. Immunological mechanisms in the pathogenesis of diabetic retinopathy. Semin Immunopathol. 2008;30(2):65–84.
- Adamis AP, Miller JW, Bernal MT, et al. Increased vascular endothelial growth factor levels in the vitreous of eyes with proliferative diabetic retinopathy. Am J Ophthalmol. 1994;118:445–50.
- 5. Adler AJ, Severin KM. Proteins of the bovine interphotoreceptor matrix tissues of origin. Exp Eye Res. 1981;2:755–69.
- 6. Aguilera G, Kiss A. Regulation of the hypothalamic-pituitary-adrenal axis and vasopressin secretion. Role of angiotensin II. Adv Exp Med Biol. 1996;396:105–12.

- 7. Aiello LP, Bursell SE, Clermont A, et al. Vascular endothelial growth factor-induced retinal permeability is mediated by protein kinase C in vivo and suppressed by an orally effective beta-isoform-selective inhibitor. Diabetes. 1997;46:1473–80.
- 8. Antcliff RJ, Marshall J. The pathogenesis of edema in diabetic maculopathy. Semin Ophthalmol. 1999;14:223–32.
- Antonetti DA, Barber AJ, Bronson SK, Freeman WM, Gardner TW, Jefferson LS, Kester M, Kimball SR, Krady JK, LaNoue KF, et al. Diabetic retinopathy: seeing beyond glucoseinduced microvascular disease. Diabetes. 2006;55:2401–11.
- Antonetti DA, Barber AJ, Khin S, et al. Vascular permeability in experimental diabetes is associated with reduced endothelial occludin content: vascular endothelial growth factor decreases occludin in retinal endothelial cells. Penn State Retina Research Group. Diabetes. 1998;47:1953–9.
- Archer D, Krill AE, Newell FW. Fluorescein studies of normal choroidal circulation. Am J Ophthalmol. 1970;69:543–54.
- Arend O, Remky A, Harris A, et al. Macular microcirculation in cystoid maculopathy of diabetic patients [see comments]. Br J Ophthalmol. 1995;79:628–32.
- 13. Bainbridge JW, Mistry A, De Alwis M, et al. Inhibition of retinal neovascularization by gene transfer of soluble VEGF receptor sFlt-1. Gene Ther. 2002;9:320–6.
- 14. Ballantyne AJ, Loewenstein A. The pathology of diabetic retinopathy. Trans Ophthalmol Soc. 1943;63:95–113.
- Barber AJ, Antonetti DA, Kern TS, Reiter CE, Soans RS, Krady JK, Levison SW, Gardner TW, Bronson SK. The Ins2Akita mouse as a model of early retinal complications in diabetes. Invest Ophthalmol Vis Sci. 2005;46:2210–8.
- 16. Bates DO. The chronic effect of vascular endothelial growth factor on individually perfused frog mesenteric microvessels. J Physiol. 1998;513:225–33.
- Bates DO, Curry FE. Vascular endothelial growth factor increases hydraulic conductivity of isolated perfused microvessels. Am J Physiol. 1996;271:H2520–8.
- Bates DO, Hillman NJ, Williams B, et al. Regulation of microvascular permeability by vascular endothelial growth factors. J Anat. 2002;200:581–97.
- 19. Bek T, Lund-Andersen H. Localised blood-retinal barrier leakage and retinal light sensitivity in diabetic retinopathy. Br J Ophthalmol. 1990;74:388–92.
- Beltramo E, Pomero F, Allione A, D'Alu F, Ponte E, Porta M. Pericyte adhesion is impaired on extracellular matrix produced by endothelial cells in high hexose concentrations. Diabetologia. 2002;45:416–9.
- 21. Benjamin LE. Glucose, VEGF-A, and diabetic complications. Am J Pathol. 2001;158:1181-4.
- Bolton SJ, Anthony DC, Perry VH. Loss of the tight junction proteins occludin and zonula occludens-1 from cerebral vascular endothelium during neutrophil-induced blood-brain barrier breakdown in vivo. Neuroscience. 1998;86:1245–57.
- 23. Bolton WK, Cattran DC, Williams ME, et al. Randomized trial of an inhibitor of advanced glycation end products in diabetic nephropathy. Am J Nephrol. 2004;24:32–40.
- 24. Brooks Jr HL, Caballero Jr S, Newell CK, et al. Vitreous levels of vascular endothelial growth factor and stromal-derived factor 1 in patients with diabetic retinopathy and cystoid macular edema before and after intraocular injection of triamcinolone. Arch Ophthalmol. 2004;122:1801–7.
- 25. Brownlee M. Biochemistry and molecular cell biology of diabetic complications. Nature. 2001;414:813–20.
- 26. Brucklacher RM, Patel KM, Vanguilder HD, Bixler GV, Barber AJ, Antonetti DA, Lin CM, Lanoue KF, Gardner TW, Bronson SK, Freeman WM. Whole genome assessment of the retinal response to diabetes reveals a progressive neurovascular inflammatory response. BMC Med Genomics. 2006;1:26.
- Budzynski E, Wangsa-Wirawan N, Padnick-Silver L, Hatchell D, Linsenmeier R. Intraretinal pH in diabetic cats. Curr Eye Res. 2005;30:229–40.
- Bursell SE, Clermont AC, Kinsley BT, Simonson DC, Aiello LM, Wolpert HA. Retinal blood flow changes in patients with insulin-dependent diabetes mellitus and no diabetic retinopathy. Invest Ophthalmol Vis Sci. 1996;37:886–97.

- 29. Chakravarthy U, Gardiner TA, Anderson P, et al. The effect of endothelin I on the retinal microvascular pericyte. Microvasc Res. 1992;43:241–54.
- Charonis AS, Reger LA, Dege JE, et al. Laminin alterations after in vitro nonenzymatic glycosylation. Diabetes. 1990;39:807–14.
- 31. Chaturvedi N, Sjolie AK, Stephenson JM, et al. Effect of lisinopril on progression of retinopathy in normotensive people with type 1 diabetes. The EUCLID Study Group. EURODIAB Controlled Trial of Lisinopril in insulin-Dependent Diabetes Mellitus. Lancet. 1998;351:28–31.
- 32. Chen P, Guo AM, Edwards PA, et al. Role of NADPH oxidase and ANG II in diabetes-induced retinal leukostasis. Am J Physiol Regul Integr Comp Physiol. 2007;293:R1619–29.
- Ciulla TA, Harris A, Latkany P, Piper HC, Arend O, Garzozi H, Martin B. Ocular perfusion abnormalities in diabetes. Acta Ophthalmol Scand. 2002;80:468–77.
- Clarke H, Marano CW, Peralta Soler A, Mullin JM. Modification of tight junction function by protein kinase C isoforms. Adv Drug Deliv Rev. 2000;41:283–301.
- 35. Cogan DG, Toussaint D, Kuwabara T. Retinal vascular patterns. Part IV. Diabetic retinopathy. Arch Ophthalmol. 1961;66:366–78.
- 36. Cox O, Stitt AW, Simpson DA, Gardiner TA. Sources of PDGF expression in murine retina and the effect of short-term diabetes. Mol Vis. 2003;10:665–72.
- 37. Craven PA, Phillip SL, Melham MF, et al. Overexpression of Mn²⁺ superoxide dismutase increases in collagen accumulation induced by culture in mesangial cells in high-media glucose. Metabolism. 2001;50:1043–8.
- Craven PA, Studer RK, Felder J, et al. Nitric oxide inhibition of transforming growth factorbeta and collagen synthesis in mesangial cells. Diabetes. 1997;46:671–81.
- 39. Cringle S, Yu DY, Alder V, Su EN. Oxygen tension and blood flow in the retina of normal and diabetic rats. Adv Exp Med Biol. 1992;317:787–91.
- 40. Culman J, Hohle S, Qadri F, et al. Angiotensin as neuromodulator/neurotransmitter in central control of body fluid and electrolyte homeostasis. Clin Exp Hypertens. 1995;17:281–93.
- 41. Cunha-Vaz J, Faria de Abreu JR, Campos AJ. Early breakdown of the blood-retinal barrier in diabetes. Br J Ophthalmol. 1975;59:649–56.
- 42. Curcio CA, Allen KA. Topography of ganglion cells in human retina. J Comp Neurol. 1990;300:5–25.
- Curtis TM, Gardiner TA, Stitt AW. Microvascular lesions of diabetic retinopathy: clues towards understanding pathogenesis? Eye (Lond). 2009;23:1496–508.
- Degenhardt TP, Thorpe SR, Baynes JW. Chemical modification of proteins by methylglyoxal. Cell Mol Biol. 1998;44:1139–45.
- 45. de Gooyer TE, Stevenson KA, Humphries P, et al. Retinopathy is reduced during experimental diabetes in a mouse model of outer retinal degeneration. Invest Ophthalmol Vis Sci. 2006;47:5561–8.
- 46. Dodge AB, Hechtman HB, Shepro D. Microvascular endothelial-derived autacoids regulate pericyte contractility. Cell Motil Cytoskeleton. 1991;18:180–8.
- Dowling JE, Boycott BB. Organization of the primate retina electron microscopy. Proc R Soc Ser B. 1966;166:80–111.
- 48. Droge W. Free radicals and the physiological control of cell function. Physiol Rev. 2002;83:47–95.
- 49. Du XL, Edelstein D, Dimmeler S, et al. Hyperglycemia inhibits endothelial nitric oxide synthase activity by posttranslational modification at the Akt site. J Clin Invest. 2001;108:1341–8.
- 50. Du XL, Edelstein D, Rossetti L, et al. Hyperglycemia-induced mitochondrial superoxide overproduction activates the hexosamine pathway and induces plasminogen activator inhibitor-1 expression by increasing Sp1 glycosylation. Proc Natl Acad Sci U S A. 2000;97:12222–6.
- Engerman RL, Kern TX, Larson ME. Nerve conduction and aldose reductase inhibition during 5 years of diabetes or galactosaemia in dogs. Diabetologia. 1994;37:141–4.
- Ernest JT. Macrocirculation and microcirculation of the retina. In: Ryan SJ, Ogden TE, editors. Retina, vol. 1. St. Louis: CV Mosby; 1989. p. 65–6.
- Fanning AS, Ma TY, Anderson JM. Isolation and functional characterization of the actin binding region in the tight junction protein ZO-1. FASEB J. 2002;16:1835–7.

- Feeney-Burns L, Burns RP, Gao C-L. Age-related macular changes in humans over 90 years old. Am J Ophthalmol. 1990;109:265–78.
- 55. Felinski EA, Antonetti DA. Glucocorticoid regulation of endothelial cell tight junction gene expression: novel treatments for diabetic retinopathy. Curr Eye Res. 2005;30:949–57.
- Ferrara N. Role of vascular endothelial growth factor in the regulation of angiogenesis. Kidney Int. 1999;56:794–814.
- Ferrara N. Vascular endothelial growth factor: basic science and clinical progress. Endocr Rev. 2004;25:581–611.
- Ferrari-Dileo G, Davis EB, Anderson DR. Response of retinal vasculature to phenylephrine. Invest Ophthalmol Vis Sci. 1990;30:1181–2.
- 59. Fine BS, Brucker AJ. Macular edema and cystoid macular edema. Am J Ophthalmol. 1981;92:466–81.
- Fong DS, Aiello L, Gardner TW, et al. American Diabetes Association. Retinopathy in diabetes. Diabetes Care. 2004;27(Suppl 4):S84–7.
- Fruttiger M. Development of the mouse retinal vasculature: angiogenesis versus vasculogenesis. Invest Ophthalmol Vis Sci. 2002;43:522–7.
- 62. Fukumura D, Gohongi T, Kadambi A, et al. Predominant role of endothelial nitric oxide synthase in vascular endothelial growth factor-induced angiogenesis and vascular permeability. Proc Natl Acad Sci U S A. 2001;98:2604–9.
- Fukushima I, McLeod DS, Lutty GA. Intrachoroidal microvascular abnormality: a previously unrecognized form of choroidal neovascularization. Am J Ophthalmol. 1997;124:473–87.
- 64. Funatsu H, Yamashita H, Ikeda T, et al. Angiotensin II and vascular endothelial growth factor in the vitreous fluid of patients with proliferative diabetic retinopathy. Br J Ophthalmol. 2002;86:311–5.
- 65. Funatsu H, Yamashita H, Ikeda T, et al. Vitreous levels of interleukin-6 and vascular endothelial growth factor are related to diabetic macular edema. Ophthalmology. 2003;110:1690–6.
- Gardiner TA, Archer DB, Curtis TM, Stitt AW. Arteriolar involvement in the microvascular lesions of diabetic retinopathy: implications for pathogenesis. Microcirculation. 2007;14:25–38.
- Gardiner TA, Stitt AW, Anderson HR, Archer DB. Selective loss of vascular smooth muscle cells in the retinal microcirculation of diabetic dogs. Br J Ophthalmol. 1994;78:54–60.
- Gardner TW, Lieth E, Khin SA, et al. Astrocytes increase barrier properties and ZO-1 expression in retinal vascular endothelial cells. Invest Ophthalmol Vis Sci. 1997;38:2423–7.
- Gass JDM, Anderson DR, Davis EB. A clinical, fluorescein angiographic, and electron microscopic correlation of cystoid macular edema. Am J Ophthalmol. 1985;100:82–6.
- Gaudry M, Bregerie O, Andrieu V, et al. Intracellular pool of vascular endothelial growth factor in human neutrophils. Blood. 1997;90:4153–61.
- Gehlbach P, Demetriades AM, Yamamoto S, et al. Periocular gene transfer of sFlt-1 suppresses ocular neovascularization and vascular endothelial growth factor-induced breakdown of the blood-retinal barrier. Hum Gene Ther. 2003;14:129–41.
- 72. Gerhardinger C, Dagher Z, Sebastiani P, Park YS, Lorenzi M. The transforming growth factor-β pathway is a common target of drugs that prevent experimental diabetic retinopathy. Diabetes. 2009;58:1659–67.
- Giardino I, Edelstein D, Brownlee M. BCL-2 expression or antioxidants prevent hyperglycemiainduced formation of intracellular advanced glycation end-products in bovine endothelial cells. J Clin Invest. 1996;97:1422–8.
- Giebel SJ, Menicucci G, McGuire PG, Das A. Matrix metalloproteinases in early diabetic retinopathy and their role in alteration of the blood-retinal barrier. Lab Invest. 2005;85:597–607.
- Gilbert RD, Kelly DJ, Cox AJ, et al. Angiotensin converting enzyme inhibition reduces retinal overexpression of vascular endothelial growth factor and hyperpermeability in experimental diabetes. Diabetologia. 2000;43:1360–7.
- 76. Gillies MC, Su T, Stayl J, et al. Effect of high glucose on permeability of retinal capillary endothelium in vitro. Invest Ophthalmol Vis Sci. 1997;38:635–42.
- Greene DA, Arezzo JC, Brown MB. Effect of aldose reductase inhibition on nerve conduction and morphometry in diabetic neuropathy. Zenarestat Study Group. Neurology. 1999;53:580–91.

- 78. Grunwald JE, Riva CE, Baine J, Brucker AJ. Total retinal volumetric blood flow rate in diabetic patients with poor glycemic control. Invest Ophthalmol Vis Sci. 1992;33:356–63.
- Guigliano D, Ceriello A, Paolisso G. Oxidative stress and diabetic vascular complications. Diabetes Care. 1996;19:257–67.
- Hammes HP, Lin J, Renner O, Shani M, Lundqvist A, Betsholtz C, Brownlee M, Deutsch U. Pericytes and the pathogenesis of diabetic retinopathy. Diabetes. 2002;51:3107–12.
- Handelman GJ, Snodderly DM, Krinsky NI, et al. Biological control of primate macular pigment – biochemical and densitometric studies. Invest Ophthalmol Vis Sci. 1991;32:257–67.
- Hart GW. Dynamic O-linked glycosylation of nuclear and cytoskeletal proteins. Annu Rev Biochem. 1997;66:315–35.
- Hayflick L. The limited *in vitro* lifetime of human diploid cell strains. Exp Cell Res. 1965;37:614–36.
- 84. Henkind P. New observations on the radial peripapillary capillaries. Invest Ophthalmol. 1967;6:103.
- Hewitt AT, Adler R. The retinal pigment epithelium and interphotoreceptor matrix structure and specialized functions. In: Ryan SJ, Ogden TE, editors. Retina, vol. 1. St. Louis: CV Mosby; 1989. p. 57–64.
- Hoffmann J, Feng Y, von Hagen F, Hillenbrand A, Lin J, Erber R, Vajkoczy P, Gourzoulidou E, Waldmann H, Giannis A, et al. Endothelial survival factors and spatial completion, but not pericyte coverage of retinal capillaries determine vessel plasticity. FASEB J. 2005;19:2035–6.
- Hogan MJ, Alvarado JA, Weddell JE. The retina. In: Histology of the human eye. Philadelphia: WB Saunders; 1971. p. 393–522.
- Hogan MJ, Feeney L. Ultrastructure of the retinal vessels. Part 1. The larger vessels. J Ultrastruct Res. 1963;9:10–28.
- Horie K, Miyata T, Maeda K, et al. Immunohistochemical colocalization of glycoxidation products and lipid peroxidation products in diabetic renal glomerular lesions. Implication for glycoxidative stress in the pathogenesis of diabetic nephropathy. J Clin Invest. 1997;100:2995–3004.
- Hseuh WA, Law RE. Cardiovascular risk continuum: implications of insulin resistance and diabetes. Am J Med. 1998;105:4S–14S.
- 91. Huijberts MSP, Wolffenbuttel BH, Boudier HA, et al. Aminoguanidine treatment increases elasticity and decreases fluid filtration of large arteries from diabetic rats. J Clin Invest. 1993;92:1407–11.
- 92. Hussain SP, Harris CC. Inflammation and cancer: an ancient link with novel potentials. Int J Cancer. 2007;121:2372–80.
- 93. Ishii H, Jirousek MR, Koya D, et al. Amelioration of vascular dysfunctions in diabetic rats by an oral PKC beta inhibitor. Science. 1996;272:728–31.
- 94. Ito M, Oliverio MI, Mannon PJ, et al. Regulation of blood pressure by the type 1A angiotensin II receptor gene. Proc Natl Acad Sci U S A. 1995;92:3521–5.
- Jaakkola P, Mole DR, Tian YM, et al. Targeting of HIF-alpha to the von Hippel-Lindau ubiquitylation complex by O₂-regulated prolyl hydroxylation. Science. 2001;292:468–72.
- Jerdan JA, Kao L, Glaser BM. The inner limiting membrane: a modified basement membrane? Invest Ophthalmol Vis Sci. 1986;27(suppl):230a.
- Jones CW, Cunha-Vaz J, Zweig KO, Stein M. Kinetic vitreous fluorophotometry in experimental diabetes. Arch Ophthalmol. 1979;97:1941–3.
- Joussen AM, Poulaki V, Le ML, Koizumi K, Esser C, Janicki H, Schraermeyer U, Kociok N, Fauser S, Kirchhof B, et al. A central role for inflammation in the pathogenesis of diabetic retinopathy. FASEB J. 2004;18:1450–2.
- 99. Joussen AM, Poulaki V, Mitsiades N, et al. Nonsteroidal anti-inflammatory drugs prevent early diabetic retinopathy *via* TNF alpha suppression. FASEB J. 2002;16:438–40.
- Kaiser N, Sasson S, Feener EP, et al. Differential regulation of glucose transport and transporters by glucose in vascular endothelial and smooth muscle cells. Diabetes. 1993;42:80–9.
- 101. Kato H, Suzuki H, Tajima S, et al. Angiotensin II stimulates collagen synthesis in cultured vascular smooth muscle cells. J Hypertens. 1991;9:17–22.

- 102. Kawamura H, Kobayashi M, Li Q, et al. Effects of angiotensin II on the pericyte-containing microvasculature of the rat retina. J Physiol. 2004;561:671–83.
- 103. Keough RJ, Dunlop ME, Larkins RG. Effect of inhibition of aldose reductase on glucose flux, diacylglycerol formation, protein kinase C, and phospholipase A2 activation. Metabolism. 1997;46:41–7.
- Khan ZA, Farhangkhoee H, Chakrabarti S. Towards newer molecular targets for chronic diabetic complications. Curr Vasc Pharmacol. 2006;4:45–57.
- Kim S, Iwao H. Molecular and cellular mechanisms of angiotensin II-mediated cardiovascular and renal diseases. Pharmacol Rev. 2000;52:11–34.
- 106. Kiuchi-Saichin Y, Gotoh S, Furuse M, et al. Differential expression patterns of claudins, tight junction membrane proteins, in mouse nephron segments. J Am Soc Nephrol. 2002;13:875–86.
- 107. Klein R, Klein BE, Moss SE, Wong TY, Hubbard L, Cruickshanks KJ, Palta M. Retinal vascular abnormalities in persons with type 1 diabetes: the Wisconsin Epidemiologic Study of Diabetic Retinopathy: XVIII. Ophthalmology. 2003;110:2118–25.
- Klein R, Myers CE, Lee KE, Gangnon R, Klein BE. Changes in retinal vessel diameter and incidence and progression of diabetic retinopathy. Arch Ophthalmol. 2012;130:749–55.
- 109. Kocak N, Alacacioglu I, Kaynak S, Ozcan MA, Celik O, Yuksel F, Piskin O, Oner H, Saatci AO, Ergin M. Comparison of vitreous and plasma levels of vascular endothelial growth factor, interleukin-6 and hepatocyte growth factor in diabetic and non-diabetic retinal detachment cases. Ann Ophthalmol (Skokie). 2010;42 Spec No:10–4.
- 110. Kohner EM. The retinal blood flow in diabetes. Diabetes Metab. 1993;19:401-4.
- Kohner EM, Dollery CT. Fluorescein angiography of the fundus in diabetic retinopathy. Br Med Bull. 1970;26:166–70.
- 112. Kohner EM, Hamilton AM, Saunders SJ, Sutcliffe BA, Bulpitt CJ. The retinal blood flow in diabetes. Diabetologia. 1975;11:27–33.
- 113. Kohner EM, Henkind P. Correlation of fluorescein angiogram and retinal digest in diabetic retinopathy. Am J Ophthalmol. 1970;69:403–14.
- 114. Kohner EM, Patel V, Rassam SM. Role of blood flow and impaired autoregulation in the pathogenesis of diabetic retinopathy. Diabetes. 1995;44:603–7.
- 115. Kokame GT, de Leon MD, Tanji T. Serous retinal detachment and cystoid macular edema in hypotony maculopathy. Am J Ophthalmol. 2001;131:384–6.
- 116. Koya D, Haneda M, Nakagawa H, et al. Amelioration of accelerated diabetic mesangial expansion by treatment with a PKC beta inhibitor in diabetic db/db mice, a rodent model for type 2 diabetes. FASEB J. 2000;14:439–47.
- Koya D, King GL. Protein kinase C activation and the development of diabetic complications. Diabetes. 1998;47:859–66.
- 118. Kristinsson JK, Gottfredsdotter MS, Stefannson E. Retinal vessel dilatation and elongation precedes diabetic macular oedema. Br J Ophthalmol. 1997;81:274–8.
- 119. Kuiper EJ, Witmer AN, Klaassen I, Oliver N, Goldschmeding R, Schlingemann RO. Differential expression of connective tissue growth factor in microglia and pericytes in the human diabetic retina. Br J Ophthalmol. 2004;88:1082–7.
- 120. Langham ME, Grebe R, Hopkins S, et al. Choroidal blood flow in diabetic retinopathy. Exp Eye Res. 1991;52:167–73.
- 121. Lee JH, Lee W, Kwon OH, et al. Cytokine profile of peripheral blood in type 2 diabetes mellitus patients with diabetic retinopathy. Ann Clin Lab Sci. 2008;38:361–7.
- 122. Lewis H, Abrams GW, Blumenkranz MS, Campo RV. Vitrectomy for diabetic macular traction and edema associated with posterior hyaloids traction. Ophthalmology. 1992;99:753–9.
- 123. Li G, Veenstra AA, Talahalli RR, Wang X, Gubitosi-Klug RA, Sheibani N, Kern TS. Marrowderived cells regulate the development of early diabetic retinopathy and tactile allodynia in mice. Diabetes. 2012;61:3294–303.
- 124. Lieth E, Barber AJ, Xu B, Dice C, Ratz MJ, Tanase D, Strother JM. Glial reactivity and impaired glutamate metabolism in short-term experimental diabetic retinopathy. Diabetes. 1998;47:815–20.

- Linsenmeier RA, Braun RD, McRipley MA, et al. Retinal hypoxia in long-term diabetic cats. Invest Ophthalmol Vis Sci. 1998;39:1647–57.
- Linskens MH, Harley CB, West MD, Campisi J, Hayflick L. Replicative senescence and cell death. Science. 1995;267:17.
- 127. Liu H, Ren JG, Cooper WL, et al. Identification of the antivasopermeability effect of pigment epithelium-derived factor and its active site. Proc Natl Acad Sci U S A. 2004;101:6605–10.
- 128. Ljubimov AV, Burgeson RE, Butkowski RJ, Couchman JR, Zardi L, Ninomiya Y, Sado Y, Huang ZS, Nesburn AB, Kenney MC. Basement membrane abnormalities in human eyes with diabetic retinopathy. J Histochem Cytochem. 1996;44:1469–79.
- 129. Lu M, Kuroki M, Amano S, et al. Advanced glycation end products increase retinal vascular endothelial growth factor expression. J Clin Invest. 1998;101:1219–24.
- Lu M, Perez VL, Ma N, et al. VEGF increases retinal vascular ICAM-1 expression in vivo. Invest Ophthalmol Vis Sci. 1999;40:1808–12.
- 131. Lucis AJ. Atherosclerosis. Nature. 2000;407:233-41.
- 132. Matsugi T, Chen Q, Anderson DR. Contractile responses of cultured bovine retinal pericytes to angiotensin II. Arch Ophthalmol. 1997;115:1281–5.
- 133. Matter K, Balda MS. Occludin and the functions of tight junctions. Int Rev Cytol. 1999;186:117-46.
- 134. Matthews DR, Stratton IM, Aldington SJ, et al. Risks of progression of retinopathy and vision loss related to tight blood pressure control in type 2 diabetes mellitus: UKPDS 69. Arch Ophthalmol. 2004;122:1631–40.
- McLeod D. Why cotton wool spots should not be regarded as retinal nerve fiber layer infarcts. Br J Ophthalmol. 2005;89:229–37.
- 136. Miller JW, Adamis AP, Aiello LP. Vascular endothelial growth factor in ocular neovascularization and proliferative diabetic retinopathy. Diabetes Metab Rev. 1997;13:37–50.
- Mizutani M, Gerhardinger C, Lorenzi M. Muller cell changes in human diabetic retinopathy. Diabetes. 1998;47:445–9.
- 138. Mizutani M, Kern TS, Lorenzi M. Accelerated death of retinal microvascular cells in human and experimental diabetic retinopathy. J Clin Invest. 1996;97:2883–90.
- Nagai N, Izumi-Nagai K, Oike Y, et al. Suppression of diabetes-induced retinal inflammation by blocking the angiotensin II type 1 receptor or its downstream nuclear factor-κB pathway. Invest Ophthalmol Vis Sci. 2007;48:4342–50.
- 140. Nagai N, Noda K, Urano T, et al. Selective suppression of pathologic, but not physiologic, retinal neovascularization by blocking the angiotensin II type 1 receptor. Invest Ophthalmol Vis Sci. 2005;46:1078–84.
- 141. Nitta T, Hata M, Gotoh S, et al. Size-selective loosening of the blood-brain barrier in claudin-5 deficient mice. J Cell Biol. 2003;161:653–60.
- 142. Ogden TE. The glia of the retina. In: Ryan SJ, Ogden TE, editors. Retina, vol. 1. St. Louis: CV Mosby; 1989. p. 53–6.
- 143. Otani A, Takagi H, Oh H, et al. Angiotensin II induces expression of the Tie2 receptor ligand, angiopoietin-2, in bovine retinal endothelial cells. Diabetes. 2001;50:867–75.
- 144. Otani A, Takagi H, Suzuma K, Honda Y. Angiotensin II potentiates endothelial growth factor-induced angiogenic activity in retinal microcapillary endothelial cells. Circ Res. 1998;82:619–28.
- 145. Ozaki H, Hayashi H, Vinores SA, et al. Intravitreal sustained release of VEGF causes retinal neovascularization in rabbits and breakdown of the blood-retinal barrier in rabbits and primates. Exp Eye Res. 1997;64:505–17.
- 146. Padayatti PS, Jiang C, Glomb MA, Uchida K, Nagaraj RH. High concentrations of glucose induce synthesis of argpyrimidine in retinal endothelial cells. Curr Eye Res. 2001;23:106–15.
- 147. Pan HZ, Zhang H, Chang D, et al. The change of oxidative stress products in diabetes mellitus and diabetic retinopathy. Br J Ophthalmol. 2008;92:548–51.
- 148. Patel JI, Tombran-Tink J, Hykin PG, et al. Vitreous and aqueous concentrations of proangiogenic, antiangiogenic factors and other cytokines in diabetic retinopathy patients with macular edema: implications for structural differences in macular profiles. Exp Eye Res. 2006;82:798–806.

- Paul SA, Simons JW, Mabjeesh NJ. HIF at the crossroads between ischemia and carcinogenesis. J Cell Physiol. 2004;200:20–30.
- 150. Pfeffer BA. Improved methodology for cell culture of human and monkey retinal pigment epithelium. Prog Retinal Res. 1991;10:251.
- 151. Pieper GM, Riaz-ul-Haq J. Activation of nuclear factor-kappa B in cultured endothelial cells by increased glucose concentration: prevention by calphostin C. J Cardiovasc Pharmacol. 1997;30:528–32.
- 152. Pollock SC, Miller NR. The retinal nerve fiber layer. Int Ophthalmol Clin. 1986;26:201-21.
- 153. Portilla D, Dai G, Peters JM, et al. Etomoxir-induced PPARalpha-modulated enzymes protect during acute renal failure. Am J Physiol Renal Physiol. 2000;278:F667–75.
- 154. Poulaki V, Qin W, Joussen AM, et al. Acute intensive insulin therapy exacerbates diabetic blood-retinal barrier breakdown via hypoxia-inducible factor-1alpha and VEGF. J Clin Invest. 2002;109:805–15.
- 155. Puro DG. Diabetes-induced dysfunction of retinal Muller cells. Trans Am Ophthalmol Soc. 2002;100:339–52.
- 156. Reuter S, Gupta SC, Chaturvedi MM, Aggarwal BB. Oxidative stress, inflammation, and cancer: how are they linked? Free Radic Biol Med. 2010;49(11):1603–16.
- 157. Roy MS, Gunkel RD, Podgor MJ. Color vision defects in early diabetic retinopathy. Arch Ophthalmol. 1986;104:225–8.
- 158. Roy S, Maiello M, Lorenzi M. Increased expression of basement membrane collagen in human diabetic retinopathy. J Clin Invest. 1994;93:438–42.
- Roy S, Tonkiss J, Roy S. Aging increases retinal vascular lesions characteristic of early diabetic retinopathy. Biogerontology. 2010;11:447–55.
- Rungger-Brandle E, Dosso AA, Leuenberger PM. Glial reactivity, an early feature of diabetic retinopathy. Invest Ophthalmol Vis Sci. 2000;41:1971–80.
- Safran M, Kaelin Jr WG. HIF hydroxylation and the mammalian oxygen-sensing pathway. J Clin Invest. 2003;111(6):779–83.
- 162. Saishin Y, Saishin Y, Takahashi K, et al. VEGF-TRAP (R1R2) suppresses choroidal neovascularization and VEGF-induced breakdown of the blood-retinal barrier. J Cell Physiol. 2003;195:241–8.
- Schmetterer L, Wolzt M. Ocular blood flow and associated functional deviations in diabetic retinopathy. Diabetologia. 1999;42:387–405.
- 164. Schmidt AM, Hori O, Chen JX, et al. Advanced glycation end-products interacting with their endothelial receptor induce expression of vascular cell adhesion molecule-1 (VCAM-1) in cultured human endothelial cells and in mice: a potential mechanism for the accelerated vasculopathy of diabetes. J Clin Invest. 1995;96:1395–403.
- 165. Schroedl C, McClintock DS, Budinger GR, Chandel NS. Hypoxic but not anoxic stabilization of HIF-1alpha requires mitochondrial reactive oxygen species. Am J Physiol Lung Cell Mol Physiol. 2002;283:L922–31.
- 166. Semenza G. Signal transduction to hypoxia-inducible factor 1. Biochem Pharmacol. 2002;64:993-8.
- 167. Shakib M, Cunha-Vaz JG. Studies on the permeability of the blood-retinal barrier. IV. Junctional complexes of the retinal vessels and their role in the permeability of the blood-retinal barrier. Exp Eye Res. 1966;5:229–34.
- 168. Shin DH, Tsai CS, Parrow KA, et al. Vasoconstrictive effect of topical timolol on human retinal arteries. Graefes Arch Clin Exp Ophthalmol. 1991;229:298–9.
- 169. Sigelman J, Ozanics V. Retina. In: Duane's foundations of clinical ophthalmology. Philadelphia: Lippincott; 1990.
- 170. Simo R, Carrasco E, Garcia-Ramirez M, Hernandez C. Angiogenic and antiangiogenic factors in proliferative diabetic retinopathy. Curr Diabetes Rev. 2006;2:71–98.
- 171. Sinclair SH. Macular retinal capillary hemodynamics in diabetic patients. Ophthalmology. 1991;98:1580–6.
- 172. Skovborg F, Nielsen AV, Lauritzen E, Hartkopp O. Diameters of the retinal vessels in diabetic and normal subjects. Diabetes. 1969;18:292–8.

- 173. Smelser GK, Ishikawa T, Pei YF. Electron microscopic studies of intra-retinal spaces diffusion of particulate materials. In: Rohen JW, editor. Structure of the eye, II Symp. Stuttgart: Schattauer-Verlay; 1965. p. 109–21.
- 174. Smith RT, Lee CM, Charles HC, et al. Quantification of diabetic macular edema. Arch Ophthalmol. 1987;105:218–22.
- 175. Sokol S, Moskowitz A, Skarf B, Evans R, Molitch M, Senior B. Contrast sensitivity in diabetics with and without background retinopathy. Arch Ophthalmol. 1985;103:51–4.
- 176. Soulis-Liparota T, Cooper M, Papazoglou D, et al. Retardation by aminoguanidine of development of albuminuria, mesangial expansion, and tissue fluorescence in streptozotocininduced diabetic rat. Diabetes. 1991;40:1328–34.
- 177. Stasek Jr JE, Patterson CE, Garcia JG. Protein kinase C phosphorylates caldesmon 77 and vimentin and enhances albumin permeability across cultured bovine pulmonary artery endothelial cell monolayers. J Cell Physiol. 1992;153:62–75.
- 178. Stefansson E. The therapeutic effects of retinal laser treatment and vitrectomy. A theory based on oxygen and vascular physiology. Acta Ophthalmol Scand. 2001;79:435–40.
- Stefansson E. Ocular oxygenation and the treatment of diabetic retinopathy. Surv Ophthalmol. 2006;51:364–80.
- Stefansson E, Landers 3rd MB, Wolbarsht ML. Increased retinal oxygen supply following pan-retinal photocoagulation and vitrectomy and lensectomy. Trans Am Ophthalmol Soc. 1981;79:307–34.
- Stevenson BR, Keon BH. The tight junction: morphology to molecules. Annu Rev Cell Dev Biol. 1998;14:89–109.
- 182. Stitt A, Gardiner TA, Alderson NL, Canning P, Frizzell N, Duffy N, Boyle C, Januszewski AS, Chachich M, Baynes JW, Thorpe SR. The AGE inhibitor pyridoxamine inhibits development of retinopathy in experimental diabetes. Diabetes. 2002;51:2826–32.
- 183. Stitt AW, Anderson HR, Gardiner TA, Archer DB. Diabetic retinopathy: quantitative variation in capillary basement membrane thickening in arterial or venous environments. Br J Ophthalmol. 1994;78:133–7.
- 184. Stitt AW, Hughes SJ, Canning P, Lynch O, Cox O, Frizzell N, Thorpe SR, Cotter TG, Curtis TM, Gardiner TA. Substrates modified by advanced glycation end-products cause dysfunction and death in retinal pericytes by reducing survival signals mediated by platelet-derived growth factor. Diabetologia. 2004;47:1735–46.
- 185. Stitt AW, Li YM, Gardiner TA, et al. Advanced glycation end products (AGEs) co-localize with AGE receptors in the retinal vasculature of diabetic and of AGE-infused rats. Am J Pathol. 1997;150:523–8.
- 186. Stratton IM, Kohner EM, Aldington SJ, et al. UKPDS 50: risk factors for incidence and progression of retinopathy in Type II diabetes over 6 years from diagnosis. Diabetologia. 2001;44:156–63.
- 187. Studer RK, Craven PA, DeRubertis FR. Role for protein kinase C in the mediation of increased fibronectin accumulation by mesangial cells grown in high-glucose medium. Diabetes. 1993;42:118–26.
- Suzuki Y, Ruiz-Ortega M, Lorenzo O, et al. Inflammation and angiotensin II. Int J Biochem Cell Biol. 2003;35:881–900.
- 189. Talahalli R, Zarini S, Tang J, Li G, Murphy R, Kern TS, Gubitosi-Klug RA. Leukocytes regulate retinal capillary degeneration in the diabetic mouse via generation of leukotrienes. J Leukoc Biol. 2013;93:135–43.
- 190. Tamura K, Nyui N, Tamura N, et al. Mechanism of angiotensin II-mediated regulation of fibronectin gene in rat vascular smooth muscle cells. J Biol Chem. 1998;273:26487–96.
- 191. Tang J, Kern TS. Inflammation in diabetic retinopathy. Prog Retin Eye Res. 2011;30:343–58.
- 192. Thornalley PJ. The glyoxalase system: new developments towards functional characterization of a metabolic pathway fundamental to biological life. Biochem J. 1990;269:1–11.
- 193. Tso MO. Pathological study of cystoid macular oedema. Trans Ophthalmol Soc U K. 1980;100:408–13.

- 194. Tso MO. Pathology of cystoid macular edema. Ophthalmology. 1982;89:902-15.
- 195. Vinores SA, Van Niel E, Swerrdloff IL, et al. Electron microscopic immunocytochemical evidence for the mechanism of blood-retinal barrier breakdown in galactosemic rats and its association with aldose reductase expression and inhibition. Exp Eye Res. 1993;57:723–35.
- 196. Wagener HP, Story DTD, Wilder RM. Retinitis in diabetes. N Engl J Med. 1934;211:1131–7.
- 197. Wald G, Brown PK. Human rhodopsin. Science. 1958;127:222-6.
- 198. Wang GL, Jiang BH, Rue EA, Semenza GL. Hypoxia-inducible factor 1 is a basic-helix-loop-helix-PAS heterodimer regulated by cellular O₂ tension. Proc Natl Acad Sci U S A. 1995;92:5510–4.
- 199. Wang W, Dentler WL, Borchardt RT. VEGF increases BMED monolayer permeability by affecting occludin expression and tight junction assembly. Am J Physiol Heart Circ Physiol. 2001;280:H434–40.
- Weinberger D, Fink-Cohen S, Gaton DD, et al. Non-retinovascular leakage in diabetic maculopathy. Br J Ophthalmol. 1995;79:728–31.
- Weiner CM, Booth G, Semenza GL. In vivo expression of mRNAs encoding hypoxiainducible factor 1. Biochem Biophys Res Commun. 1996;225:485–8.
- 202. Wells-Knecht KJ, Zyzak DV, Litchfield JE, et al. Mechanism of autoxidative glycosylation: identification of glyoxal and arabinose as intermediates in the autoxidative modification of proteins by glucose. Biochemistry. 1995;34:3702–9.
- 203. Wilkinson-Berka J. Angiotensin and diabetic retinopathy. Int J Biochem Cell Biol. 2006;38:752–65.
- 204. Williams B, Gallacher B, Patel H, Orme C. Glucose-induced protein kinase C activation regulates vascular permeability factor mRNA expression and peptide production by human vascular smooth muscle cells *in vitro*. Diabetes. 1997;46:1497–503.
- Williamson JR, Chang K, Frangos M, et al. Hyperglycemic pseudohypoxia and diabetic complications. Diabetes. 1993;42:801–13.
- 206. Wise GN, Dollery CT, Henkind P. The retinal circulation. New York: Harper & Row; 1971.
- 207. Wong A, Merritt S, Butt AN, et al. Effect of hypoxia on circulating levels of retina-specific messenger RNA in type 2 diabetes mellitus. Ann NY Acad Sci. 2008;1137:243–52.
- 208. Wu HM, Huang Q, Yuan Y, Granger HJ. VEGF induces NO-dependent hyperpermeability in coronary venules. Am J Physiol. 1996;271:H2735–9.
- 209. Yamada E. Some structural features of the fovea centralis in the human retina. Arch Ophthalmol. 1969;82:151–9.
- Yamagishi SI, Edelstein D, Du XL, Brownlee M. Hyperglycemia potentiates collageninduced platelet activation through mitochondrial superoxide overproduction. Diabetes. 2001;50:1491–4.
- Yerneni KK, Bai W, Khan BV, et al. Hyperglycemia-induced activation of nuclear transcription factor kappa B in vascular smooth muscle cells. Diabetes. 1999;48:855–64.
- Yonemura D, Aoki T, Tsuzuki K. Electroretinogram in diabetic retinopathy. Arch Ophthalmol. 1962;68:19–24.
- Zeng HY, Green WR, Tso MO. Microglial activation in human diabetic retinopathy. Arch Ophthalmol. 2008;126:227–32.
- Zeng XX, Ng YK, Ling EA. Neuronal and microglial response in the retina of streptozotocininduced diabetic rats. Vis Neurosci. 2000;17:463–71.
- Zhang J, Gerhardinger C, Lorenzi M. Early complement activation and decreased levels of glycosylphosphatidylinositol-anchored complement inhibitors in human and experimental diabetic retinopathy. Diabetes. 2002;51:3499–504.
- 216. Zhang J-Z. Captopril inhibits capillary degeneration the early stages of diabetic retinopathy. Curr Eye Res. 2007;32:883–9.
- 217. Zhang SX, Wang JJ, Gao G, et al. Pigment epithelium-derived factor downregulates vascular endothelial growth factor (VEGF) expression and inhibits VEGF-VEGF receptor 2 binding in diabetic retinopathy. J Mol Endocrinol. 2006;37:1–12.
- 218. Zinn KM, Marmor MF, editors. The retinal pigment epithelium. Cambridge: Harvard University Press; 1979.

Chapter 3 Treatment of Diabetic Retinopathy: A Historical Perspective

3.1 Introduction

The lower prevalence of metabolic syndrome throughout the early part of the twentieth century meant that a greater proportion of patients suffered from type 1 diabetes and the total number of patients affected with type 2 diabetes was much lower than it is today. The introduction of insulin in 1922 extended survival for patients with type 1 diabetes mellitus (DM), but life expectancy remained poor because many died of disease-related complications. Since few patients lived long with DM, the incidence of diabetic retinopathy (DR) was low and early diagnosis and treatment were a low priority. In addition, only a small number of ophthalmologists served the population and their ability to accurately diagnose retinopathy was far more limited than it is today.

The development of DR was frequently accompanied by a steady decline in vision often with progression to legal blindness. Most physicians believed that tight glucose control decreased the incidence and progression of retinopathy but conclusive evidence did not emerge until completion of the Diabetes Complications and Control Trial and the United Kingdom Prospective Diabetes Study in the 1990s. Despite best attempts to control glucose, success was limited by inadequate dietary guidelines, lack of home glucose monitoring systems and hemoglobin A1c testing, and a limited choice of medications.

Several treatments for vision-threatening DR were tried, but efficacy was never demonstrated for some of them and significant side effects plagued others. This chapter begins by briefly discussing a couple of DR treatments that were tried and abandoned during the latter half of the twentieth century [46, 61, 131]. Most of the chapter focuses on the use of laser photocoagulation and vitrectomy for the treatment of diabetic macular edema (DME) and proliferative diabetic retinopathy (PDR). Laser and vitrectomy were introduced in the 1960s and 1970s, respectively, and still play an important role in the treatment of DR. Recent technological advances in the forms of micropulsed and navigated laser and small gauge vitrectomy have made these treatments easier to apply with potentially less damage to

surrounding tissues. Current research with these modalities attempts to determine how they fit into today's pharmacology-based treatment regimens.

3.2 Pituitary Ablation

Evidence supporting the notion that pituitary ablation alters the course of DM surfaced in the 1930s after pituitary glands had been removed from dogs that had undergone pancreatectomies [56] and cats that had undergone hypophysectomies and adrenalectomies [81]. Some observers believed that a "fundamental neural mechanism" was responsible for stabilization of retinopathy [156], while others suggested that pituitary ablation decreases or eliminates growth hormone, adrenocorticotrophin, and thyroid-stimulating hormone, which directly or indirectly through their downstream effects worsen DR [51]. Additionally, vision-threatening DR was rarely observed in growth hormone-deficient diabetic dwarfs. Support for pituitary ablation as a treatment for DR originally came from a case report (1953) of a woman whose DR regressed after she developed postpartum pituitary necrosis.

Few studies have compared the results of pituitary ablation for DR with control patients. In one report, 4 of 19 patients that underwent radioactive yttrium ablation of the pituitary experienced improved vision after 1 year, compared to 0 of 22 controls. Neovascularization of the retina decreased in 30% of ablated patients but in none of the controls [60]. In another report, 9 of 33 patients that underwent surgical resection of the pituitary stalk experienced decreased vision, compared to 19 of 33 controls. Retinopathy improved in 16 ablated patients but in only 9 controls [125]. In a review of 385 ablated patients, retinopathy was stabilized or improved in one-half [9]. Data from one series of patients suggested that improvement in retinopathy does not start until 12 months after ablation, so the authors recommended that ablation be performed early in the proliferative phase of retinopathy [5]. Pooled results from a large symposium reported that 75% of 708 patients had stabilized vision during the first 6 months after ablation [137]. Long-term visual results after ablation have not been well established, but up to one-third of patients with 5-year follow-ups experienced progression of retinopathy after an initial response to surgery [33].

Pituitary ablation was performed via surgical removal or resection, irradiation with yttrium or proton beam, or cryohypophysectomy (Fig. 3.1) [77, 122]. Surgeons believed that a more rapid and complete hypophysectomy had a greater effect on retinopathy [61]. Complications of pituitary ablation included cranial nerve palsies, cerebrospinal fluid rhinorrhea, meningitis, and herpes simplex encephalitis [133]. Following ablation, patients required lifelong hormonal replacement with corticosteroids and possibly thyroid and sex hormones.

Predicting which patients with DR would respond favorably to ablation was often difficult. Vascular lesions associated with both background and proliferative retinopathy were often observed to regress, but fluorescein angiography demonstrated that ischemia improved slowly [69]. Contraindications to pituitary ablation included retinal detachment, preretinal fibrosis, advanced age with cardiovascular Fig. 3.1 This intraoperative photograph shows exposure of the pituitary gland (*beneath the white star*) via a trans-nasal approach. Trans-nasal hypophysectomies are no longer performed for diabetic retinopathy but are frequently employed for large pituitary tumors



disease, and the inability to tolerate long-term, complex hormonal replacement. The ideal patient was below the age of 40 years with good renal function, but 75% of patients failed to meet the surgical criteria.

Pituitary ablation was performed on hundreds of patients, but firm evidence supporting its efficacy was never obtained. Due to its lack of established efficacy and unacceptable complication rates, pituitary ablation had been discontinued at some medical centers by the mid-1960s. After the introduction and widespread use of laser photocoagulation, nearly all centers had discontinued pituitary ablation by the mid-1970s.

3.3 Interferon

Interferon is a pluripotent endogenous protein with antiviral, antiproliferative, and immunogenic activity that has been shown to limit endothelial cell motility and proliferation in vitro [38, 147]. Recombinant α -interferon has been used to treat pulmonary hemangiomatosis and human immunodeficiency-related Kaposi sarcoma [134, 153]. Since interferon only mildly interferes with glucose tolerance, it would be unlikely to interfere with metabolic control in patients with DM [151].

In a 4-month prospective study, 3 patients with proliferative diabetic retinopathy were treated with systemic interferon [131]. Subjects experienced mild fatigue, nausea, myalgias, leukopenia, thrombocytopenia, and transient ataxia. Fasting hyperglycemia and postprandial hyperglycemia increased from 2 to 6% despite a 17 to 68% increase in insulin dosage. During treatment with interferon, DR was stable without new hemorrhages, but all patients had new retinal bleeding after interferon was discontinued.

Interferon has also been evaluated for the treatment of neovascular age-related macular degeneration [48], but since these studies failed to show a therapeutic effect, no further testing of interferon in patients with DR was performed.

3.4 Laser Photocoagulation

Damage to the retina by excessive light has been recognized for centuries. Socrates warned against directly viewing solar eclipses and advised his pupils to indirectly view the reflected image of the eclipse off of a pool of water. Solar maculopathy was first described in the seventeenth century [50], and it has been suggested that the artist Degas suffered solar damage to the macula when standing watch on a sunny day. Hundreds of solar burns occurred in Europe during a solar eclipse in 1912.

These events represent the results of a photochemical effect on the retina and not the thermal or photocoagulative effect that occurs with photocoagulation. These observations, however, inspired researchers to develop light-based devices to photocoagulate the retina. Experimental photocoagulation of rabbit eyes with a concave mirror and convex lens was first performed in 1867 [18] and duplicated in 1882 [21]. A carbon arc lamp produced the same effect in1893 [154].

The first experimental photocoagulation of the human retina (1927) was performed in an eye scheduled for enucleation because of a choroidal melanoma. Sunlight focused on the retina for 10 min caused hyperemia and retinal edema [88]. The first instrument specifically designed for therapeutic photocoagulation of the retina (1949) focused the sun's rays with a Galilean telescope and mirror [93]. This instrument was relatively unreliable because it depended on weather conditions. A next-generation device used a high-intensity carbon arc [94]. Though much more reliable, the carbon arc released soot and undesirable particles. The next light source was the xenon lamp, which produced a bright light with long-duration burns of moderate intensity over proliferating blood vessels [95, 96]. The xenon lamp was an effective photocoagulator for years, but it was eventually replaced by the ruby laser, the argon blue-green and green lasers, the krypton laser, the tunable dye laser, and most recently the infrared diode and frequency-doubled diode laser (Table 3.1). These lasers produce monochromatic light with small spot sizes and better control of power and pulse duration. They enable physicians to treat not only proliferative retinopathy with broad areas of photocoagulation but also macular edema with precise placement of small spots. The diode lasers are highly efficient as they derive power from standard electrical outlets without the need for specially installed, high voltage outlets.

Each of these instruments produces a collimated beam of coherent light that is transmitted through the ocular media to the retina where it is absorbed by retinal chromophores – xanthophyll, hemoglobin, and melanin. Light absorption produces a photothermal effect that heats the surrounding tissues and induces inflammation and edema that eventually causes localized tissue necrosis. High-intensity laser pulses produce burns that affect all levels of the retina, retinal pigment epithelium (RPE), and choroid, whereas low-energy pulses affect primarily the photoreceptors and RPE but spare the inner retina [17, 43]. Longer wavelengths penetrate deeper into the choroid because they are absorbed poorly by xanthophyll chromophores. Histopathologic studies confirm that the resultant scars vary according to wavelength, but the clinical importance of this has not been proven.

3.4 Laser Photocoagulation

0.1

~

Common types of laser used for retinal photocoagulation		
Long-duration laser systems		
Argon	Argon gas in replaceable tube Wavelengths in blue (488 nm) and blue-green (514 nm) part of spectrum Use has diminished because of high-power requirements and need to periodically replace tube	
Krypton	Krypton gas in replaceable tube Wavelengths in red (627 nm) part of spectrum Use has diminished because of high-power requirements and need to periodically replace tube	
Tunable dye	Dye allows modification of wavelength including 577 nm, highest uptake of hemoglobin	
Frequency-doubled YAG (solid state)	Solid-state laser uses 120 volt power output Spots at wavelength of 532 nm Device has long lifespan	
NAVILAS	Computer-guided system delivers spots directly to spots identified on fluorescein angiography Some surgeons believe that NAVILAS delivery may be better than surgeon directed photocoagulation	
Micropulse laser syste	ems	
Visible and infrared	Subthreshold laser delivery. Repeated delivery may be performed. Some surgeons suggest that subfoveal delivery is safe Has not been subjected to randomized, multicenter, masked trials	

Table 3.1 Commonly used laser systems are listed

Several advantages and disadvantages are discussed

While physicians were evaluating laser photocoagulation, retinal photography was also being developed. High-quality, stereoscopic color fundus photographs enabled physicians to identify microaneurysms and areas of diffuse macular thickening. Stereoscopic photos were critical for enrolling patients in prospective studies for the treatment of DME. Fluorescein angiography showed that DME develops from both leakage of microaneurysms and diffuse leakage from incompetent capillary beds.

3.4.1 Techniques for Performing Laser Photocoagulation

When choosing to perform laser photocoagulation of the retina, ophthalmologists can choose among different machines and techniques. Several manufacturers produce user-friendly, reliable lasers that produce excellent retinal burns. Most current laser systems have abandoned the use of Nobel gases (argon and krypton) and tunable dyes and have migrated to diode technology. Diodes are reliable, durable, and portable and require minimal servicing, and unlike the old gas-generated lasers, diodes work with 120 volt electrical outlets, thereby making installation easier and use in the clinic and operating rooms safer. They produce a 532 nm beam that is

moderately absorbed by oxyhemoglobin and minimally absorbed by retinal xanthophylls.

The most common laser delivery methods include the slit-lamp biomicroscope, indirect laser ophthalmoscope, and intraocular probe. In the outpatient or clinic setting, slit-lamp delivery systems are most commonly used, particularly for the treatment of DME (Fig. 3.2). The retina is visualized through a high-diopter condensing lens or a contact lens which may be flat (for the macula), may invert images (such as the panfundus or Mainster lenses that are generally used for panretinal or scatter photocoagulation), or may be mirrored (for treatment of the peripheral retina). Panfundus and Mainster lenses magnify the spot size by factors of 1.75 and 2.2, respectively. Topical anesthesia is usually sufficient for most treatments though retrobulbar anesthesia improves pain control and limits ocular motility.

Macular edema is usually treated with the slit-lamp delivery system through a flat lens. Microaneurysms are focally treated with $50-100 \mu m$ spots and sufficient power to blanch them. A complete or modified grid pattern of spots is used to treat broader areas of capillary leakage (Fig. 3.3). A typical laser treatment includes $50-100 \mu m$ spot sizes, power to produce light burns with only mild blanching of the

Fig. 3.2 Laser treatment of diabetic macular edema and proliferative diabetic retinopathy is most commonly performed with the slit-lamp delivery system. The surgeon directly visualizes the retina through a contact lens and focuses the laser beam on microaneurysms, areas of capillary leakage, and areas of peripheral retina





Fig. 3.3 This composite photograph shows scarring from a previous modified grid laser pattern in the macula. Note that the spots are not confluent and they do not completely encircle the macula

retina and pigment epithelium, and separation of burns by 2 spot diameters. Spots are not placed within 500 μ m of the fovea during the initial treatment, but if edema persists and visual acuity remains decreased, subsequent spots can be placed to within 300 μ m of the fovea but outside of the foveal avascular zone. If the foveal avascular zone is larger than normal because of capillary non-perfusion, spots should be kept further from the fovea to avoid areas of perifoveal non-perfusion.

Treatment recommendations have changed since the older ETDRS guidelines when spots were larger, closer together, and more powerful (Fig. 3.4). The new guidelines produce a subtler treatment pattern that is believed to cause fewer permanent scotomas. Since the spacing between spots is increased, there remains sufficient space for repeat laser treatments, if needed.

Treatment of non-center-involving edema, even in the anti-VEGF drug era, usually includes focal treatment of microaneurysms and a modified grid pattern to areas of edema. Some studies suggest that focal treatment of microaneurysms may not be required as treatment of the RPE may produce the desired effect [115]. Extrafoveal edema often resorbs after laser and the effect may be durable, thereby preventing the subsequent need for a series of anti-VEGF injections. If necessary, laser treatments can be repeated at 3–4-month interval. When a retreatment is planned, fluorescein angiography may help identify areas of persistent leakage.

Complications of focal/grid laser are unusual but can be serious. Laser photocoagulation permanently damages photoreceptors and retinal pigment epithelium, and large, heavily pigmented laser scars may cause paracentral scotomas. Densely applied treatment patterns decrease color vision and contrast sensitivity. Laser spots can enlarge and encroach on the fovea – referred to as "laser creep." In one series of patients with DME, enlarging laser scars reached the fovea in 11 of 203



Fig. 3.4 The figure on the left shows confluent laser spots resulting from large, intense laser spots that may have been placed too close together. These spots had been placed according to the original ETDRS guidelines. The figure on the right shows a treatment pattern that was done in accordance with the new modified treatment guidelines. The spots are smaller, less intense, and spaced farther apart. This decreases the likelihood of spot enlargement or "laser creep"



Fig. 3.5 This composite fundus photograph shows mature laser scars following a completed panretinal photocoagulation treatment. The treatment has been extended posteriorly to the vascular arcades to achieve maximum effect

eyes [127]. Laser scars in the posterior pole expand more than those in the periphery possibly because the greater density of photoreceptors facilitates heat transfer [87]. Surgeons must be careful not to use too much energy when using small spot sizes because of the risk of an RPE rupture. Rupture can be followed by choroidal neovascularization with accompanying subretinal hemorrhage, exudation, and disciform scarring [80, 82].

Several pathophysiologic mechanisms have been proposed to explain why macular laser improves DME. Laser is believed to increase local oxygen tension [135], decrease autoregulatory vasoconstriction [45], decrease the entire area of abnormal leakage [155], restore the outer blood-retinal barrier [42, 89], decrease the production of vasoactive cytokines (primarily VEGF), and increase phagocytosis by the RPE and retinal glial cells. RPE cells at the margins of burns may modulate the production of cytokines through their interactions with photoreceptors [28, 100].

Proliferative diabetic retinopathy has been treated with scatter or panretinal photocoagulation (PRP) for the past 40 years. Moderate-intensity spots of 350–500 µm in diameter and 0.075–0.2 s. in duration are delivered through an inverting, panfundus contact lens or 3-mirror contact lens to the mid-peripheral retina. Spots are spaced 1 diameter apart and are applied from just outside the major vascular arcades to the far peripheral retina (Fig. 3.5). Complete treatment usually consists of 1200– 3000 spots, but more may be delivered either primarily or for resistant cases. In the Diabetic Retinopathy Study (DRS), photocoagulation was applied directly to patches of new blood vessels, but since this does not adequately address the underlying cause of neovascularization and fails to promote regression, this strategy has been abandoned. Panretinal photocoagulation can also be performed with an indirect ophthalmoscopic delivery system, which enables physicians to more completely treat the far peripheral retina (Fig. 3.6). Treatments are often divided into 2–3 sessions to improve patient comfort. Many physicians choose to photocoagulate the inferior retina first so that if the post-laser course is complicated by a Fig. 3.6 Panretinal photocoagulation can be delivered through the indirect ophthalmoscope with the patient in the supine position. Since ultra-widefield angiography has shown that broad areas of peripheral retinal non-perfusion may contribute to the development of neovascularization, many surgeons advocate the routine use of indirect ophthalmoscopic laser delivery in all cases of proliferative diabetic retinopathy



vitreous hemorrhage, the view of the superior retina is less likely to be obstructed and the course of treatment can be completed. Smaller spot sizes and longer wavelengths (such as 810 nm) penetrate vitreous hemorrhages better.

Preretinal neovascularization usually regresses within 3 weeks of adequate PRP (Fig. 3.7) [24], and patients who have a favorable early response usually have a better visual outcome than those who do not [149]. Persistent or progressive neovascularization despite the performance of a standard course of PRP may respond to additional laser. Indirect laser-assisted treatment with up to 8000 spots [150] has been described. Ultra-widefield fluorescein angiography can visualize persistent neovascularization and identify untreated areas of peripheral retinal ischemia. These images can be used to guide the placement of supplemental photocoagulation.

Panretinal photocoagulation that spares the posterior retina and preserves the paracentral visual field is sufficient to cause neovascular regression in many cases [7]. Ultra-widefield fluorescein angiography has subsequently shown that most retinal non-perfusion is found in the mid- and far periphery, thereby validating the logic of this approach.

Fig. 3.7 This photograph of the posterior pole shows panretinal photocoagulation scars that have been extended to the vascular arcades. A near-complete ring of preretinal fibrosis extends around the macula, and an epiretinal membrane is causing mild macular wrinkling. This represents successful post-laser regression of what were previously active neovascular fronds



Exudative choroidal detachments may complicate intensive, widespread PRP, but this rarely occurs after split-session treatments [57]. High-intensity burns can rupture Bruch's membrane and lead to retinal and vitreous hemorrhages and the ingrowth of choroidal neovascular membranes [11]. Peripheral visual field loss and night blindness can follow high-intensity PRP. Worsening of preexisting vitreoretinal traction can occur after intense photocoagulation [92], but treatment with divided sessions usually avoids this complication.

Broad areas of mid-peripheral and peripheral retina must usually be ablated to successfully involute neovascularization. Physicians have long appreciated that the severity of neovascularization correlates with the amount of retinal ischemia [98], but the biochemical processes responsible for angiogenesis were not understood. The recent discovery of vascular endothelial growth factor (VEGF) [16, 32], the understanding of its pivotal role in angiogenesis [10, 31, 47, 138], and its detection in the vitreous in eyes with PDR suggest that it plays an important role in eyes with PDR.

Several lines of evidence associate VEGF upregulation with retinal neovascularization. Tissue hypoxia is the most important upregulator of VEGF synthesis [26], and advanced DR is characterized by broad areas of capillary closure with retinal ischemia. The cell's oxygen sensor, hypoxia-inducible factor (HIF)-1 α , is stabilized when oxygen tension is low, enabling it to upregulate the synthesis of VEGF (Fig. 3.8) [44, 148]. Administration of exogenous VEGF into monkey eyes induces retinal neovascularization and other findings typical of DR [144]. Panretinal photocoagulation is thought to improve overall retinal oxygen tension by either destroying large areas of ischemic retina or increasing oxygen diffusion from the choroid into the remaining retina. Improved retinal oxygenation downregulates VEGF and involutes immature neovascular vessels.



3.5 Treatment of Background Diabetic Retinopathy

The Early Treatment of Diabetic Retinopathy Study (ETDRS) produced level I evidence supporting the use of laser photocoagulation for patients with clinically significant macular edema (CSME) due to diabetic retinopathy (Table 3.2). The ETDRS was designed to test the theory that laser photocoagulation is superior to observation for patients with CSME [29]. Diabetic patients were eligible for enrollment if they had biomicroscopic evidence of CSME, which was defined as follows:

- 1. Retinal thickening within 500 µm of the fovea
- 2. Hard exudates within 500 μ m of the fovea that were accompanied by adjacent thickening
- 3. Retinal thickening of 1 disc diameter in area part of which was within 1 disc diameter of the fovea

Table 3.2 The definitions of clinically significant macular edema (CSME) and diabetic macular edema (DME) are listed

Clinically significant macular edema is diagnosed if ANY of the following is met:

1. Thickening at or within 500 µm of the fovea

2. Hard exudates within 500 µm of the fovea with adjacent retinal thickening

3. An area of macular thickening at least 1 disc area in size, any part of which is within 1 disc diameter of the fovea

**To make the diagnosis of CSME, macular thickening is diagnosed by slit-lamp biomicroscopy *Diabetic macular edema* is diagnosed if the following is met:

1. Thickening within 2 disc diameters of the fovea

For clinical purposes, DME is classified as:

1. Center-involving or non-center-involving

2. Diffuse or focal

The presence of clinically significant macular edema was used to guide laser photocoagulation, but this definition has become less important since the introduction of retinal pharmacotherapy

Fig. 3.9 This venous phase frame from a fluorescein angiogram shows areas of dye leakage in the superior macula from microaneurysms and capillary beds. Macular edema is usually found in areas that display these findings



Eligible eyes had baseline visual acuities that ranged from 20/20 to 20/200. Fluorescein angiography (FA) was performed at baseline, but angiographic findings were not used to define CSME (Fig. 3.9). Angiography served two main purposes in patients with diabetic macular edema. First, it identified microaneurysms and areas of diffuse capillary leakage, information that helped physicians accurately place laser spots. Second, the FA identified areas of capillary non-perfusion, information that helped physicians set appropriate visual expectations with the patients and helped them avoid treating areas of non-perfusion.

Enrolled eyes were randomized to observation or fluorescein angiography-guided laser photocoagulation. Treatment was applied focally to microaneurysms and in a grid pattern to areas of diffuse capillary leakage. Microaneurysms were treated directly with spot sizes of $50-100 \,\mu\text{m}$ and durations of $0.05-0.1 \,\text{s}$ until either darkening

or whitening of the microaneurysms was observed; areas of diffuse capillary leakage received a grid pattern of laser with moderate-intensity burns, spot sizes of 100 μ m separated by 100 μ m, and duration of 0.05–0.5 s. Spots were not applied within 500 μ m of the foveal center, and treatment could be extended out past 3000 μ m from the fovea. Eyes were eligible for retreatment on a quarterly basis to areas of persistent or new leakage that were untreated or judged to be incompletely treated. Argon laser was used during the ETDRS, but other wavelengths are likely to be as effective [116]

Overall in the ETDRS, 30% of laser-treated patients experienced some improvement in BCVA. Patients receiving laser had a 50% decrease in their risk of moderate vision loss (15 letters) over the course of 3 years. Unfortunately, only 3% of laser-treated patients improved by +15 letters (three lines) at 3 years and 40% of laser-treated eyes had persistent CSME at 12 months [30]. Nonetheless, the trial results suggested that all eyes with CSME should be treated with laser, including those with 20/20 visual acuity.

Persistent DME following laser photocoagulation is not uncommon as the closure rate of microaneurysms with focal laser is only 70% [78], and the need for repeat treatment to areas of diffuse capillary leakage is common. Unless the underlying disease process is halted, new areas of macular thickening frequently develop.

3.6 Treatment of Proliferative Diabetic Retinopathy (PDR)

Randomized clinical trials have validated the efficacy of panretinal photocoagulation for proliferative diabetic retinopathy, and many of these were summarized in a 2007 systematic review [102]. An early British trial used the xenon arc photocoagulator [12] to treat PDR, and this was soon followed by the National Eye Institutesponsored Diabetic Retinopathy Study (DRS; ClinicalTrials.gov number, NCT00000160). The DRS was one of the first multicenter, randomized studies in medicine and served as a model for subsequent collaborative trials [139]. The DRS enrolled 1758 patients with visual acuities of 20/100 or better in each eye and with either proliferative diabetic retinopathy in at least one eye or severe nonproliferative diabetic retinopathy in both eyes. Each patient was randomly assigned to undergo panretinal photocoagulation with either the xenon arc photocoagulator or argon laser in one eye, with the other eye serving as the untreated control. Treatment consisted of 800-1600 burns of 0.1 s duration and new blood vessels were treated directly. Panretinal photocoagulation reduced the risk of severe vision loss (defined as visual acuity of 20/800 or worse at two consecutive 4-month visits) from 14.0 to 6.2% over a 2-year period [139] and from 33.0 to 13.9% over a 5-year period [142].

There were no significant differences between the xenon arc and argon groups regarding the primary outcome, but eyes treated with argon maintained better visual acuities and fewer visual field and night vision defects. Panretinal photocoagulation reduced the overall incidence of severe vision loss by at least 50% [140]. Eyes at high risk of vision loss included those with moderate neovascularization of the disc (covering at least ¼ of the disc area) and those with any neovascularization of the disc (NVD) or elsewhere (NVE) that was accompanied by preretinal or vitreous hemorrhage [141].

The ETDRS (NCT00000151) enrolled 3711 patients with mild-to-severe nonproliferative or early proliferative diabetic retinopathy in both eyes. Each patient was assigned to undergo panretinal photocoagulation in one eye, while treatment in the fellow eye was deferred until the development of high-risk PDR. Rates of severe vision loss at 5 years were 2.6% following early treatment and 3.7% following deferred treatment [30]. The ETDRS showed that performing PRP in eyes with advanced pre-proliferative diabetic retinopathy generally does not change the visual outcome. The Diabetic Retinopathy Clinical Research Network published guidelines for the treatment of proliferative diabetic retinopathy [23]. Treatment should be initiated promptly when an eye develops high-risk PDR as delaying treatment increases the short-term risk of vision loss because vitreous hemorrhage may develop.

Treatment of coexistent diabetic macular edema and high-risk PDR is particularly challenging. Following PRP, 25–43% of eyes may develop DME or experience an increase in macular leakage with resulting loss of visual acuity [54, 90, 91]. The mechanism responsible for the development of DME after PRP is not well understood, but the accumulation of leukocytes in the posterior pole and the increased release of permeability promoting chemokines and cytokines (including VEGF) due to laser damage may play a role [40, 113, 146]. The ETDRS previously recommended macular laser photocoagulation for DME either before or at the time of PRP [1, 30]. Macular laser photocoagulation performed at these times decreased the risk of moderate visual acuity loss by 50%, but patients experienced minimal visual improvement and only 3% experienced three-line improvements in VA at 3 years. Unfortunately, 12% of eyes lost three lines of VA at 3 years, and 40% of eyes with baseline macular edema had persistent edema at 12 months.

These treatment guidelines were formulated before the introduction of retinal pharmacotherapy, and recent studies have demonstrated that the intravitreal injection of triamcinolone acetonide or anti-VEGF agents prior to PRP minimizes the exacerbation of DME and in many cases with pre-existing edema actually improves VA. Furthermore, pharmacotherapy is synergistic with panretinal photocoagulation in promoting the regression of retinal neovascularization [4, 14, 145, 163]. In a randomized, comparative study, intravitreal triamcinolone was superior to bevacizumab at resolving macular edema and preventing PRP-induced VA loss in eyes with preexisting macular edema, while both drugs were effective at preventing the development of edema in eyes with dry maculas [15]. Recent findings suggest that ranibizumab monotherapy effectively treats PDR through 2 years [157], so anti-VEGF injections may be increasingly used to treat both DME and PDR over extended periods of time, thereby delaying or even avoiding the need for macular laser or PRP.

3.7 New Laser Technologies

The ETDRS provided physicians with a proven strategy to treat CSME, but concerns regarding treatment intensity and subsequent complications lingered. Large spot size, insufficient burn separation, and excessive treatment intensity may all limit visual acuity gains and induce post-treatment scotomas. Clinicians have modified treatment strategies, and laser manufacturers have developed new technologies in attempts to improve efficacy and safety.

3.7.1 Micropulse Laser

A new laser delivery method featuring short bursts or "micropulses" rather than a continuous wave was introduced in 1990 [118]. Compared to standard pulse thermal laser, micropulse spots heat the target tissues less, spare the overlying photoreceptors, and theoretically reduce iatrogenic side effects such as scarring and fibrosis [19]. Micropulse does not adversely affect the neurosensory retina and leaves it undamaged. The mechanism by which micropulse may exert a treatment effect is unclear though "normalizing RPE function" and decreasing cytokine production have been suggested [20, 41].

The primary determinant of tissue heating is the duty cycle or frequency of micropulses. Longer periods between pulses – lower duty cycle – heat the tissue less. A lower duty cycle limits heat diffusion, produces smaller spot sizes, and creates less thermal tissue damage. Since longer wavelengths (647 and 810 nm) are absorbed less by the retinal chromophores, more energy reaches the RPE without affecting the overlying retina. By micropulsing an infrared laser, energy is delivered directly to the RPE [27, 118].

Early micropulse treatment of DME produced faint scars because physicians sought to mimic the more familiar results that they achieved with thermal laser [39, 103, 109]. They subsequently learned to decrease the duty cycle to achieve a clinical effect without the formation of permanently visible RPE scarring. By not creating a tissue scar, micropulse preserves the outer retina and avoids the adverse effects that characterize thermal laser [99]. Over the years, there has been a slow increase in the popularity of micropulse laser for DME as new delivery devices have become more refined.

Treatment of eyes with macular edema is said to follow a "low-intensity/highdensity" pattern. Near confluent spots are delivered throughout the macula with some authors treating to within 200 μ m of the fovea. Typical treatment parameters include the following: spot size of 75–125 μ m, pulse duration of 300 ms with a 15% duty cycle, and power of 1000 mW. If a tissue reaction is seen, the power is reduced by 200 mW increments until no further reaction is noted [117].

A 2005 pilot study with the subthreshold diode micropulsar (SDM) laser, which uses an 810 nm diode laser at suboptimum intensity, suggested that SDM laser photocoagulation produces results similar to those achieved with thermal laser photocoagulation without causing iatrogenic retinal damage [84]. Small, retrospective studies have shown improvements in macular edema and visual acuity [73, 112].

In a large prospective study with 220 patients, mean CMT improved from 353 to 215 μ m at 1 year and mean visual acuity was stable (0.21 LogMAR to 0.18 LogMAR) [117]. A 2009 prospective, randomized controlled trial compared the

efficacy of SDM laser photocoagulation with conventional argon green laser delivered according to the modified ETDRS protocol. The BCVA remained stable in both arms at 12 months but laser scars were identified in only 13.9% of eyes treated with SDM compared to 59% of eyes treated with argon laser [34]. In a study of 24 eyes that had undergone SDM laser, fundus autofluorescence at 1 year failed to detect damage to the RPE [152].

In a long-term (median follow-up of 47 months) study of 252 eyes treated with micropulse, none of the eyes treated with low irradiance (<350 W/cm²) had detectable damage on fundus photography, fluorescein angiography, or SD-OCT [85]. Unfortunately, visual acuity results from this study were not reported.

The absence of observable damage to the photoreceptors and RPE paved the way for high-density SDM treatment of the macula. By delivering approximately 900 spots to the macular in a confluent pattern that spares the fovea, lateral spread of thermally stimulated cells is promoted in an effort to improve visual outcomes. A 2011 study suggested that high-density SDM produces superior improvements in macular edema and BCVA compared to both low-density SDM and conventional laser therapy [76]. The 577 nm micropulse may also be superior to the 810 nm laser at closing microaneurysms [58].

Micropulse laser treatment of DME appears promising, but current data does not suggest that its results are superior to those of standard thermal photocoagulation. Large, randomized, controlled trials against standard laser photocoagulation and in combination with anti-VEGF therapy are needed to better understand the role of micropulse in the management of DME.

3.7.2 Navigated Laser

Another newly developed method of delivering subthermal laser to the macula involves placing spots in a horseshoe-shaped grid pattern with a 577 nm semiautomated patterned scanning laser (PASCAL, Topcon, Capelle aan den IJessel, the Netherlands). This system allows surgeons to deliver multiple spots simultaneously to the macula or peripheral retina in a preprogrammed pattern. The PASCAL uses burns of 10 or 20 ms that localize to the outer retina, thereby avoiding damage to the RPE and inner retina [105, 106]. Since the PASCAL uses visible wavelength, burns can be seen with SD-OCT 1 h after treatment. Treatment effects are localized to the junction of the photoreceptor inner and outer segments and RPE apices. At 12 months, SD-OCT shows treatment effects at the RPE apices but without overlying changes in the photoreceptors [68].

Panretinal photocoagulation treatment technique has not changed since the DRS, but the PASCAL laser shortens treatment duration and decreases pain while producing the same outcomes [2, 8]. The PETER PAN study showed that the PASCAL laser produced results similar to standard PRP but with lower power settings and less vision loss [107]. Single-session, complete PRP with the PASCAL appears as effective as multi-session single-spot PRP treatments with no difference in the incidence

of macular edema [108]. This technology has the potential to improve patient compliance and decrease the overall cost of care by requiring fewer treatment visits.

The navigated laser photocoagulator (NAVILAS®, NAVILAS Laser System, Irvine, CA, USA) tracks the eye and simultaneously integrates digital fluorescein angiographic information. This stabilizes the real-time retinal image and overlays the acquired fundus photograph to improve the accuracy of laser delivery [70]. In a small retrospective study of seven patients with DME treated with the NAVILAS system, median central foveal thickness decreased (248–220 μ m) and LogMAR VA improved (0.695–0.477) [62]. In the 12-month CAVNAV study for treatment of DME, patients receiving anti-VEGF therapy combined with navigated laser required fewer anti-VEGF injections than did patients treated with anti-VEGF injections and standard thermal laser photocoagulation [6].

3.8 Pars Plana Vitrectomy

Approximately 60% of patients with PDR achieve satisfactory regression of neovascularization within 3 months of PRP [149], but many patients require additional laser and 4.5% ultimately undergo pars plana vitrectomy because of vitreous hemorrhage and/or traction retinal detachments [37]. Vitreous hemorrhage is a cause of significant vision loss in patients with PDR, and over the course of 3–10 years, vitreous hemorrhage persists or worsens in two-thirds of patients [164]. For recurrent or non-clearing vitreous hemorrhage, vitrectomy is indicated to clear the visual axis, allow adequate laser photocoagulation to be applied to prevent further bleeding, and address fibrovascular traction [160].

Pars plana vitrectomy has been used to treat DME for more than two decades. In the first vitrectomy for DME series, nine of ten patients with biomicroscopically evident vitreomacular traction experienced improved vision [79]. Encouraged by these results, surgeons began operating on eyes with DME, without concern for the condition of the posterior hyaloid. (Fig. 3.10). Many early studies reported improved postoperative VA, but these were often retrospective, with poor VA measurement protocols and without OCT images. These served as encouraging pilot studies but provided insufficient data for surgeons to make firm treatment recommendations.

Subsequent studies enrolled patients prospectively and measured macular thickness with time-domain optical coherence tomography (OCT), and some included control arms. Dozens of studies have been published with visual acuity results that vary considerably, but most show objective evidence of postoperative macular thinning by OCT. Few of the studies included spectral domain OCT analysis of outer retinal integrity, and none has produced level I evidence supporting the use of vitrectomy for the treatment of DME.

Landers et al. performed a literature review and composite analysis of vitrectomy for DME [74, 75] that included 37 manuscripts published between 2002 and 2012 [3, 13, 25, 35, 36, 49, 52, 53, 55, 63–67, 71, 72, 83, 86, 101, 110, 111, 119–121, 123, 124, 126, 128–130, 132, 136, 143, 158, 159, 161, 162]. These studies reported



Fig. 3.10 This figure shows an eye with diabetic macular edema that underwent pars plana vitrectomy. Prior to vitrectomy (*top row*), the eye had a prominent epiretinal membrane and thickened macula. At 1 month (*second row*) and 3 months (*third row*) postoperatively, the macular edema had improved but had not completely resolved. The visual acuity at 5 months had dropped to 20/150 because of a nuclear sclerotic cataract. Following cataract removal, the visual acuity improved to 20/40. By 36 months postoperatively the macular edema had resolved. This eye did not require pharmacotherapy at any time. *CE/IOL* cataract extraction with posterior chamber lens insertion

excellent mean thinning of the macula $(-187 \ \mu m)$, which compares favorably with anti-VEGF therapy [22, 97, 99, 114]. Improvements in VA were generally favorable but highly variable.

The DRCR.net performed a prospective, single-arm vitrectomy study that enrolled 241 patients who, in the opinion of the primary investigator, would not benefit from additional standard therapy [36]. The mean subfield thickness decreased from 412 to 278 µm at 6 months with more thinning in eyes that had greater preoperative thickness, the presence of an epiretinal membrane, and the presence of vitreomacular traction. Although the average VA (20/80) of the entire cohort did not change, significant visual acuity improvements were seen in eyes with worse baseline vision (P < 0.001) and in those requiring the removal of epiretinal membranes (P = 0.006). The authors concluded that vitrectomy may have a role in eyes with vitreomacular interface abnormalities but that vitrectomy for eyes without traction could not be recommended. Most eyes that were enrolled in the DRCR.net study had advanced DME that had been previously treated with laser and/or corticosteroids. Many of these eyes may have already suffered irreversible macular damage that would not have improved with vitrectomy or any other treatment. Chhablani et al. correlated visual acuity improvement after vitrectomy with the preoperative integrity of the external limiting membrane (ELM) and inner segment/outer segment (IS/OS) [13]. They found that ELM integrity on SD-OCT correlated with better final visual acuity and IS/OS integrity correlated with the greatest improvement in vision. Spectral domain OCT was not available to the DRCR.net study, so detailed evaluation of outer segment health was not possible.

In one novel study of 20 patients with chronic DME and subfoveal fluid, saline was injected beneath the fovea, after which an air-fluid exchange was performed followed by facedown positioning [104]. Eighty-five percent of patients had complete resolution of edema, and the VA improved by three lines in 65% of patients.

3.9 Conclusion

Laser photocoagulation still plays an important role in the management of PDR though anti-VEGF therapy has been recently shown to be equally effective. Laser photocoagulation remains the preferred treatment for most eyes with macula-sparing DME, but its role in the management of center-involving DME is unclear. Vitrectomy for DME is usually reserved for eyes that fail other therapies, but its role as an early treatment for DME continues to be evaluated [59].

References

- Aiello LM, Cavallerano J, Aiello LP. Diagnosis, management and treatment of nonproliferative diabetic retinopathy and macular edema. In: Albert DM, Jakobiec FA, editors. Principles and Practice of Ophthalmology. 2nd edn. Philadelphia: WB Saunders; 2000. p. 1900–14.
- Al-Hussainy S, Dodson PM, Gibson JM. Pain response and follow-up of patients undergoing panretinal laser photocoagulation with reduced exposure times. Eye. 2008;22:96–9.
- Bahadir M, Ertan A, Mertoglu O. Visual acuity comparison of vitrectomy with and without internal limiting membrane removal in the treatment of diabetic macular edema. Int Ophthalmol. 2005;26(1–2):3–8.
- Bandello F, Polito A, Pognuz DR, et al. Triamcinolone as adjunctive treatment to laser panretinal photocoagulation for proliferative diabetic retinopathy. Arch Ophthalmol. 2006;124: 643–50.
- 5. Behrendt T, Field RA, Duane TD. Pituitary ablation: results in diabetic retinopathy. Trans Am Ophthalmol Soc. 1968;66:62–73.
- Barteselli G, Kozak I, El-Emam S, Chhablani J, Cortes MA, Freeman WR. 12-month results of the standardised combination therapy for diabetic macular oedema: intravitreal bevacizumab and navigated retinal photocoagulation. Br J Ophthalmol. 2014;98:1036–41.
- Blankenship GA. A clinical comparison of central and peripheral argon laser panretinal photocoagulation for proliferative diabetic retinopathy. Ophthalmology. 1988;95:170.

- Blumenkranz MS, Yellachich D, Andersen DE, et al. Semiautomated patterned scanning laser for retinal photocoagulation. Retina. 2006;26:370–6.
- 9. Bradley RF, Rees SB, Fager CA. Pituitary ablation in the treatment of diabetic retinopathy. Med Clin N Amer. 1965;49:1105.
- Byrne AM, Bouchier-Hayes DJ, Harmey JH. Angiogenic and cell survival functions of vascular endothelial growth factor (VEGF). J Cell Mol Med. 2005;9:777–94.
- Chandra SR, Bresnick GH, Davis MD, et al. Choroidovitreal neovascular ingrowth after photocoagulation for proliferative diabetic retinopathy. Arch Ophthalmol. 1980;98:1593.
- Cheng H. Multicentre trial of xenon-arc photocoagulation in the treatment of diabetic retinopathy. A Randomized controlled study. Interim report. Trans Ophthalmol Soc U K. 1975;95:351–7.
- Chhablani JK, Kim JS, Cheng L, et al. External limiting membrane as a predictor of visual improvement in diabetic macular edema after pars plana vitrectomy. Graefes Arch Clin Exp Ophthalmol. 2012;250(10):1415–20.
- 14. Cho WB, Moon JW, Kim HC, et al. Panretinal photocoagulation combined with intravitreal bevacizumab in high-risk proliferative diabetic retinopathy. Retina. 2009;29:516–22.
- Cho WB, Moon JW, Kim HC. Intravitreal triamcinolone and bevacizumab as adjunctive treatments to panretinal photocoagulation in diabetic retinopathy. Br J Ophthalmol. 2010;94:858–63.
- Connolly DT, Heuvelman DM, Nelson R, et al. Tumor vascular permeability factor stimulates endothelial cell growth and angiogenesis. J Clin Invest. 1989;84:1470–8.
- Curtin VT, Norton EWD. Early pathological changes of photocoagulation in the human retina. Arch Ophthalmol. 1963;69:744.
- 18. Czerny: Ber Wien Acad Wiss 1912;11:56.
- Das A, Stroud S, Mehta A, Rangasamy S. New treatments for diabetic retinopathy. Diabetes Obesity Metabolism. 2015;17:219–30.
- Del Maschio A, Zanetti A, Corada M, et al. Polymorphonuclear leukocyte adhesion triggers the disorganization of endothelial cell-to-cell adherens junctions. J Cell Biol. 1996;135:497–510.
- Deutschmann R. Ueber die blending der netzhaut durch directes sonnenlicht. Von Graefe's Arch Ophthalmol. 1882;28:241.
- 22. Diabetes Retinopathy Clinical Research Network. Expanded 2-year follow-up of ranibizumab plus prompt or deferred laser or triamcinolone plus prompt laser for diabetic macular edema. Ophthalmology. 2011;118(4):609–14.
- 23. Diabetic Retinopathy Clinical Research Network. An observational study of the development of diabetic macular edema following scatter laser photocoagulation. http://drcrnet.jaeb.org/ Studies.aspx?RecID=142.
- 24. Doft BH, Blankenship G. Retinopathy risk factor regression after laser panretinal photocoagulation for proliferative diabetic retinopathy. Ophthalmology. 1984;91:1453.
- 25. Doi N, Sakamoto T, Sonoda Y, et al. Comparative study of vitrectomy versus intravitreous triamcinolone for diabetic macular edema on randomized paired-eyes. Graefes Arch Clin Exp Ophthalmol. 2012;250(1):71–8.
- Dor Y, Porat R, Keshet E. Vascular endothelial growth factor and vascular adjustments to perturbations in oxygen homeostasis. Am J Physiol. 2001;280:C1367–74.
- 27. Dorin G. Subthreshold and micropulse photocoagulation. Sem Opthalmol. 2003;18:147–53.
- Duh EJ, Yang HS, Suzuma I, Miyagi M, et al. Pigment epithelium-derived factor suppresses ischemia-induced retinal neovascularization and VEGF-induced migration and growth. Invest Ophthalmol Vis Sci. 2002;43:821–9.
- Early Treatment Diabetic Retinopathy Study Research Group. Photocoagulation for diabetic macular edema. Early Treatment Diabetic Retinopathy Study Research Group report number 1. Arch Ophthalmol. 1985;103(12):1796–806.
- Early Treatment Diabetic Retinopathy Study Research Group. Early photocoagulation for diabetic retinopathy: ETDRS report number 9. Ophthalmology. 1991;98:Suppl:766–85.
- Elicieri BP, Paul R, Schwartzberg PL, Hood JD, Leng J, Cheresh DA. Selective requirement for Src kinases during VEGF-induced angiogenesis and vascular permeability. Mol Cell. 1999;4:915–24.

- Ferrara N, Henzel WJ. Pituitary follicular cells secrete a novel heparin-binding growth factor specific for vascular endothelial cells. Biochem Biophys Res Commun. 1989;161:851–8.
- Field RA, McMeel JW, Sweet WH, Schepens CL. Symposium on the Treatment of Diabetic Retinopathy, Goldberg MF, Fine SL. Washington, Government Printing Office; 1969. p. 213.
- 34. Figueira J, Khan J, Nunes S, et al. Prospective randomised controlled trial comparing subthreshold micropulse diode laser photocoagulation and conventional green laser for clinically significant diabetic macular oedema. Br J Ophthalmol. 2009;93:1341–4.
- Figueroa MS, Contreras I, Noval S. Surgical and anatomical outcomes of pars plana vitrectomy for diffuse nontractional diabetic macular edema. Retina. 2008;28(3):420–6.
- 36. Flaxel CJ, Edwards AR, Aiello LP, et al. Factors associated with visual acuity outcomes after vitrectomy for diabetic macular edema: diabetic retinopathy clinical research network. Retina. 2010;30(9):1488–95.
- Flynn Jr HW, Chew EY, Simons BD, et al. Pars plana vitrectomy in the early treatment diabetic retinopathy study. ETDRS report number 17. Ophthalmology. 1992;99:1351–7.
- 38. Folkman J. Successful treatment of an angiogenic disease. [Editorial]. N Engl J Med. 1989;320:1211–2.
- Friberg TR, Karatza EC. The treatment of macular disease using a micropulsed and continuous wave 810-nm diode laser. Ophthalmology. 1997;104:2030–8.
- 40. Funatsu H, Yamashita H, Ikeda T, et al. Relation of diabetic macular edema to cytokines and posterior vitreous detachment. Am J Ophthalmol. 2003;135:321–7.
- Gao X, Xing D. Molecular mechanisms of cell proliferation induced by low power laser irradiation. J Biomed Sci. 2009;16:4.
- 42. Glaser BM, Campochiaro PA, Davis JLJ, Jerdan JA. Retinal pigment epithelial cells release inhibitors of neovascularization. Ophthalmology. 1987;94:780–4.
- Geeraets WJ, Williams RC, Chan G, et al. The relative absorption of thermal energy in retina and choroid. Invest Ophthalmol. 1962;1:340.
- 44. Gerber HP, McMurtrey A. Kowalski, et al. VEGF regulates endothelial cell survival by the PI3-kinase/Akt signal transduction pathway. Requirement for Flk-1/KDR activation. J Biol Chem. 1998;273:30343–66.
- 45. Gottfredsdottir MS, Stefansson E, Jonasson F, Gislason I. Retinal vasoconstriction after laser treatment for diabetic macular edema. Am J Ophthalmol. 1993;115:64–7.
- 46. Grant MB, Mames RN, Fitzgerald C, et al. The efficacy of octreotide in the therapy of severe nonproliferative and early proliferative diabetic retinopathy: a randomized controlled study. Diabetes Care. 2000;23(4):504–9.
- 47. Guo D, Jia Q, Song HY, Warren RS, Donner DB. Vascular endothelial cell growth factor promotes tyrosine phosphorylation of mediators of signal transduction that contain SH2 domains. Association with endothelial cell proliferation. J Biol Chem. 1995;270: 6729–33.
- Guyer DR, Tiedeman J, Yannuzzi LA, Slakter JJ, Parke D, Kelley J, Tang RA, Marmor M, Abrams G, Miller JW, Gragoudas ES. Interferon-associated retinopathy. Arch Ophthalmol. 1993;111:350–6.
- 49. Haller JA, Qin H, Apte RS, et al. on behalf of the Diabetic Retinopathy Clinical Research Network (DRCR.net). Vitrectomy outcomes in eyes with diabetic macular edema and vitreomacular traction. Ophthalmology. 2010;117(6):1087–93.
- 50. Hamm H. Zentralskotom nach Sonnenblendung, dissertation, Hamburg, 1947.
- Hardy J, Panisset A, Marchildon A, Lanthier A. Microsurgical selective anterior pituitary ablation for diabetic retinopathy. Can Med Assn J. 1969;100(17):785–92.
- 52. Hartley KL, Smiddy WE, Flynn HW, Murray TG. Pars plana vitrectomy with internal limiting membrane peeling for diabetic macular edema. Retina. 2008;28(3):410–9.
- 53. Hernandez-DaMota SE, Chacon-Lara A, Hernandez-Vazquez E. Use of triamcinolone and bevacizumab in 25 ga phaco-vitrectomy surgery for the treatment of cataract and diabetic macular edema. Arch Soc Esp Oftalmol. 2008;83(5):293–300.
- Higgins KE, Meyers SM, Jaffe MJ, et al. Temporary loss of foveal contrast sensitivity associated with panretinal photocoagulation. Arch Ophthalmol. 1986;104:997–1003.

- 55. Hoerauf H, Bruggemann A, Muecke M, et al. Pars plana vitrectomy for diabetic macular edema. Internal limiting membrane delamination vs. posterior hyaloid removal. A prospective randomized trial. Graefes Arch Clin Exp Ophthalmol. 2011;249(7):997–1008.
- Houssay BA, Biasotti A. La diabetes pancreatica de los perros hipofisoprivos. Rev Soc Argent Biol. 1930;6:251–96.
- 57. Huamonte FJ, Peyman GA, Goldberg MF, et al. Immediate fundus complications after retinal scatter photocoagulation. Ophthalmic Surg. 1974;93:287.
- Inagaki K, Ohkoshi K, Ohde S, Deshpande GA, Ebihara N, Murakami A. Comparative efficacy of pure yellow (577-nm) and 810-nm subthreshold micropulse laser photocoagulation combined with yellow (561-577-nm) direct photocoagulation for diabetic macular edema. Jpn J Ophthalmol. 2015;59(1):21–8.
- International Consortium Investigating Early Vitrectomy in Diabetic Macular Edema Patients (ICV-DME) ClinicalTrials.gov Identifier: NCT02639507.
- Joplin GF, Fraser R, Hill DW, Oakley NW, Scott DJ, Doyle FH. Pituitary ablation for diabetic retinopathy. Quart J Med. 1965;34:443.
- 61. Joplin GF, Oakley NW, Hill DW, Kohner EM, Fraser TR. Diabetologia. 1967;3:406.
- 62. Jung JJ, Gallego-Pinazo R, Lleó-Pérez A, Huz JI, Barbazetto IA. NAVILAS laser system focal laser treatment for diabetic macular edema – one year results of a case series. Open Ophthalmol J. 2013;7:48–53.
- Kamura Y, Sato Y, Isomae T, Shimada H. Effects of internal limiting membrane peeling in vitrectomy on diabetic cystoid macular edema patients. Jpn J Ophthalmol. 2005;49(4):297–300.
- 64. Kang SW, Park SC, Cho HY, Kang JH. Triple therapy of vitrectomy, intravitreal triamcinolone, and macular laser photocoagulation for intractable diabetic macular edema. Am J Ophthalmol. 2007;144(6):878–85.
- 65. Kim YM, Lee SY, Koh HJ. Prediction of postoperative visual outcome after pars plana vitrectomy based on preoperative multifocal electroretinography in eyes with diabetic macular edema. Graefes Arch Clin Exp Ophthalmol. 2010;248(10):1387–93.
- 66. Kim YT, Kang SW, Kim SJ, et al. Combination of vitrectomy, IVTA, and laser photocoagulation for diabetic macular edema unresponsive to prior treatments; 3-year results. Graefes Arch Clin Exp Ophthalmol. 2012;250(5):679–84.
- 67. Kimura T, Kiryu J, Nishiwaki H, et al. Efficacy of surgical removal of the internal limiting membrane in diabetic cystoid macular edema. Retina. 2005;25(4):454–61.
- 68. Klein R, Klein B. The epidemiology of diabetic retinopathy. In: Ryan SJ, editor. Retina II. London: Elsevier; 2013. p. 907–24.
- 69. Kohner EM, Dollery CT, Paterson JW, Fraser TR. Excerpta med (Amst) Internadonal Congress Series. 1967;140:13.
- Kozak I, Oster SF, Cortes MA, et al. Clinical evaluation and treatment accuracy in diabetic macular edema using navigated laser photocoagulator NAVILAS. Ophthalmology. 2011;118:1119–24.
- Kumagai K, Furukawa M, Ogino N, et al. Long-term follow-up of vitrectomy for diffuse nontractional diabetic macular edema. Retina. 2009;29(4):464–72.
- 72. Kumar A, Sinha S, Azad R, et al. Comparative evaluation of vitrectomy and dye-enhanced ILM peel with grid laser in diffuse diabetic macular edema. Graefes Arch Clin Exp Ophthalmol. 2007;245(3):360–8.
- Kwon YH, Lee DK, Kwon OW. The short-term efficacy of subthreshold Micropulse yellow (577-nm) laser photocoagulation for diabetic macular edema. Korean J Ophthalmol. 2014;28:379–85.
- 74. Landers MB III, Kon Graversen VA, Stewart MW. Early vitrectomy for DME: does it have a role? Sometimes vitrectomy can be first-line treatment. Part 1 of 2. Retinal Physician. 2013:46–53.
- Landers MB III, Kon Graversen VA, Stewart MW. Early vitrectomy for DME: does it have a role? Sometimes vitrectomy can be first-line treatment. Part 2 of 2. Retinal Physician 2013:56–60.

- 76. Lavinsky D, Cardillo JA, Melo Jr LA, et al. Randomized clinical trial evaluating mETDRS versus normal or high-density micropulse photocoagulation for diabetic macular edema. Invest Ophthalmol Vis Sci. 2011;52:4314–23.
- 77. Lawrence JH, Tobias CA, Linfoot JA, Born JL, Gottschalk A, Kling RP. Diabetes. 1963;12:490.
- Lee SN, Chhablani J, Chan CK, et al. Characterization of microaneurysm closure after focal laser photocoagulation in diabetic macular edema. Am J Ophthalmol. 2013;155(5): 905–12.
- Lewis H, Abrams GW, Blumenkranz MS, Campo RV. Vitrectomy for diabetic macular traction and edema associated with posterior hyaloidal traction. Ophthalmology. 1992;99(5):753–9.
- Lewis H, Schachat A, Haimann MH, et al. Choroidal neovascularization after laser photocoagulation for diabetic macular edema. Ophthalmology. 1990;97:503.
- Long CNH, Lukens FDW. The effects of adrenalectomy and hypophysectomy upon experimental diabetes in the cat. J Exp Med. 1936;63:465–90.
- Lovestam-Adrian M, Agardh E. Photocoagulation of diabetic macular edema—complications and visual outcome. Acta Ophthalmol Scand. 2000;78(6):667–71.
- Lovestam-Adrian M, Larsson J. Vitrectomy seems to be beneficial for advanced diffuse diabetic macular oedema not responding to laser treatment. Int Ophthalmol. 2005;26(1–2): 21–6.
- Luttrull JK, Musch DC, Mainster MA. Subthreshold diode micro- pulse photocoagulation for the treatment of clinically significant diabetic macular oedema. Br J Ophthalmol. 2005;89: 74–80.
- Luttrell JK, Sramek C, Palanker D, Spink CJ, Musch DC. Long-term safety, high-resolution imaging, and tissue temperature modeling of subvisible diode micropulse photocoagulation for retinovascular macular edema. Retina. 2012;32:375–86.
- Ma J, Yao K, Jiang J, et al. Assessment of macular function by multifocal electroretinogram in diabetic macular edema before and after vitrectomy. Doc Ophthalmol. 2004;109(2): 131–7.
- Maeshima K, Utsugi-Sutoh N, Otani T, Kishi S. Progressive enlargement of scattered photocoagulation scars in diabetic retinopathy. Retina. 2004;24:507–11.
- 88. Maggiore L. Socl Ital Oftal, Rome, 1927.
- Matsumoto M, Yoshimura N, Honda Y. Increased production of transforming growth factor -β2 from cultured human retinal pigment epithelial cells by photocoagulation. Invest Ophthalmol Vis Sci. 1994;35:4645–52.
- 90. McDonald HR, Schatz H. Macular edema following panretinal photocoagulation. Retina. 1985;5:5e10.
- McDonald HR, Schatz H. Visual loss following panretinal photocoagulation for proliferative diabetic retinopathy. Ophthalmology. 1985;92:388–93.
- McMeel JW. Phootocoagulation approach with various diabetic vitreoretinal problems. In: L'Esperance F, editor. Current Diagnosis and Management of Chorioretinal Diseases. CV Mosby: St. Louis; 1977. p. 269.
- 93. Meyer-Schwickerath G: Ber Vers Deutsch Ophthl Ges Heidelberg. 1949;55:256.
- 94. Meyer-Schwickerath G: Ber Vers Deutsch Ophthl Ges Heidelberg. 1951;57:144.
- 95. Meyer-Schwickerath G. History and development of photocoagulation. Am J Ophthalmol. 1967;63:1812–4.
- 96. Meyer-Schwickerath G, Schott K. Diabetic retinopathies and light coagulation. Ophthalmologica. 1969;158:605–14.
- Michaelides M, Kaines A, Hamilton RD, et al. A prospective randomized trial of intravitreal bevacizumab or laser therapy in the management of diabetic macular edema (BOLT Study). Ophthalmology. 2010;117(6):1078–86.
- Michaelson IC. The mode of development of the vascular system of the retina with some observations on its significance for certain retinal disorders. Trans Ophthalmol Soc UK. 1948;68:1625–710.

- Mitchell P, Bandello F, Schmidt-Erfurth U, et al. RESTORE study group. The RESTORE study: ranibizumab monotherapy or combined with laser versus laser monotherapy for diabetic macular edema. Ophthalmology. 2011;118:615–25.
- Miura Y, Treumer F, Klettner A, et al. VEGF and PEDF secretions over time following various laser irradiations on an RPE organ culture. Invest Ophthalmol Vis Sci. 2010;51:469.
- 101. Mochizuki Y, Hata Y, Enaida H, et al. Evaluating adjunctive surgical procedures during vitrectomy for diabetic macular edema. Retina. 2006;26(2):143–8.
- 102. Mohamed Q, Gillies MC, Wong TY. Management of diabetic retinopathy: a systematic review. JAMA. 2007;298:902–16.
- 103. Moorman CM, Hamilton AM. Clinical applications of the Micro-Pulse diode laser. Eye. 1999;13:145–50.
- 104. Morizane Y, Kimura S, Hosokawa M, Shiode Y, Hirano M, Doi S, Hosogi M, Fujiwara A, Inoue Y, Shiraga F. Planned foveal detachment technique for the resolution of diffuse diabetic macular edema. Jpn J Ophthalmol. 2015;59:279–87.
- 105. Muqit MM, Gray JC, Marcellino GR, et al. Barely visible 10-millisecond pascal laser photocoagulation for diabetic macular edema: observations of clinical effect and burn localization. Am J Ophthalmol. 2010;149:979–86.
- 106. Muqit MM, Henson DB, Young LB, et al. Laser tissue interactions. Ophthalmology. 2010;117:2039.
- 107. Muqit MM, Marcellino GR, Henson DB, et al. Single-session vs multiple-session pattern scanning laser panretinal photocoagulation in proliferative diabetic retinopathy. Arch Ophthalmol. 2010a;128:525–33.
- 108. Muqit MM, Young LB, McKenzie R, et al. Pilot randomised clinical trial of Pascal TargETEd Retinal versus variable fluence PANretinal 20 ms laser in diabetic retinopathy: PETER PAN Study. Br J Ophthalmol. 2013;97:220–7.
- 109. Nakamura Y, Mitamura Y, Ogata K, et al. Functional and morphological changes of macula after subthreshold micropulse diode laser photocoagulation for diabetic macular oedema. Eye. 2010;24:784–8.
- Naito T, Matsushita S, Sato H, et al. Results of submacular surgery to remove diabetic submacular hard exudates. J Med Invest. 2008;55(3–4):211–5.
- 111. Navarro A, Pournaras JAC, Hoffart L, et al. Vitrectomy may prevent the occurrence of diabetic macular edema. Acta Ophthalmologica. 2010;88(4):483–5.
- 112. Nicolò M, Musetti D, Traverso CE. Yellow micropulse laser in diabetic macular edema: a short-term pilot study. Eur J Ophthalmol. 2014;24:885–9.
- 113. Nonaka A, Kiryu J, Tsujikawa A, et al. Inflammatory response after scatter laser photocoagulation in nonphotocoagulated retina. Invest Ophthalmol Vis Sci. 2002;43:1204–9.
- 114. Nguyen QD, Shah SM, Khwaja AA, et al. Two-year outcomes of the ranibizumab for edema of the macula in diabetes (READ-2) study. Ophthalmology. 2010;117(11):2146–51.
- 115. Olk RJ. Modified grid argon laser photocoagulation for diffuse diabetic macular edema. Ophthalmology. 1986;93:938.
- 116. Olk RJ. Argon green (514 nm) versus krypton red (647 nm) modified grid laser photocoagulation for diffuse diabetic macular edema. Ophthalmology. 1990;997:1101.
- 117. Othman IS, Eissa SA, Kotb MS, Sadek SH. Subthreshold diode-laser micropulse photocoagulation as a primary and secondary line of treatment in management of diabetic macular edema. Clin Ophthalmol. 2014;8:653–9.
- Pankratov MM. Pulsed delivery of laser energy in experimental thermal retinal photocoagulation. Proc Soc Photo-Optical Instrum Eng. 1990;1202:205–13.
- 119. Park JH, Woo SJ, Ha YJ, Yu HG. Effect of vitrectomy on macular microcirculation in patients with diffuse diabetic macular edema. Graefes Arch Clin Ophthalmol. 2009;247(8):1009–17.
- Patel JI, Hykin PG, Schadt M, et al. Pars plana vitrectomy for diabetic macular oedema: OCT and functional correlations. Eye. 2006;20(6):674–80.
- 121. Patel JI, Hykin PG, Schadt M, et al. Pars plana vitrectomy with and without peeling of the inner limiting membrane for diabetic macular edema. Retina. 2006;26(1):5–13.

- 122. Rand RW, Dashe AM, Paglia DE, Conway LW, Solomon DH. JAMA. 1964;189:255.
- 123. Recchia FM, Ruby AJ, Carvalho Rechia CA. Pars plana vitrectomy with removal of the internal limiting membrane in the treatment of persistent diabetic macular edema. Am J Ophthalmol. 2005;139(3):447–54.
- 124. Rosenblatt BJ, Shah GK, Sharma S, Bakal J. Pars plana vitrectomy with internal limiting membranectomy for refractory diabetic macular edema without a taut posterior hyaloid. Graefes Arch Clin Exp Ophthalmol. 2005;243(1):20–5.
- 125. Rucker CW, Gastineau CF, Svien HJ. Mayo Clin Proc. 1967;42:409.
- 126. Sakamoto A, Nishijima K, Kita M, et al. Association between foveal photoreceptor status and visual acuity after resolution of diabetic macular edema by pars plana vitrectomy. Graefes Arch Clin Exp Ophthalmol. 2009;247(10):1325–30.
- 127. Schatz H, Madeira D, McDonald HR, Johnson RN. Progressive enlargement of laser scars following grid laser photocoagulation for diffuse diabetic macular edema. Arch Ophthalmol. 1991;109:1549–51.
- 128. Shah SP, Patel M, Thomas M, et al. Factors predicting outcome of vitrectomy for diabetic macular oedema: results of a prospective study. Br J Ophthalmol. 2006;90(1):33–6.
- 129. Shiba T, Kamura Y, Yagi F, Sato Y. Comparison of surgical procedures for vitreous surgery in diabetic macular edema. Jpn J Ophthalmol. 2009;53(2):120–4.
- 130. Shimonagano Y, Makiuchi R, Miyazaki M, et al. Results of visual acuity and foveal thickness in diabetic macular edema after vitrectomy. Jpn J Ophthalmol. 2007;51(3):204–9.
- 131. Skowsky WR, Siddiqui T, Hodgetts D, Lambrou Jr FH, Stewart MW, Foster Jr MT. A pilot study of chronic recombinant interferon-alfa 2a for diabetic proliferative retinopathy: metabolic effects and ophthalmologic effects. J Diabetes Complications. 1996;10:94–9.
- 132. Song SH, Sohn JH, Park KH. Evaluation of the efficacy of vitrectomy for persistent diabetic macular edema and associated factor predicting outcome. Kor J Ophthalmol. 2007;21(3):146–50.
- 133. Speakman JS, Mortimer CB, Briant TD, Ezrin C, Lougheed WM, Clarke WTW. Canad Med Ass J. 1966;94:627.
- 134. Speigel RJ. The alpha interferons: Clinical overview. Semin Oncol. 1989;14(suppl 2):1–12.
- 135. Stefansson E. The therapeutic effects of retinal laser treatment and vitrectomy. A theory based on oxygen and vascular physiology. Acta Ophthalmol Scand. 2001;79:435–40.
- 136. Stolba U, Binder S, Gruber D, et al. Vitrectomy for persistent diffuse diabetic macular edema. Am J Ophthalmol. 2005;140(2):295–301.
- 137. Symposium on the Treatment of diabetic retinopathy. Goldberg MF, Fine SL, editors. Washington: Government Printing Office, 1969. p. 378.
- Takahashi T, Ueno H, Shibuya M. VEGF activates protein kinase C-dependent, but Rasindependent Raf-MEF-MAP kinase pathway for DNA synthesis in primary endothelial cells. Oncogene. 1999;18:2221–30.
- 139. The Diabetic Retinopathy Study Research Group. Preliminary report on effects of photocoagulation therapy. Am J Ophthalmol. 1976;81:383–96.
- 140. The Diabetic Retinopathy Study Research Group: Photocoagulation treatment of proliferative diabetic retinopathy: the second report of diabetic retinopathy study findings. Trans Am Acad Ophthalmol Otolaryngol. 1978;85:82.
- 141. The Diabetic Retinopathy Study Research Group: Photocoagulation treatment of proliferative diabetic retinopathy: clinical application of diabetic retinopathy study (DRS) findings, DRS report No. 8. Ophthalmology. 1981;88:583.
- 142. The Diabetic Retinopathy Study Research Group. Indications for photocoagulation treatment of diabetic retinopathy: diabetic retinopathy study report no. 14. Int Ophthalmol Clin. 1994;27:239–53.
- 143. Thomas D, Bunce C, Moorman C, Laidlaw DAH. A randomized controlled feasibility trial of vitrectomy versus laser for diabetic macular oedema. Br J Ophthalmol. 2005;89(1):81–6.
- 144. Tolentino MJ, McLeod DS, Taomoto M. OtsujiT, Adamis AP, Lutty GA. Pathologic features of vascular endothelial growth factor-induced retinopathy in the nonhuman primate. Am J Ophthalmol. 2002;133(3):373–85.
- 145. Tonello M, Costa RA, Almeida FP, et al. Panretinal photocoagulation versus PRP plus intravitreal bevacizumab for high-risk proliferative diabetic retinopathy (IBeHi study). Acta Ophthalmol. 2008;86:385–9.
- 146. Tsujikawa A, Kiryu J, Dong J, et al. Quantitative analysis of diabetic macular edema after scatter laser photocoagulation with the scanning retinal thickness analyzer. Retina. 1999;19:59–64.
- 147. Tsuruoka N, Sugiyama M, Tawaragy Y, Tsujimoto M, Nishihara T, Goto T, Sato N. Inhibition of in vitro angiogenesis by lymphotoxin on interferon gamma. Biochem Biophys Res Commun. 1988;155:429–35.
- 148. Tuder RM, Flook BE, Voelkel NF. Increased gene expression for VEGF and the VEGF receptors KDR/Flk and Flt in lungs exposed to acute or chronic hypoxia. J Clin Invest. 1995;95:1798–807.
- 149. Vander JF, Duker JS, Benson WE, et al. Long term stability and visual outcome after favorable initial response for proliferative diabetic retinopathy to panretinal photocoagulation. Ophthalmology. 1991;98:1575–9.
- 150. Vine AK. The efficacy of additional argon laser photocoagulation for persistent, severe proliferative diabetic retinopathy. Ophthalmology. 1985;92:1532.
- 151. Volvista V, Pelkonen R, Cantell K. Effects of interferon on glucose tolerance and insulin sensitivity. Diabetes. 1989;38:641–7.
- 152. Vujosevic S, Bottega E, Casciano M, et al. Microperimetry and fundus autofluorescence in diabetic macular edema: subthreshold micropulse diode laser versus modified early treatment diabetic retinopathy study laser photocoagulation. Retina. 2010;30:908–16.
- 153. White CW, Sondheimer HM, Crouch EC, Wilson H, Fan LL. Treatment of pulmonary hemangiomatosis with recombinant interferon alfa 2. N Engl J Med. 1989;30:1197–200.
- 154. Widmark. Skand Arch Physiol. 1983;4:281.
- 155. Wilson DJ, Finkelstein D, Quigley HA, Green RW. Macular grid photocoagulation. An experimental study on the primate retina. Arch Ophthalmol. 1988;106:100–5.
- 156. Wolter JH, Knoblich RR. Br J Ophthalmol. 1965;49:246.
- 157. Writing Committee for the Diabetic Retinopathy Clinical Research Network, Gross JG, Glassman AR, Jampol LM, Inusah S, Aiello LP, Antoszyk AN, Baker CW, Berger BB, Bressler NM, Browning D, Elman MJ, Ferris FL 3rd, Friedman SM, Marcus DM, Melia M, Stockdale CR, Sun JK, Beck RW. Panretinal Photocoagulation vs Intravitreous Ranibizumab for Proliferative Diabetic Retinopathy: A Randomized Clinical Trial. JAMA 2015;314(20):2137–46.
- 158. Yamamoto T, Hitani K, Tsukahara I, et al. Early postoperative retinal thickness changes and complications after vitrectomy for diabetic macular edema. Am J Ophthalmol. 2003;135(1):14–9.
- 159. Yamamoto T, Takeuchi S, Sato Y, Yamashita H. Long-term follow-up results of pars plana vitrectomy for diabetic macular edema. Jpn J Ophthalmol. 2007;51(4):285–91.
- 160. Yang CM. Surgical treatment for diabetic retinopathy: 5-year experience. J Formos Med Assoc. 1998;97(7):477-84.
- Yanyali A, Horozoglu F, Celik E, et al. Long-term outcomes of pars plana vitrectomy with internal limiting membrane removal in diabetic macular edema. Retina. 2007;27(4):557–66.
- 162. Yanyali A, Nohutcu AF, Horozoglu F, Celik E. Modified grid laser photocoagulation versus pars plana vitrectomy with internal limiting membrane removal in diabetic macular edema. Am J Ophthalmol. 2005;139(5):795–801.
- 163. Zein WM, Noureddin BN, Jurdi FA, et al. Panretinal photocoagulation and intravitreal triamcinolone acetonide for the management of proliferative diabetic retinopathy with macular edema. Retina. 2006;26:137–42.
- 164. Ziemianski MC, McMeel JW, Franks EP. Natural history of vitreous hemorrhage in diabetic retinopathy. Ophthalmology. 1980;87(4):306–12.

Chapter 4 Targeting Vascular Endothelial Growth Factor

4.1 Introduction

Laser photocoagulation was the preferred treatment for the complications of diabetic retinopathy – diabetic macular edema (DME) and proliferative diabetic retinopathy (PDR) – for several decades with level I evidence emerging from the Early Treatment of Diabetic Retinopathy Study [45] and the Diabetic Retinopathy Study (DRS) [166–168]. Laser reduced the risk of severe vision loss (<5/200) in eyes with PDR and moderate vision loss (>15 letters) in eyes with DME by one half, but significant improvements in visual acuity (>15 letters) were unusual.

In an attempt to improve the treatment of macular edema, physicians began injecting triamcinolone acetate into the vitreous of the eyes with macular edema due to diabetic retinopathy and retinal vein occlusions [77, 91, 114] early in the twenty-first century. Corticosteroids improved macular edema dramatically but the high incidences of posterior subcapsular cataracts and elevated intraocular pressures limited their widespread adoption. Surgeons frequently performed pars plana vitrectomy for advanced complications of PDR (vitreous hemorrhage, traction retinal detachment, traction/rhegmatogenous retinal detachment) [169] and even for DME that was accompanied by vitreomacular traction [106]. Surgical outcomes for PDR were superior to the natural history of the disease [169], but vitrectomy only benefited a small number of patients with vision loss due to DME [60].

Effective pharmacotherapy for DME eventually came from a surprising source – tumor biology [24, 56]. The discovery of vascular endothelial growth factor (VEGF) and the early understanding of its role in neovascular age-related macular degeneration, macular edema due to retinal vein occlusions, DME, and PDR [5] set in motion a pharmacologic revolution in ophthalmology. The development of potent ocular pharmaceuticals created a paradigm shift in the treatment of center-involving diabetic macular edema and more recently in the treatment of PDR (Fig. 4.1).

Many manuscripts describing the treatment of DME with anti-VEGF drugs have been published, but most are flawed with small numbers of patients, short durations of treatment and follow-up, lack of control arms and randomization, and inadequate



Fig. 4.1 Important milestones in the discovery of vascular endothelial growth factor and the development of VEGF-inhibitory drugs. Each of the VEGF-inhibitory drugs was initially approved by the US Food and Drug Administration for the treatment of neovascular age-related macular degeneration (nAMD) with approvals for diabetic macular edema lagging by 6 years (ranibizumab) and 3 years (aflibercept)

masking. Fortunately, several high-quality, randomized, controlled, double-masked, multicenter trials have been published, and these constitute the focus of this chapter.

4.2 Vascular Endothelial Growth Factor

Pathologic ocular neovascularization has long been recognized as the sequela to severe vascular injury. Michaelson (1948) postulated the existence of a soluble factor X that he believed was responsible for new blood vessel growth in conditions such as neovascular glaucoma (Fig. 4.2) [119]. Folkman (1971) noted that both primary and metastatic solid tumors in children were accompanied by leashes of blood vessels through which the tumor obtained oxygen and nutrients [61]. He postulated that solid tumors synthesized a pro-angiogenic substance that promoted blood vessel growth necessary to sustain tumors.

Vascular permeability factor (VPF) was discovered in 1983 [155], and Ferrara and Connolly independently discovered vascular endothelial growth factor (VEGF) [25, 56] 6 years later. Sequencing analysis showed that both investigators had identified the same molecule, which turned out to be the previously discovered VPF. This set in motion a multi-specialty, international research initiative that characterized VEGF, its receptors, and its downstream pathways.

VEGF is actually a group of closely related molecules that segregate into seven families: VEGF-A, VEGF-B, VEGF-C, VEGF-D, the orf virus encoded VEGF-E,



Fig. 4.2 Slit lamp photograph showing severe neovascularization of the iris. Contraction of the neovascular membrane on the anterior surface of the iris everts the pupillary margin and exposes the posterior pigmented epithelium of the iris (ectropion uvea)

VEGF-F, and placental growth factor (PIGF) (Table 4.1). Most of these families consist of several isoforms with similar binding sequences.

VEGF-A isoforms are the most important contributors to ocular angiogenesis. The 14 kb VEGF-A gene resides on chromosome 6p21.3 [174]. Splicing of eight exons and seven introns creates at least six major (VEGF₁₂₁, VEGF₁₄₅, VEGF₁₆₅, VEGF₁₈₃, VEGF₁₈₉, and VEGF₂₀₆) and several minor isoforms within the VEGF-A family [84, 170]. In addition, the split product VEGF₁₁₀ is produced when tissue-sequestered isoforms with 165 or more amino acids are cleaved by plasmin and matrix metalloproteinase-3. The shorter non-heparin-binding isoforms are biologically active but with only 10–20% the biological activity of VEGF₁₆₅. VEGF naturally exists as a homodimer, so it presents two receptor-binding sites to its environment [95, 104].

Each of the VEGF families controls a different type of angiogenesis (Table 4.1). VEGF-B stimulates coronary blood vessel growth, and VEGF-C and VEGF-D promote lymphangiogenesis by activating VEGF receptor (VEGFR) 3 [94]. The exact role of PIGF in angiogenesis is unclear, but its binding to VEGFR1 may force VEGF-A to preferentially bind to VEGFR2 [137], the receptor primarily responsible for ocular angiogenesis. PIGF may play an active role in the development of DR as it has been found in the retinas of diabetic rats [89, 99].

The common receptor-binding sequence of all VEGF-A isoforms resides in the amino acid region 82–93, whereas the heparin-binding sequence is found in the 111–165 amino acid region of longer isoforms. Molecular charge determines isoform diffusibility with the acidic VEGF₁₂₁ isoform being freely diffusible whereas the basic VEGF₂₀₆ isoform is nearly totally bound to interstitial matrix. Fifty to

Vascular endothelial g	rowth factors and their major functions	
VEGF-A	Responsible for most angiogenesis in adults	
$\begin{array}{l} (\text{VEGF}_{121}, \text{VEGF}_{145}, \\ \text{VEGF}_{165}) \end{array}$	Responsible for most ocular neovascularization and breakdown of the blood retinal barrier	
VEGF-B	Responsible for coronary angiogenesis	
	Promotes tumor metastasis	
	Found in choroidal neovascular membranes	
VEGF-C	Main function is lymphangiogenesis	
	Found to participate in angiogenesis, perhaps including choroidal	
	neovascular membranes	
	Participates in neural development and blood pressure regulation	
VEGF-D	Main function is lymphangiogenesis surrounding pulmonary bronchioles	
VEGF-E	Encoded by orf viruses	
VEGF-F	Expressed in the venom of the habu snake	
Placental growth	Found in choroidal neovascular membranes	
factor 1 and 2	Important for vasculogenesis and angiogenesis during ischemia,	
	inflammation, wound healing, and cancer	

 Table 4.1
 The various VEGF families, the most important isoforms, and their major physiologic effects

The isoforms of VEGF-A are most important for ocular angiogenesis. Placental growth factor has been found in experimental models of diabetic retinopathy and in the aqueous of eyes with diabetic retinopathy, but its contribution to the development of diabetic retinopathy in humans is unknown

seventy percent of the electrically neutral VEGF₁₆₅ is bound to the interstitial matrix, and heparin-binding sequences enable isoforms of 165 amino acids and longer to bind to interstitial matrix heparin moieties. Sequestration of these longer molecules provides a readily available reservoir of VEGF when injury or inflammation induces a protease-mediated breakdown of the interstitium. The heparin-binding domain also enables VEGF₁₆₅ to bind to the transmembrane neuropilin-1 co-receptor.

Diffusible VEGF binds to the extracellular binding domains of three transmembrane receptors: VEGFR1 (flt-1), VEGFR2 (flk-1), and VEGFR3 (flt-4). Each of these receptors is composed of three major parts: the seven extracellular immunoglobulin-like VEGF-binding domains, the single transmembrane region, and the intracellular tyrosine kinase moieties [28, 165]. VEGFR1 binds VEGF-A, VEGF-B, and placental growth factor; VEGFR2 binds VEGF-A and VEGF-C; and VEGFR3 binds VEGF-C, VEGF-D, VEGF-E, and VEGF-F. VEGFR-1 binds VEGFR4 with a much higher affinity than does VEGFR2. Binding of VEGF by VEGFR1 and VEGFR2 occurs at the second and third extracellular domains. Alternative splicing of the RNA transcript for the binding domains of the VEGF receptors creates soluble VEGFR2 (sVEGFR2) and VEGFR1 (sVEGFR1), which inhibit vascular development by promoting vascular maturation and maintaining corneal avascularity [137].

The role of VEGFR1 in angiogenesis is unclear because its downstream activity varies with both the age of the individual and the cell type. Activation of VEGFR1 during embryological development stimulates angiogenesis, but during adulthood it dampens VEGF activity by acting as a decoy receptor.



Fig. 4.3 Diffusible dimers of vascular endothelial growth factor (VEGF) bind to transmembrane receptors causing them to dimerize. Receptor dimerization leads to conformational changes and activation of the cytoplasmic tyrosine kinase moieties. These phosphorylate several cytoplasmic proteins, which upregulate downstream biochemical pathways. Each of the four available VEGF-inhibitory drugs binds to diffusible VEGF to prevent receptor binding

VEGF-A activity occurs when either naturally occurring VEGF₁₂₁ or VEGF₁₆₅, or the fibrin-split product VEGF₁₁₀, binds to VEGFR2 (Fig. 4.3). Each of two VEGFR2 receptors binds to the VEGF dimer causing conformational changes in the 3-dimensional structures of the receptor molecules and activating their intracytoplasmic tyrosine kinase moieties (Fig. 4.2). This phosphorylates several intracellular enzymes that activate downstream pathways, which lead to VEGF's potent effects (Fig. 4.3) [21, 48, 79, 164].

Hypoxic environments created by processes such as tumor growth and occlusive vasculopathies upregulate VEGF [42]. Tissue hypoxia reduces hydroxylation of the cell's "oxygen sensor" HIF-1 α , preventing it from binding to the von-Hippel Lindau factor, and undergoing ubiquination and subsequent destruction within proteosomes. Stable HIF-1 α dimerizes with HIF-1 β to form a complex that enters the cell nucleus where it activates the promoter region of the VEGF gene and also upregulates both VEGFR1 and VEGFR2 [71, 172] synthesis (Fig. 4.4).

VEGF is upregulated during starvation and by several metabolic factors, chemokines, and cytokines. A variety of metabolic regulators including ROS (reactive oxygen species such as superoxide (O_2^-)) increase the synthesis of both VEGF and its receptors [173]. Several growth factors including epidermal growth factor, tumor necrosis factor (TNF)- α , transforming growth factor (TGF)- β , basic fibroblast growth factor, interleukin-6, insulin-like growth factor-1, keratinocyte growth



Fig. 4.4 Hypoxia inducible factor (HIF)-1 α is normal tissue oxygen conditions and is subsequently metabolized in the proteosomes. Under hypoxic conditions and in the presence of various chemokines and cytokines, HIF-1 α becomes stabilized by dimerizing with HIF-1 β , enters the cell nucleus, and binds to the promoter region of the vascular endothelial growth factor gene

factor, and platelet-derived growth factor upregulate VEGF and may work in conjunction with tissue hypoxia [128]. Oncologic mutations that stimulate RAS, a family of ubiquitous intracellular GTP-ases, upregulate VEGF [78]. Mechanical forces such as sheer stress and stretch that occur with vitreomacular traction promote VEGF production [46].

VEGF has been detected in both the neuroretina and pigment epithelium [97], and it can be produced by several cell types including pericytes, glia, vascular endothelial cells, invading leukocytes, and retinal pigment epithelium [2, 134]. Some VEGF isoforms modulate the synthesis of others by affecting HIF-1 α stability [183], and VEGF inhibitors probably function via a similar mechanism.

PIGF and VEGF-B binding to VEGFR1 activates signaling pathways implicated in monocyte chemotaxis [36] and inflammatory-related angiogenesis. Though PIGF appears to be a critical mediator of inflammatory angiogenesis, it does not appear to directly influence embryologic or physiologic adult angiogenesis [59]. Tyrosine kinase activity of VEGFR1 is relatively weak but several downstream signaling molecules including phospholipase C, the p85 subunit of PI-3 kinase, growth factor receptor bound protein, SHP-2, and NcK are associated with VEGFR1 phosphorylation sites. These pathways result in eNOS activation and calcium influx in vascular endothelial cells.

VEGFR2 activation is responsible for the full spectrum of VEGF-A-mediated responses within human endothelial cells (survival, proliferation, migration, and formation of a vascular tube) by stimulating Raf-MEP-ERK and MAPK pathways [164]. These pathways phosphorylate several endothelial cell proteins including phospholipase-3, PI-3 kinase, RAS GTP-ase activating protein and the Src family [48, 79]. They stimulate cell migration via FAK and paxillin, PI-3 kinase, and MAPK and prolong survival via PI-3 kinase and Akt/PKB [21].



Fig. 4.5 The many functions of vascular endothelial growth factor A

VEGF-A isoforms also bind to the heparin-binding domains of neuropilin-1 and neuropilin-2, transmembrane proteins with small, non-catalytic tails that present VEGF₁₆₅ favorably to VEGFR2. The neuropilin-modulated binding of VEGF₁₆₅ increases its biological activity by tenfold over that of VEGF₁₂₁, which cannot interact with neuropilin [63].

VEGF-A upregulates gene expression that promotes the division, migration, and survival of vascular endothelial cells, regulates vascular dilation and permeability, and stabilizes immature blood vessels (Fig. 4.5) [55]. VEGF-A enhances vascular permeability 50,000 times more than does histamine, increases nitric oxide production, and stimulates the synthesis of several molecules that are important to angiogenesis, such

as matrix metalloproteinases, which break down the interstitial matrix to allow advancement of proliferating endothelial cells [175]. VEGF₁₆₅ induces endothelial cell growth by activating the Raf-MEK-Erk pathway [52]. VEGF degrades the blood retinal barrier (BRB) in venules but not capillaries [126] by causing a calcium-mediated increase in hydraulic conductivity [14], increase in endothelial cell fenestrations [147], and opening of intercellular tight junctions by phosphorylating several junctional proteins [67]. Increased permeability allows plasma to extravasate and form an interstitial fibrin gel into which endothelial cells can proliferate and migrate [44]. VEGF-A increases hexose transport to meet the metabolic demands of neovascularization [138]. VEGF-A produces a dose-dependent vasodilation in vitro via nitric oxide synthesis [102] and retinal vasodilation in animal models and human retinal vein occlusions [51, 171].

VEGF-A promotes the survival of rat endothelial cells in vitro and in vivo, while VEGF blockade causes extensive apoptotic changes in the vasculature of neonatal but not adult mice [73]. Newly formed but not mature vessels are VEGF dependent [16, 181], and a complete vestment of pericytes is key to establishing VEGF independence [16]. Nonetheless, VEGF-A plays a role in maintaining the choriocapillaris in adults [53]. VEGF-A administration to hypoxic vascular endothelial cells upregulates phosphatidylinositol (PI)-3 kinase-Akt pathways and the expression of antiapoptotic proteins Bl-2 and A1 and prevents apoptosis and regression of the vasculature [6, 72]. The effects of VEGF on the retinal pigment epithelium remain controversial as some investigators have noted breakdown of the outer blood-retinal barrier whereas others have not [1, 74, 85, 122].

VEGF-A promotes monocyte chemotaxis and colony formation by granulocytemacrophage progenitor cells. VEGF-A upregulates the synthesis of intercellular adhesion molecule (ICAM)-1 and vascular cellular adhesion molecule (VCAM)-1, which allows natural killer (NT) cells to adhere to vascular endothelial cells.

Exogenous administration of VEGF into primate eyes induces changes that resemble diabetic retinopathy [171]. VEGF has been successfully blocked in animals with soluble decoy receptors [93], tyrosine kinase inhibitors, and antibodies directed against the VEGF dimer or its receptors [101]. In animal models, blocking VEGF pathways appears to be more effective than blocking other angiogenic pathways [52]. Drugs that block VEGF work primarily by reestablishing the blood-retinal barrier, "normalizing" the circulation, and preventing further new blood vessel growth.

Healthy human eyes have vitreous VEGF concentrations of 8.8 +/- 9.9 ng/ml [5], whereas patients with DME and other retinal vascular conditions have elevated intraocular VEGF concentrations [5, 80]. Following pars plana vitrectomy, the intravitreal half-life of VEGF in rabbit eyes decreases by 75%.

4.3 Anti-VEGF Drugs

Currently available drugs that inhibit the actions of vascular endothelial growth factor are listed in Table 4.2. Included are the major binding and pharmacokinetic properties of each drug.

Important characteristics of drugs that bind vascular endothelial growth factor				
Characteristic	Aflibercept	Bevacizumab	Pegaptanib	Ranibizumab
Description	Fusion protein with receptor sequences bound to Fc fragment of IgG	Recombinant, humanized, murine antibody to VEGF-A	Pegylated aptamer	Recombinant, humanized, murine antibody fragment to VEGF-A
Molecular weight (kDa)	115	149	50	48
Isoforms and families bound	VEGF-A, VEGF-B, placental growth factor	VEGF-A	VEGF ₁₆₅	VEGF-A
Binding affinity for VEGF ₁₆₅ (pmol)	0.5	58-1000	46–192	50
Intravitreal half-life (days)	9 (estimated)	9.8	7 (estimated)	7.1
Serum half-life (days)	6	21	0.2	2-4

Table 4.2 Drugs that bind to and inhibit the actions of vascular endothelial growth factor that have been used to treat ocular angiogenesis

The major binding and pharmacokinetic properties of each drug are detailed

4.3.1 Pegaptanib

Pegaptanib is a 28-base oligonucleotide – a modified RNA fragment or aptamer – that attaches to the heparin-binding domain of VEGF₁₆₅. It has a molecular weight of 50 kDa, a negative charge of 28, and a hairpin structure. The addition of a 40 kDa polyethylene glycol chain (pegylation) increased the drug's binding affinity ($K_D = 50$ pM for VEGF₁₆₅) and stability by protecting it against the action of endonucle-ases [152]. After injection into monkey eyes, the intravitreal concentration of pegaptanib decreases according to a first-order kinetics model with an intraocular half-life of 94 h. Pegaptanib accumulation in the serum is limited by both metabolism (endo-and exonuclease actions) and urinary excretion.

Drug developers had been concerned that pan-VEGF suppression might increase the risk of serious systemic adverse events so they developed pegaptanib to inhibit the actions of diffusible VEGF₁₆₅ but not the shorter natural VEGF₁₂₁ and fibrin split product VEGF₁₁₀. Pegaptanib's exact mechanism of action is not fully known, but it may prevent VEGF binding to neuropilin-1 (by obstructing the heparin-binding site), or it may stereoscopically interfere with VEGF's receptor-binding site. Pegaptanib reduces endothelial cell proliferation [15], completely inhibits VEGFmediated vascular permeability in a guinea pig model, and reduces experimental neovascularization, leukostasis, and BRB breakdown in diabetic rats [86]. In vitro studies suggest that pegaptanib inhibits VEGF-mediated processes as effectively as monoclonal antibodies [50]. A preclinical study in rhesus monkeys determined that several doses of pegaptanib were safe, and intraocular samples obtained 7 and 28 days after intravitreal injections demonstrated no molecular instability [43].

The important pegaptanib drug trials and their key findings are listed in Table 4.3.

The VEGF Inhibition Study in Ocular Neovascularization Clinical Trial Group (VISION) trials led to United States Food and Drug Administration (US FDA) approval of pegaptanib for the treatment of neovascular age-related macular degeneration (AMD). Compared to observation, patients receiving injections of 0.3 mg pegaptanib every 6 weeks experienced only half as much vision loss at 1 year (-14 vs. -7 letters) [76]. In the post-approval Evaluation of Efficacy and Safety in Maintaining Visual Acuity with Sequential Treatment of Neovascular AMD (LEVEL) trial, maintenance therapy with pegaptanib sustained most of the gains in vision previously acquired with the more potent, nonselective VEGF inhibitors bevacizumab and ranibizumab [62].

Following the nAMD trials, pegaptanib underwent clinical testing for the treatment of DME. In a phase II trial, 172 patients were randomized to receive one of three pegaptanib doses (0.3 mg, 1 mg, or 3 mg) or sham injections every 6 weeks for 36 weeks. A greater proportion of patients receiving 0.3 mg pegaptanib than sham improved by at least 10 letters (34% vs. 10%; P = 0.003) and at least 15 letters (18% vs. 7%; P = 0.12). Patients receiving 0.3 mg pegaptanib as compared to sham

Important diabetic macular edema trials with pegaptanib			
Trial and phase	Cohorts	Key findings	
Macugen DME trial Phase II 172 patients	Pegaptanib 0.3 mg 1.0 mg 3.0 mg Sham	 At 36 weeks, patients treated with 0.3 mg pegaptanib compared to sham/laser: 1. Greater likelihood of 10-letter gain (34% vs. 10%, <i>P</i> = 0.003) 2. Trend toward 15-letter gain (18% vs. 7%, <i>P</i> = 0.12) 3. Greater improvement in mean BCVA (+4.7 vs0.4 letters, <i>P</i> = 0.02) 4. Greater improvement in macular thickness (-68 μm vs. +4 μm, <i>P</i> = 0.02) 	
Querques et al. Retrospective 63 patients	Single cohort	At 36 weeks compared to baseline, patients had significant improvements in: 1. BCVA (<i>P</i> < 0.019) 2. Macular thickness (<i>P</i> < 0.001)	
Macugen study group Phase II/III 260 patients		 At 102 weeks, patients treated with 0.3 mg pegaptanib compared to laser: 1. Greater likelihood of 10-letter gain (36.8% vs. 19.7%, P = 0.0047) 2. Trend toward 15-letter gain (16.5% vs. 10.2%, P = 0.2466) 3. Greater improvement in mean BCVA (+6.1 vs. +1.3 letters, P < 0.01) 4. Fewer patients required laser for persistent edema (25.2% vs. 45.5%, P = 0.003) 	

Table 4.3 Important trials with pegaptanib and details their key findings

experienced a greater mean improvement in BCVA (+4.7 vs. -0.4 letters; P = 0.04) and greater mean decrease in macular thickness ($-68 \mu m vs. +4 \mu m$; P = 0.02). One case of endophthalmitis occurred after 652 injections [113].

In a 6-month retrospective study of 63 eyes with DME, patients receiving pegaptanib experienced significant improvements in BCVA (P = 0.019) and macular thickness (P < 0.001) compared to baseline. Patients were treated every 6 weeks PRN for recurrent edema, and most eyes required at least three injections. Compared to the phase II trial, patients in this retrospective study experienced similar rates of 1-line improvement in vision (55.6% vs. 59%) and reduction in CRT (42% vs. 29.6%) [143].

A phase II/III trial randomized 260 patients to receive pegaptanib or sham injections every 6 weeks [163]. Significantly more patients receiving pegaptanib than sham injections gained at least 10 letters (36.8% vs. 19.7%; P = 0.0047) though the proportions gaining at least 15 letters were not significantly different (16.5% vs. 10.2%; P = 0.2466). At week 102, patients receiving pegaptanib had greater mean gains in VA (+6.1 vs. +1.3 letters; P < 0.01), and fewer pegaptanib-treated patients required grid laser photocoagulation by week 54 (23.3% vs. 41.7%; P = 0.002) and week 102 (25.2% vs. 45.0%; P = 0.003) for persistent edema. A slightly higher proportion of pegaptanib-treated patients experienced a 25% reduction in CRT. Pegaptanib appeared safe since only two patients suffered cerebrovascular accidents, compared to one patient in the sham group.

Self-reported quality of life (QoL) was assessed with the 25-item National Eye Institute Visual Function Questionnaire (NEI VFQ 25) and the EQ-5D [109]. The NEI VFQ 25 domains of near vision, distance vision, and social functioning (week 54) and distance vision, social functioning, mental health, and composite score (week 102) demonstrated clinically meaningful (>5-point between-group difference) and statistically significant (P < 0.05) benefits favoring pegaptanib. No significant difference in the mean change in generic EQ-5D-weighted utility scores was seen. The authors concluded that the VA improvement from pegaptanib treatment versus sham is accompanied by improved vision-related QoL as reported by the DME patient.

Pegaptanib causes marked regression of new vessels in eyes with proliferative diabetic retinopathy (PDR) [3, 13, 101, 116]. Pegaptanib was evaluated as an adjunct to vitrectomy in eyes with recurrent vitreous hemorrhage due to PDR. Fifteen eyes of 14 patients each received one to three pegaptanib injections prior to undergoing vitrectomy for persistent vitreous hemorrhage (VH) or progressive traction retinal detachment. In the majority of patients with VH, pegaptanib enabled surgeons to place additional laser prior to vitrectomy. Surgery after pegaptanib injections was felt to be faster and less challenging compared with conventional vitrectomy for recurrent VH due to PDR [83].

Pegaptanib showed promise in the early DME trials but further development was halted because it was inferior to the pan-VEGF-A blockers bevacizumab and ranibizumab for the treatment of nAMD. The drug's developer appeared unwilling to devote the resources necessary to run large, multicenter, phase III DME trials. Some authors have argued that $VEGF_{165}$ -specific binding may have doomed pegaptanib for the treatment of nAMD because it was unable to prevent the pro-angiogenic effects of VEGF₁₂₁ and VEGF₁₁₀. Since these isoforms appear to be less important in the development of DME, pegaptanib may have been relatively more effective in this condition. The theoretical safety advantages of VEGF₁₆₅ specific binding over pan-VEGF-A binding in a diabetic population that is at risk of Antiplatelet Trialists' Collaborative (APTC)-defined events may also have favored pegaptanib use in patients with DME. Unfortunately, pegaptanib never received US FDA approval for DME, and plans for registration trials are unlikely to be resurrected. Its frequency of use in patients with nAMD and DME remains insignificant.

4.3.2 Bevacizumab

Bevacizumab is a humanized (93% of protein sequences are human), recombinant, murine-derived, monoclonal antibody that binds all isoforms of VEGF-A. Bevacizumab has an impressive dissociation constant for VEGF₁₆₅ (K_D = 56–1100 pM) though its binding affinity is less than that of the approved anti-VEGF drugs [136, 140]. Bevacizumab maximally inhibits proliferation of human umbilical vascular cells and improves their survival when the bevacizumab/VEGF ratio exceeds 2.6:1 [176].

The intravitreal half-life of bevacizumab in human eyes varies between 6.7 and 9.82 days, with an average of 8.25 days [117, 161]. Very poor intraocular penetration occurs after topical administration as only small quantities of drug reach the iris, ciliary body, and retina. Subconjunctival and intravitreal injections in rabbits produce similar serum concentrations, suggesting that the drug passes into the systemic circulation without being altered within the eye. Bevacizumab possesses a long serum half-life (21 days) because of its Fc fragment, and it significantly lowers serum VEGF concentrations [10].

Bevacizumab reaches the subretinal space in rabbit eyes within 2 h after intravitreal injections and reaches the inner retina, choroid, and serum in monkeys within the first day [35, 81, 156]. The peak concentrations in the iris and retina are twice those in the vitreous, suggesting that the drug undergoes active transport and binding within the tissues. Bevacizumab can be internalized by RPE cells though toxicity has not been noted [81].

Bevacizumab was developed for the systemic treatment of solid tumors [57] and is currently approved for advanced colorectal carcinoma, small cell lung cancer, renal cell carcinoma, cervical carcinoma, ovarian carcinoma, and glioblastoma. In a small pilot study, bevacizumab was administered intravenously to patients with nAMD [120]. Patients experienced anatomic and functional improvements, but the incidence of systemic side effects precluded further use. In 2005, bevacizumab was injected intravitreally into patients with macular edema due to central retinal vein occlusion and nAMD [150, 151].

The important bevacizumab diabetic retinopathy trials and their key findings are listed in Table 4.4.

Important diabetic macular edema trials with bevacizumab			
Trial and phase	Cohorts	Key findings	
DRCR.net Phase II	Bevacizumab 1.25 mg 2.5 mg Laser	Q6wk injections, 12-week endpoint Bevacizumab-treated patients averaged one-line BCVA gain better than those receiving laser Bevacizumab patients had 11% improvement in macular thickness at week 3, but stable through week 12	
Pan-American Collaborative Retina Group 139 eyes	Bevacizumab 1.25 mg 2.5 mg	 24 month, retrospective study Patients receiving 1.25 mg and 2.5 mg had improved 1. BCVA (20/150 to 20/75, <i>P</i> < 0.0001; 20/168 to 20/114, <i>P</i> = 0.02) 2. Central macular thickness (466.5 μm to 286.6 μm, <i>P</i> < 0.0001; 423.4 μm to 271.8 μm, <i>P</i> = 0.001) 	
Soheilian et al. (2009) 150 eyes Prospective	3 treatment arms: Bevacizumab Bevacizumab + IVT Laser	 At 36 weeks 1. 2 lines improvement in BCVA in 37, 25, and 14.8% of eyes. 2. Significant improvements in central retinal thickness only at 6 weeks 	
BOLT study Phase II 80 eyes Prospective	2 treatment arms: Bevacizumab q6wk Laser q4month	 At 1 year 1. More eyes receiving bevacizumab improved by 15 letters (11.9% vs. 5.3%) 2. More eyes receiving bevacizumab improved by ten letters (31% vs. 7.9%) 3. Bevacizumab eyes had greater improvements in mean BCVA (+8.0 vs0.5 letters, <i>P</i> = 0.0002) At 2 years 1. More eyes receiving bevacizumab improved by 15 letters (32% vs. 4%) 2. More eyes receiving bevacizumab improved by ten letters (49% vs. 7%) 3. Bevacizumab eyes had greater improvements in mean BCVA (+8.6 vs0.5 letters) 	

Table 4.4 The important diabetic retinopathy trials with bevacizumab and their key findings

The DRCR.net evaluated the short-term (12-week primary endpoint) efficacy of q6week bevacizumab in a phase II DME trial. One hundred twenty-one patients were randomized to receive laser photocoagulation or intravitreal injections of 1.25 mg or 2.5 mg bevacizumab, with or without laser. Patients receiving bevacizumab experienced a one-line improvement in visual acuity compared to those treated with laser. Approximately one-half of the bevacizumab-treated patients experienced a greater than 11% decrease in macular thickness at week 3, but additional improvements through week 12 were not seen. The authors concluded that 6-week injection intervals may be too long and that combining bevacizumab with laser photocoagulation provides no short-term advantage over bevacizumab mono-therapy [30].

The Pan-American Collaborative Retina Study (PACORES) reported the results of a 24-month retrospective study of 139 eyes (115 patients) with DME [8]. Patients who received 1.25 mg or 2.5 mg bevacizumab experienced significant improvements in mean VA (20/150 to 20/75, P < 0.0001; 20/168 to 20/114, P = 0.02) and central macular thickness (466.5 µm to 286.6 µm, P < 0.0001; 423.4 µm to 271.8 µm, P = 0.001).

In a 36-week prospective study, 150 eyes with DME were randomized to receive bevacizumab monotherapy (q12week injections), bevacizumab in combination with intravitreal triamcinolone, or macular laser photocoagulation [158]. Two Snellen lines of visual acuity improvement were achieved by 37% (bevacizumab monotherapy), 25% (bevacizumab + triamcinolone), and 14.8% (laser) of eyes. Compared to baseline, VA improved significantly at all visits in the bevacizumab monotherapy group, only at weeks 6 and 12 in the combination therapy group, but at no visits in the laser group. Significant improvements in central retinal thickness were seen only at 6 weeks. The authors concluded that quarterly bevacizumab injections are superior to laser photocoagulation but that triamcinolone provides no added benefit.

The Bevacizumab Or Laser Therapy (BOLT) in the management of diabetic macular edema trial best demonstrated the superiority of bevacizumab over laser for the treatment of DME [118, 157]. This prospective, single-center, 2-year trial randomized 80 eyes to receive q6week PRN bevacizumab or q4week PRN laser. At 1 year, more patients receiving bevacizumab than laser gained >15 letters (11.9% vs. 5.3%) and >10 letters (31% vs. 7.9%), and fewer lost >15 (2.4% vs. 26.3%) and >30 (0% vs. 5.3%) letters. Patients receiving bevacizumab experienced greater mean VA improvements compared to laser at 1 year (+8.0 vs. -0.5 letters, *P* = 0.0002) and 2 years (+8.6 vs. -0.5 letters). At 2 years, 49% of bevacizumab-treated patients improved by at least +10 letters and 32% by at least +15 letters, compared to only 7 and 4% of laser-treated eyes. Fewer bevacizumab than laser patients lost 15 letters (0% vs. 14%, *P* = 0.03). Eyes receiving bevacizumab had greater mean decreases in macular thickness compared to laser (-146 µm vs. -118 µm). The median number of treatments through 2 years was 13 bevacizumab injections and 4 laser treatments.

A post hoc analysis of the BOLT data showed that eyes with subretinal fluid at baseline were most likely to have persistent edema at 2 months [157], and the authors noted that resolution of edema by 4 months is a strong predictor of a favorable long-term response. They found that 20% of eyes with persistent edema at 12 months achieved dry retinas at 24 months with VA improvements of at least 15 letters. They stated that although the 4-month response may be predictive of long-term outcome, it should not lead to withholding of therapy.

Intravitreal injections of bevacizumab cause regression of optic disk neovascularization due to PDR [9, 159], but the effect is transient as neovascularization recurs by 12 weeks after single intravitreal injections [92]. Several studies have reported resolution of chronic vitreous hemorrhage after intravitreal injections of bevacizumab [4, 29, 47, 123], but these results must be viewed with caution since bevacizumab likely does not promote the clearing of vitreous hemorrhage, but rather it allows hemorrhage to clear while it prevents further bleeding.



Fig. 4.6 Ultra-widefield photograph showing an eye with proliferative diabetic retinopathy. Fibrovascular proliferation can be seen along the superotemporal arcade causing a traction retinal detachment that threatens the fovea. The vertical optical coherence tomography scan (*upper left corner of image*) was displaced superiorly to better demonstrate the traction retinal detachment in the superior macula that threatens the fovea. Fibrovascular proliferation just outside the inferotemporal arcade features active neovascularization (*black arrow*). An intravitreal injection of bevacizumab was given 3 days prior to pars plana vitrectomy to reduce intraoperative bleeding

Intravitreal injections of bevacizumab have been used as adjuvants to improve surgical outcomes by decreasing intraoperative hemorrhage, facilitating fibrovascular membrane dissection (Fig. 4.6) [22, 27, 29, 87, 108, 123, 135, 146, 148, 180] and reducing the incidence of postoperative vitreous hemorrhage [4, 29, 148, 179]. These studies, however, were limited by relatively small numbers of patients, heterogeneous retinal pathology (TRD and VH were studied together), and varying surgical techniques (multiple surgeons and different gauge vitrectomies). Many surgeons remain concerned that the preoperative administration of bevacizumab may worsen fibrovascular traction [7, 90, 124, 135] and may cause enlargement of the foveal vascular zone [23, 104, 105]. Some surgeons recommend that bevacizumab should be administered only within a few days of planned surgery so that prompt action can be taken if a traction retinal detachment worsens or a traction-rhegmatogenous detachment develops.

In a much larger study, 99 eyes of 90 patients scheduled for diabetic vitrectomy were randomized to receive intravitreal bevacizumab preoperatively or no injection [139]. Thirty-four patients received IVB on an average of 11.5 (range, 3–30) days before vitrectomy. Visual acuity improved significantly after surgery in both groups: from 20/617 to 20/62 in the IVB group, and from 20/443 to 20/86 in the non-IVB group (P = 0.11 between groups). Surgery time and postoperative complications (glaucoma, RD, and repeat vitrectomy rate) were similar in both groups. In patients under the age of 40 years, operating time was shorter in the bevacizumab group (P = 0.02) with a trend toward better VA. The authors concluded that preoperative

bevacizumab may be a useful adjunct to vitrectomy for severe PDR complicated by TRD, particularly in younger patients.

A meta-analysis of randomized controlled trials compared the safety and functional outcomes of vitrectomy with or without preoperative intravitreal bevacizumab for PDR [182]. Eight trials with 414 eyes of 394 patients were included. The authors found that vitrectomy with preoperative bevacizumab shortened overall surgical time (mean difference of 26.89 min, 95% confidence interval (CI) 31.38– 22.39; P < 0.00001) and reduced the number of required endodiathermy applications (mean difference of 3.46, P = 0.02) compared to vitrectomy alone. The bevacizumab group also experienced less intraoperative bleeding (odds ratio [OR] 0.10; 95% CI 0.02–0.46; P = 0.003) and recurrent vitreous hemorrhage within the first month (OR 0.35; 95% CI 0.21–0.58; P < .0001), but the incidence of recurrent vitreous hemorrhage after the first month was comparable between the two groups. There were no significant differences in other complication rates between the two groups, except for iatrogenic retinal breaks, which were more likely to occur in the vitrectomy-alone group (OR 0.27, 95% CI 0.12–0.63; P = 0.003).

4.3.3 Ranibizumab

Scientists at Genentech recognized that VEGF suppression could be used to treat nAMD, DME, and macular edema due to retinal vein occlusions, but they worried that a full-length antibody such as bevacizumab might have limited efficacy and an unfavorable safety profile. Monkey studies showed that the full-length HER2 antibody (molecular weight 149 kDa) was unable to cross the inner retina after intravitreal injection [125], an observation consistent with the proposed inner retina exclusion limit of 76 kDa [88]. They also worried that the antibody's Fc fragment might incite intraocular inflammation or cause systemic adverse events because of a prolonged serum half-life. For these reasons, ranibizumab, an affinity enhanced, humanized, single binding site antibody fragment (Fab), was created in 1996 [54] from a murine antibody to human VEGF. Ranibizumab (MW, 48 kDa) binds all isoforms of VEGF-A, but with a molar affinity for VEGF₁₆₅ that is 5–20 times that of bevacizumab. Its dissociation constant for the binding of VEGF₁₆₅ is 44–192 pM [111, 136]. After successfully completing phase III trials [17, 149], ranibizumab was approved by the US FDA in 2006 for the treatment of neovascular AMD.

Ranibizumab concentrations in the retina are 2.2- to 3-fold lower than in the vitreous [70] but since they are 3000-fold greater than VEGF concentrations, maximum binding of VEGF probably occurs within the retina. The terminal intraocular half-life of ranibizumab in monkey eyes appears to be concentration dependent, ranging from 2.63 days (0.5 mg dose) to 3.95 days (2.0 mg dose) [69]. The intravit-real half-life of ranibizumab in rabbits is slightly shorter at 2.1–3 days [70]. In human eyes, the intravitreal half-life ranges from 7.1 days in a study that sampled aqueous concentrations to 9 days in a population kinetics study [96, 178]. Ranibizumab appears to pass unaltered through the trabecular meshwork and cho-

roid into the systemic circulation where it has a half-life of only 2 h because of rapid ultrafiltration by the kidneys. Serum concentrations are 10,000-fold lower than those in the vitreous [68], and ranibizumab does not appear to suppress serum concentrations of VEGF [10].

Ranibizumab reduces the VEGF-mediated in vitro proliferation of human vascular endothelial cells (HUVAC) in a dose-dependent manner [110] and inhibits VEGF-mediated vascular permeability in guinea pigs. Both ranibizumab and bevacizumab increase vascular endothelial cell apoptosis and decrease their proliferation, migration, and assembly into vascular structures [26]. Both drugs decrease VEGF expression, VEGFR2 phosphorylation, and Akt expression [26]. Ranibizumab causes a greater reduction in endothelial cell proliferation whereas bevacizumab has a greater effect on migration, tube formation, and VEGFR2 phosphorylation [26]. In vitro proliferation of pig choroidal endothelial cells is decreased by ranibizumab (44.1%), bevacizumab (38.2%), and pegaptanib (35.1%) [160].

Pharmacokinetic studies suggested that clinically achievable ranibizumab concentrations within the retina after intravitreal injections were capable of suppressing angiogenic activity for 1 month [68, 100]. This was confirmed by the nAMD pilot studies so monthly injections were used in both the nAMD and DME trials [17, 149]. Excellent results were achieved with monthly dosing of patients with nAMD, but attempts to extend the treatment interval to 3 months resulted in progressive loss of vision [144]. Eyes that were treated with 3 monthly injections and observed for 2 months also showed vision loss of three to four letters [153], suggesting that monthly dosing of patients with nAMD was required to achieve the best results. Therefore, the monthly treatment strategy was carried over to the treatment of DME.

Evidence supporting the use of anti-VEGF therapy for DME emerged from several sources. Elevated concentrations of both VEGF and VEGFR2 were found in animal models of diabetic retinopathy [75]. Exogenous administration of VEGF into monkey eyes produces retinal changes similar to those of diabetic retinopathy, with retinal hemorrhages, increased capillary permeability, and neovascularization [147, 171]. The discovery that these changes could be prevented by the coadministration of VEGF-Trap A40 [142] was important to drug development.

Aqueous VEGF concentrations are higher in patients with DME than in those with nAMD and vein occlusions [64], and VEGF concentrations are higher in patients with severe as opposed to mild DME [65]. In eyes with PDR, vitreous VEGF concentrations fall after successful panretinal photocoagulation [5]. Retinal hypoxia may be the most potent stimulus for VEGF production in diabetes, and treating patients with continuous oxygen supplementation for 3 months significantly reduces DME [132].

4.3.3.1 Ranibizumab Clinical Trials

The important ranibizumab diabetic retinopathy trials and their key findings are listed in Table 4.5.

Important diabetic macular edema trials with ranibizumab (ran)			
Trial and phase	Cohorts	Key findings	
READ-2	Treatment arms:	RAN injections at baseline, 1, 3, 5 months	
Phase II	RAN	At 6 months	
126 patients	RAN + laser	1. Mean \triangle BCVA were +7, +4, 0 letters	
6 months	Laser	2. Mean changes in central retinal thickness were	
		$-95 \ \mu\text{m}, -82 \ \mu\text{m}, \text{and} -117 \ \mu\text{m}$	
		At 24 months	
		1. Mean \triangle BCVA were +8, +7, +5 letters (laser group	
		eligible for q3month RAN at 6 months)	
RISE and	Treatment arms:	At 24 months:	
RIDE	Monthly RAN	1. More RAN patients had three-line improvement in $DCNA (2AG) = A(G) = 12G = 12G$	
Phase III	U.5 Monthly DAN	BUVA $(34\% - 40\% \text{ VS}, 12\% - 18\%)$	
7.59 patients Registration		2. KAN patients had greater improvements in mean $BCVA (\pm 11 \text{ to } \pm 13 \text{ ys} \pm 0.5 \text{ to } \pm 3 \text{ letters})$	
trials	Sham/laser	3 RAN natients had greater improvements in macular	
trais	Shaniy laser	thickness ($-250 \text{ µm to } -270 \text{ µm vs.} -125 \text{ µm to}$	
		-133 um)	
		4. RAN patients had significant improvements in	
		diabetic retinopathy severity scores.	
		At 36 months:	
		1. No significant changes in BCVA or macular	
		thickness in RAN arms.	
		2. Sham/laser crossed over to RAN and improved by	
	T		
DRCR.net	PAN Llocar	At 12 months:	
Phase III	RAN + def	+3 letters	
854 patients	laser	2. BCVA improvements in IVT + laser group that were	
I	IVT + laser	pseudophakic at baseline were similar to RAN arms.	
	Laser	At 36 months:	
		1. Patients treated with RAN + def laser improved by	
		2.9 letters more than RAN + laser	
		2. Compared to patients in RAN + laser, patients in	
		three fewer lasers	
RESOLVE	Treatment arms:	At 12 months:	
Phase II	RAN03	1 Mean BCVA improved by +10.3 letters in pooled	
152 patients	RAN 0.5	RAN groups vs1 in sham/laser group	
F	Sham/laser	2. 2-line and 3-line improvements in 60.8 and 33% of	
	RAN groups	RAN patients and 18.4 and 5% of sham patients	
	were eligible	3. 68% of RAN patients required higher drug dose at	
	for dose	some point	
	doubling		
RESTORE	Treatment arms:	At 12 months:	
Phase III	RAN	1. Mean BCVA improvements were $+6.1$, $+5.9$, and	
345 patients	KAN + laser	+0.8 letters.	
	Laser	2. We an improvements in macular thickness were $-118.7 \text{ µm} = 128.3 \text{ µm}$ and -61.3 µm	
		At 36 months: $-120.5 \mu m$, and $-01.5 \mu m$	
		1. BCVA in laser arm caught up to RAN + laser after	
		crossover to RAN at 12 months	

 Table 4.5 Important diabetic retinopathy trials with ranibizumab and details their key findings

Human nAMD trials were evaluating ranibizumab well before pilot studies tested its use in patients with DME. The efficacy and safety of ranibizumab in DME proceeded along three distinct clinical lines: Diabetic Retinopathy Clinical Research Network (DRCR.net), READ/RISE/RIDE, and RESOLVE/RESTORE. The US National Eye Institute, which funded DRCR.net group, was the first to demonstrate the superiority of ranibizumab with immediate or deferred laser over standard of care (laser photocoagulation). The READ trials followed by RISE and RIDE were North American registration trials that led to the approval of monthly 0.3 mg ranibizumab by the US FDA. The RESOLVE/RESTORE trials were performed in the eastern hemisphere and led to the approval of 0.5 mg ranibizumab by the European Medicines Agency (EMA).

Two small pilot studies produced early evidence that intravitreal ranibizumab reduces macular thickening and improves VA in patients with DME. In the READ-1 study, ten eyes were injected at baseline and 1, 2, 4, and 6 months [133]. At the 7-month examination, the mean excess in foveal thickness had decreased by 85% from 503 µm to 257 µm, macular volume had decreased from 9.22 mm² to 7.47 mm² (a 77% reduction in excess volume), and the mean VA improved by +12.3 letters. Several patients experienced a rapid reduction in macular edema (median of -88μ m, mean of -130μ m) by day 7, and a strong correlation between macular thinning and visual improvement was noted ($r^2 = 0.78$). No safety signals emerged though patients experienced slight increases in blood pressure.

In a second pilot study, ten patients were treated with 0.3 mg or 0.5 mg of ranibizumab at baseline and at months 1 and 2 [24]. The primary endpoints were the incidence and severity of adverse events at 3 months, with secondary endpoints being changes in VA and appearance of the retina (based on fundus photography, fluorescein angiography, and OCT). Two patients experienced mild, sterile inflammation, which was insufficient to alter the performance of subsequent trials. Improvements in visual acuity and macular thickness in the 0.3 mg and 0.5 mg groups were +12 and +7.8 letters and -45.3 μ m and -197.8 μ m, respectively. Improvements in VA diminished somewhat between 3 and 6 months but were better sustained in the group receiving 0.3 mg.

4.3.3.2 READ-2 and READ-3 Trials

The phase II READ-2, the first prospective, randomized, double-masked, multicenter ranibizumab DME trial, followed on the heels of the phase I READ-1 trial [130]. One hundred twenty-six patients were randomized to three treatment arms: intravitreal injections of 0.5 mg ranibizumab at baseline and months 1, 3, and 5, focal/grid laser photocoagulation of the macula at baseline and at month 3 if needed, and intravitreal 0.5 mg ranibizumab at baseline followed by laser 1 week later. At the 6-month primary temporal endpoint, mean visual acuity improvements were +7, 0, and +4 letters, and mean changes in central macular thickness (CMT) were $-95 \mu m$, $-82 \mu m$, and $-117 \mu m$. A protocol modification at 6 months allowed patients in the first two groups to receive ranibizumab every 2 months PRN, while patients in the third group became eligible for laser and ranibizumab every 3 months for macular thickness >250 μ m. At 24 months, the mean VA improvements in the three arms were +8, +5, and +7 letters, and the mean CMTs were 340, 286, and 258 μ m. Visual acuity improvements in the laser group approached that of the ranibizumab group, but the thick maculas in the ranibizumab monotherapy group indicated that these patients were undertreated [131].

A second protocol modification at 24 months allowed patients to receive monthly ranibizumab PRN during year 3. Between the 24- and 36-month visits, patients achieved mean gains of +3, -2, and +2 letters, with each group receiving an average of 5, 2, and 3 ranibizumab injections during the third year [39].

During the exploratory dosing trials for nAMD, a quadruple dose of ranibizumab (2.0 mg) was tested but quickly abandoned because of its high viscosity. Years later, the 2.0 mg dose was reintroduced in the HARBOR trial [20] for nAMD and the READ-3 trial [41] for DME. Patients with DME received either the 2.0 mg or the standard 0.5 mg dose monthly for 6 months then PRN through 12 months. The 0.5 mg dose produced greater improvement in vision than the 2.0 mg dose (+10.88 letters vs. +7.39 letters), and patients receiving the 2.0 mg dose experienced more deaths due to myocardial infarction (3% vs. 0%). Based upon the results from the READ-3 and HARBOR (in which the 0.5 mg and 2.0 mg doses performed comparably), further development of the 2.0 mg dose was halted.

4.3.3.3 RISE and RIDE

READ-2 demonstrated that intravitreal ranibizumab, either with or without accompanying macular laser photocoagulation, was superior to laser for the treatment of DME and had an acceptable safety profile. The disappointing VA results from years 2 and 3 suggested that fixed treatment intervals of 2–3 months were probably too long. Lessons from READ-2 were incorporated into the randomized, multicenter, double-masked, phase III RISE and RIDE registration trials that compared the efficacy of 0.3 mg and 0.5 mg ranibizumab against sham/laser for patients with centerinvolving DME [129]. The 3-year trials randomized 759 patients to three treatment arms: monthly 0.3 mg ranibizumab, monthly 0.5 mg ranibizumab, or sham. Patients in each group became eligible for rescue laser photocoagulation at 3 months if CRT was >250 μ m and if the change in CRT following the previous injection was <50 μ m. The primary functional endpoint was the proportion of eyes improving by at least 15 letters. Secondary endpoints included improvement in BCVA, improvement in macular thickness, safety measures, and improvement in VFQ-25 quality of life scores.

At 24 months, significant proportions of patients receiving 0.3 mg ranibizumab, 0.5 mg ranibizumab, and sham injections improved by at least 15 letters in RISE (45%, 39%, 18%) and RIDE (34%, 46%, 12%). Mean visual acuity improvements were particularly impressive in the ranibizumab treated groups in RISE (+13, +12,

+3) and RIDE (+12, +11, +0.5). The corresponding mean improvements in central foveal thickness were -250, -253, and $-133 \ \mu m$ (RISE) and -259, -270, and $-125 \ \mu m$ (RIDE). Patients receiving ranibizumab required fewer lasers (mean of 0.3–0.8) than those randomized to sham injections (means of 1.8 and 1.6).

The median diabetic retinopathy severity scores (DRSS) in the sham/laser group remained at moderately severe NPDR throughout the study, whereas in patients receiving ranibizumab they improved from moderately severe NPDR to mild NPDR. Fewer patients receiving ranibizumab compared to sham/laser experienced a two-step worsening in the DRSS score (1.7% to 2.1% vs. 9.6%), and fewer ranibizumab eyes developed vitreous hemorrhage. Since visual acuity changes were the same for patients receiving the 0.3 and 0.5 mg doses of ranibizumab, and there appeared to be a dose-dependent risk of stroke, the US FDA approved the 0.3 mg ranibizumab dose for the treatment of center-involving DME in 2012. In 2015, the label was expanded to include the treatment of diabetic retinopathy in eyes with DME.

Patients originally randomized to ranibizumab continued receiving monthly injections during year 3, but those randomized to sham were eligible to cross over to monthly PRN 0.5 mg ranibizumab [18]. Patients in the ranibizumab arms had stable BCVA during year 3 whereas those in the sham arms improved to +4 (RISE) and +5 (RIDE) letters from baseline.

After the 36-month visit, 582 patients from RISE and RIDE were followed in the extension study. All patients were eligible to receive 0.5 mg ranibizumab every 4 weeks if DME was identified by the investigator or BCVA worsened by at least five letters compared to month 36. A mean of 4.5 injections/patient (annualized 3.8) was given during a mean follow-up of 14.1 months. Twenty-five percent of patients did not require any injections during the extension. Best corrected visual acuity in all groups remained stable throughout the extension period and mean CFT increased slightly. Few patients developed PDR during the extension, and those originally randomized to sham/laser.

Patients with macular non-perfusion at baseline had lower VA scores than those with good perfusion. But by the end of the trial, patients with non-perfusion had greater improvements in VA and many actually caught up to those with good perfusion. Areas of non-perfusion did not increase in size when exposed to anti-VEGF therapy. These patients also experienced similar two-step improvements in DRSS compared to patients with good macular perfusion.

4.3.3.4 **RESOLVE**

The phase II RESOLVE trial and phase III RESTORE trials were performed in Europe, Asia, and Australia. In the 12-month RESOLVE trial, 152 patients were randomized to receive monthly 0.3 mg ranibizumab, 0.5 mg ranibizumab, or sham injections [115]. After 1 month, the dose of the injected drug could be doubled if the CRT was greater than 300 μ m or greater than 225 μ m if the CRT reduction was less than 50 μ m following the previous injection. At 3 months, patients were eligible for

rescue laser and additional monthly PRN injections or sham. At the 12-month primary temporal endpoint, BCVA improved by +10.3 letters in the pooled ranibizumab groups but declined by -1 letter in the sham group. Vision gains of +2 lines and +3 lines were achieved by 60.8 and 33% of ranibizumab-treated eyes but only 18.4 and 5% of sham-treated eyes. Mean improvements in CST were $-194 \mu m$ for the pooled ranibizumab groups and $-48 \mu m$ for the sham group. Eighty-six percent of all eyes and 68% of those receiving ranibizumab (70–78% of these at the 1-month exam) required the higher drug dose at some point during the trial. The mean number of administered injections was 10.2, and only 4.9% of ranibizumab-treated eyes (compared to 34.7% of sham eyes) required rescue laser.

4.3.3.5 **RESTORE**

RESTORE was the multicenter (75 sites) phase III ranibizumab registration trial in the eastern hemisphere. Three hundred forty-five patients were randomized to receive ranibizumab + sham laser, ranibizumab + laser, or sham injections + laser. Ranibizumab injections were given monthly \times 3 then PRN; laser was performed at baseline then every 3 months PRN [121]. The primary objective of the trial was to demonstrate the superiority of ranibizumab monotherapy or ranibizumab + laser over laser monotherapy as measured by mean improvements in BCVA over 12 months. Secondary objectives included the proportion of patients achieving BCVA of 73 letters (20/40), the time course of the mean change in BCVA and CRT, patient-reported outcomes relative to laser photocoagulation, and safety measures.

At the 12-month primary endpoint, patients in the ranibizumab monotherapy, ranibizumab + laser, and sham/laser groups had improvements in mean BCVA (+6.1, +5.9, and +0.8 letters), BCVA score >73 letters (53, 44.9, and 23.6%), and CRT (-118.7 μ m, -128.3 μ m, and -61.3 μ m). Health-related quality of life (measured by the NEI VFQ-25 questionnaire) improved more in the ranibizumab monotherapy and ranibizumab + laser groups compared to sham/laser (*P* < 0.05 for each). Subgroup analyses showed that patients with baseline BCVA of >73 ETDRS letters or CRT <400 μ m saw as well after laser as after ranibizumab. Overall, patients received a mean of seven ranibizumab/sham injections. No cases of endophthalmitis occurred, and ranibizumab therapy was not associated with an increased incidence of cardiovascular or cerebrovascular events.

After the 12-month primary endpoint, 240 patients rolled into the 24-month extension trial. All patients were eligible to receive 0.5 mg ranibizumab according to BCVA and disease progression criteria and at the investigators' discretion. Additional laser photocoagulation was allowed according to ETDRS guidelines. At the preplanned 24-month interim analysis, patients originally receiving ranibizumab monotherapy and ranibizumab + laser maintained improvements in mean best corrected visual acuity (+7.9 letters, +6.7 letters), CRT ($-140.6 \mu m$, $-133.0 \mu m$), and NEI VFQ-25 composite scores (5.6, 5.8) [103]. Between months 12 and 24, significant improvements in these measures were seen in patients originally treated with sham/laser (+5.4 letters, $-126 \mu m$, 4.3). Similar numbers of injections were

performed in each group (3.9, 3.5, and 4.1). No cases of endophthalmitis were reported and the incidences of non-ocular SAEs were low.

Two hundred eight (86.7%) patients completed the 24-month extension study. As-needed ranibizumab treatment led to an overall maintenance of VA and CRT between months 12 and 36 [154]. Best corrected visual acuities at the completion of the study according to initial treatments were +8.0 letters (ranibizumab only), +6.7 letters (ranibizumab + previous laser), and +6.0 letters (laser). Patients in the three treatment arms required between 4.0 and 6.8 (mean for each group) injections over the final 2 years.

4.3.3.6 DRCR.net Protocol I

The DRCR.net Protocol I trial provided the first level I evidence that supported the use of ranibizumab in eyes with DME. Protocol I was a 5-year, multicenter study that randomized 854 eyes with center-involving DME to receive 0.5 mg ranibizumab with prompt macular laser photocoagulation, 0.5 mg ranibizumab with deferred laser (for at least 6 months), intravitreal triamcinolone with prompt laser, or sham injections with prompt laser [31]. During the first year, patients received ranibizumab injections according to the 4:2:7 rule – 4 monthly injections, followed by two injections if fluid persisted, followed by seven visits during which the drug could be administered at the investigator's discretion if there was insufficient improvement. Laser photocoagulation and intravitreal triamcinolone (4 mg) could be repeated quarterly as needed. Patients randomized to the deferred laser group were not obligated to receive laser if the macula was dry at and beyond 6 months.

At 1 year, the median improvements in VA in the ranibizumab + prompt laser, ranibizumab + deferred laser, triamcinolone + laser, and sham + laser groups were +9, +9, +4, and +3 letters, respectively, with most of the gains in BCVA occurring by the 8-week visit. During the first 3 months, patients receiving triamcinolone experienced improvements in VA similar to those receiving ranibizumab, but VA then worsened through 12 months because of the development of steroid-induced cataracts. In the triamcinolone group that was pseudophakic at baseline, 1-year improvements in VA were comparable to those in the ranibizumab groups. There were no significant differences in visual outcomes when the following subgroup analyses were performed: prior treatment for DME, baseline VA, baseline CST, and baseline severity of DR. Improvements in macular thickness were comparable in the groups receiving ranibizumab and triamcinolone, all of which were superior to the group receiving sham/laser. Eyes treated with ranibizumab were less likely to experience progression of diabetic retinopathy. Three patients receiving ranibizumab (0.8%) developed endophthalmitis, and cataracts and elevated intraocular pressure were most commonly seen in patients receiving triamcinolone.

During year 2, the visit interval could be extended to 8 weeks if treatment was deferred at three consecutive visits and to 16 weeks if treatment was not given at the 8-week visit. Patients in the triamcinolone + laser and laser/sham groups were eligible to receive ranibizumab as early as week 74 for persistent edema without

improved vision. The 2-year outcomes were similar to those at 1 year in which 50% of ranibizumab-treated eyes improved by at least 10 letters and 33% improved by at least 15 letters [32]. Compared to the sham/laser group, the mean changes in BCVA in patients receiving ranibizumab + prompt laser, ranibizumab + deferred laser, and triamcinolone + prompt laser were +3.7, +5.8, and -1.5 letters. Forty-three eyes in the sham/prompt laser group were switched to alternative therapy (ranibizumab) within the first 2 years because of "failure," whereas none of the patients randomized to ranibizumab required switching. Elevated intraocular pressures were more common in the triamcinolone/laser group.

At the 3-year visit, the median numbers of injections in the ranibizumab + prompt laser and ranibizumab + deferred laser groups were 12 and 15, respectively [33], and the median numbers of lasers were 3 and 0, respectively. By the 3-year visit, 46% of patients in the ranibizumab + deferred laser group had been treated with laser. Patients treated with ranibizumab + deferred laser improved by +2.9 letters more than those receiving ranibizumab + prompt laser (P = 0.02). In the two ranibizumab groups, the percentage of eyes with CST <250 µm was 36% in both groups.

At the 5-year visit, the mean BCVA improvements from baseline were +7.2 letters in the ranibizumab + prompt laser group and +9.8 letters in the ranibizumab + deferred laser group (P = 0.09) [49]. In the two groups, there was vision loss of >10 letters in 9 and 8%, improvement of vision by >10 letters in 46 and 58%, and improvement in vision of >15 letters in 27 and 38%. From baseline through 5 years, 56% of patients in the deferred group did not require laser. The mean numbers of ranibizumab injections during the trial were 13 and 17, 54 and 45% did not receive ranibizumab during year four, and 62 and 52% did not receive injections during year 5.

4.3.3.7 Retain

The RETAIN trial explored the viability of a treat-and-extend strategy for DME [141]. Three hundred seventy-two patients were randomized to receive treat-and-extend 0.5 ranibizumab plus laser (G1), treat-and-extend 0.5 mg ranibizumab (G2), or monthly PRN 0.5 mg ranibizumab (G3). All patients received monthly injections until vision stabilized, after which patients in G1 and G2 could be extended at 1 month intervals up to a maximum of 3 months. At 24 months, median BCVA changes in the G1, G2, and G3 groups were +8, +7, and +8 letters. Compared to PRN treatment, patients receiving treat-and-extend required 40% fewer clinic visits, and 70% were extended to a treatment interval of at least 2 months.

4.3.3.8 Ranibizumab for PDR

Ranibizumab not only restores the blood-retinal barrier, thereby allowing the resorption of macular edema, but it also possesses potent anti-angiogenesis activity. It has been used to treat PDR, both as monotherapy and in conjunction with panretinal photocoagulation. Ranibizumab regresses retinal neovascularization in patients with PDR but since its effects appear to be transient, PRP may still be necessary in some cases to permanently close new vessels.

In a prospective study, ranibizumab combined with PRP was compared to PRP alone for the treatment of high-risk PDR in 40 patients that had not received previous laser photocoagulation [58]. Split-session PRP was administered to all patients at weeks 0 and 2, and the "PRP Plus" group received 0.5 mg ranibizumab at the completion of the first laser session and again at weeks 16 and 32 if needed. Patients were evaluated with ETDRS visual acuity measurements, fluorescein angiography (FA), and OCT. Eyes in both groups experienced significant reductions in FA leakage at all visits though 48 weeks, but this was significantly greater in the PRP Plus group. Best corrected visual acuity worsened in the laser monotherapy group at 16, 32, and 48 weeks but was unchanged in the PRP Plus group. The CMT increased at all visits in the laser group but decreased at week 16 and stabilized through week 48 in the PRP Plus group. The authors concluded that ranibizumab after PRP results in superior reduction in PDR-related FA leakage and improved VA and CMT compared to laser alone.

The DRCR.net performed a randomized clinical trial (Protocol S) with 305 patients at 55 sites to compare panretinal photocoagulation with intravitreal 0.5 mg ranibizumab for PDR [177]. PRP was performed at baseline and ranibizumab was given at baseline followed by q4wk PRN for the PDR, and eyes in both groups with DME were eligible to receive ranibizumab. The primary outcome was change in BCVA and the secondary outcomes included VA area under the curve, peripheral visual field loss (as measured on Humphrey automated visual field testing), incidence of vitrectomy, development of DME, and persistent or new neovascularization. Improvements in BCVA for the ranibizumab and PRP groups were +2.2 and +0.2 letters, respectively (95% CI, -0.5 to +5.0). The group receiving ranibizumab experienced less peripheral visual field sensitivity loss (-23 dB vs. -422 dB; 95% CI, 213-531 dB; P < 0.001), fewer vitrectomies (4% vs. 15%; 95% CI 4%-15%; P < 0.001), and less development of DME (9% vs. 28%). A median of seven ranibizumab injections was administered through year 1 and ten injections through year 2. Forty-five percent of eyes required additional PRP after the initial treatment, and 53% of eyes in the PRP group required ranibizumab for DME. Only one eye developed endophthalmitis after a ranibizumab injection. The authors concluded that ranibizumab may be a reasonable alternative to PRP over the course of 2 years.

4.3.4 Aflibercept

Aflibercept, previously referred to as the VEGF Trap-eye, is a 115 kD, recombinant, high-affinity, soluble, decoy-receptor molecule. Aflibercept is composed of the second extracellular binding domain from VEGFR1, and the third binding domain from VEGFR2 fused to the constant region (Fc) of a human IgG1 molecule [82]. As a result, aflibercept binds all isoforms of VEGF-A, VEGF-B, and PIGF. Unlike ranibizumab and bevacizumab, aflibercept is composed entirely of human amino acid sequences. It possesses a high binding affinity for VEGF₁₆₅ ($K_D = 0.45$ pM) and PIGF2 ($K_D = 45$ pM) [136] because of its "two-fisted" grasp of the VEGF dimer that is 100-fold stronger than bevacizumab and ranibizumab and exceeds even that of the native VEGFR1 and VEGFR2 receptors. In luciferase-binding assays, aflibercept is 45–92 times as potent as bevacizumab and ranibizumab at preventing VEGF-A isoforms from binding to VEGFR1 and 33–51 times as potent at preventing binding to VEGFR2 [136]. Aflibercept blocked the VEGF-A-induced activation of VEGFR1 and VEGFR2 in human endothelial cells with greater potency than bevacizumab and also blocked VEGF-A- and PIGF-induced migration of endothelial cells [136].

Aflibercept has an intravitreal half-life of 4.7 days in rabbits [66], considerably longer than either ranibizumab (2.88 days) [11] or bevacizumab (4.32 days) [12], but its half-life in human eyes has not been determined. Mathematical modeling based on binding affinities and estimated intravitreal half-lives predicted that aflibercept would have a longer duration of action than bevacizumab and ranibizumab [162]. After intravitreal injection, aflibercept passes unaltered into the systemic circulation where its half-life is approximately 6 days. Aflibercept binds plasma VEGF and lowers serum concentrations to below 10 pg/ml (the lower detectable limit of some assays) for at least 7 days [10]. Clearance of aflibercept from the serum via renal clearance of bound complexes and Fc-mediated pinocytotic pathways resembles that of full-length antibodies [37].

The important affibercept diabetic retinopathy trials and their key findings are listed in Table 4.6.

Important diabetic macular edema trials with aflibercept		
Trial and		
phase	Cohorts	Key findings
DA VINCI	Treatment arms:	At 1 year
Phase II	IAI 0.5 mg q4wk	1. Improvements in mean BCVA were +11.0, +13.1,
221 patients	IAI 2.0 mg q4wk	+9.7, +12.0, and -1.3 letters
	IAI 2.0 mg q8wk	2. Improvements in mean CST were $-165.4 \mu m$,
	IAI 2.0 mg PRN	$-227.4 \ \mu m$, $-187.8 \ \mu m$, $-180.3 \ \mu m$, and
	Laser	-58.4 μm
		3. Fewer laser in AFL groups (0.5–0.8) compared to
		laser group (2.5)
VIVID and	Treatment arms:	At 1 year
VISTA	IAI 2 mg q4wk	1. Improvements in mean BCVA were +10.5 and
Phase III	IAI 2 mg q8wk	+12.5, +10.7 and 10.7, and +0.2 and +1.2 letters
872 patients	Laser/sham	At 2 years
		1. Improvements in mean BCVA were +11.5 and
		+11.4, +9.4 and +11.1, and +0.9 and +0.7 letters

Table 4.6 Important trials with aflibercept and details their key findings

IAI aflibercept

4.3.4.1 Pilot Study

In a small pilot study, five patients with DME received single intravitreal injections of 4 mg aflibercept. At 4 weeks, the mean excess macular thickness decreased from 108 μ m to 59 μ m, and the BCVA improved by an average of +9 letters. At 6 weeks, four of the five eyes still had improved excess thickness (median improved from 108 μ m to 74 μ m; two were normal), and the mean BCVA had improved by +3 letters [40].

4.3.4.2 DA VINCI

The prospective, multicenter, phase II DA VINCI trial randomized 221 patients with center-involving DME to five treatment arms: intravitreal 0.5 mg affibercept every 4 weeks (0.5q4), 2 mg every 4 weeks (2q4), 2 mg every 8 weeks after 3 monthly loading injections (2q8), 2 mg PRN after 3 monthly loading doses (2PRN), and quarterly laser PRN/sham [38]. Patients receiving affibercept were not offered rescue laser until 6 months. At 1 year, the mean visual acuity improvements were +11.0, +13.1, +9.7, +12.0, and -1.3 letters for each arm; the proportions improving by at least 15 letters were 40.9, 45.5, 23.8, 42.2, and 11.4%; and the mean improvements in CST were -165.4, -227.4, -187.8, -180.3, and -58.4 μ m. Patients in the 2PRN and 2q8 groups received an average of 7.4 and 7.2 injections. The laser/sham group received a mean of 2.5 laser procedures compared to 0.5–0.8 in the aflibercept patients but only 12% of laser patients, whereas worsening in DRSS was seen in 0–14% of aflibercept patients compared to 24% of laser-treated patients.

4.3.4.3 VIVID and VISTA

The Study of Intravitreal Administration of VEGF Trap-Eye in Patients with DME (VISTA; NCT01363440) and the VEGF Trap-Eye in Vision Impairment due to DME (VIVID; NCT01331681) [98] were similarly designed, double-blind, randomized, phase III trials that enrolled 872 patients (eyes) (VISTA, 466; VIVID, 406) with center-involving DME. VISTA-DME was performed in the United States and Canada whereas VIVID-DME was run in Australia, Europe, and Japan. Eligible patients had type 1 or 2 diabetes mellitus with BCVA from 24 to 73 letters (20/40 to 20/320) and central macular thickening on OCT. Eyes were randomized 1:1:1 to receive intravitreal aflibercept injections (IAI) 2 mg q4wk, IAI 2 mg q8wk after 5 monthly loading doses, or laser photocoagulation/sham injection. Patients were eligible for laser retreatment every 12 weeks if ETDRS-defined edema was present. All study eyes were eligible for rescue treatment beginning at 24 weeks if they lost >10 letters of BCVA on two consecutive visits or >15 letters at any visit from the previous best measurement, and the BCVA was worse than baseline. For lasertreated eyes, additional treatment consisted of 5 monthly doses of 2 mg IAI followed by injections every 8 weeks, and for IAI treated eyes, active laser was performed.

The primary temporal endpoint was at 52 weeks, but patients receiving IAI will continue to receive therapy through 148 weeks, and patients randomized to laser/ sham will be eligible to cross over to IAI during year 3. The primary efficacy endpoint was the mean improvement in ETDRS BCVA at 52 weeks. Secondary efficacy endpoints included the proportions of patients gaining ≥ 15 letters, the proportions of patients gaining ≥ 10 letters, the proportions of eyes experiencing a two-step improvement in the ETDRS Diabetic Retinopathy Severity Scale (DRSS) score, the mean changes in central retinal thickness (CRT) as measured by OCT, the change from baseline in the National Eye Institute Visual Function Questionnaire-25 (NEI VFQ-25) near activities subscale score, and the change from baseline in the NEI VFQ-25 distance activities subscale score. VISTA enrolled a greater proportion of Black patients and VIVID enrolled a greater proportion of Asian patients. More eyes in VISTA, compared to VIVID, had previously received anti-VEGF injections (42.9% vs. 8.9%).

Mean BCVA changes from baseline to 52 weeks for the groups receiving IAI 2 mg q4wk, IAI 2 mg q8wk, and laser/sham were +12.5, +10.7, and +0.2 letters (P < 0.0001) in VISTA and +10.5, +10.7, and +1.2 letters (P < 0.0001) in VIVID. When eyes receiving additional rescue therapy were included in the analvsis, those in the IAI groups changed by +10.7 to +12.4 letters from baseline whereas those in the laser groups changed by +4.2 and +3.5 letters. Visual acuity gains were significantly greater in the IAI groups in both patients who had and had not received prior anti-VEGF therapy. The corresponding proportions improving by ≥ 10 letters were 64.9, 58.3, and 19.5% (P < 0.0001) in VISTA and 54.4, 53.3, and 25.8% (P < 0.0001) in VIVID. The proportions improving by >15 letters were 41.6, 31.1, and 7.8% (P < 0.0001) in VISTA and 32.4, 33.3, and 9.2% (P < 0.0001) in VIVID. The proportions that lost >15 letters were 0.6, 0.7, and 9.1% (P < 0.0001) in VISTA and 0.7, 0, and 10.6% (P < 0.0001) in VIVID. Compared to laser, most patients receiving IAI did not lose any letters from baseline: 94.2, 92.7, and 57.1% in VISTA and 94.1, 91.9, and 62.9% (P < 0.0001) in VIVID. Significantly more patients treated with IAI q4wk and q8wk than laser experienced a two-step improvement in DRSS in both VISTA (33.8% and 29.1% vs.14.3%) and VIVID (33.3% and 27.7% vs. 7.5%). Mean changes in CRT were -185.9 µm, -183.1 µm, and -73.3 µm in VISTA and -195.0 µm, -192.4 µm, and -66.2 µm in VIVID. The mean +/- SD in NEI VFQ-25 scores for the IAI q4k groups were significantly different from the laser groups only for the near activities subscale scores in VISTA (9.0 +/- 20.6 vs. 5.4 +/- 20.4; P = 0.0168). For patients treated with laser/sham, the mean numbers of lasers were 2.7 and 2.1 in VISTA and VIVID, respectively. More patients in the laser group than the IAI groups received additional (rescue) therapy (VISTA, 31.2% vs. 0.7% and 2.6%; VIVID, 24.1% vs. 4.4% and 8.1%).

Incidences of ocular and non-ocular adverse events and serious adverse events including Antiplatelet Trialists' Collaborative-defined vascular events and deaths were similar among all groups. The incidences of ocular and non-ocular adverse events were similar across all treatment groups. Serious non-ocular adverse events were uncommon (hypertension: 9.7%; cerebrovascular accidents: 1.1%; and myo-cardial infarction: 1.1%). Incidences of intraocular inflammation were 0.2% (4/1832 injections), 0.1% (1/1284 injections), and 0.5% (1/212 injections) in VISTA and 0.2% (4/1566 injections), 0.4% (5/1186 injections), and 0.7% (1/135 injections) in VIVID. Both laser patients developed inflammation before receiving rescue aflibercept. There were no cases of endophthalmitis. The incidences of congestive heart failure and anemia were higher in the aflibercept groups, and the incidences of myocardial infarction and osteoarthritis were higher in the laser groups. The total numbers of vascular deaths were 2, 2, and 2, and the total numbers of deaths were 2, 4, and 2 due to additional deaths from B-cell lymphoma and lung carcinoma in the 2 mg q8wk group.

The mean improvements in BCVA from baseline to week 100 in the 2q4, 2q8, and laser arms in VISTA (+11.5, +11.1, and +0.9 letters) and VIVID (+11.4, +9.4, and +0.7 letters) resembled those at the 52-week primary endpoint [19]. The proportions of eyes that gained >15 letters were 38.3, 33.1, and 13.0% (P < 0.001) in VISTA and 38.2, 31.1, and 12.1% (P < 0.001) in VIVID. Significantly more eyes receiving aflibercept than laser achieved >two-step improvements in DRSS in both VISTA (37.0, 37.1, and 15.6%) and VIVID (29.3, 32.6, and 8.2%).

Eyes in the aflibercept arms of both VISTA and VIVID had sustained improvements in VA and CRT through the 148-week visit. After week 100, eyes in the laser groups crossed over to receive monthly aflibercept but experienced mean VA improvements of only 1 letter. Rescue aflibercept had been allowed for laser treated eyes that initially lost at least 10 letters. In the VIVID trial, 82% of eyes required either rescue or as-needed aflibercept as did 87% of eyes in VISTA. Eyes in the laser groups that required rescue therapy achieved better final visual acuities than those that did not require rescue therapy [Justus Ehlers, Macula Society, Miami Beach, FL, February 26, 2016]. Only 12% of aflibercept treated eyes required laser at some point during the trials. Eyes with limited responses at 12 weeks (<10% improvement in CRT) ultimately went on to mean visual acuity improvements of +7.8 letters [Rishi Singh, Macula Society, Miami Beach, FL, February 25, 2016].

At week 100, eyes in the laser groups averaged $-84 \ \mu\text{m}$ of macular thinning and after crossover to aflibercept this increased to $-110 \ \mu\text{m}$ by week 148. In contrast, the IAI 2q4wk and 2q8wk groups achieved significantly better thinning of the macula ($-200 \ \text{and} -190 \ \mu\text{m}$).

In 2014, aflibercept received US FDA approval for the treatment of centerinvolving DME and soon thereafter for the treatment of diabetic retinopathy in patients with DME. It has been approved by the European Medicines Agency and with the recent approval in Egypt is now available for the treatment of DME in 31 countries. The ability of intravitreal aflibercept to prevent DR progression in eyes without DME and PDR is being evaluated in the PANORAMA trial. Patients with moderately severe BDR are randomized to receive sham injections or aflibercept every 8 or 16 weeks. This trial resembles the DRCR.net Protocol W trial in which ranibizumab is being evaluated.

4.4 Other Studies

In a retrospective analysis of eyes receiving anti-VEGF therapy, Wycoff noted that two-line improvements in DRSS occurred most frequently in eyes with moderately severe DR, as opposed to those with milder NPDR or with PDR [Charles Wycoff, Retina Society, Paris, FR, October 8, 2015]. The reason for this finding is not known, but it may indicate that the reversibility of DR is related to disease severity and may be influenced by the contribution of chemokines or cytokines other than VEGF.

The best results from anti-VEGF therapy appear to occur when monthly injections are continued for at least 5 months. Unfortunately, a small proportion of eyes (15%) will fail to respond well to monthly therapy and post hoc analyses suggest that these can be reliably identified by poor resolution in macular thickness (<10%) or improvement in BCVA (< 5 letters) at 3 months [David Boyer, Retina Society, Paris, FR. October 8, 2015]. A small retrospective series found that the 1-month response to bevacizumab injections could be reliably determined by the 1 h response (542.3 ± 127.7 µm at baseline to 516.9 ± 123.4 µm 1 h after injection, P < 0.001, and to 345.5 ± 110.0 µm at 1 month after injection, P < 0.001; r = 0.515) [112].

None of the registration trials allowed switching of anti-VEGF drugs in eyes that respond poorly to initial therapy and reliable data regarding this strategy is limited. A retrospective study of 21 eyes evaluated the efficacy of switching from bevacizumab or ranibizumab to aflibercept. Patients had received a median of 6 injections before the switch and 3 aflibercept after the switch. The mean CFT decreased from $453.52 \pm 143.39 \,\mu\text{m}$ immediately prior to the switch to $362.57 \pm 92.82 \,\mu\text{m}$ (Wilcoxon signed-rank test, P < 0.001) after the first injection and to $324.17 \pm 98.76 \,\text{mm}$ (P < 0.001) at the study's conclusion. The mean visual acuity was $0.42 \pm 0.23 \,\text{logMAR}$ just prior to the switch, $0.39 \pm 0.31 \,\text{logMAR}$ after one aflibercept injection, and $0.37 \pm 0.22 \,\text{logMAR}$ at the end of follow-up. The final visual acuity was significantly better than visual acuity before the switch (P = 0.04). The authors concluded that eyes with DME that are unresponsive to multiple ranibizumab or bevacizumab injections can demonstrate anatomic and visual improvement after switching to aflibercept [107].

A post hoc analysis of the DRCR.net trials showed that eyes with poor responses at 3 or 6 months eventually achieved 5–7 letters of improvement in BCVA simply by continuing intensive intravitreal therapy [Daniel Martin, Macula Society, Miami Beach, FL, February 26, 2016]. The authors stated that the perceived improvement with switching anti-VEGF drugs may actually be the result of continued, intensive anti-VEGF therapy.

4.5 Comparison Trials

Prior to the publication of DRCR.net Protocol I, a Bayesian network meta-analysis compared the efficacies of ranibizumab and aflibercept for the treatment of DME [145]. For 10-letter gains, the results slightly favored ranibizumab (RR 1.59, 95% credible interval 0.61–5.37).

A 48-week, randomized, prospective trial of 45 patients compared intravitreal bevacizumab with ranibizumab [127]. Patients in both groups experienced significant improvements in BCVA at all visits (P < 0.05), but those receiving ranibizumab had greater improvements at weeks 8 (P = 0.032) and 32 (P = 0.042). Mean CST significantly improved in both arms at all visits with no difference noted between the drugs. Patients receiving bevacizumab required more injections than those receiving ranibizumab (means 9.84 vs. 7.67).

The only trial to directly compare aflibercept, bevacizumab, and ranibizumab treatment of DME was the prospective, randomized, multicenter DRCR.net Protocol T trial [34]. Six hundred sixty patients at 89 sites received 1.25 mg bevacizumab, 0.3 mg ranibizumab, or 2 mg aflibercept. Entry criteria included BCVA from 20/32 to 20/320 with center-involving DME by clinical exam and OCT. Patients were treated every 4 weeks unless the BCVA reached 20/20 or better with a CST below the eligibility threshold, or over the past two injections there was no BCVA change of 5 letters or more or a 10 % change in CST. Beginning at week 24, injections were withheld if the BCVA change was < 5 letters and the CRT change was < 10 % over two injections irrespective of BCVA. Patients were eligible for laser photocoagulation at 24 weeks for persistent edema.

At 52 weeks, mean numbers of injections were 9 (affibercept), 10 (bevacizumab), and 10 (ranibizumab) (P = 0.045). Laser photocoagulation was performed in 37, 56, and 46% of eyes (P < 0.001). Mean changes in BCVA were +13.3 letters (affibercept), +9.7 letters (bevacizumab), and +11.2 letters (ranibizumab) (P < 0.001, affibercept vs. bevacizumab; P = 0.03, affibercept vs. ranibizumab). A preplanned subgroup analysis was critical to uncovering significant differences in efficacy among the drugs. For eyes with baseline BCVA of 20/32 to 20/40, mean changes were +8.0 (affibercept), +7.5 (bevacizumab), and +8.3 letters (ranibizumab). When baseline VA was $\leq 20/50$, mean changes in BCVA were +18.9 (affibercept), +11.8 (bevacizumab), and +14.2 letters (ranibizumab). The average changes in CST were -169 µm, -101 µm, and -147 µm. Only two eyes developed endophthalmitis, and there were no significant differences in the rates of serious adverse events (P = 0.40), hospitalization (P = 0.51), death (P = 0.72), or major cardiovascular events.

Visual acuity gains were sustained in all groups at 2 years, but the differences in BCVA gains between the drugs narrowed. Aflibercept produced significantly better

vision gains than bevacizumab but was not statistically better than ranibizumab [Wells, Mac Society].

4.6 Conclusions and Outstanding Questions

The phase III registration trials showed conclusively that treatment with bevacizumab, ranibizumab, or aflibercept is superior to sham/laser for the treatment of DME. Treated patients are also more likely to experience significant improvements in DRSS.

Important questions regarding treatment of DR persist. It is not known how effective anti-VEGF therapy will be on DRSS progression and regression if therapy is administered less frequently. For example, patients in BOLT and Protocol I were treated less frequently than in RISE/RIDE and VIVID/VISTA, and their improvement in DRSS were also less.

The recent DRCR.net Protocol S provided the first evidence that anti-VEGF therapy may be as good as or better than panretinal photocoagulation for PDR. Both longer duration and confirmatory studies with other anti-VEGFs need to be performed.

The best treatment strategy for poor responders has not been determined. Many options are now available to treating physicians, but randomized controlled trials are difficult to structure and have not yet been performed. Management strategies for these challenging eyes will be discussed in detail in Chap. 6.

References

- Ablonczy Z, Crosson CE. VEGF modulation of retinal pigment epithelium resistance. Exp Eye Res. 2007;85:762–71.
- Adamis AP, Miller JW, Bernal MT, et al. Vascular endothelial growth factor in ocular fluid of patients with diabetic retinopathy and other retinal disorders. Am J Ophthalmol. 1994;118:445–50.
- 3. Adamis AP, Altaweel M, Bressler NM, Cunningham ET, Davis MD, Goldbaum M, et al. Changes in retinal neovascularization after pegaptanib (macugen) therapy in diabetic individuals. Ophthalmology. 2006;113(1):23–8.
- Ahmadieh H, Shoeibi N, Entezari M, Monshizadeh R. Intravitreal bevacizumab for prevention of early postvitrectomy hemorrhage in diabetic patients a randomized clinical trial. Ophthalmology. 2009;116:1943–8.
- Aiello LP, Avery RL, Arrigg PG, et al. Vascular endothelial growth factor in ocular fluid of patients with diabetic retinopathy and other retinal disorders. N Engl J Med. 1994;331:1480–7.
- Alon T, Hemo I, Itin A, Pe'er J, Stone J, Keshet E. Vascular endothelial growth factor acts as a survival factor for newly formed retinal vessels and has implications for retinopathy of prematurity. Nat Med. 1995;1:1024–8.
- Arevalo JF, Maia M, Flynn Jr HW, Saravia M, Avery RL, Wu L, et al. Tractional retinal detachment following intravitreal bevacizumab (Avastin) in patients with severe proliferative diabetic retinopathy. Br J Ophthalmol. 2008;92:213–6.

- Arevalo JF, Sanchez JG, Wu L, et al. Primary intravitreal bevacizumab for diffuse diabetic macular edema. The Pan-American Collaborative Retina Study Group at 24 months. Ophthalmology. 2009;116:1488–97.
- 9. Avery RL. Regression of retinal and iris neovascularization after intravitreal bevacizumab (Avastin) treatment. Retina. 2006;26:352–4.
- Avery RL, Castellarin AA, Steinle NC, Dhoot DS, Pieramici DJ, See R, Couvillion S, Nasir MA, Rabena MD, Le K, Maia M, Visich JE. Systemic pharmacokinetics following intravitreal injections of ranibizumab, bevacizumab or aflibercept in patients with neovascular AMD. Br J Ophthalmol. 2014;98(12):1636–41.
- 11. Bakri SJ, Snyder MR, Reid JM, Pulido JS, Ezzat MK, Singh RJ. Pharmacokinetics of intravitreal ranibizumab (Lucentis). Ophthalmology. 2007;114:2179–82.
- Bakri SJ, Snyder MR, Reid JM, Pulido JS, Singh RJ. Pharmacokinetics of intravitreal bevacizumab (Avastin). Ophthalmology. 2007;114:855–9.
- Bansal AG, Narayanan R, Majji AB, Thomas R. Neovascular changes after pegaptanib in diabetics. Ophthalmology. 2007;114(3):615–6.
- 14. Bates DO, Curry FE. Vascular endothelial growth factor increases microvascular permeability via a Ca(2+)-dependent pathway. Am J Physiol. 1997;273:H687–94.
- Bell C, Lynam E, Landfair DJ, et al. Oligonucleotide NX1838 inhibits VEGF₁₆₅-mediated cellular responses in vitro. In Vitro Cell Dev Biol Anim. 1999;35:533–42.
- Benjamin LE, Golijanin D, Itin A, Pode D, Keshet E. Selective ablation of immature blood vessels in established human tumors follows vascular endothelial growth factor withdrawal. J Clin Invest. 1999;103:159–65.
- Brown DM, Kaiser PK, Michels M, et al. Ranibizumab versus verteporfin for neovascular agerelated macular degeneration. N Engl J Med. 2006;334:1432–44.
- Brown DM, Nguyen QD, Marcus DM, et al. Long-term outcomes of ranibizumab therapy for diabetic macular edema: the 36-month results from two phase III trials. Ophthalmology. 2013;120:2013–22.
- Brown DM, Schmidt-Erfurth U, Do DV, Holz FG, Boyer DS, Midena E, Heier JS, Terasaki H, Kaiser PK, Marcus DM, Nguyen QD, Jaffe GJ, Slakter JS, Simader C, Soo Y, Schmelter T, Yancopoulos GD, Stahl N, Vitti R, Berliner AJ, Zeitz O, Metzig C, Korobelnik JF. Intravitreal aflibercept for diabetic macular edema: 100-week results from the VISTA and VIVID studies. Ophthalmology. 2015;122(10):2044–52.
- Busbee BG, Ho AC, Brown DM, Heier JS, Suñer IJ, Li Z, Rubio RG, Lai P, HARBOR Study Group. Twelve-month efficacy and safety of 0.5 mg or 2 mg ranibizumab in patients with subfoveal neovascular age-related macular degeneration. Ophthalmology. 2013;120(5):1046–56.
- Byrne AM, Bouchier-Hayes DJ, Harmey JH. Angiogenic and cell survival functions of vascular endothelial growth factor (VEGF). J Cell Mol Med. 2005;9:777–94.
- Chen E, Park CH. Use of intravitreal bevacizumab as a preoperative adjunct for tractional retinal detachment repair in severe proliferative diabetic retinopathy. Retina. 2006;26:699–700.
- Chen E, Hsu J, Park CH. Acute visual acuity loss following intravitreal bevacizumab for diabetic macular edema. Ophthalmic Surg Lasers Imaging. 2009;40:68–70.
- Chun DW, Heier JS, Topping TM, Duker JS, Bankert JM. A pilot study of multiple intravitreal injections of ranibizumab in patients with center-involving clinically significant diabetic macular edema. Ophthalmology. 2006;113:1706–12.
- Connolly DT, Heuvelman DM, Nelson R, et al. Tumor vascular permeability factor stimulates endothelial cell growth and angiogenesis. J Clin Invest. 1989;84:1470–8.
- Costa R, Carneiro A, Rocha A, et al. Bevacizumab and ranibizumab on microvascular endothelial cells: a comparative study. J Cell Biol. 2009;108:1410–7.
- 27. da R Lucena D, Ribeiro JA, Costa RA, Barbosa JC, Scott IU, de Figueiredo-Pontes LL, et al. Intraoperative bleeding during vitrectomy for diabetic tractional retinal detachment with versus without preoperative intravitreal bevacizumab (IBeTra study). Br J Ophthalmol. 2009;93:688–91.
- de Vries C, Escobedo JA, Ueno H, Houck K, Ferrara N, Williams LT. The fms-like tyrosine kinase, a receptor for vascular endothelial growth factor. Science. 1992;255:989–91.

- 29. di Lauro R, De Ruggiero P, di Lauro R, di Lauro MT, Romano MR. Intravitreal bevacizumab for surgical treatment of severe proliferative diabetic retinopathy. Graefes Arch Clin Exp Ophthalmol. 2010;248:785–91.
- Diabetic Retinopathy Clinical Research Network. A phase II randomized clinical trial of intravitreal bevacizumab for diabetic macular edema. Ophthalmology. 2007;114:1860–7.
- Diabetic Retinopathy Clinical Research Network. Randomized trial evaluating ranibizumab plus prompt or deferred laser or triamcinolone plus prompt laser for diabetic macular edema. Ophthalmology. 2010;117:1064–77.
- 32. Diabetic Retinopathy Clinical Research Network Writing Committee, Elman MJ, Bressler NM, Qin H, Beck RW, Ferris III FL, Friedman SM, Glassman AR, Scott IU, Stockdale CR, Sun JK. Expanded 2-Year Follow-up of Ranibizumab Plus Prompt or Deferred Laser or Triamcinolone Plus Prompt Laser for Diabetic Macular Edema. Ophthalmology. 2011;118:609–14.
- 33. Diabetic Retinopathy Clinical Research Network Writing Committee, Elman MJ, Qin H, Aiello LP, Beck RW, Bressler NM, Ferris III FL, Glassman AR, Maturi RK, Melia M. Intravitreal ranibizumab for diabetic macular edema with prompt versus deferred laser treatment. Three-year randomized trial results. Ophthalmology. 2012;119(11):2312–8.
- 34. Diabetic Retinopathy Clinical Research Network, Wells JA, Glassman AR, Ayala AR, Jampol LM, Aiello LP, Antoszyk AN, Arnold-Bush B, Baker CW, Bressler NM, Browning DJ, Elman MJ, Ferris FL, Friedman SM, Melia M, Pieramici DJ, Sun JK, Beck RW. Aflibercept, bevacizumab, or ranibizumab for diabetic macular edema. N Engl J Med. 2015;372(13): 1193–203.
- Dib E, Maia M, Longo-maugeri L, et al. Subretinal bevacizumab detection after intravitreous injection in rabbits. Invest Ophthalmol Vis Sci. 2008;49:1097–100.
- 36. Ding Y, Huang Y, Song N, et al. NFAT1 mediates placental growth factor-induced myelomonocytic cell recruitment via the induction of TNF-α. J Immunol. 2010;184:2593–601.
- Dixon JA, Oliver SC, Olson JL, Mandava N. VEGF Trap-eye for the treatment of neovascular age-related macular degeneration. Expert Opin Investig Drugs. 2009;18(10):1573–80.
- Do DV, Nguyen QD, Boyer D, et al. One-year outcomes of the DA VINCI Study of VEGF Trap-Eye in eyes with diabetic macular edema. Ophthalmology. 2012;119(8): 1658–65.
- Do DV, Nguyen QD, Khwaja AA, et al. Ranibizumab for edema of the macula in diabetes study: 3-year outcomes and the need for prolonged frequent treatment. JAMA Ophthalmol. 2013;131(2):139–45.
- 40. Do DV, Nguyen QD, Shah SM, et al. An exploratory study of the safety, tolerability and bioactivity of a single intravitreal injection of vascular endothelial growth factor Trap-Eye in patients with diabetic macular oedema. Br J Ophthalmol. 2009;93:114–49.
- 41. Do DV, Sepah YJ, Boyer D, Callanan D, Gallemore R, Bennett M, Marcus DM, Halperin L, Sadiq MA, Rajagopalan N, Campochiaro PA, Nguyen QD. Month-6 primary outcomes of the READ-3 study (Ranibizumab for Edema of the mAcula in Diabetes-Protocol 3 with high dose). Eye (Lond). 2015;29(12):1538–44.
- Dor Y, Porat R, Keshet E. Vascular endothelial growth factor and vascular adjustments to perturbations in oxygen homeostasis. Am J Physiol. 2001;280:C1367–74.
- Drolet DW, Nelson J, Tucker CE, et al. Pharmacokinetics and safety of an anti-vascular endothelial growth factor aptamer (NX1838) following injection into the vitreous humor of rhesus monkeys. Pharm Res. 2000;17:1503–10.
- 44. Dvorak HF, Harvey VS, Estrella P, Brown LF, McDonagh J, Dvorak AM. Fibrin containing gels induce angiogenesis: implications for tumor stroma generation and wound healing. Lab Invest. 1987;57:673–86.
- 45. Early Treatment Diabetic Retinopathy Study Research Group. Photocoagulation for diabetic macular edema. Early Treatment Diabetic Retinopathy Study Research Group report number 1. Arch Ophthalmol. 1985;103(12):1796–806.
- Eggington S, Zhou AL, Brown MD, Hudlicka O. Unorthodox angiogenesis in skeletal muscle. Cardiovasc Res. 2003;49:634–46.

- El-Batarny AM. Intravitreal bevacizumab as an adjunctive therapy before diabetic vitrectomy. Clin Ophthalmol. 2008;2(4):709–16.
- Elicieri BP, Paul R, Schwartzberg PL, Hood JD, Leng J, Cheresh DA. Selective requirement for Src kinases during VEGF-induced angiogenesis and vascular permeability. Mol Cell. 1999;4:915–24.
- Elman MJ, Ayala A, Bressler NM, Browning D, Flaxel CJ, Glassman AR, Jampol LM, Stone TW, Diabetic Retinopathy Clinical Research Network. Intravitreal ranibizumab for diabetic macular edema with prompt vs. deferred laser treatment: 5-year randomized trial results. Ophthalmology. 2015;122(2):375–81.
- 50. Eyetech Study Group. Preclinical and phase 1A clinical evaluation of an anti-VEGF pegylated aptamer (EYE001) for the treatment of exudative age-related macular degeneration. Retina. 2002;22:143–52.
- Ferrara DC, Koizumi H, Spaide RF. Early bevacizumab treatment of central retinal vein occlusion. Am J Ophthalmol. 2007;144(6):864–71.
- 52. Ferrara N. Vascular endothelial growth factor: basic science and clinical progress. Endocr Rev. 2004;25:581–611.
- Ferrara N, Chen H, Davis-Smyth T, et al. Vascular endothelial growth factor is essential for corpus luteum angiogenesis. Nat Med. 1998;4:336–40.
- 54. Ferrara N, Damico L, Shams N, Lowman H, Kim R. Development of ranibizumab, an antivascular endothelial growth factor antigen binding fragment, as therapy for neovascular agerelated macular degeneration. Retina. 2006;26(8):859–70.
- 55. Ferrara N, Gerber HP, LeCouter J. The biology of VEGF and its receptors. Nat Med. 2003;9:669–76.
- Ferrara N, Henzel WJ. Pituitary follicular cells secrete a novel heparin-binding growth factor specific for vascular endothelial cells. Biochem Biophys Res Commun. 1989;161:851–8.
- 57. Ferrara N, Hillan KJ, Gerber HP, Novotny W. Discovery and development of bevacizumab, an anti-VEGF antibody for treating cancer. Nat Rev Drug Discov. 2004;3:391–400.
- Filho JA, Messias A, Almeida FP, Ribeiro JA, Costa RA, Scott IU, Jorge R. Panretinal photocoagulation (PRP) versus PRP plus ranibizumab for high-risk proliferative diabetic retinopathy. Acta Ophthalmol. 2011;89:e567–72.
- 59. Fischer C, Mazzone M, Jonckx B, Carmeliet P. FLT1 and its ligands VEGFB and PIGF: drug targets for anti-angiogenic therapy? Nat Rev Cancer. 2008;8:942–56.
- Flaxel CJ, Edward AR, Aiello LP, et al. Factors associated with visual acuity outcomes after vitrectomy for diabetic macular edema: diabetic retinopathy clinical research network. Retina. 2010;30(9):1488–95.
- 61. Folkman J. Tumor angiogenesis: therapeutic implications. N Engl J Med. 1971;285:1182-6.
- 62. Friberg TR, Tolentino M. Pegaptanib sodium as maintenance therapy in neovascular agerelated macular degeneration: the LEVEL study. Br J Ophthalmol. 2010;94:1611–7.
- Fujisawa H, Kitsukawa T, Kawakami A, et al. Roles of a neuronal cell-surface molecule, neuropilin, in nerve fiber fasciculation and guidance. Cell Tissue Res. 1997;290:465–70.
- 64. Funatsu H, Yamashita H, Noma H, Mimura T, Yamashita T, Hori S. Increased levels of vascular endothelial growth factor and interleukin-6 in the aqueous humor of diabetics with macular edema. Am J Ophthalmol. 2002;133:70–7.
- 65. Funatsu H, Yamashita H, Ikeda T, Mimura T, Eguchi S, Hori S. Vitreous levels of interleukin-6 and vascular endothelial growth factor are related to diabetic macular edema. Ophthalmology. 2003;110:1690–6.
- 66. Furfine E, Coppi A, Koehler-Stec E, Zimmer E, Tu W, Struble C. Pharmacokinetics and ocular tissue penetration of VEGF Trap after intravitreal injections in rabbits. Invest Ophthalmol Vis Sci. 2006;47: E-abstract 1430.
- 67. Garrido-Urbani S, Bradfield PF, Lee BP, Imhof BA. Vascular and epithelial junctions: a barrier for leucocyte migration. Biochem Soc Trans. 2008;36:203–11.
- 68. Gaudreault J, Fei D, Beyer JC, et al. Pharmacokinetics and retinal distribution of ranibizumab, a humanized antibody fragment directed against VEGF-A, following intravitreal administration in rabbits. Retina. 2007;27:1260–6.
- Gaudreault J, Fei D, Rusit J, Suboc P, Shiu V. Preclinical pharmacokinetics of ranibizumab (rhuFabV2) after a single intravitreal administration. Invest Ophthalmol Vis Sci. 2005;46: 726–33.
- 70. Gaudreault J, Webb W, Van Hoy M, et al. Pharmacokinetics and retinal distribution of AMD rhufab V2 after intravitreal administration in rabbits. AAPS Pharm Sci. 1999;(suppl 1): 3207.
- Gerber HP, Condorelli F, Park J, Ferrara N. Differential transcriptional regulation of the two VEGF receptor genes. Flt-1, but not Flk-1/KDR, is up-regulated by hypoxia. J Biol Chem. 1997;272:23659–67.
- Gerber HP, McMurtrey A, Kowalski, et al. VEGF regulates endothelial cell survival by the PI3-kinase/Akt signal transduction pathway. Requirement for Flk-1/KDR activation. J Biol Chem. 1998;273:30343–66.
- Gerber HP, Hillan KJ, Ryan AM, et al. VEGF is required for growth and survival in neonatal mice. Development. 1999;126:1149–59.
- 74. Ghassemifar R, Lai CM, Raoczy PE. VEGF differentially regulates transcription and translation of ZO-1alpha+ and ZO-1alpha- and mediates trans-epithelial resistance in cultured endothelial and epithelial cells. Cell Tissue Res. 2006;323:117–25.
- 75. Gilbert RE, Vranes D, Berka JL, et al. Vascular endothelial growth factor/vascular permeability factor is temporally and spatially correlated with ocular angiogenesis in a primate model. Am J Pathol. 1998;145:574–84.
- Gragoudas ES, Adamis AP, Cunningham ET, Feinsod M, Guyer DR. Pegaptanib for neovascular age-related macular degeneration. N Engl J Med. 2004;351:2805–16.
- Greenberg PB, Martidis A, Rogers AH, Duker JS, Reichel E. Intravitreal triamcinolone acetonide for macular oedema due to central retinal vein occlusion. Br J Ophthalmol. 2002;86(2):247–8.
- Grugel S, Finkenzeller G, Weindel K, Barleon B, Marme D. Both v-Ha-Ras and v-Raf stimulate expression of the vascular endothelial growth factor in HIH 3 T3 cells. J Biol Chem. 1995;270:25915–9.
- 79. Guo D, Jia Q, Song HY, Warren RS, Donner DB. Vascular endothelial cell growth factor promotes tyrosine phosphorylation of mediators of signal transduction that contain SH2 domains. Association with endothelial cell proliferation. J Biol Chem. 1995;270:6729–33.
- Hattori K, Dias S, Heissig B, et al. Vascular endothelial growth factor and angiopoietin-1 stimulate postnatal hematopoiesis by recruitment of vasculogenic and hematopoietic stem cells. J Exp Med. 2001;193:1005–14.
- Heiduschka P, Fietz H, Hofmeister S, et al. Penetration of bevacizumab through the retina after intravitreal injection in the monkey. Invest Ophthalmol Vis Sci. 2007;48:2814–23.
- Holash J, Davis S, Papadoupoulos N, et al. VEGF-Trap: a VEGF blocker with potent antitumor effects. Proc Natl Acad Sci U S A. 2002;99:11393–8.
- Hornan D, Edmeades N, Krishnan R, Khan J, Lochhead J. Use of pegaptanib for recurrent and non-clearing vitreous haemorrhage in proliferative diabetic retinopathy. Eye (Lond). 2010;24:1315–9.
- 84. Houck KA, Ferrara N, Winer J, Cachianes G, Li B, Leung DW. The vascular endothelial growth factor family: identification of a fourth molecular species and characterization of alternative splicing of RNA. Mol Endocrinol. 1991;5:1806–14.
- 85. Ida H, Tobe T, Nambu H, Matsumura M, Uyama M, Campochiaro PA. RPE cells modulate subretinal neovascularization, but do not cause regression in mice with sustained expression of VEGF. Invest Ophthalmol Vis Sci. 2003;44:5430–7.
- 86. Ishida S, Usui T, Yamashiro K, et al. VEGF164 is proinflammatory in the diabetic retina. Invest Ophthalmol Vis Sci. 2003;44:2155–62.
- 87. Ishikawa K, Honda S, Tsukahara Y, Negi A. Preferable use of intravitreal bevacizumab as a pretreatment of vitrectomy for severe proliferative diabetic retinopathy. Eye (Lond). 2009;23:108–11.
- Jackson TL, Antcliff RJ, Hillenkamp J, Marshall J. Human retinal molecular weight exclusion limit and estimate of species variation. Invest Ophthalmol Vis Sci. 2003;44:2141–6.

- Jonas JB, Jonas RA, Neumaier N, Findeisen P. Cytokine concentration in aqueous humor of eyes with diabetic macular edema. Retina. 2012;32(10):2150–7.
- Jonas JB, Schmidbauer M, Rensch F. Progression of tractional retinal detachment following intravitreal bevacizumab. Acta Ophthalmol. 2009;87:571–2.
- Jonas JB, Söfker A. Intraocular injection of crystalline cortisone as adjunctive treatment of diabetic macular edema. Am J Ophthalmol. 2001;132(3):425–7.
- 92. Jorge R, Costa RA, Calucci D, et al. Intravitreal bevacizumab (Avastin) for persistent new vessels in diabetic retinopathy (IBEPE Study). Retina. 2006;26:1006–13.
- Kamiya K, Konno H, Tanaka T, et al. Antitumor effect on human gastric cancer and induction of apoptosis by vascular endothelial growth factor neutralizing antibody. Jpn J Cancer Res. 1999;90:794–800.
- Karkkainan MJ, Makinen T, Alitalo K. Lymphatic endothelium: a new frontier of metastasis research. Nat Cell Biol. 2002;4:E2–5.
- Keyt B, Berleau L, Nguyen H, et al. The carboxyl-terminal domain (111-165) of VEGF is critical for mitogenic potency. J Biol Chem. 1996;271:7788–95.
- Krohne TU, Liu Z, Holz FG, Meyer CH. Intraocular pharmacokinetics of ranibizumab following a single intravitreal injection in humans. Am J Ophthalmol. 2012;154(4):682–6.
- 97. Kim I, Ryan AM, Rohan R, et al. Constitutive expression of VEGF, VEGFR-1 and VEGFR-2 in normal eyes. Invest Ophthalmol Vis Sci. 1999;40:2115–21.
- 98. Korobelnik JF, Do DV, Schmidt-Erfurth U, Boyer DS, Holz FG, Heier JS, Midena E, Kaiser PK, Terasaki H, Marcus DM, Nguyen QD, Jaffe GJ, Slakter JS, Simader C, Soo Y, Schmelter T, Yancopoulos GD, Stahl N, Vitti R, Berliner AJ, Zeitz O, Metzig C, Brown DM. Intravitreal aflibercept for diabetic macular edema. Ophthalmology. 2014;121(11):2247–54.
- Kowalczuk L, Touchard E, Omri S, et al. Placental growth factor contributes to microvascular abnormalization and blood-retinal barrier breakdown in diabetic retinopathy. PLoS One. 2011;6(3):e17462.
- Krzystolik MG, Afshari MA, Adamis AP, et al. Prevention of experimental choroidal neovascularization with intravitreal anti-vascular endothelial growth factor antibody fragment. Arch Ophthalmol. 2002;120:338–46.
- 101. Krzystolik MG, Filippopoulos T, Ducharme JF, Losenstein JI. Pegaptanib as an adjunctive treatment for complicated neovascular diabetic retinopathy. Arch Ophthalmol. 2006;124(6):920–1.
- 102. Ku DD, Zaleski JK, Liu S, Brock TA. Vascular endothelial growth factor induces EDRFdependent relaxation in coronary arteries. Am J Physiol. 1993;265:H586–92.
- 103. Lang GE, Berta A, Eldem BM, Simader C, Sharp D, Holz FG, Sutter F, Gerstner O, Mitchell P, RESTORE Extension Study Group. Two-year safety and efficacy of ranibizumab 0.5 mg in diabetic macular edema. Interim analysis of the RESTORE extension study. Ophthalmology. 2013;120(10):2004–12.
- 104. Lee S, Jilani SM, Nikolova GV, et al. Processing of VEGF-A by matrix metalloproteinases regulates bioavailability and vascular patterning in tumors. J Cell Biol. 2005;169:681–91.
- 105. Lee SJ, Koh HJ. Enlargement of the foveal avascular zone in diabetic retinopathy after adjunctive intravitreal bevacizumab (Avastin) with pars plana vitrectomy. J Ocul Pharmacol Ther. 2009;25:173–4.
- Lewis H, Abrams GW, Blumenkranz MS, Campo RV. Vitrectomy for diabetic macular traction and edema associated with posterior hyaloidal traction. Ophthalmology. 1992;99(5):753–9.
- 107. Lim LS, Ng WY, Mathur R, Wong D, Wong EY, Yeo I, Cheung CM, Lee SY, Wong TY, Papakostas TD, Kim LA. Conversion to aflibercept for diabetic macular edema unresponsive to ranibizumab or bevacizumab. Clin Ophthalmol. 2015;9:1715–8.
- 108. Lo WR, Kim SJ, Aaberg Sr TM, Bergstrom C, Srivastava SK, Yan J, et al. Visual outcomes and incidence of recurrent vitreous hemorrhage after vitrectomy in diabetic eyes pretreated with bevacizumab (Avastin). Retina. 2009;29:926–31.
- 109. Loftus JV, Sultan MB, Pleil AM, Macugen 1013 Study Group. Changes in vision- and healthrelated quality of life in patients with diabetic macular edema treated with pegaptanib sodium or sham. Invest Ophthalmol Vis Sci. 2011;52:7498–505.

- Lowe J, Araujo J, Palma M, et al. RhuFab V2 inhibits VEGF-isoforms-stimulated HUVEC proliferation. Invest Ophthalmol Vis Sci. 2003;44:[E-abstract no. 1828].
- 111. Lowe J, Araujo J, Yang J, Reich M, Oldendorp A, Shiu V, Quarmby V, Lowman H, Lien S, Gaudreault J, Maia M. Ranibizumab inhibits multiple forms of active vascular endothelial growth factor in vitro and in vivo. Exp Eye Res. 2007;85(4):425–30.
- 112. Ma DJ, Park KH, Woo SJ. Predicting 1-month response of macular edema to intravitreal bevacizumab from 1-hour response. Can J Ophthalmol. 2014;49(3):267–72.
- 113. Macugen Diabetic Retinopathy Study Group. A phase II randomized double-masked trial of pegaptanib, an anti-vascular endothelial growth factor aptamer, for diabetic macular edema. Ophthalmology. 2005;112:1747–57.
- 114. Martidis A, Duker JS, Greenberg PB, Rogers AH, Puliafito CA, Reichel E, Baumal C. Intravitreal triamcinolone for refractory diabetic macular edema. Ophthalmology. 2002;109(5):920–7.
- 115. Massin P, Bandello F, Garweg JG, et al. Safety and efficacy of ranibizumab in diabetic macular edema (RESOLVE Study): a 12-month, randomized, controlled, double-masked, multi-center phase II study. Diabetes Care. 2010;33:2399–405.
- 116. Mendrinos E, Donati G, Pournaras CJ. Rapid and persistent regression of severe new vessels on the disc in proliferative diabetic retinopathy after a single intravitreal injection of pegaptanib. Acta Ophthalmol. 2009;87(6):683–4.
- 117. Meyer CH, Krohne TU, Holz FG. Intraocular pharmacokinetics after a single intravitreal injection of 1.5 mg versus 3.0 mg of bevacizumab in humans. Retina. 2011;31:1877–84.
- 118. Michaelides M, Kaines A, Hamilton RD, et al. A prospective randomized trial of intravitreal bevacizumab or laser therapy in the management of diabetic macular edema (BOLT Study). Ophthalmology. 2010;117:1078–86.
- 119. Michaelson IC. The mode of development of the vascular system of the retina with some observations on its significance for certain retinal disorders. Trans Ophthalmol Soc UK. 1948;68:1625–710.
- 120. Michels S, Rosenfeld PJ, Puliafito CA, Marcus EN, Venkatraman AS. Systemic bevacizumab (Avastin) therapy for neovascular age-related macular degeneration twelve-week results of an uncontrolled open-label clinical study. Ophthalmology. 2005;112(6):1035–47.
- 121. Mitchell P, Bandello F, Schmidt-Erfurth U, Lang GE, Massin P, Schlingemann RO, Sutter F, Simader C, Burian G, Gerstner O, Weichselberger A, RESTORE study group. The RESTORE Study. Ranibizumab monotherapy or combined with laser versus laser monotherapy for diabetic macular edema. Ophthalmology. 2011;118(4):615–25.
- 122. Miyamoto N, de Kozak Y, Normand N, et al. PIGF-1 and VEGFR-1 pathway regulation of the external epithelial hemato-ocular barrier. Ophthalmic Res. 2008;40:203–7.
- 123. Modarres M, Nazari H, Falavarjani KG, Naseripour M, Hashemi M, Parvaresh MM. Intravitreal injection of bevacizumab before vitrectomy for proliferative diabetic retinopathy. Eur J Ophthalmol. 2009;19:848–52.
- 124. Moradian S, Ahmadieh H, Malihi M, Soheilian M, Dehghan MH, Azarmina M. Intravitreal bevacizumab in active progressive proliferative diabetic retinopathy. Graefes Arch Clin Exp Ophthalmol. 2008;246:1699–705.
- 125. Mordenti J, Cuthbertson RA, Ferrara N, et al. Comparisons of the intraocular tissue distribution, pharmacokinetics, and safety of ¹²⁵I-labeled full-length and Fab antibodies in rhesus monkeys following intravitreal administration. Toxicol Pathol. 1999;27:536–44.
- 126. Nagy JA, Benjamin L, Zeng H, Dvorak AM, Dvorak HF. Vascular permeability, vascular hyperpermeability and angiogenesis. Angiogenesis. 2008;11:109–19.
- 127. Nepomuceno AB, Takaki E, Paes de Almeida FP, et al. A prospective randomized trial of intravitreal bevacizumab versus ranibizumab for the management of diabetic macular edema. Am J Ophthalmol. 2013;156:502–10.
- 128. Neufeld G, Cohen T, Gengrinovitch S, Poltorak Z. Vascular endothelial growth factor (VEGF) and its receptors. FASEB J. 1999;13:9–22.
- 129. Nguyen QD, Brown DM, Marcus DM, et al. Ranibizumab for diabetic macular edema: results from 2 phase III randomized trials: RISE and RIDE. Ophthalmology. 2012;119:789–801.

- 130. Nguyen QD, Shah SM, Heier JS, Do DV, Lim J, Boyer D, Abraham P, Campochiaro PA, READ-2 Study Group. Primary End Point (Six Months) Results of the Ranibizumab for Edema of the mAcula in diabetes (READ-2) study. Ophthalmology. 2009;116(11): 2175–81.
- 131. Nguyen QD, Shah SM, Khwaja AA, READ-2 Study Group, et al. Two-year outcomes of the ranibizumab for edema of the macula in diabetes (READ-2) study. Ophthalmology. 2010;117:2146–51.
- 132. Nguyen QD, Shah SM, Van Anden E, Sung JU, Vitale S, Campochiaro PA. Supplemental oxygen improves diabetic macular edema: a pilot study. Invest Ophthalmol Vis Sci. 2004;45(2):617–24.
- 133. Nguyen QD, Tatlipinar S, Shah SM, et al. Vascular endothelial growth factor is critical stimulus for diabetic macular edema. Am J Ophthalmol. 2006;142:961–9.
- 134. Nomura M, Yamagishi S, Harada S, et al. Possible participation of autocrine and paracrine vascular endothelial growth factors in hypoxia-induced proliferation of endothelial cells and pericytes. J Biol Chem. 1995;270:28316–24.
- 135. Oshima Y, Shima C, Wakabayashi T, Kusaka S, Shiraga F, Ohji M, et al. Microincision vitrectomy surgery and intravitreal bevacizumab as a surgical adjunct to treat diabetic traction retinal detachment. Ophthalmology. 2009;116:927–38.
- 136. Papadopoulos N, Martin J, Ruan Q, et al. Binding and neutralization of vascular endothelial growth factor (VEGF) and related ligands by VEGF Trap, ranibizumab, and bevacizumab. Angiogenesis. 2012;15(2):171–85.
- 137. Park JE, Chen HH, Winer J, Houck KA, Ferrara N. Placental growth factor. Potentiation of vascular endothelial growth factor bioactivity, in vitro and in vivo, and high affinity binding to Flt-1 but not Flk-1/KDR. J Biol Chem. 1994;269:25646–54.
- 138. Pekala P, Marlow M, Heuvelman D, Connolly D. Regulation of hexose transport in aortic endothelial cells by vascular permeability factor and tumor necrosis factor-alpha, but not by insulin. J Biol Chem. 1990;265:18051–4.
- 139. Pokroy R, Desai UR, Li Y, Edwards P. Bevacizumab prior to vitrectomy for diabetic traction retinal detachment. Eye. 2011;25:989–97.
- 140. Presta LG, et al. Humanization of an anti-vascular endothelial growth factor monoclonal antibody for the therapy of solid tumors and other disorders. Cancer Res. 1997;57(20): 4593–9.
- 141. Prünte C, Fajnkuchen F, Mahmood S, et al. Ranibizumab 0.5 mg treat-and-extend regimen for diabetic oedema: the RETAIN study. Br J Ophthalmol. 2015;100:787–95. [Epub before print]
- 142. Qaum T, Xu Q, Joussen AM, et al. VEGF-initiated blood-retinal barrier breakdown in early diabetes. Invest Ophthalmol Vis Sci. 2001;42:2408–13.
- 143. Querques G, Bux AV, Martinelli D, Iaculli C, Noci ND. Intravitreal pegaptanib sodium (Macugen®) for diabetic macular oedema. Acta Ophthalmol. 2009;87:623–30.
- 144. Regillo CD, Brown DM, Abraham P, et al. Randomized, double-masked, sham-controlled trial of ranibizumab for neovascular age-related macular degeneration: PIER study year 1. Am J Ophthalmol. 2008;145(2):239–48.
- 145. Régnier S, Malcolm W, Allen F, Wright J, Bezlyak V. Efficacy of anti-VEGF and laser photocoagulation in the treatment of visual impairment due to diabetic macular edema: a systemic review and network meta-analysis. PLoS One. 2014;9(7):e102309.
- 146. Rizzo S, Genovesi-Ebert F, Di Bartolo E, Vento A, Miniaci S, Williams G. Injection of intravitreal bevacizumab (Avastin) as a preoperative adjunct before vitrectomy surgery in the treatment of severe proliferative diabetic retinopathy (PDR). Graefes Arch Clin Exp Ophthalmol. 2008;246:837–42.
- 147. Roberts WG, Palade GE. Increased microvascular permeability and endothelial fenestration induced by vascular endothelial growth factor. J Cell Sci. 1995;108:2369–79.
- 148. Romano MR, Gibran SK, Marticorena J, Wong D, Heimann H. Can a preoperative bevacizumab injection prevent recurrent postvitrectomy diabetic vitreous haemorrhage? Eye. 2009;23:1698–701.

- Rosenfeld PJ, Brown DM, Heier JS, et al. Ranibizumab for age-related macular degeneration. N Engl J Med. 2006;355:1419–31.
- 150. Rosenfeld PJ, Fung AE, Puliafito CA. Optical coherence tomography findings after an intravitreal injection of bevacizumab (Avastin) for macular edema from central retinal vein occlusion. Ophthalmic Surg Lasers Imaging. 2005;36(4):336–9.
- 151. Rosenfeld PJ, Moshfeghi AA, Puliafito CA. Optical coherence tomography findings after an intravitreal injection of bevacizumab (Avastin) for neovascular age-related macular degeneration. Ophthalmic Surg Lasers Imaging. 2005;36(4):331–5.
- 152. Ruckman J, Green LS, Beeson J, et al. 2'-Fluoropyrimidine RNA-based aptamers to the 165amino acid form of vascular endothelial growth factor (VEGF-165). Inhibition of receptor binding and VEGF-induced vascular permeability through interactions requiring the exon7endoded domain. J Biol Chem. 1998;273:20556–67.
- 153. Schmidt-Erfurth U, Eldem B, Guymer R, et al., EXCITE Study Group, et al. Efficacy and safety of monthly versus quarterly ranibizumab treatment in neovascular age-related macular degeneration: the EXCITE study. Ophthalmology. 2011;118(5):831–9.
- 154. Schmidt-Erfurth U, Lang GE, Holz FG, et al., RESTORE extension study group, et al. Threeyear outcomes of individualized ranibizumab treatment in patients with diabetic macular edema: the RESTORE extension study. Ophthalmology. 2014;121:1045–53.
- 155. Senger DR, Galli SJ, Dvorak AM, Perruzzi CA, Harvey VS, Dvorak HF. Tumor cells secrete a vascular permeability factor that promotes accumulation of ascites fluid. Science. 1983;12:983–5.
- 156. Shahar J, Avery RL, Heilweil G, et al. Electrophysiologic and retinal penetration studies following intravitreal injection of bevacizumab (Avastin). Retina. 2006;26:262–9.
- 157. Sivaprasad S, Crosby-Nwaobi R, Heng LZ, Peto T, Michaelides M, Hykin P. Injection frequency and response to bevacizumab monotherapy for diabetic macular oedema (BOLT Report 5). Br J Ophthalmol. 2013;97(9):1177–80.
- 158. Soheilian M, Ramezani A, Obudi A, et al. Randomized trial of intravitreal bevacizumab alone or combined with triamcinolone versus macular photocoagulation in diabetic macular edema. Ophthalmology. 2009;116:1142–50.
- 159. Spaide R, Fisher YL. Intravitreal bevacizumab (Avastin) treatment of proliferative diabetic retinopathy complicated by vitreous hemorrhage. Retina. 2006;26:275–8.
- 160. Spitzer MS, Yoeruek E, Sierra A, et al. Comparative antiproliferative and cytotoxic profile of bevacizumab (Avastin), pegaptanib (Macugen) and ranibizumab (Lucentis) on different ocular cells. Graefes Arch Clin Exp Ophthalmol. 2007;245(12):1837–42.
- 161. Stewart MW. What are the half-lives of ranibizumab and affibercept (VEGF Trap-eye) in human eyes? Calculations with a mathematical model. Eye Reports. 2011;1:e5.
- 162. Stewart MW, Rosenfeld PJ. Predicted biological activity of intravitreal VEGF Trap. Br J Ophthalmol. 2008;92(5):667–8.
- 163. Sultan MB, Zhou D, Loftus J, Dombi T, Ice KS, Macugen 1013 Study Group. A phase 2/3, multicenter, randomized, double-masked, 2-year trial of pegaptanib sodium for the treatment of diabetic macular edema. Ophthalmology. 2011;118:1107–18.
- 164. Takahashi T, Ueno H, Shibuya M. VEGF activates protein kinase C-dependent, but Rasindependent Raf-MEP-ERK kinase pathway for DNA synthesis in primary endothelial cells. Oncogene. 1999;18:2221–30.
- 165. Terman BI, Carrion ME, Kovacs E, Rasmussen BA, Eddy RL, Shows TB. Identification of a new endothelial cell growth factor receptor tyrosine kinase. Oncogene. 1991;6:1677–83.
- 166. The Diabetic Retinopathy Study Research Group. Preliminary reports on effects of photocoagulation therapy. Am J Ophthalmol. 1976;81:383.
- 167. The Diabetic Retinopathy Study Research Group. Photocoagulation treatment of proliferative diabetic retinopathy: the second report of diabetic retinopathy study findings. Trans Am Acad Ophthalmol Otolaryngol. 1978;85:82.
- 168. The Diabetic Retinopathy Study Research Group. Photocoagulation treatment of proliferative diabetic retinopathy: clinical application of diabetic retinopathy study (DRS) findings, DRS report No. 8. Ophthalmology. 1981;88:583.

- 169. The Diabetic Retinopathy Vitrectomy Study Research Group. Early vitrectomy for severe vitreous hemorrhage in diabetic retinopathy. Two-year results of a randomized trial. Diabetic Retinopathy Vitrectomy Study report 2. Arch Ophthalmol. 1985;103(11):1644–52.
- 170. Tischer E, Mitchell R, Hartman T, et al. The human gene for vascular endothelial growth factor. Multiple protein forms are encoded through alternative exon splicing. J Biol Chem. 1991;266:11947–54.
- 171. Tolentino MJ, McLeod DS, Taomoto M, Otsuji T, Adamis AP, Lutty GA. Pathologic features of vascular endothelial growth factor-induced retinopathy in the nonhuman primate. Am J Ophthalmol. 2002;133(3):373–85.
- 172. Tuder RM, Flook BE, Voelkel NF. Increased gene expression for VEGF and the VEGF receptors KDR/Flk and Flt in lungs exposed to acute or chronic hypoxia. J Clin Invest. 1995;95:1798–807.
- 173. Ushio-Fukai M, Nakamura Y. Reactive oxygen species and angiogenesis: NADPH oxidase as target for cancer therapy. Cancer Lett. 2008;266:37–52.
- 174. Vincenti V, Cassano C, Rocchi M, Persico G. Assignment of the vascular endothelial growth factor gene to the human chromosome 6p21.3. Circulation. 1996;93:1493–5.
- 175. Wang H, Keiser A. Vascular endothelial growth factor upregulates the expression of matrix metalloproteinases in vascular smooth muscle cells: role of flt-1. Circ Res. 1998;83: 832–40.
- 176. Wang Y, Fei D, Vanderlaan M, Song A. Biological activity of bevacizumab, a humanized anti-VEGF antibody in vitro. Angiogenesis. 2004;7(4):335–45.
- 177. Writing Committee for the Diabetic Retinopathy Clinical Research Network, Gross JG, Glassman AR, Jampol LM, Inusah S, Aiello LP, Antoszyk AN, Baker CW, Berger BB, Bressler NM, Browning D, Elman MJ, Ferris FL 3rd, Friedman SM, Marcus DM, Melia M, Stockdale CR, Sun JK, Beck RW. Panretinal photocoagulation vs intravitreous ranibizumab for proliferative diabetic retinopathy. A randomized clinical trial. JAMA. 2015;314(20):2137–46.
- 178. Xu L, Lu T, Tuomi L, Jumbe N, Lu J, Eppler S, Kuebler P, Damico-Beyer LA, Joshi A. Pharmacokinetics of ranibizumab in patients with neovascular age-related macular degeneration: a population approach. Invest Ophthalmol Vis Sci. 2013;54(3):1616–24.
- 179. Yang CM, Yeh PT, Yang CH, Chen MS. Bevacizumab pretreatment and long-acting gas infusion on vitreous clear-up after diabetic vitrectomy. Am J Ophthalmol. 2008;146:211–7.
- 180. Yeoh J, Williams C, Allen P, Buttery R, Chiu D, Clark B, et al. Avastin as an adjunct to vitrectomy in the management of severe proliferative diabetic retinopathy: a prospective case series. Clin Experiment Ophthalmol. 2008;36:449–54.
- 181. Yuan F. Time-dependent vascular regression and permeability changes in established human tumor xenografts induced by an anti-vascular endothelial growth factor/vascular antibody. Proc Natl Acad Sci U S A. 1996;93:14765–70.
- 182. Zhang Z-H, Liu H-Y, Hernandez-da Mota SE, Romano MR, Falavarjani KC, Ahmadieh H, Xu X, Liu K. Vitrectomy with or without preoperative intravitreal bevacizumab for proliferative diabetic retinopathy: a meta-analysis of randomized controlled trials. Am J Ophthalmol. 2013;156:106–15.
- 183. Zhao B, Ma A, Cai J, Boulton M. VEGF-A regulates the expression of VEGF-C in human retinal pigment epithelial cells. Br J Ophthalmol. 2006;90:1052–9.

Chapter 5 Corticosteroids: Targeting Multiple Cytokines and Chemokines

5.1 Introduction

Drugs that inhibit vascular endothelial growth factor (VEGF) are effective for center-involving diabetic macular edema (DME) since most eyes respond well to regimens that include repeated intravitreal injections. A significant minority of patients (approximately 40%), however, exhibit an inadequate response and require additional or alternate therapy. In the phase III anti-VEGF registration trials, laser photocoagulation served as secondary therapy for patients with inadequate responses after 3–6 months of monthly anti-VEGF injections [52, 66]. Supplemental laser usually resolves persistent edema slowly, but it has not been shown to incrementally improve VA in most cases [26, 62].

Glucocorticoids were the first drug class shown to improve DME in randomized clinical trials [34–36]. Triamcinolone acetonide, the first drug to be tested (Kenalog[®]; Bristol-Myers Squibb, New York, NY), was not originally formulated for intraocular use, and it was associated with a high risk of cataract and elevated intraocular pressure. Topical and periocular corticosteroids [24] do not significantly improve DME because they do not penetrate the sclera sufficiently to produce therapeutic concentrations within the retina. Intravitreally injected corticosteroids resolve macular edema and improve visual acuity in both treatment-naïve eyes and those previously treated with laser and/or anti-VEGF therapy. As a result, significant interest in treating DME with intravitreal corticosteroids has developed, and this chapter reviews the current state-of-the art corticosteroid use in the management of patients with DME.

5.2 Characteristics of Corticosteroids

The primary structure of VEGF – a long amino acid sequence – prevents it from crossing the cell membrane's lipid bilayer, whereas the lipophilic ring structures of corticosteroids enable them to pass easily. Corticosteroids bind to heat shock

proteins and several soluble steroid receptors within the cytoplasm [20] and the resultant complex migrates to the cell nucleus (Fig. 5.1). The heat shock protein dissociates, leaving the corticosteroid-receptor complex to bind to the promoter or repressor regions of several genes [11, 29]. The corticosteroid-receptor complex can



Fig. 5.1 This drawing shows the mechanisms through which corticosteroids exert their effects. Corticosteroids dissociate from serum transport proteins, enter target cells by crossing cell membranes, and bind to cytoplasmic receptors that include heat shock proteins (hsp90). The corticosteroid-receptor complexes enter cell nuclei, interact with DNA receptor sites, and suppress nuclear factor (NF)- κ B. This increases the expression of many species of mRNA and suppresses others; both mechanisms affect the synthesis of thousands of proteins. Corticosteroids also act as survival factors for neuroglia and directly promote vasoconstriction

also produce signal transduction within the cytoplasm, stimulating the release of molecules such as annexin-1 [77], a modulator of leukocyte migration [23].

Corticosteroids improve macular edema via several mechanisms. They bind to the promoter region of the VEGF gene and downregulate VEGF synthesis [33, 74]. The resultant free-VEGF levels drop significantly, though they remain 100-fold higher than after the intravitreal injection of anti-VEGF drugs. Steroids decrease exudation by favorably altering the Starling's law equilibrium through vasoconstriction. Steroids decrease average capillary lumen pressure, decrease the mean transcapillary exudative pressure, and reduce the hydrostatic pressure gradient.

Inflammation contributes significantly to the pathogenesis of diabetic retinopathy, and corticosteroids are potent anti-inflammatory molecules. Corticosteroids repress several key pro-inflammatory transcription markers such as nuclear factorkappa B (NF κ B) and activator protein 1, thereby disrupting the pro-inflammatory feedback loop that is critical for the development of DME [80, 89]. Corticosteroids inhibit phospholipase A2 [79], which is upregulated in the retinas of streptozotocininduced diabetic rates [40, 57] and has been associated with increased ICAM-1, VEGF, and TNF- α expression, and the formation of DME [57, 67]. Corticosteroids are known to reduce tissue edema and downregulate the release of prostaglandins and histamines [87]. Corticosteroids inhibit the synthesis of endothelial nitric oxide synthase (eNOS), a potent vasodilator [56]. Each of these factors causes vasodilation and/or edema and leads to choroidal thickening. Corticosteroids decrease the synthesis of several chemokines, intercellular adhesion molecules, and growth factors that inhibit the migration and margination of polymorphonuclear leukocytes [55]. These are capable of disrupting the VEGF feedback loop and downregulating several VEGF-mediated processes. Corticosteroids also upregulate the expression of anti-inflammatory proteins that downregulate the inflammatory response: interleukin (IL)-10, adenosine, and IκBα (the natural inhibitor of NFκB).

Steroids maintain and restore the blood-retinal barrier by preventing phosphorylation of tight junction proteins [39, 86]. Corticosteroids improve tight junction integrity by upregulating junctional protein expression and facilitating their translocation to the cellular borders [3] and by protecting against oxidative stress-induced phosphorylation of tight junction proteins in RPE cells [63]. Corticosteroids increase fluid movement through the retina by stabilizing Müller cells and improving aquaporin-4 (AQP-4) and potassium channels [70, 90].

The neuroprotective capabilities of a constant infusion of low-dose fluocinolone were demonstrated in transgenic S334ter-4 and Royal College of Surgeons rats. Fluocinolone preserves outer nuclear layer cell morphology, stabilizes a- and b-wave electroretinography amplitudes, and reduces neuroinflammation [37, 38].

Triamcinolone, dexamethasone, and fluocinolone are all fluorinated glucocorticoids that lack mineralocorticoid activity. They differ according to their glucocorticoid-receptor binding affinities (dexamethasone > triamcinolone > fluocinolone) and their lipophilicity (triamcinolone > fluocinolone > dexamethasone). These characteristics may be partially responsible for their relative potencies (triamcinolone = 5, dexamethasone = 25, fluocinolone = 25, compared to cortisol = 1) (Table 5.1).

Characteristics of corticosteroids that are used to treat diabetic retinopathy				
Characteristic	Dexamethasone	Fluocinolone	Triamcinolone	
Molecular weight (kDa)	0.392	0.452	0.394	
Binding affinity (nmol)	5.4	2.0	1.5	
Intravitreal half-life	5.5 h	Unknown	18 days (crystalline)	
Relative potencies(cortisol = 1)	25	25	5	

 Table 5.1
 This table lists the important characteristics of the corticosteroids used for the treatment of diabetic macular edema

Corticosteroids differ according to the proteins that they regulate. Triamcinolone, dexamethasone, and fluocinolone each upregulate over 6000 proteins in each of two types of retinal pigment epithelial cells, but only 15–25% of these proteins are upregulated by all three corticosteroids. Within trabecular meshwork (TM 86) cells, dexamethasone upregulates transcripts associated with RNA posttranscriptional modifications, fluocinolone affects lipid metabolism, and triamcinolone affects cell morphology; within TM 93 cells dexamethasone affects histone methylation, fluocinolone affects the cell cycle, and triamcinolone affects cell adhesion [65].

Solubilized fractions of the 3 corticosteroids have brief intravitreal half-lives of only 2–5 h [29]. Crystalline deposits of triamcinolone acetonide are minimally soluble in aqueous, and they slowly release steroid with a half-life of approximately 18 days [75]. Since triamcinolone and dexamethasone are less lipophilic than fluocinolone, some authors speculate that fluocinolone may accumulate less in the anterior chamber and cause fewer steroid-related intraocular pressure rises [69].

Corticosteroid administration by several routes (topical, periocular, intraocular, systemic) frequently increases intraocular pressure and causes posterior subcapsular cataracts [5, 6, 43]. Cataract formation and IOP elevation may be due to activation of a corticosteroid receptor found in both the lens and trabecular meshwork [34].

5.3 Corticosteroid Delivery to the Eye

The timeline in Fig. 5.2 shows important milestones in the use of intraocular corticosteroids for the treatment of DME.

Since corticosteroids are small and lipophilic, they easily cross cell membranes and the blood-retinal barrier to reach glucocorticoid receptors in target tissues. Parenterally administered corticosteroids have large volumes of distribution, so high doses would need to be given to improve retinal vascular conditions and significant side effects would be frequent. Corticosteroid drops and periocular depot injections would be unlikely to cause systemic adverse events, but they must penetrate several tissue layers to reach the retina in therapeutic concentrations [44].

The composition of tears together with their rapid elimination through the lacrimal drainage system via the blink mechanism constitutes the major barriers to topically administered drug penetration into the eye. Drug that manages to penetrate the



Fig. 5.2 This figure details the timeline for the intraocular use of corticosteroids. Many of the most important milestones in the history of corticosteroid use and delivery system creation are listed

corneal and conjunctival epithelium may be rapidly eliminated by conjunctival lymphatics and uveal blood flow. Unfortunately, topically administered drugs generally have intravitreal bioavailability of less than 0.001% [60].

Topical corticosteroids are effective against several chorioretinal vascular conditions in rats because the diffusion distance from the cornea to the retina is very short. The larger anteroposterior diameter of the human eye (approximately 24 mm), however, makes it nearly impossible to achieve clinically active concentrations within the retina. Corticosteroid formulations that may enhance ocular penetration and facilitate diffusion to the macula will be discussed in Chap. 8.

Sub-Tenon's injections of triamcinolone acetonide treat uveitis and postoperative cystoid macular edema but are ineffective against DME [24]. Drug must penetrate several anatomic barriers (episclera, sclera, choroid, and retinal pigment epithelium) to reach the retinal vasculature. Furthermore, flow within the conjunctival lymphatics and vitreous moves drug away from the retina [50]. Together, these diffusion barriers decrease intravitreal bioavailability after a sub-Tenon's injection to 0.01-0.1% [30]. Drug injections directly into the vitreous bypass the outer bloodretinal barrier and produce maximum bioavailability [30].

Several sustained-release reservoirs and extended duration strategies have been developed to prolong the retention of steroids within the vitreous. Triamcinolone sustained-release systems (Kenalog[®]; Trivaris[®], Allergan, Irvine, CA, USA; Triesence[®], Alcon, Fort Worth, TX, USA) are currently available for periocular and intraocular use (Fig. 5.3). Attempts to create sustained-release triamcinolone devices have thus far been unsuccessful, and further development has been halted. Biodegradable nanocarrier systems (liposomes, nanoparticles, nanocrystals, and nanosystems) are in various stages of development and may ultimately be used to treat posterior-segment disease [48]. Three sustained-release corticosteroid devices have been successfully developed and approved by the US Food and Drug Administration (US FDA) – the dexamethasone phosphate insert (DEX, Ozurdex[®],



Fig. 5.4 This ultrawidefield fundus photograph shows a dexamethasone insert in the inferior vitreous (*black arrow*)



Allergan, Irvine, CA, USA) (Fig. 5.4), the fluocinolone acetonide insert (Iluvien[®], Alimera Sciences, Alpharetta, GA, USA), and the fluocinolone acetonide implant (Retisert[®], Bausch & Lomb, Rochester, NY, USA). These will be discussed later in this chapter.

5.4 Triamcinolone

Elevated aqueous levels of interleukin (IL)-8, interferon-induced protein-10 (IP-10), monocyte chemoattractant protein-1 (MCP-1), and VEGF were found in 22 eyes of 11 patients with DME compared to 6 control eyes [76]. In the 11 patients

macula

Fig. 5.3 This fundus photograph was taken 1 day after intravitreal injection of triamcinolone acetonide. Drug suspended in the posterior vitreous can be seen over the

Important diabet	ic macular edema trial	ls with triamcinolone
Trial and phase	Cohorts	Key findings
DRCR.net Protocol B	Treatment arms IVT 1 mg IVT 4 mg Laser	 At 2 years 1. Improvements in mean BCVA were -3, -2, and +1 letters 2. Elevated intraocular pressures seen in 16%, 33%, and 13% 3. Cataracts developed in 23%, 61%, and 13%
Gillies et al. (2006)	Treatment arms IVT Laser	 At 2 years 1. Eyes treated with IVTA were twice as likely to achieve 10 letter improvement in BCVA 2. 44% of IVT eyes required intraocular pressure lowering medications
Kriechbaum et al. (2014)	Treatment arms IVT (8 mg) Bevacizumab (2.5 mg)	Comparable improvements in BCVA and macular thickness at 3 months At 12 months, VA stable in bevacizumab arm but decreased in IVT arm due to cataracts
DRCR.net Protocol I 854 patients	Treatment arms RAN + laser RAN + def laser IVT + laser Laser	 At 12 months 1. Eyes receiving IVT + laser achieved 5 fewer letters of visual improvement compared to RAN groups 2. Pseudophakic eyes receiving IVT + laser had comparable BCVA improvements as RAN groups

 Table 5.2
 This table lists the important diabetic macular edema trials with triamcinolone acetonide and details their key findings

IVT intravitreal triamcinolone

with DME, one eye was injected with 4 mg of triamcinolone acetonide (TA), and the other was injected with 1.25 mg of bevacizumab. Four weeks later, steroidinjected eyes showed a reduction in IP-10, MCP-1, VEGF, IL-6, and platelet-derived growth factor-AA (PDGF-AA), whereas only VEGF was reduced in bevacizumabinjected eyes. Significant clinical trials with intravitreal triamcinolone are detailed in Table 5.2.

Pilot studies showed that intravitreal injections of triamcinolone acetonide (IVTA) effectively reduce DME [4, 59]. In the prospective, randomized DRCR.net Protocol B trial, 1 and 4 mg intravitreal TA every 4 months were compared to laser photocoagulation. At 4 months, patients receiving 4 mg of TA were more likely to achieve +10-letter improvements in BCVA compared to laser (27% vs. 17%) and had greater mean decreases in central retinal thickness (CRT) ($-98 \mu m vs. -39 \mu m$). At the 2-year primary endpoint, however, the mean BCVA improvement in the laser group was +1 ± 17 letters compared to -3 ± 22 letters and -2 ± 18 letters in the 1 and 4 mg TA groups, respectively. Elevated IOP was seen in 33% of eyes receiving 4 mg triamcinolone, 16% of those receiving 1% triamcinolone, and 13% of those treated with laser. At 2 years, more patients receiving triamcinolone required glaucoma medications (4 mg, 13%; 1 mg, 6%; laser, 3%) and more developed cataracts (4 mg, 61%; 1 mg, 23%, laser, 13%) [25].

In a 2-year randomized clinical trial, IVTA was administered as often as every 6 months [36]. Eyes receiving IVTA were twice as likely as those treated with laser monotherapy to achieve +10-letter improvements in BCVA. The majority of eyes

developed IOP problems: >5 mmHg rise in IOP (68%), need for glaucoma medication (44%), and need for cataract surgery (54%).

The effect of triamcinolone and bevacizumab on subfoveal choroidal thickness (SFCT) was investigated in a prospective, randomized study of 51 eyes with DME [78]. Compared to baseline, the mean SFCT in eyes receiving IVTA was significantly reduced at 24 h (-5.2%) and 12 weeks (-8.2%) (P < 0.01 for both). Eyes receiving single injections of bevacizumab experienced significant decreases in SFCT from 24 h to 4 weeks, but not at 8 or 12 weeks. The authors suggest that the presence of inflammatory factors other than VEGF may be responsible for the differences between IVTA and bevacizumab observed after 4 weeks. A more likely explanation is that the shorter intraocular half-life of bevacizumab produces a duration of action of less than 8 weeks.

A 12-month randomized, prospective trial compared 3 monthly injections of bevacizumab (2.5 mg) with a single injection of IVTA (8 mg) followed by monthly and q4month PRN injections, respectively [53]. Eyes in both the bevacizumab and triamcinolone arms experienced improved BCVA (0.30–0.23 logMAR for bevacizumab vs. 0.32–0.26 logMAR for triamcinolone) and reduced macular thickness (505–358 μ m for bevacizumab vs. 490–308 μ m for triamcinolone) at 3 months. Visual acuity continued to improve through 12 months in the bevacizumab group (0.18 LogMAR) but had decreased in the triamcinolone group (0.36 LogMAR) because of cataracts.

The efficacy of IVTA was evaluated in 20 eyes that had been unresponsive to at least 3 monthly intravitreal bevacizumab injections [46]. At least 2 months after the last intravitreal bevacizumab, an aqueous humor sample was obtained, and vascular endothelial growth factor, IL-2, IL-6, IL-8, tumor necrosis factor- α , and transforming growth factor- β 2 concentrations were measured with a multiplex cytokine array. Triamcinolone was injected intravitreally, and BCVA and central subfield thickness were evaluated through 3 months. The mean BCVA was 47.1 ± 18.9 letters at baseline, 53.3 ± 19.7 letters at 1 month (P = 0.002), and 52.4 ± 19.1 letters at 2 months (P = 0.204). A mean decrease in central subfield thickness (-11%) was seen in 12 eyes at 1 month. Multivariate analysis showed that the intraocular concentration of IL-8 was an independent factor for anatomic response at 1 month (P = 0.006). The authors concluded that intravitreal triamcinolone may be an attractive treatment option for patients who have poor short-term responses to bevacizumab.

Diabetic macular edema may develop or worsen after panretinal photocoagulation (PRP) for PDR. In a prospective study of eyes with severe diabetic retinopathy, 91 eyes (46 eyes with DME; 45 eyes without DME) of 76 patients underwent PRP with IVTA (30 eyes), PRP with intravitreous bevacizumab (31 eyes), or PRP alone (30 eyes) [19]. The primary outcome measures included changes in BCVA and central macular thickness (CMT) at 1 and 3 months. Secondary outcome measures were the proportions of eyes with BCVA gain or loss, and decreased or increased CMT. In eyes with DME, there was significant worsening in BCVA (P =0.031) in the PRP group but significant improvement in BCVA (P = 0.012) in the IVTA group. In eyes without CSME, those receiving PRP alone experienced significant worsening in BCVA from 0.18 to 0.26 at 1 month (P = 0.008) and to 0.27 at 3 months (P = 0.005). In eyes without DME, there was significant increase in CMT from 209.75 to 259.00 µm at 1 month (P = 0.023) and to 276.14 µm at 3 months (P = 0.011) in the PRP group; in eyes with DME, the proportion of eyes with improved BCVA and decreased CMT was significantly higher in the IVTA group (75% and 100%, respectively) than in the IVB group (37.5% and 62.5%, respectively). The authors concluded that IVTA and bevacizumab may be effective adjunctive treatments to PRP by minimizing the risk of PRP-induced macular edema and visual loss.

The DRCR.Net Protocol I trial randomized 854 patients to receive ranibizumab + prompt laser, ranibizumab + deferred laser, IVT + prompt laser or prompt laser + sham. At 1 year, the median improvements in BCVA in each of the four treatment arms were +9, +9, +4, and +3 letters, respectively [26]. Patients receiving triamcinolone experienced BCVA improvements during the first 3 months that were similar to patients receiving ranibizumab, but BCVA then worsened through 12 months because steroid-related cataracts developed. One-year BCVA improvements in the triamcinolone subgroup that was pseudophakic at baseline were comparable to those in the ranibizumab groups.

Eyes randomized to prompt laser + triamcinolone were eligible to receive ranibizumab as early as week 74 for persistent DME with no improvement in BCVA. At the 5-year concluding visit, mean improvements in BCVA in the four arms were +10, +8, +7, and +5 letters [31]. During the first 2 years of the trial, pseudophakic eyes that were randomized to IVT + laser had better improvements in BCVA compared to the entire group receiving triamcinolone. By the 5-year visit, however, the BCVA gains in the pseudophakic eyes receiving triamcinolone resembled those in the entire triamcinolone arm.

5.5 Dexamethasone

The dexamethasone posterior-segment drug delivery system (DEX, Ozurdex[®], Allergan Inc., Irvine, CA, USA) is a biodegradable, sustained-release reservoir. The DEX contains 0.7 mg of micronized, preservative-free dexamethasone in a biodegradable cylinder (6 mm long by 0.46 mm diameter) made of a copolymer of poly (lactic-co-glycolic) acid that slowly breaks down into glycolic acid and water while simultaneously releasing dexamethasone over the course of 3 months. The cylinder is preloaded into a single-use applicator and then injected across the pars plana into the anterior vitreous through a 22-gauge needle. The needle is advanced through the sclera in a tunneled fashion, and the tract self-seals after removal. The procedure is performed in the clinic under aseptic conditions after the instillation of topical and subconjunctival anesthesia.

A prospective, comparative study showed that pain after dexamethasone injection under topical anesthesia was comparable to that after anti-VEGF injections that were delivered with much smaller needles [64]. The insert should not be injected into aphakic and pseudophakic eyes that lack an intact barrier between the vitreous and anterior chamber because it can migrate into the anterior chamber and damage the corneal endothelium [49, 51, 58]. After injection, the insert usually settles into the inferior vitreous base without being noticed by patients. After approximately 3 months, the insert dissolves completely, and intraocular dexamethasone concentrations fall to near zero. In addition to rare cases of endophthalmitis and vitreous hemorrhage, injections have been complicated by inadvertent insert placement into the crystalline lens [7, 21, 32] or fragmentation of the insert [1, 27, 71, 72].

Drug release from the insert follows a biphasic pattern with intraocular concentrations peaking at 2–6 weeks (vitreous, 100–1000 μ g/mL; retina, 100–1000 μ g/gm), followed by a sharp drop during the third month, a lower plateau for 3–4 months (vitreous, 0.1–1 ng/mL; retina, 0.1–1 ng/gm), and non-detectable levels by months 7–8 [17]. The intraocular dexamethasone concentrations are higher and steadier during the initial plateau than can be achieved with topical [83], subconjunctival [81], periocular [84], and oral administration [82].

Previous vitrectomy accelerates clearance of drugs from the eye, but rabbit studies show that time-related intravitreal dexamethasone concentrations after the injection of the insert were the same in vitrectomized and non-vitrectomized eyes [18]. The CHAMPLAIN study was a prospective, open-label, 26-week trial of DEX in 55 eyes with DME that had previously undergone vitrectomy. At 8 weeks after injection, the mean BCVA improved by +6.0 letters, with 30.4% improving by >10 letters, and mean CRT changed by $-156 \mu m$ [9]. These improvements were similar to those achieved in non-vitrectomized eyes in the phase III MEAD trial [10], suggesting that DEX functions well following vitrectomy.

Plasma dexamethasone concentrations after intravitreal injection of the DEX are below the lower limit of quantification (50 pg/mL), so the likelihood of patients developing systemic corticosteroid-related effects is very low [28].

In the ORVO study [14], protein arrays were performed on aqueous samples taken before and 4 weeks after a dexamethasone insert. In many patients, aqueous hepatocyte growth factor and EG-VEGF reductions at 4 weeks mirrored the reduction in macular edema.

Important trials that featured dexamethasone inserts are detailed in Table 5.3.

In an initial 6-month trial, single applications of 0.35 and 0.7 mg dexamethasone inserts were performed in eyes with macular edema due to diabetic retinopathy, retinal vein occlusions, posterior uveitis, and the Irvine-Gass syndrome [54]. A subset analysis showed that patients with uveitis and Irvine-Gass responded equally well to the insert compared to eyes with the other diagnoses [85].

Patients in this trial with DME (n = 171) were fully reported in a separate analysis [42]. Enrollment criteria for this subgroup included patients with DME and baseline BCVA measurements from 20/40 to 20/200. Key exclusion criteria included a history of vitrectomy, moderate or severe glaucoma, poorly controlled systemic arterial hypertension (systolic blood pressure >160 mmHg and/or diastolic blood pressure >90 mmHg), and poorly controlled diabetes mellitus (HbA1c >13%). Patients were randomized 1:1:1 to receive the high-dose insert (700 µg), the lowdose insert (350 µg), or observation. The insert (20-gauge) was inserted through the

Important diabetic macular edema trials with the dexamethasone insert		
Trial and phase	Cohorts	Key findings
Macular edema trial 171 patients	Treatment arms DEX 700 µg DEX 350 µg Observation	 At 90 days 1. Proportions of patients improving by 10 letters were 33%, 21%, and 12% (<i>P</i> = 0.007) 2. Greater improvements in central macular thickness in DEX 700 μg compared to laser (<i>P</i> = 0.03) At 6 months 1. Proportions of patients improving by 10 letters were 30%, 19%, and 23%
PLACID trial Phase II	Treatment arms DEX + laser Laser	More patients receiving DEX + laser improved by 10 letters at 1 month (<i>P</i> < 0.001) and 9 months (31.7% vs. 17.3%; <i>P</i> = 0.0076) but not at 12 months Area under the curve analysis showed greater BCVA improvements in DEX + laser group
CHAMPLAIN trial open-label 55 patients	Single-arm DEX in patients with previous vitrectomy	Mean change in central retinal thickness was $-156 \mu m$ at 8 weeks ($P < 0.001$) but only $-39 \mu m$ at 26 weeks ($P = 0.004$) Mean change in BCVA was +6 letters at 8 weeks ($P < 0.001$) but only +3 letters at 26 weeks ($P = 0.046$)
MEAD trials Phase III 253 patients	Treatment arms DEX 700 μg DEX 350 μg Sham	 At 3 years 1. More patients receiving the DEX improved by at least 15 letters (22.2%, 18.4%, 12.0%; <i>P</i> < 0.018) 2. Mean improvement in BCVA in pseudophakic eyes receiving DEX was +7 letters 3. 29.7% of DEX group had IOP >25 mmHg, but only one patient required incisional glaucoma surgery 4. 66% of DEX group developed cataracts and 61% underwent surgery

 Table 5.3 This table lists the important diabetic macular edema trials with the dexamethasone insert and details their key findings

DEX dexamethasone insert, BCVA best corrected visual acuity, IOP intraocular pressure

pars plana with suturing of the incision sites. Eyes that lost 15 letters of BCVA were eligible for whatever treatment was proposed by the investigator. The primary outcome measure for these groups was a +10-letter improvement in BCVA at day 90. Additional outcome measures included the proportion of eyes improving by at least +15 letters, changes in dye leakage on fluorescein angiography, reduction in CRT, and safety measures.

At day 90, more patients in the group receiving the 700 µg insert than the 350 µg insert or observation groups experienced 10-letter improvements in BCVA (33.3% vs. 21.1% vs. 12.3%) (P = 0.007, 700 µg group vs. laser). At day 180, the corresponding +10-letter BCVA improvements were 30%, 19%, and 23% (P > 0.4 for treated vs. observed eyes). At day 60, more patients receiving the 700 µg insert than observation achieved +15-letter improvements (P = 0.01). There were significantly greater improvements in CRT for treated vs. observed eyes (P = 0.03 at day 90). A dose-response trend was noted at all time points, but there were no significant differences between the 700 and 350 µg DEX arms.

Compared to the observation group at day 90, patients receiving the 700 μ g DEX had significantly greater improvements in CRT (+30.2 μ m vs. -132.3 μ m) and fluorescein dye leakage. Four patients in the observation group and two in the 350 μ g DEX group were treated with rescue laser or triamcinolone, but none in the 700 μ g DEX group.

A subset analysis explored the relationship between macular thickness and VA improvements at sites that used OCT during the trial [8]. The macular thickness at baseline inversely correlated with BCVA (r = 20.406, P < 0.001), and patients treated with 350 or 700 µg DDS experienced significant decreases in macular thickness at day 90 (P = 0.002). There was a modest inverse correlation between changes in macular thickness from baseline to day 90 and improvement in BCVA in the 700 µg DEX group (r = 20.530, P = 0.009), but the correlation was weaker and not statistically significant in the 350 µg DEX group (r = 20.206, P = 0.304). The authors concluded that the correlation between baseline BCVA and macular thickness in patients with persistent macular edema was modest and that improvement in BCVA after treatment with the 700 µg DEX was consistent with changes in macular thickness measured using OCT.

The DEX was tolerated well, and there were no cases of endophthalmitis and no differences in cataract progression among the three groups. By the 90-day visit, an IOP of >25 mmHg had been measured in 12.7% of patients in the 350 μ g DEX group, 7.5% in the 700 μ g group, but none in the observation group. By day 180, the comparable proportions were 16.4%, 13.2%, and 0%. Most of the IOP increases were seen in the first week of the study, and many were single occurrences. No eyes required laser trabeculoplasty, and none required incisional glaucoma surgery.

Compared to patients enrolled in the DRCR.net triamcinolone trial, patients receiving DEX had worse baseline VA and longer durations of edema at enrollment. These differences make direct comparisons of these corticosteroid trials problematic. Inserts had to be surgically implanted in this study, but for the subsequent phase III trials, a single-use 22-gauge applicator was used [41].

A randomized, double-masked, multicenter, phase II trial (PLACID) compared the intravitreal injection of the dexamethasone insert + laser photocoagulation against laser monotherapy in 253 patients with DME [12]. Dexamethasone/sham was injected at baseline, and laser was performed 1 month later. During the 12-month trial, patients could receive up to three additional lasers and one additional insert. The primary efficacy variable was the percentage of patients who achieved a +10-letter improvement in BCVA. Other key efficacy measures included the change in BCVA from baseline and the change in vessel leakage as determined by fluorescein angiography. Safety outcomes included adverse events and elevations in intraocular pressure. Significantly more patients in the DEX + laser group than in the laser monotherapy group improved by +10 letters at month 1 (P < 0.001) and month 9 (31.7% vs. 17.3%, P = 0.0076), but not at month 12 (27.8% vs. 23.6%). The mean improvement in BCVA in the combination group exceeded that in the laser group (up to +7.9 vs. +2.3 letters) at all time points through month 9 (P = 0.013). Area under the curve analysis showed significantly more improvement in BCVA in eyes receiving DEX over both the 0-6-month and 0–12-month periods (P < 0.001). The central retinal thickness improved significantly more in the DEX combination group at 4 months, and the vascular leakage decreased more in the combination group at 12 months. Angiographically measured decreases in the area of diffuse capillary leakage were significantly greater in the DEX group (P = 0.041).

Elevations in IOP were modest, and all were managed with topical pressurelowering medications. Among eyes treated with DEX plus laser, 15.2% (19/125) had IOP increases of at least 10 mmHg from baseline, 16.8% (21/125) had an IOP of at least 25 mmHg, and 4.0% (5/125) had an IOP of at least 35 mmHg at some point during the study. By month 12, the IOP in all eyes had returned to less than 25 mmHg. In all cases, the elevated IOP was managed with IOP-lowering medication or observation as none required surgery. Cataract development occurred more commonly in the DEX group than in the laser group (95% vs. 22%, P = 0.017). Cataract surgery was performed in 4 eyes in the DEX group and 5 eyes in the laser group. The authors concluded that DEX plus laser effectively reduces edema and improves vision better than laser monotherapy. The 12-month results, however, suggest that 6-month treatment intervals may be too long to produce optimal results [12].

The CHAMPLAIN trial was a prospective, multicenter, open-label, 26-week study in eyes that had undergone previous vitrectomy [9]. Enrolled patients had undergone PPV an average of 31 months before study entry, most commonly for vitreous hemorrhage, proliferative DR, epiretinal membrane, DME, or vitreomacular traction syndrome. Fifty-five patients with treatment-resistant DME (average duration of 43 months) received a single intravitreal injection of 0.7 mg DEX. The primary efficacy outcome measure was the change in CRT from baseline to week 26. The mean change from baseline central retinal thickness (403 μ m) was -156 μ m at week 8 (P < 0.001) and $-39 \,\mu\text{m}$ at week 26 (P = 0.004). The mean increase in BCVA from baseline was +6.0 letters at week 8 (P < 0.001) and +3.0 letters at week 26 (P = 0.046). At week 8, 30.4% of patients had gained +10 letters in BCVA. Conjunctival hemorrhage, conjunctival hyperemia, eye pain, and increased IOP were the most common adverse events. The authors concluded that treatment with the dexamethasone intravitreal insert led to statistically and clinically significant improvements in both vision and vascular leakage from DME in previously vitrectomized eyes. Comparable efficacy in vitrectomized eyes was found in a recent, retrospective study of 58 patients [61].

The long-term safety and efficacy of the dexamethasone insert was compared to sham injections in two randomized, double-blind, multicenter, phase III registration trials (MEAD) [10]. From baseline to 3 years, more patients receiving the 0.7 mg and 0.35 mg inserts compared to sham experienced >15-letter improvements in VA (22.2% vs. 18.4% vs. 12.0%; P < 0.018). The mean improvement in VA among patients receiving the 0.7 mg insert was +7 letters. Patients that were pseudophakic at baseline had relatively stable improvements in VA throughout the trial, whereas those who were phakic at baseline experienced a rapid improvement in VA, a decline beginning at week 24 due to the formation of cataracts, and finally an improvement in VA after week 52 as cataracts were removed.

More patients receiving the insert experienced IOP readings of at least 25 mmHg (29.7% vs. 4.3% sham). In most cases, the increase in IOP was transient as the incidence peaked at 6 weeks following DEX insertion and returned to baseline in 6 months. The incidence and severity of IOP elevation after each DEX injection was consistent throughout the 3-year study. In most cases, IOP elevation was managed with topical medications or by observation, and only three patients (0.4%) required incisional glaucoma surgery. IOP elevations tended to occur early, as 75% of spikes were diagnosed after the first 2 insertions and 85% after the first 3.

Among phakic patients, 66.0% of patients treated with DEX experienced development or progression of cataracts (cortical, nuclear, or subcapsular) compared with 20.4% of sham-treated patients. Nearly 56% of dexamethasone insert-treated patients compared with 7.2% of sham-treated patients required cataract surgery during the study. The incidence of cataract-related adverse effects increased throughout the duration of the study with most cataract surgeries performed during the second and third years.

In addition to cataract progression and IOP elevation, the most frequent adverse events were conjunctival hemorrhage (23.5%), vitreous hemorrhage (10.0%), macular fibrosis (8.3%), conjunctival hyperemia (7.2%), eye pain (6.1%), vitreous detachment (5.8%), and dry eye (5.8%). Retinal tear, retinal detachment, vitreous loss, and endophthalmitis occurred in approximately 2% of patients.

Use of the dexamethasone insert is contraindicated in patients with active or suspected ocular or periocular infections, including viral, mycobacterial, and fungal diseases, and glaucoma patients with cup-to-disk ratios of at least 0.8. The use of the dexamethasone insert may increase the risk of secondary infections.

The UK National Institute for Health and Care Excellence has recommended the use of the dexamethasone intravitreal implant for pseudophakic patients with diabetic macular edema (DME) who are unresponsive to or unsuitable for non-corticosteroid therapy. In 2014, the US FDA approved the DEX for the treatment of DME in eyes that were pseudophakic or were scheduled to undergo cataract surgery. Shortly thereafter, the indications were expanded to include phakic eyes.

A 12-month head-to-head comparison of bevacizumab and the dexamethasone insert included 88 eyes of 61 patients with center-involving DME. Eyes were randomized to receive intravitreal injections of bevacizumab q4weeks PRN or the dexamethasone insert q16weeks PRN. Improvement of at least +10 letters was seen in 40% of eyes receiving bevacizumab and 41% of eyes receiving the dexamethasone insert (P = 0.83). None of the bevacizumab-treated eyes lost 10 letters, compared to 11% of the dexamethasone-treated eyes, primarily because of cataracts. Dexamethasone-treated eyes had greater decreases in CMT ($-187 \mu m vs. -122 \mu m$, P = 0.015). Not surprisingly, bevacizumab-treated eyes required more injections (8.6 vs. 2.7). An elevation in IOP of at least 5 mmHg was seen at some point in 46% of dexamethasone-treated eyes but only 19% of bevacizumab-treated eyes. The authors concluded that the dexamethasone insert and bevacizumab lead to comparable improvements in VA, but dexamethasone produces superior drying of the macula with fewer injections.

5.6 Fluocinolone

Two sustained-release fluocinolone devices have been approved for the treatment of posterior-segment conditions. The 0.59 mg fluocinolone acetonide implant (Retisert[®], Bausch & Lomb, Rochester, NY, USA) is surgically implanted through a 4 mm pars plana incision and sutured to the sclera for fixation. This device was originally based on the design of the sustained-release ganciclovir reservoir (Vitrasert[®]) but was modified in 2011 to include a silicone elastomer strut (as opposed to polyvinyl alcohol) because of spontaneous detachment of the reservoir after several years within the eye. This implant was developed, tested, and US FDA approved for the treatment of chronic noninfectious posterior uveitis [45]. The device elutes fluocinolone at a rate of 0.6 μ g/day during the first month, which drops to 0.3–0.4 μ g/day for the next 35 months. The implant precludes the need for regularly administered immunosuppressive therapy and thereby avoids the side effects that often accompany systemic drug use. Pricing of the implant reflects this tradeoff as the implant wholesales for \$17,000 US.

The fluocinolone implant was studied in 80 eyes with DME [69]. The results from this feasibility study served as the basis for a subsequent 4-year, 23-center trial that randomized (2:1) 159 patients with center-involving DME to receive the 0.59 mg fluocinolone implant or standard-of-care (SOC) treatment (laser or observation at the discretion of the investigating physician) [68]. Eligible patients had macular thickening at least 1 disk area in diameter that involved the fovea. Baseline visual acuities varied from 20 letters to 68 letters (20/50 to 20/400). All eyes had previously received laser photocoagulation >12 weeks prior to enrollment. The primary outcome was the proportion of patients improving by at least 15 ETDRS letters at 6 months, and the secondary outcomes included the improvement in macular thickness, the mean change in BCVA, the change in leakage on fluorescein angiography, and the diabetic retinopathy severity scores (DRSS). Safety evaluation included the incidences of adverse events. The implant was surgically placed at day 1, but no subsequent surgery (including cataract extraction) was allowed for 6 months. Laser photocoagulation for diffuse macular edema could not be performed within the first 6 months, but focal treatment of microaneurysms was allowed earlier.

Visual acuity improved by >3 lines in more patients receiving the implant than SOC at 6 months (16.8% vs. 1.4%; P = 0.0012), 1 year (16.4% vs. 8.1%; P = 0.0012), and 2 years (31.8% vs. 9.3%; P = 0.0016). The lack of a significant difference in VA at 3 years (31.1% vs. 20.0%; P = 0.1566) probably reflects depletion of the implants. The number of eyes with the FA implant with no evidence of foveal thickening was lower than those receiving SOC at 6 months (P < 0.0001), 1 year (P < 0.0001), 2 years (P = 0.016), but not 3 years (P = 0.861). More eyes receiving the implant experienced improvements in DRSS from 6 months through 3 years. At 1 year, 20% of implanted eyes experienced a two-step improvement in DRSS, but this had fallen to 12% by 3 years. By 3 years, 20 implanted eyes required 32 laser treatments, whereas 28 SOC eyes required 52 laser treatments. Intraocular pressure >30 mmHg was noted in 61.4% of implanted eyes compared to only 5.8% of SOC

Important diabetic macular edema trials with the fluocinolone insert		
Trial and		
phase	Cohorts	Key findings
FAMOUS	Treatment arms	At 12 months
trial	FA 0.2 µg/day	1. Excellent sustained-release of fluocinolone
Phase I	FA 0.5 µg/day	2. Reduction of macular edema
35 patients		3. Improvement in visual acuity
FAME trials	Treatment arms	At 24 months (primary endpoint)
Phase III	FA 0.2 µg/day	1. The proportions of eyes improving by 15 letters were
956 patients	FA 0.5 µg/day	28.7%, 28.6%, and 16.2% (<i>P</i> = 0.002)
	Sham	2. Improvements in mean BCVA of +4.4, +5.4, and +1.7 letters
		3. 25% of patients required a second insert before
		36 months
		4. Eyes with chronic edema (>3 years) had greater chance of three-line improvement

 Table 5.4
 This table lists the important diabetic macular edema trials with the fluocinolone insert and details their key findings

FA fluocinolone acetonide insert, BCVA best corrected visual acuity

eyes, and by 4 years, 33.8% of implanted eyes required incisional glaucoma surgery to lower the pressure. Cataract removal was performed in 91.4% of phakic eyes but only 20% of SOC eyes by 4 years. Because of the high rate of incisional glaucoma surgeries, further development of the fluocinolone implant for DME was halted.

The fluocinolone insert (Iluvien) is a nonbiodegradable cylindrical tube $(3.5 \times 0.37 \text{ mm})$ that is injected with a 25-gauge preloaded inserter. The inserts are made of polyvinyl alcohol, similar to the original design of the fluocinolone implant. The drug release characteristics of the fluocinolone acetonide insert (FA) were originally studied in rabbits. After injection of the 0.2 µg insert, drug concentrations in ocular tissues peaked between days 2 and 8, reached a steady-state level by 3 months, and slowly decreased through the remainder of the study. Fluocinolone was found in most ocular tissues through 2 years, but it was not detected in the anterior chamber at most time points [47].

Important trials that featured the fluocinolone insert are detailed in Table 5.4.

The insert was originally studied in a 1-year pharmacokinetic study (FAMOUS). Inserts that eluted either 0.2 μ g/day or 0.5 μ g/day provided consistent sustained drug delivery for 1 year with associated reduction in DME [15].

The insert underwent testing for DME in two parallel, 101 center, randomized, phase III registration trials (FAME) [13]. A total of 956 patients were randomized to receive sham (185), a 0.2 mg (375), or a 0.5 mg insert (393). At entry, eyes had BCVA between 20/50 and 20/400 despite having received at least one previous macular laser treatment. Patients with preexisting glaucoma were excluded from the trial. Six weeks after randomization, subjects were eligible for rescue laser. One year after randomization, additional inserts or sham injections could be given if necessary. The primary outcome was an improvement in BCVA of >15 ETDRS letters at month 24. Secondary outcomes included other measures of visual function and foveal thickness.

The mean duration of DME at baseline was 3.5–3.9 years. Significant visual acuity improvements were noted in both FA treatment groups at 3 weeks and at every

time point thereafter. The proportions of patients reaching the primary endpoint were 28.7% (low dose), 28.6% (high dose), and 16.2% (sham; P = 0.002 for each). The mean improvements in BCVA at month 24 were +4.4, +5.4, and +1.7 letters (P = 0.02 and P = 0.016 compared to sham). A final VA of 20/40 or better was achieved in 33%, 31%, and 16% of eyes (P = 0.0185 and P = 0.0064 compared to sham), whereas a final acuity of <20/200 was achieved in 14% of insert eyes and 12% of sham eyes. For eyes with at least 3 years of DME prior to the study, 34% in the lowdose group improved by at least 15 letters (versus 13.4% of sham; P < 0.001). The treatment benefit in these eyes did not appear to depend on baseline anatomic characteristics [Cunha-Vaz]. Eves receiving the insert had significantly greater improvements in foveal thickness at all time points. Final CST of <250 µm was achieved in 40%, 47%, and 51% of eyes. Because of recurrent or persistent edema, 23.5% (low dose) and 26.4% (high dose) of eves required at least 2 insert injections. Fewer insert than sham patients required laser photocoagulation treatments (36.7%, 35.2%, and 58.9%). Significantly more phakic patients receiving the insert (74.9% and 84.5% vs. 23.1% sham) required cataract surgery and their final BCVA improvements were similar to eyes that were already pseudophakic at baseline. Eyes that underwent cataract surgery by the conclusion of the trial had better average visual acuities than when the trials began [88]. Incisional surgery to control glaucoma was required in 3.7%, 7.6%, and 0.5% of eyes. A secondary analysis showed that eyes receiving the 0.2 mg insert experienced less progression of PDR compared to controls (17% vs. 31%; P < 0.0001) [2].

A preplanned subgroup analysis of the FAME trials compared VA improvements in eyes with chronic (>3 years) versus non-chronic (<3 years) DME. At the 36-month endpoint, more eyes with chronic edema treated with inserts improved by 3 lines (34% vs. sham (13.4%); P < 0.001) compared to eyes with non-chronic edema (22.3% vs. sham (27.8%); P = 0.275). The authors speculated that chronic edema is more sensitive to corticosteroids because it is chemokine-driven, whereas nonchronic edema is less responsive to corticosteroids because it is VEGF-driven [22].

A pharmacokinetic analysis of aqueous fluocinolone concentrations that had been obtained during the previously discussed fluocinolone implant and insert trials was subsequently performed [16]. At 1 month after insertion, the anterior chamber fluocinolone concentrations were 2.17 ng/ml (0.2 μ g/day insert), 3.03 ng/ml (0.5 μ g/day insert), and 6.12 ng/ml (0.6 μ g/day implant). At 3 months, mean FAc levels were 1.76, 2.15, and 6.12 ng/ml, respectively. The low-dose and high-dose inserts produced relatively stable aqueous drug concentrations (0.45–1.18 ng/ml and 0.84–1.50 ng/ml) between 6 and 30 months after injection that were far below the concentrations produced by the implant (>6 ng/ml) through 15 months. The authors concluded that the inserts provide high fluocinolone concentrations for the first 6 months that reach a lower steady state through 3 years. The higher aqueous fluocinolone concentrations from the implant may result from its relatively anterior scleral fixation point immediately behind the lens, which probably explains the considerably higher incidence of glaucoma.

Because the insert has an extended duration of action and has only recently been approved, very little post-approval data is available. A small retrospective study of 15 eyes with chronic edema unresponsive to other therapies showed improved VA in 11, stable VA in 2, and decreased VA in 2 after injection of the low-dose insert. Elevated IOP occurred in 2 eyes with one requiring cyclocryodestruction to control the intraocular pressure [73].

The 0.19 mg insert has been approved for the treatment of diffuse DME in 17 countries through the European Repeat-Use application procedure (Austria, Belgium, Czech Republic, Denmark, Finland, France, Germany, Ireland, Italy, Luxembourg, the Netherlands, Norway, Poland, Portugal, Spain, Sweden, and the United Kingdom) as well as the USA (2014). The package insert specifies that Iluvien is indicated in eyes that do not have steroid-responsive intraocular pressures, but the required method of steroid challenge is not specified. Some investigators propose that a sufficient challenge includes the administration of topical corticosteroids, the peri- or intraocular administration of triamcinolone, or the intravitreal use of the dexamethasone insert.

5.7 Conclusions

Intraocular corticosteroids successfully reduce excess macular thickness in eyes with DME and improve BCVA in most cases. The currently available drugs and devices possess longer mean durations of action than the anti-VEGF drugs, but their use is complicated by higher incidences of cataracts and glaucoma. Because of this less favorable safety profile, corticosteroids are regarded as second-line therapy by most physicians. Considerable work needs to be done to better define the role of steroids as primary, secondary, or combination therapy for DME. The current role of corticosteroids in the management of DME will be discussed further in Chap. 6.

References

- 1. Agrawal R, Fernandez-Sanz G, Bala S, Addison PK. Desegmentation of Ozurdex implant in vitreous cavity: report of two cases. Br J Ophthalmol. 2014;98(7):961–3.
- Alimera Sciences. http://investor.alimerasciences.com/releasedetail.cfm?ReleaseID=942828. Accessed 24 Nov 2015.
- Aveleira CA, Lin CM, Abcouwer SF, Ambrosio AF, Antonetti DA. TNF-alpha signals through PKCzeta/NF-kappaB to alter the tight junction complex and increase retinal endothelial cell permeability. Diabetes. 2010;59(11):2872–82.
- Bakri SJ, Shah A, Falk NS, Beer PM. Intravitreal preservative free triamcinolone acetonide for the treatment of macular oedema. Eye. 2005;19:686–8.
- Bandi N, Kompella UB. Budesonide reduces vascular endothelial growth factor secretion and expression in airway (Calu-1) and alveolar (A549) epithelial cells. Eur J Pharmacol. 2001;425:109–16.
- Becker MD, Smith JR, Max R, Fiehn C. Management of sight-threatening uveitis: new therapeutic options. Drugs. 2005;65:497–519.
- Berarducci A, Sian IS, Ling R. Inadvertent dexamethasone implant injection into the lens body management. Eur J Ophthalmol. 2014;24(4):620–2.
- Blumenkranz MS, Haller JA, Kuppermann BD, et al. Correlation of visual acuity and macular thickness measured by optical coherence tomography in patients with persistent macular edema. Retina. 2010;30:1090–4.

- Boyer DS, Faber D, Gupta S, et al. for the CHAMPLAIN Study Group. Dexamethasone intravitreal implant for treatment of diabetic macular edema in vitrectomized patients. Retina. 2011;31:915–23.
- Boyer DS, Yoon YH, Belfort Jr R, et al. Three-year, randomized, sham-controlled trial of dexamethasone intravitreal implant in patients with diabetic macular edema. Ophthalmology. 2014;121(10):1904–14.
- 11. Busillo JM, Cidlowski JA. The five Rs of glucocorticoid action during inflammation: ready, reinforce, repress, resolve, and restore. Trends Endocrinol Metab. 2013;24(3):109–19.
- Callanan DG, Gupta S, Boyer DS, et al. for the Ozurdex PLACID Study Group. Dexamethasone intravitreal implant in combination with laser photocoagulation for the treatment of diffuse diabetic macular edema. Ophthalmology. 2013;120(9):1843–51.
- 13. Campochiaro PA, Brown DM, Pearson A, et al. for the FAME Study Group. Sustained delivery fluocinolone acetonide vitreous inserts provide benefit for at least 3 years in patients with diabetic macular edema. Ophthalmology. 2012;119(10):2125–32.
- 14. Campochiaro PA, Hafiz G, Mir TA, et al. Pro-Permeability Factors After Dexamethasone Implant in Retinal Vein Occlusion; the Ozurdex for Retinal Vein Occlusion (ORVO) Study. Am J Ophthalmol. 2015;160(2):313–21. e19
- 15. Campochiaro PA, Hafiz G, Shah SM, et al. Famous Study Group. Sustained ocular delivery of fluocinolone acetonide by an intravitreal insert. Ophthalmology. 2010;117:1393–9.
- Campochiaro PA, Nguyen QD, Hafiz G, et al. for the FAMOUS Study Group. Aqueous levels of fluocinolone acetonide after administration of fluocinolone acetonide inserts or fluocinolone acetonide implants. Ophthalmology. 2013;120:583–7.
- Chang-Lin JE, Attar M, Acheampong AA, et al. Pharmacokinetics and pharmacodynamics of a sustained-release dexamethasone intravitreal implant. Invest Ophthalmol Vis Sci. 2011;52(1):80–6.
- Chang-Lin JE, Burke JA, Peng Q, et al. Pharmacokinetics of a sustained-release dexamethasone intravitreal implant in vitrectomized and nonvitrectomized eyes. Invest Ophthalmol Vis Sci. 2011;52(7):4605–9.
- Cho WB, Moon JW, Kim HC. Intravitreal triamcinolone and bevacizumab as adjunctive treatments to panretinal photocoagulation in diabetic retinopathy. Br J Ophthalmol. 2010;94:858–63.
- Clark AR, Belvisi MG. Maps and legends: the quest for dissociated ligands of the glucocorticoid receptor. Pharmacol Ther. 2012;134(1):54–67.
- Coca-Robinot J, Casco-Silva B, Armada-Maresca F, Garcia-Martinez J. Accidental injections of dexamethasone intravitreal implant (Ozurdex) into the crystalline lens. Eur J Ophthalmol. 2014;24(4):633–6.
- Cunha-Vaz J, Ashton P, Iezzi R, et al. for the FAME Study Group. Sustained delivery fluocinolone acetonide vitreous implants: long-term benefit in patients with chronic diabetic macular edema. Ophthalmology. 2014;121(10):1892–903.
- Dalli J, Norling LV, Renshaw D, Cooper D, Leung KY, Perretti M. Annexin 1 mediates the rapid anti-inflammatory effects of neutrophil-derived microparticles. Blood. 2008;112(6):2512–9.
- 24. Diabetic Retinopathy Clinical Research Network. Randomized trial of peribulbar triamcinolone acetonide with and without focal photocoagulation for mild diabetic macular edema: a pilot study. Ophthalmology. 2007;114:1190–6.
- 25. Diabetic Retinopathy Clinical Research Network. A randomized trial comparing intravitreal triamcinolone acetonide and focal/grid photocoagulation for diabetic macular edema. Ophthalmology. 2008;115:1447–9.
- 26. Diabetic Retinopathy Clinical Research Network. Randomized trial evaluating ranibizumab plus prompt or deferred laser or triamcinolone plus prompt laser for diabetic macular edema. Ophthalmology. 2010;117:1064–77.
- 27. Donmez O, Parlak M, Yaman A, Saatci AO. Splitting of a dexamethasone implant (Ozurdex) following the injection. Case Rep Ophthalmol Med. 2013;2013:247949.
- Dugel PU, Bandello F, Lowenstein A. Dexamethasone intravitreal implant in the treatment of diabetic macular edema. Clin Ophthalmol. 2015;9:1321–35.
- 29. Edelman JL. Differentiating intraocular glucocorticoids. Ophthalmologica. 2010;224(suppl 1):25–30.

- Edelhauser HF, Rowe-Rendleman CL, Robinson MR, et al. Ophthalmic drug delivery systems for the treatment of retinal diseases: basic research to clinical applications. Invest Ophthalmol Vis Sci. 2010;51(11):5403–20.
- Elman MJ, Ayala A, Bressler NM, Browning D, Flaxel CJ, Glassman AR, Jampol LM, Stone TW, for the Diabetic Retinopathy Clinical Research Network. Intravitreal ranibizumab for diabetic macular edema with prompt vs. deferred laser treatment: 5-year randomized trial results. Ophthalmology. 2015;122(2):375–81.
- Fasce F, Battaglia Parodi M, Knutsson KA, et al. Accidental injection of dexamethasone intravitreal implant in the crystalline lens. Acta Ophthalmol. 2014;92(4):e330–1.
- 33. Gardner TW, Antonetti DA, Barber AJ, et al. Penn State Retina Research Group. Diabetic retinopathy: more than meets the eye. Surv Ophthalmol. 2002;47(suppl):S253–62.
- Gillies MC, Kuzniarz M, Craig J, et al. Intravitreal triamcinolone-induced elevated intraocular pressure is associated with the development of posterior subcapsular cataract. Ophthalmology. 2005;112:139–43.
- 35. Gillies MC, Simpson JM, Gaston C, et al. Five-year results of a randomized trial with open-label extension of triamcinolone acetonide for refractory diabetic macular edema. Ophthalmology. 2009;116:2182–7.
- 36. Gillies MC, Sutter FK, Simpson JM, et al. Intravitreal triamcinolone for refractory diabetic macular edema: two-year results of a double-masked, placebo-controlled, randomized clinical trial. Ophthalmology. 2006;113:1533–8.
- Glybina IV, Kennedy A, Ashton P, Abrams GW, Iezzi R. Photoreceptor neuroprotection in RCS rats via low-dose intravitreal intravitreal sustained-delivery of fluocinolone acetonide. Invest Ophthalmol Vis Sci. 2009;50(10):4847–57.
- Glybina IV, Kennedy A, Ashton P, Abrams GW, Iezzi R. Intravitreous delivery of the corticosteroid fluocinolone acetonide attenuates retinal degeneration in S334ter-4rats. Invest Ophthalmol Vis Sci. 2010;51(8):4243–52.
- 39. Gomez-Ulla F, Marticorena J, Alfaro V, et al. Intravitreal triamcinolone in the treatment for diabetic macular edema. Curr Diabetes Rev. 2006;1:99–112.
- 40. Gong Y, Jin X, Wang QS, et al. The involvement of high mobility group 1 cytokine and phospholipases A2 in diabetic retinopathy. Lipid Health Dis. 2014;13:156.
- Haller JA, Dugel P, Weinberg DV, Chou C, Whitcup SM. Evaluation of the safety and performance of an applicator for a novel intravitreal dexamethasone drug delivery system for the treatment of macular edema. Retina. 2009;29(1):46–51.
- 42. Haller JA, Kupperman BD, Blumenkranz MS, et al. for the Dexamethasone DDS Phase II Study Group. Randomized controlled trial of an intravitreous dexamethasone drug delivery system in patients with diabetic macular edema. Arch Ophthalmol. 2010;128(3):289–96.
- Holekamp NM, Thomas MA, Pearson A. The safety profile of long-term, high-dose intraocular corticosteroid delivery. Am J Ophthalmol. 2005;139:421–8.
- Hughes PM, Olejnik O, Chang-Lin JE, Wilson CG. Topical and systemic drug delivery to the posterior segment. Adv Drug Deliv Rev. 2005;57(14):2010–32.
- 45. Jaffe GJ, Martin D, Callanan D, et al. Fluocinolone acetonide implant (Retisert) for noninfectious posterior uveitis: thirty four-week results of a multicenter randomized clinical study. Ophthalmology. 2006;113:1020–7.
- 46. Jeon S, Lee WK. Effect of intravitreal triamcinolone in diabetic macular edema unresponsive to intravitreal bevacizumab. Retina. 2014;34:1606–11.
- 47. Kane FE, Green ME. Ocular pharmacokinetics of fluocinolone acetonide following Iluvien implantation in the vitreous humor of rabbits. J Ocul Pharmacol Ther. 2015;31(1):11–6.
- 48. Kaur IP, Kakkar S. Nanotherapy for posterior eye disease. J Control Release. 2014;193:100–12.
- 49. Khurana RN, Appa SN, McCannel CA, et al. Dexamethasone implant anterior chamber migration: risk factors, complications, and management strategies. Ophthalmology. 2014;121(1):67–71.
- Kim SH, Lutz RJ, Wang NS, Robinson MR, et al. Transport barriers in transscleral drug delivery for retinal diseases. Ophthalmic Res. 2007;39(5):244–54.

- 51. Kishore SA, Schaal S. Management of anterior chamber dislocation of dexamethasone implant. Ocul Immunol Inflamm. 2013;21(1):90–1.
- 52. Korobelnik JF, Do DV, Schmidt-Erfurth U, Boyer DS, Holz FG, Heier JS, Midena E, Kaiser PK, Terasaki H, Marcus DM, Nguyen QD, Jaffe GJ, Slakter JS, Simader C, Soo Y, Schmelter T, Yancopoulos GD, Stahl N, Vitti R, Berliner AJ, Zeitz O, Metzig C, Brown DM. Intravitreal aflibercept for diabetic macular edema. Ophthalmology. 2014;121(11):2247–54.
- 53. Kriechbaum K, Prager S, Mylonas G, Scholda C, Rainer G, Funk M, Kundi M, Schmidt-Erfurth U, Diabetic Retinopathy Research Group. Intravitreal bevacizumab (Avastin) versus triamcinolone (Volon A) for treatment of diabetic macular edema: one-year results. Eye (Lond). 2014;28(1):9–15.
- 54. Kuppermann BD, Blumenkranz MS, Haller JA. et al.; Dexamethasone DDS Phase II Study Group. Randomized controlled study of an intravitreous dexamethasone drug delivery system in patients with persistent macular edema. Arch Ophthalmol. 2007;125(3):309–17.
- 55. Kupperman BD, Zacharias LC, Kenney MC. Steroid differentiation: the safety profile of various steroids on retinal cells in vitro and their implications for clinical use (an American Ophthalmology Society Thesis). Trans Am Ophthalmol Soc. 2014;112:116–41.
- 56. Liu Y, Mladinov D, Pietrusz JL, et al. Glucocorticoid response elements and 11 betahydroxysteroid dehydrogenases in the regulation of endothelial nitric oxide synthase expression. Cardiovasc Res. 2009;81:140–7.
- 57. Lupo G, Motta C, Giurdanella G, et al. Role of phospholipases A2 in diabetic retinopathy: in vitro and in vivo studies. Biochem Pharmacol. 2013;86(11):1603–13.
- 58. Malclès A, Janin-Manificat H, Yhuel Y, et al. Anterior chamber migration of intravitreal dexamethasone implant (Ozurdex®) in pseudophakic eyes: report of three cases. J Franc Ophtalmol. 2013;36(4):362–7. French.
- 59. Martidis A, Duker JS, Greenberg PB, et al. Intravitreal triamcinolone for refractory diabetic macular edema. Ophthalmology. 2002;109:920–7.
- 60. Maurice DM. Drug delivery to the posterior segment from drops. Surv Ophthalmol. 2002;47(suppl 1):S41–52.
- 61. Medeiros MD, Alkabes M, Navarro R, Garcia-Arumí J, Mateo C, Corcóstegui B. Dexamethasone intravitreal implant in vitrectomized versus nonvitrectomized eyes for treatment of patients with persistent diabetic macular edema. J Ocul Pharmacol Ther. 2014;30(9):709–16.
- 62. Mitchell P, Bandello F, Schmidt-Erfurth U, Lang GE, Massin P, Schlingemann RO, Sutter F, Simader C, Burian G, Gerstner O, Weichselberger A. on behalf of the RESTORE study group. The RESTORE Study. Ranibizumab monotherapy or combined with laser versus laser monotherapy for diabetic macular edema. Ophthalmology. 2011;118(4):615–25.
- Miura Y, Roider J. Triamcinolone acetonide prevents oxidative stress-induced tight junction disruption of retinal pigment epithelial cells. Graefes Arch Clin Exp Ophthalmol. 2009;247(5):641–9.
- Moisseiev E, Regenbogen M, Rabinovitch T, Barak A, Loewenstein A, Goldstein M. Evaluation of pain during intravitreal Ozurdex injections vs intravitreal bevacizumab injections. Eye. 2014;28(8):980–5.
- 65. Nehmé A, Lobenhofer EK, Stamer WD, Edelman JL. Glucocorticoids with different chemical structures but similar glucocorticoid receptor potency regulate subsets of common and unique genes in human trabecular meshwork cells. BMC Medical Genomics. 2009;2:58.
- 66. Nguyen QD, Brown DM, Marcus DM, et al. Ranibizumab for diabetic macular edema: results from 2 phase III randomized trials: RISE and RIDE. Ophthalmology. 2012;119:789–801.
- 67. Pannicke T, Iandiev I, Wurm A, et al. Diabetes alters osmotic swelling characteristics and membrane conductance of glial cells in rat retina. Diabetes. 2006;55(3):633–9.
- Pearson PA, Comstock TL, Ip M, et al. Fluocinolone acetonide intravitreal implant for diabetic macular edema: a 3-year multicenter, randomized, controlled clinical trial. Ophthalmology. 2011;118(8):1580–7.
- Pearson P, Levy B, Fluocinolone acetonide implant study group. Fluocinolone acetonide intravitreal implant to treat DME: 2-year results of a multi-center clinical trial. Ophthalmol Vis Sci. 2005;46:E-Abstract 4673.

- Reichenbach A, Wurm A, Pannicke T, Iandiev I, Wiedemann P, Bringmann A. Muller cells as players in retinal degeneration and edema. Graefes Arch Clinical Exp Ophthalmol. 2007;245(5):627–36.
- 71. Rishi P, Mathur G, Rishi E. Fractured Ozurdex[™] implant in the vitreous cavity. Ind J Ophthalmol. 2012;60(4):337–8.
- 72. Roy R, Hegde S. Split Ozurdex implant: a caution. Can J Ophthalmol. 2013;48(1):e15-6.
- 73. Schmit-Eilenberger VK. A novel intravitreal fluocinolone acetonide implant (Iluvien[™]) in the treatment of patients with chronic diabetic macular edema that is insufficiently responsive to other medical treatment options: a case series. Clin Ophthalmol. 2015;9:801–11.
- 74. Sears JE, Hoppe G. Triamcinolone acetonide destabilizes VEGF mRNA in Muller Cells under continuous cobalt stimulation. Invest Ophthalmol Vis Sci. 2005;46:4336–41.
- Sobrin L, D'Amico DJ. Controversies in intravitreal triamcinolone acetonide use. Intl Ophthalmol Clinics. 2005;45:133–41.
- Sohn HJ, Han DH, Kim IT, et al. Changes in aqueous concentrations of various cytokines after intravitreal triamcinolone versus bevacizumab for diabetic macular edema. Am J Ophthalmol. 2011;152(4):686–94.
- 77. Solito E, Mulla A, Morris JF, Christian HC, Flower RJ, Buckingham JC. Dexamethasone induces rapid serine-phosphorylation and membrane translocation of annexin 1 in a human folliculostellate cell line via a novel nongenomic mechanism involving the glucocorticoid receptor, protein kinase C, phosphatidylinositol 3-kinase, and mitogen-activated protein kinase. Endocrinology. 2003;144(4):1164–74.
- Sonada S, Sakamoto T, Yamashita T, et al. Effect of intravitreal triamcinolone acetonide or bevacizumab on choroidal thickness in eyes with diabetic macular edema. Invest Opthalmol Clin Sci. 2014;55:3979–85.
- 79. Stewart MW. Corticosteroid use for diabetic macular edema: old fad or new trend? Curr Diab Rep. 2012;12(4):364–75.
- 80. Tang J, Kern TS. Inflammation in diabetic retinopathy. Prog Retin Eye Res. 2011;30(5):343-58.
- Weijtens O, Feron EJ, Schoemaker RC, et al. High concentration of dexamethasone in aqueous and vitreous after subconjunctival injection. Am J Ophthalmol. 1999;128(2):192–7.
- 82. Weijtens O, Schoemaker RC, Cohen AF, et al. Dexamethasone concentration in vitreous and serum after oral administration. Am J Ophthalmol. 1998;125(5):673–9.
- Weijtens O, Schoemaker RC, Romijn FP, Cohen AF, Lentjes EG, van Meurs JC. Intraocular penetration and systemic absorption after topical application of dexamethasone disodium phosphate. Ophthalmology. 2002;109(10):1887–91.
- Weijtens O, van der Sluijs FA, Schoemaker RC, et al. Peribulbar corticosteroid injection: vitreal and serum concentrations after dexamethasone disodium phosphate injection. Am J Ophthalmol. 1997;123(3):358–63.
- 85. Kupperman BD, et al. Dexamethasone DDS Phase II Study Group. Dexamethasone posteriorsegment drug delivery system in the treatment of macular edema resulting from uveitis or Irvine-Gass syndrome. Am J Ophthalmol. 2009;147(6):1048–54. e1-e2
- Wilson CA, Berkowitz BA, Sato Y, et al. Treatment with intravitreal steroid reduces blood-retinal barrier breakdown due to retinal photocoagulation. Arch Ophthalmol. 1992;110:1155–9.
- 87. Yamamoto Y, Gaynor RB. Therapeutic potential of inhibition of the NF-kappaB pathway in the treatment of inflammation and cancer. J Clin Invest. 2001;107:135–42.
- Yang Y, Bailey C, Holz FG, et al. Long-term outcomes of phakic patients with diabetic macular oedema treated with intravitreal fluocinolone acetonide (FAc) implants. Eye. 2015;29(9):1173–80.
- Zhang W, Liu H, Rojas M, Caldwell RW, Caldwell RB. Anti-inflammatory therapy for diabetic retinopathy. Immunotherapy. 2011;3(5):609–28.
- 90. Zhao M, Bousquet E, Valamanesh F, et al. Differential regulations of AQP4 and Kir4.1 by triamcinolone acetonide and dexamethasone in the healthy and inflamed retina. Invest Ophthalmol Vis Sci. 2011;52(9):6340–7.

Chapter 6 Current Treatment Recommendations

6.1 Introduction

The use of the intraocular pharmaceutical agents that were discussed in Chapters 4 and 5 has transformed the treatment of diabetic retinopathy (DR). Before the introduction of potent pharmacotherapy for DR, early diagnosis of DR followed by prompt laser photocoagulation to eyes at risk of vision loss was the most effective way to stabilize vision [23]. Significant improvements in visual acuity (VA) were rarely achieved in patients with established diabetic macular edema (DME), so the primary goal of laser photocoagulation was to prevent further loss. Pars plana vit-rectomy for persistent vitreous hemorrhage in eyes with healthy maculae sometimes produced dramatic improvements in VA, but, unfortunately, these were a small minority of patients with diabetes-related vision loss [52].

Administration of intravitreal pharmacotherapy (Fig. 6.1) now gives patients with vision loss due to DR an excellent chance of achieving a clinically meaningful (>5 ETDRS letters) improvement in VA. Effective treatment regimens usually require years of frequent assessments and several administrations of one or more drugs, often in conjunction with laser photocoagulation and surgery [8, 23]. These demanding regimens can be expensive and they challenge even the most compliant patients, but for patients and physicians who adhere to a sensible treatment strategy, the results can be gratifying and life-changing.

The goal of this chapter is to synthesize the results produced by the randomized trials that were presented in previous chapters to propose data-driven – whenever possible – treatment guidelines that can be used in most clinical situations. Some new data will be introduced in this chapter, but most will be summarized from previously referenced trials. Despite the plethora of published manuscripts from the past 10 years that have provided level I and II evidence supporting the treatment of DR, conflicting results challenge certain recommendations, and significant gaps in our knowledge remain. Fortunately, however, our present knowledge enables us to make sound, evidence-based, scientific decisions when treating most patients.



Fig. 6.1 This photograph demonstrates the technique used by the author for an intravitreal injection of a vascular endothelial growth factor inhibitory drug. An eyelid speculum is not used as the surgeon manually retracts the eyelids. Injecting the drug into the vitreous offers the following advantages over systemic administration: 1. Drug is deposited close to the target tissue. 2. The vitreous acts as a slow-release depot that naturally enables extended duration therapy. 3. Systemic exposure to the drug is limited because of the low total dose

6.2 General Medical Care

The first two chapters of this volume detailed the pathophysiology of DR, its important risk factors, and exacerbating conditions. To help patients with their ophthalmic and systemic health, ophthalmologists must remember that DR represents the ocular manifestation of a multisystem disease that includes both microvascular (nephropathy and neuropathy) and macrovascular (cardiovascular and cerebrovascular) conditions. Ophthalmologists have the opportunity to not only maintain and improve visual function but, through counseling of patients and communication with primary care physicians, to improve glucose control, minimize the progression of systemic vascular diseases, reduce morbidity, and prolong life expectancy. Ophthalmologists should develop and maintain close relationships with primary care physicians, internists, and endocrinologists. Important systemic interventions that may prevent or control DR are listed in Table 6.1.

6.2.1 Diabetes Mellitus

The duration over which the patients have had diabetes mellitus (DM) and the average degree of glucose control are the major determinants of both the development and progression of DR [17]. The Diabetes Control and Complications Trial (DCCT) showed that the benefits of "tight" glycemic control on the progression of DR took at least 2 years to become manifest, but once established, these benefits persisted for years [16]. Patients with type 1 DM in the DCCT that received standard diabetes care had average hemoglobin (Hb) A1c concentrations of 9 mg/100 ml, whereas those receiving intensive care had average HbA1c concentrations of 7 mg/100 ml.

Systemic medical c	onditions and diabetic retinopathy
Study	Key findings
Diabetes mellitus	
UK Prospective Diabetes Study (UKPDS)	Type 1 diabetes mellitus 4209 patients (1977–1999) Intensive versus conventional therapy: 35% reduced progression per A1C point 47% reduction in moderate vision loss
Diabetes Control and Complications Trial (DCCT)	Type 1 diabetes 1441 patients (1983–1993) Intensive versus conventional therapy (HbA1C 7.0 vs. 9.0) 50% reduction in advanced retinopathy Intensive control can temporarily worsen retinopathy in the short term
Systemic arterial hy	pertension
Wisconsin Epidemiological Study of Diabetic Retinopathy	Progression of retinopathy was associated with: Higher diastolic blood pressure at baseline Increase in diastolic blood pressure over a 4-year period
UK Prospective Diabetes Study (UKPDS)	Systolic blood pressure <150 mmHg: Decreased the progression of DR Decreased the need for macular laser photocoagulation for DME
EUCLID study	Lisinopril decreased the progression of DR in normotensive type 1 diabetics
DIabetic REtinopathy Candesartan Trials (DIRECT)	In type 1 diabetics, 5 years of treatment: Decreased the incidence of DR Had no effect on progression of established DR In type 2 diabetics: 34% regression of DR ($P = 0.009$) Less severe retinonathy in types 1 and 2 ($P = 0.03$)
RASS trial	Evaluated 285 normotensive patients treated with enalapril, losartan, or placebo for 5 years Progression of retinopathy by 2 steps: Placebo (38%) Enalapril (25%; $P = 0.02$) Losartan (21%; $P = 0.008$) Enalapril and losartan increased the likelihood of less DR progression by 65 and 70% independent of blood pressure lowering
Hyperlipidemia	
Fenofibrate Intervention and Event Lowering in Diabetes Study (FIELD)	 Found that fenofibrate: Decreased the requirement for the first laser and the development of diabetic macular edema Decreased the need for laser treatment compared to the control group (3.4% vs 4.9%; P = 0.0002) Appeared to have protective effects independent of blood glucose, blood pressure, and baseline lipid values
ACCORD-Eye Study	 The addition of fenofibrate to basal statin therapy resulted in: A decrease in the progression of DR, in a similar manner to that observed with intensifying blood glucose control, but with a good safety profile without increasing the risk of hypoglycemia Questions regarding fenofibrate's mechanism of action and the pathogenesis of DR/DME

 Table 6.1 Systemic medical conditions that affect the development and progression of diabetic retinopathy

The UK Prospective Diabetes Study (UKPDS) demonstrated that improved glucose control in patients with type 2 DM significantly decreases the progression of DR [74]. By routinely asking patients about their HbA1c levels, ophthalmologists and their staffs reinforce the notion that glucose control is the most important modifiable risk factor for DR. Though lower HbA1c concentrations can further reduce the risk of DR development and progression, levels below 7 mg/100 ml can be difficult to achieve in patients with type 2 diabetes, are associated with higher frequencies of symptomatic hypoglycemia in patients receiving exogenous insulin, and may be associated with an increased risk of cardiovascular events and mortality [27].

Eyes with vision-threatening DME may benefit from improved glucose control since lowered HbA1c levels can delay or even prevent the need for ophthalmic interventions. The DCCT showed, however, that DME can suddenly and dramatically worsen in some patients when glucose control is improved, so these patients should be monitored carefully when tighter glucose control is being implemented. Patients should be warned that tighter glucose control can worsen DME but that the ocular benefits of tighter control accrue after 2 years.

Recent data suggest that improved control of modifiable systemic risk factors may reverse DME. In the RISE/RIDE trials, 18% of patients in the sham group had visual improvement of at least +15 letters, and 30% of patients in the sham arm never received "rescue" laser [8]. For patients with DME treated with anti-VEGF therapy, the composition of the anti-glycemic regimen (insulin vs. oral hypoglycemic) does not appear to influence the efficacy of the ocular therapy [46].

Prior to the introduction of pharmacotherapy, patients with center-threatening DME and poor glucose control would routinely receive prompt macular photocoagulation [23] because laser could not be counted on to improve VA if the edema spread to the fovea and adversely affected vision. These same eyes can now be observed while glucose control is improving, because expanding edema with worsening VA can probably be reversed by initiating pharmacotherapy.

Diabetic macular edema has been attributed to pioglitazone (Actos) and rosiglitazone (Avandia) [64], but the ACCORD trial failed to find an increase in VA loss in patients receiving thiazolidinediones [70]. Some investigators speculate that DME development due to these drugs is idiosyncratic and that ophthalmologists should consider asking that they be discontinued if a patient develops suspicious DME.

6.2.2 Systemic Arterial Hypertension

Systemic arterial hypertension (SAH) potentiates hyperglycemia-induced damage to capillary endothelial cells and results in blood-retinal barrier breakdown. Adequate blood pressure control (systolic pressure between 130 mmHg and 150 mmHg) decreases the adverse effects of SAH on DR [34, 37, 74], but tighter control may not provide additional advantage [27]. Angiotensin-converting enzyme (ACE) inhibitors should be considered to delay both the onset and progression of DR [24, 39]. Some evidence suggests that ACE inhibitors reduce the likelihood of

developing DR in diabetic patients who are not hypertensive [39]. Losartan and candesartan reduce the progression of DR in diabetic patients with SAH [39, 66].

6.2.3 Hyperlipidemia

Elevated blood lipid concentrations potentiate blood-retinal barrier breakdown and sometimes leads to the accumulation of lipid precipitates in the retina (lipemia retinalis). Though the incidence of lipemia retinalis may be dropping, large lipid deposits within the fovea can damage photoreceptors and irreversibly decrease vision. Lowering serum triglyceride levels with statin drugs decreases vision loss in patients with DR. Some investigators also recommend adding fenofibrate [70] because it further lowers serum lipid concentrations and may decrease vision loss by another unidentified mechanism [33].

6.2.4 Sleep Apnea

Obstructive sleep apnea (OSA) has been associated with DME, but treatment of OSA has not been shown to alter the course of DME. Individual cases of DME resolution after successful treatment of OSA have been reported [29].

6.3 Screening Guidelines

Diabetic retinopathy screening guidelines have been published by the American Academy of Pediatrics (AAP) [2, 44], the American Academy of Ophthalmology (AAO) [1], the American Diabetes Association (ADA) [3], and the Canadian Ophthalmological Society (COS) [10]. Guidelines vary slightly among the organizations, but the overall recommendations are remarkably similar for patients with type 1 DM. The AAP recommends that ophthalmologic examinations begin "3-5 years after DM is diagnosed if the patient is >9 years of age" followed by annual examinations [2]. The AAO Preferred Practice Pattern recommends that ophthalmologic exams begin "3-5 years after DM is diagnosed" followed by annual examinations [4]. The ADA Position Statement recommends the first eye exam "within 3–5 years after diagnosis of diabetes once the patient is 10 years of age or older" followed by yearly examinations [3]. The Canadian Ophthalmological Society recommends that screening for DR should begin "5 years following the diagnosis of diabetes" or at puberty followed by yearly examinations [10]. For patients with type 2 DM, the initial ophthalmic examination should be performed shortly after diagnosis, and this should be followed by yearly exams.

These guidelines are based on data accumulated from several studies, most of which found that DR develops between 8 and 10 years after DM is diagnosed [38, 45, 75]. Some patients, however, develop proliferative diabetic retinopathy (PDR) and peripheral neuropathy within 1–2 years of diagnosis [22, 30] despite excellent glucose control. No cases of PDR have been discovered in the first decade of life, but hormonal changes associated with puberty may increase the risk of developing DR during the second decade [22, 30, 37, 44]. For these reasons, the guidelines generally recommend that screening for patients with type 1 DM begin 3–5 years after diagnosis, once the patient is either 10 years old or has reached puberty – whichever occurs first.

Adherence to DR screening guidelines, unfortunately, is generally very poor. A study of 902 patients with type 1 DM reported that 28% had never received an eye examination and 11% of them were at high risk of vision loss [77]. A study from a tertiary care pediatric clinic reported that only 35% of diabetic children between the ages of 15 and 20 years had been referred for an eye exam [63]. During the first year after screening guidelines for DR were implemented in Australia, no change in practice patterns were noted as only 60% of at risk patients received an eye examination in 2010 [48]. A database claims study pegged the annual ophthalmic examination rate at 34%, with only 16% of patients receiving examinations during consecutive years [51].

Diabetic retinopathy can progress rapidly during pregnancy, presumably due to elevated estrogen levels and the presence of an insulin-like growth factor [36]. Patients who develop gestational diabetes are not at risk of developing DR, but if a patient with diabetic retinopathy becomes pregnant, monthly examinations should be performed until delivery.

6.3.1 Screening Methods

Color fundus photographs remain the gold standard for detecting DR, but routine fundus photos have not been proven necessary if patients receive dilated fundus examinations. Photographic screening programs have become particularly important in developing countries where an insufficient supply of physicians and long travel distances preclude the performance of regular dilated fundus examinations [55]. Telemedicine DR screening programs can be created with modestly priced standard or non-mydriatic fundus cameras, and office personnel can be trained to take high-quality fundus photographs. These programs can generate high-quality photographs that enable readers to rule out vision-threatening retinopathy with a high degree of certainty. Adequate fundus images can be obtained from 85% of patients in rural settings even with the use of non-mydriatic cameras [12]. The value of these screening programs stems from the high DR detection rates -12 to 29% in some programs.

Recently developed adapters can transform nearly any cellular telephone into a high-quality fundus camera [61]. The low cost of the adapters together with the widespread availability of cellular telephones allows the placement of inexpensive

fundus cameras into nearly any office. With this technology, patients may receive high-quality retinopathy screening evaluations in their primary physicians' offices, after which photos can be electronically transmitted to ophthalmologists' offices or reading centers for evaluation. Patients with high-risk retinopathy can be referred for complete ophthalmology examinations or scheduled for future photographs. Issues regarding Health Insurance Portability and Accountability Act (HIPAA) compliance and insurance billing can be challenging, but systems that address these concerns have been developed.

Some reading centers employ specially trained technicians to grade clinical photographs, whereas others use ophthalmologists and retina specialists. As screening programs expand and include more patients, the number of transmitted photographs may overwhelm the ability of reading centers to properly evaluate them. Computer programs are being developed to efficiently identify abnormal digital photos, make accurate diagnoses, determine the risk of vision loss, and either recommend referral to an ophthalmologist or defer for future screening [32, 76]. Developing and validating such software is a complicated task, and powerful hardware is needed to read thousands of photographs. Software developers in several countries are developing programs that may be commercially available within 5 years.

6.4 Imaging

High-resolution, stereoscopic fundus photography remains the most accurate way to assess the presence and severity of diabetic retinopathy. Diabetic seven-field photographic surveys remain the standard for assessing DR severity in both the macula and mid-periphery (Fig. 6.2), but newly available ultra-widefield photography is easy to perform, can be done through undilated pupils, and often provides excellent views of both the macula and peripheral retina. Stereoscopic macular photographs can identify the degree and extent of macular thickening, but spectral domain



Fig. 6.2 This composite photograph compares a single 50° photograph (*left*) to a seven-field composite (*right*). Note the better view of the mid-peripheral retina with the seven-field composite



optical coherence tomography (SD-OCT) is widely available, is easier to perform, and provides quantitative measurements for serial comparisons.

Fluorescein angiography identifies perfusion defects in the retinal vasculature and helps surgeons plan laser photocoagulation sessions. Laser treatments can be directed at leaking microaneurysms and capillary beds [56], and angiography is necessary if treatment with the NAVILAS system is being planned [41]. Angiography also identifies areas of perifoveal capillary non-perfusion, which should be avoided during macular laser photocoagulation. Identifying macular non-perfusion helps set reasonable expectations for visual recovery. Capillary non-perfusion of the macula is not a contraindication to pharmacotherapy, but visual outcomes in these eyes are usually poorer despite satisfactory resolution of edema.

Persistent retinal neovascularization can cause repeated vitreous hemorrhages even after the placement of what appears to be adequate PRP. Persistent peripheral NV can best be detected with ultra-widefield angiography [35], which can guide the placement of additional PRP.

Spectral domain optical coherence tomography (OCT) imaging of the macula has become indispensable for evaluating DME (Fig. 6.3). OCT produces detailed cross-sectional images with accurate thickness and volume measurements [18]. The pivotal phase III drug trials used OCT imaging to include and exclude patients. Furthermore, OCT measurements are the best way to monitor the effectiveness of treatment and determine the need for retreatment or switching therapy.

OCT angiography (OCTA) is a recently introduced technology that visualizes patent retinal and choroidal blood vessels without the use of an injected dye. OCTA detects perfusion defects in both the superficial and deep retinal capillary plexuses and may ultimately replace fluorescein angiography for evaluating retinal vascular disease.

Fig. 6.3 This composite figure shows a color fundus photograph of an eye with diabetic macular edema with a superimposed horizontal optical coherence tomography (OCT) scan through the macula. The photograph shows scattered microaneurysms and hard exudates in the temporal macula, but retinal thickening can only be appreciated with the OCT
6.5 Treatment of Diabetic Macular Edema

6.5.1 Non-Center-Involving Diabetic Macular Edema

A decision tree for the treatment of DME is shown in Fig. 6.4.

Eyes with non-center-involving DME usually have excellent visual acuity unless the foveal avascular zone is affected by capillary dropout. When evaluating these eyes, physicians must decide if treatment is indicated or if the eye can be carefully followed. None of the pivotal pharmacotherapy registration trials enrolled patients with noncenter-involving DME, so evidence-based treatment guidelines from the pharmacotherapy era are lacking. Before ocular pharmacotherapy became available, preventing foveal thickening and its associated loss of VA was critically important, but now that pharmacotherapy can resolve center-involving edema and reverse mild visual deficits in most cases, many of these eyes can be followed until edema affects the fovea.

Since publication of the ETDRS results, macular laser photocoagulation was performed when eyes developed clinically significant macular edema (CSME) [23]. Clinically significant macular edema no longer dictates the need for treatment in the pharmacotherapy era, but it helps identify eyes with non-center-involving macular edema that threatens the fovea. Eyes with hard exudates within 500 μ m of the fovea and adjacent macular thickening, or with an area of macular thickening 1 disc diameter (DD) in size any part of which is within 1 DD of the fovea, are at increased risk



Fig. 6.4 This figure shows a paradigm that can be used to guide treatment for most eyes with diabetic macular edema. Due to the complexity of the disease and the variability of responses, direct application of this paradigm may not be possible in all cases. *DME* diabetic macular edema, *VEGF* vascular endothelial growth factor, *CSME* clinically significant macular edema

of vision loss over the next 3 years and should be watched carefully. When considering treatment of eyes with CSME, physicians should remember that this diagnosis was based on contact lens biomicroscopy and not SD-OCT scanning.

Many physicians perform laser photocoagulation to eyes with non-centerinvolving CSME to avoid beginning a series of anti-VEGF injections. Low-intensity laser photocoagulation of microaneurysms and areas of capillary leakage [54] (as discussed in Chap. 3) frequently resolves edema and prevents or delays progression to center-involving edema.

6.5.2 Center-Involving Diabetic Macular Edema

The pivotal phase III drug trials demonstrated that anti-VEGF therapy with ranibizumab or affibercept improves VA and resolves center-involving edema better than laser photocoagulation. Patients enrolled in these trials had best corrected visual acuity (BCVA) measurements that ranged from 20/40 to 20/320 [40, 53], but eyes with center-involving DME and BCVA better than 20/40 have not been rigorously compared to laser in pharmacotherapeutic trials. Therefore, firm treatment guidelines for eyes with center-involving DME but good VA are not yet available. The ongoing DRCR.net Protocol V is comparing laser photocoagulation with intravitreal aflibercept for eyes with BCVA better than 20/32 [73]. Treating these eyes with anti-VEGF injections is a reasonable strategy since this rapidly resolves edema, improves vision, and avoids complications resulting from laser photocoagulation. If we extrapolate from DRCR.net Protocol T, each of the 3 anti-VEGF drugs should work equally well for this set of patients [21]. Focal or grid-pattern laser photocoagulation also remains a reasonable alternative for these eyes, particularly if patients are still asymptomatic. Since the visual acuity in these eyes is already good, laser is being performed primarily to prevent loss of vision.

The RESTORE trial showed that for eyes with CRT <400 μ m, visual improvements are the same whether treated with laser or ranibizumab [50]. These results convinced the National Institute for Health and Care Excellence (NICE) of the UK Department of Health to have physicians use ranibizumab for eyes with CST >400 μ m but not for those with CST <400 μ m [62]. Post hoc analyses of other phase III trials failed to uncover similar results, and surgeons in other countries generally use anti-VEGF drugs as first-line therapy in eyes with CMT between 300 and 400 μ m. When caring for a patient with good VA, however, one must remember that treatment should be based on the patient's complaints and wishes, and not solely on the appearance of the OCT.

Eyes with center-involving DME and BCVA of 20/32 or worse most closely resemble those enrolled in the phase III registration trials for ranibizumab and aflibercept [40, 53]. Trials with off-label bevacizumab have provided level II data with which to base clinical decisions [49, 65]. The National Eye Institute-sponsored DRCR.net Protocol T trial showed that for eyes with baseline BCVA of 20/32 to 20/40, each of the anti-VEGF drugs produces VA gains of approximately +8 letters [21]. But for eyes with baseline BCVA of 20/50 or worse, aflibercept (+18.9 letters)

produces greater gains in VA than ranibizumab (+14.2 letters) and bevacizumab (+11.8 letters) at 1 year. Aflibercept also produces greater macular thinning (-169 μ m) than either ranibizumab (-147 μ m) or bevacizumab (-101 μ m). On average, patients received fewer aflibercept (9) than ranibizumab (10) or bevacizumab (10) injections. Unfortunately, Protocol T used a complicated retreatment algorithm that is difficult for most practices to follow. The simplified version of the Protocol T algorithm says that patients should receive 5 monthly injections followed by continued injections until stable. The 2-year data has been presented but not yet published in peer-review journals. Gains in BCVA were maintained for all groups, but for patients with baseline BCVA of 20/50 or worse, the difference between the aflibercept and ranibizumab groups decreased to 2 letters and was no longer statistically significant.

The pivotal phase III anti-VEGF trials evaluated the efficacy of monthly (ranibizumab and aflibercept) or bimonthly (aflibercept) injections on center-involving DME and patients were switched to PRN ranibizumab after 12 months in RESTORE and after 36 months in RISE/RIDE. Aggressive treatment with monthly injections probably produces the best possible visual results (Fig. 6.5), but these regimens are expensive and compliance is difficult to maintain. In contrast, DRCR.net Protocols I and T featured monthly injections for 4 months or until dry, before switching to



Fig. 6.5 This figure shows the optical coherence tomography scans of an eye with a visual acuity of 20/40 and center-involving DME (*top row*). After 10 monthly injections of ranibizumab, the DME resolved and the visual acuity improved to 20/30 (*middle row*). Examinations were then performed every 4–8 weeks with additional injections given as needed for recurrent edema. During the next 3 years, the patient required only 7 injections to maintain a dry macula and stable visual acuity (*bottom row*)

monthly PRN injections that are based on retreatment criteria that many physicians believe are too complex to use in most clinical settings. Compared to monthly injections, as-needed treatment regimens reduce the number of injections but not the number of clinic visits.

Most surgeons treat DME with a treat-and-extend (T&E) regimen to decrease the number of injections and minimize the number of office visits. The 24-month, single-masked RETAIN trial compared T&E + laser, T&E, and PRN ranibizumab regimens in patients with DME [59]. Upon entering the study, patients in all groups were treated monthly until the maculas were dry, after which injections were given according to the T&E or PRN strategies. Visual acuity improvements in patients receiving T&E + laser, T&E, and PRN were similar (+5.9, +6.1, +6.2 letters). The mean numbers of injections were 12.4, 12.8, and 10.7, but patients treated with T&E required 46% fewer clinic visits compared to PRN. More than 70% of patients had treatment intervals extended to at least 2 months. Though trial design differences make it difficult to directly compare these data to those from the phase III registration trials, the results with T&E are encouraging. A multicenter, randomized trial comparing monthly therapy with T&E is needed, but its high cost will probably prevent it from being organized.

Several issues (compounding and packaging of bevacizumab, use of ETDRS visual acuity, and complex retreatment criteria) challenge surgeons' ability to apply Protocol T strategies to clinical practice. Nonetheless, many physicians treat patients according to the trial's major conclusions: for patients with baseline BCVA of 20/40 or better, each of the drugs performed the same, so surgeons use bevacizumab because of its lower cost; for eyes with VA of 20/50 or worse, surgeons use aflibercept because of its greater efficacy through 1 year. Additional trials are needed to validate these data and identify additional subgroups that may respond particularly well or poorly to treatment.

Visual acuity improvements attributed to the use of corticosteroids for centerinvolving DME are somewhat more difficult to interpret, and complication rates due to therapy are higher. DRCR.net Protocol I showed that pseudophakic eyes treated with intravitreal triamcinolone and prompt laser had 2-year improvements in BCVA comparable to eyes receiving ranibizumab together with immediate or deferred laser [19]. Protocol amendments allowed patients to cross over to ranibizumab from 1.5 to 3 years after enrollment (depending on the time of randomization), so longterm results attributed exclusively to the use of triamcinolone/laser are not available. Nonetheless, intravitreal triamcinolone remains an inexpensive treatment option for pseudophakic patients (or those scheduled for cataract surgery) who do not have a corticosteroid-induced elevation in intraocular pressure.

The dexamethasone insert (DEX, Ozurdex®, Allergan, Irvine, CA) produced mean BCVA gains of +6 to +7 letters in pseudophakic eyes at 3 years with patients receiving an average of 4.1 injections [5]. Though more than 25% of patients experienced elevations in intraocular pressure (IOP), fewer than 1% required incisional glaucoma surgery. These improvements lagged behind the +9 to +12 letter mean BCVA gains from the phase III anti-VEGF trials, but randomized, multicenter trials directly comparing DEX with anti-VEGF drugs have not been performed. DEX has

been used as first-line therapy for patients desiring a treatment with a longer duration of action than can be achieved with anti-VEGF injections.

6.6 Treatment Failures

Anti-VEGF therapy for center-involving DME usually resolves edema and improves VA, but a significant minority of eyes (approximately 25%) respond suboptimally. Some authors cite data from the pivotal phase III trials to claim that this poor response rate may be as high as 50%, but since nearly half of the eyes from these trials improved by at least +10 letters, this reference to poor responders may be overstated. Nonetheless, the term "treatment failure" is often used to describe suboptimally responsive eyes, even though a uniformly accepted definition of this term does not exist.

A post hoc analysis of VIVID and VISTA showed that 15% of eyes with poor responses (less than +5 letters improvement in VA or <10% improvement in CST) after the first 3 monthly injections experienced below-average improvements throughout the balance of the trial. Their BCVA improvements at the study endpoint averaged only +7.8 letters [6]. These data suggest that poorly responding eyes may be identified after as few as 3 anti-VEGF injections and that perhaps they should be considered for alternate therapies such as corticosteroids. The DRCR.net Protocol U trial is currently comparing 0.3 mg ranibizumab + DEX combination therapy against 0.3 mg ranibizumab monotherapy for the treatment of persistent DME [58].

Before switching strategies or adding a second medication, the surgeon should discuss with the patient whether or not the treatment has been perceived as a success and then temper expectations because changing therapies often does not improve functional results. A significant number of eyes in Protocol I had persistent edema at 1 year despite the fact that BCVA had improved from baseline. Because these patients were stable during the latter half of the first year, they received injections as-needed during the remainder of the trial. Visual acuity in this group remained stable, indicating that improved but incompletely resolved edema may constitute a visually successful and anatomically stable result.

Switching to another drug within the same pharmacologic group or to another group has been proposed for eyes that respond inadequately to initial therapy (Fig. 6.6) [78]. Most drug-switching studies have been retrospective and suffer from several methodologic shortcomings: nonstandardized visual acuity measurements, variable entry criteria, and differing retreatment strategies. In one study, 33 eyes that were refractory to bevacizumab, triamcinolone, and DEX were switched to ranibizumab. Over an average of 48 weeks, mean VA improved from 20/110 to 20/90 and CST improved from 384 μ m to 335 μ m [13]. In a 1-month prospective trial, 14 eyes that responded poorly to bevacizumab and ranibizumab were switched to aflibercept. The mean CST improved from 421 μ m to 325 μ m (P < 0.0132) [78]. In a retrospective chart review of 21 eyes that were unresponsive to bevacizumab or



Fig. 6.6 This figure shows the optical coherence tomography scans of an eye with center-involving DME (*top row*) that was treated with intravitreal injections of bevacizumab and affibercept. Despite a series of 5 monthly injections, the macular edema persisted (*center row*). After 2 injections of the dexamethasone insert, the macular edema had completely resolved (*bottom row*)

ranibizumab, the mean VA at the last follow-up examination (median of 5 months) improved from 0.42 LogMAR to 0.37 LogMAR (P = 0.04), and the mean CFT improved from 453 µm to 325 µm (P < 0.001) after switching to aflibercept [43].

Some eyes with neovascular age-related macular degeneration that respond poorly to monthly anti-VEGF injections improve when bevacizumab is injected every 2 weeks or is injected every 2 weeks with alternating injections of ranibizumab or aflibercept [68]. This strategy has not been reported for eyes with DME, but the author has seen improvement in a small number of eyes treated in this manner.

The phase III anti-VEGF registration trials allowed the use of macular laser photocoagulation for eyes with persistent edema at 3 months (RISE and RIDE) or 6 months (VIVID and VISTA). Between 20 and 30% of anti-VEGF-treated eyes subsequently received laser, but these trials did not include anti-VEGF monotherapy arms. Data suggest that early laser photocoagulation does not improve VA in eyes receiving ranibizumab and does very little to decrease the anti-VEGF treatment burden. Through 3 years in Protocol I, patients treated with ranibizumab + prompt laser required 3 fewer ranibizumab injections but 3 more laser treatments compared to those receiving ranibizumab and deferred laser [20]. Data from Protocol I and RESTORE also suggest that the addition of macular laser photocoagulation to ranibizumab therapy decreases long-term BCVA improvement by 2–3 letters



Fig. 6.7 This ultra-widefield fluorescein angiogram frame shows scattered areas of retinal neovascularization and broad areas of peripheral capillary non-perfusion. Targeted scatter laser photocoagulation to these areas has been proposed to treat persistent macular edema and retinal neovascularization but has not yet been demonstrated to be effective for either condition

compared to ranibizumab monotherapy. Nonetheless, macular laser photocoagulation for eyes with persistent edema despite 3–6 months of anti-VEGF therapy remains an accepted treatment option.

Ultra-widefield imaging systems have identified large areas of peripheral retinal non-perfusion in many eyes with DME (Fig. 6.7). Some investigators speculate that ischemia-induced upregulation of VEGF and pro-inflammatory chemokines from these areas contributes to pharmacotherapeutic resistance. Unfortunately, laser photocoagulation to these areas does not reliably improve macular edema. In one study, 52 eyes with DME were randomized to receive intravitreal bevacizumab or bevacizumab + targeted photocoagulation of the peripheral retina. Compared to eyes receiving bevacizumab, those treated with photocoagulation had better stability of the macular edema and improvement in BCVA (P < 0.05) [69]. Though results of this study appeared promising, good evidence supporting targeted photocoagulation to areas of peripheral non-perfusion has not yet been accrued. Since peripheral laser carries a low complication rate, treatment may be reasonable in selected patients.

Some investigators have attributed poorer VA results and failure to reduce the anti-VEGF burden to the inadequacies of surgeon-directed laser photocoagulation. Small studies report that navigated laser may not incrementally improve BCVA [42], but it may decrease the need for subsequent anti-VEGF injections [4, 42]. More work with navigated laser and subthreshold micropulse laser needs to be performed before firm treatment recommendations can be made.

Corticosteroids have become increasingly popular as second-line therapy for DME. Posterior subtenon's injections of triamcinolone have been given, but little evidence supports their use in DME. Sustained-release dexamethasone and fluocinolone inserts are more attractive alternatives to intravitreal triamcinolone because they cause less glaucoma. In a small retrospective study, eyes that responded poorly to 6 injections of ranibizumab experienced improved VA after receiving dexamethasone inserts [81]. Good long-term data regarding the use of these devices as salvage therapy is not yet available, but either device is an accepted treatment for anti-VEGF resistant DME [72]. Data from the FAME trials showed that eyes with chronic DME respond better to fluocinolone than those with more recent DME, perhaps because chemokine synthesis increases as the DME becomes more chronic [15]. This supports the use of corticosteroid inserts in eyes that are resistant to anti-VEGF therapy.

Drug regimens that use combination therapy (anti-VEGFs + corticosteroids) have produced mixed results compared to monotherapy [80]. A prospective, 12-month trial randomized 40 previously treated eyes to receive bevacizumab monotherapy or bevacizumab/dexamethasone insert combination therapy. Visual acuity gains at 6 months were similar in the 2 groups (+5.4 vs. + 4.9 letters), but monotherapy eves achieved less macular thinning ($-30 \ \mu m \ vs \ -45 \ \mu m; P = 0.03$). Patients in the combination therapy group required 3 fewer bevacizumab injections, but this was offset by the need for a mean of 2.1 dexamethasone injections [47]. A study of 25 patients evaluated intravitreal triamcinolone combined with bevacizumab (IVB) in eyes that had been refractory to either as monotherapy [80]. The mean BCVA was 0.8 LogMAR before enrollment, 0.6 LogMAR at 6 months, and 0.6 LogMAR at 12 months (P =0.0001 and P = 0.003 compared to baseline). The median CMT was 575 µm at baseline, 370 μ m at 6 months, and 410 μ m at 12 months (P = 0.0001 and P = 0.0001 compared to baseline). At 6 months, the BCVA of 13 (52%) patients was stabilized $(\pm 0.2 \text{ LogMAR of initial BCVA})$, and 12 (48%) patients showed significant visual acuity improvement (>0.2 LogMAR improvement from baseline). At 12 months, 10 (40%) patients had stabilized vision, 13 (52%) showed visual acuity improvement, and 2 (8%) had loss of VA. At 6 months, 18 (72%) patients showed anatomic stabilization (a 10-50% decrease from initial CMT), and 7 (28%) demonstrated anatomic success (a decrease in CMT of more than 50% or to $<250 \mu m$ at final visit). At 12 months, 13 (52%) patients showed anatomic stabilization, 10 (40%) had anatomic success, and 2 (8%) demonstrated anatomic failure (a decrease in CMT of less than 10%). The authors concluded that the combined application of IVB and triamcinolone may improve vision and decrease CMT in severe DME cases that have been refractory to previous monotherapies.

Regimens that combine corticosteroids and anti-VEGF drugs are attractive options for eyes that have failed monotherapy. True combination therapy in which DEX is administered every 3 months and an anti-VEGF is administered monthly, regardless of macular thickness, has not been adequately studied but may be considered for eyes that appear resistant to monotherapy.

Pars plana vitrectomy has been used to treat DME for over two decades, but trials directly comparing vitrectomy to anti-VEGF therapy are lacking. In a recent, prospective, short-term trial, 44 patients without vitreomacular traction were randomized to pars plana vitrectomy with internal limiting membrane removal or 3 monthly injections of bevacizumab [60]. The frequency of visual acuity improvements was similar in patients receiving vitrectomy (59.1%) and bevacizumab (72.7%) though vitrectomy resulted in greater resolution of macular edema ($-161 \mu m vs. -108 \mu m$). The study's primary temporal endpoint was at 120 days (60 days after the last bevacizumab injection), which is generally longer than is optimal for anti-VEGF

therapy. The VA remained stable between days 60 and 120, but the CMT increased slightly in the bevacizumab group between days 90 and 120. Seven eyes developed retinal breaks during vitrectomy, but none progressed to retinal detachments.

In the United States, vitrectomy is generally reserved for eyes with DME that have failed pharmacotherapy and laser photocoagulation. The DRCR.net trials showed that eyes with vitreomacular traction improved by a mean of +3 letters after vitrectomy [28], whereas eyes without traction showed no mean improvement in VA [25]. Visual acuity changes varied widely among individuals, meaning that though many eyes achieved excellent improvements in VA, a large number lost considerable VA.

The DRCR.net study enrolled eyes that, in the belief of the investigator, would not respond to other treatments. SD-OCT scanning was not available at the time of the study, so detailed evaluation of the outer retina was not possible. Recent studies have shown that eyes with defects in the external limiting membrane (ELM) and ellipsoid zone (EZ) have less improvement in VA after vitrectomy [11]. Physicians who perform vitrectomy for eyes that have failed other therapies may wish to pre-operatively evaluate the integrity of the ELM and EZ lines to set realistic expectations regarding improvements in VA.

Many surgeons in Europe and Asia who use vitrectomy as primary or early therapy for DME contend that surgery is safe, effective, and less costly than anti-VEGF therapy. Since long-term anti-VEGF therapy is expensive, even if bevacizumab is used instead of ranibizumab or affibercept, early vitrectomy has the potential to become a low-cost approach to the treatment of DME. The multicenter International Consortium Studying Vitrectomy for Diabetic Macular Edema (ICV-DME) [31] is evaluating the efficacy of vitrectomy in eyes with mostly intact ELM and EZ lines. Study proponents hope that results from this study will lead to a larger, randomized trial that directly compares vitrectomy with anti-VEGF therapy.

6.7 Cataract Surgery and DME

Patients with DME frequently have coexisting cataracts that also contribute to vision loss. Even small-incision phacoemulsification cataract extraction upregulates inflammatory molecules that may induce postoperative cystoid macular edema or exacerbate preexisting DME. Diabetic patients without retinopathy or with mild retinopathy and no macular edema can safely undergo cataract extraction without significant risk of postoperative macular edema. Postsurgical administration of topical corticosteroids or nonsteroidal anti-inflammatory drugs is usually sufficient to prevent macular edema.

In patients with foveal-threatening DME or previously treated DME, preoperative injections of anti-VEGF drugs or corticosteroids should be considered. The MEAD trial showed that patients receiving the dexamethasone insert who underwent cataract removal achieved similar improvements in VA as patients who had been pseudophakic at the study's inception [5].

The Pan American Collaborative Retina Study Group retrospectively evaluated the use of no supplemental intravitreal pharmacotherapy, intravitreal bevacizumab, and intravitreal triamcinolone in 138 patients with diabetic retinopathy who underwent cataract surgery. Mean BCVA (LogMAR) improved from 0.82 at baseline to 0.14 at 6 months (P < 0.001) in group 1, from 0.80 to 0.54 (P < 0.001) in group 2, and from 1.0 to 0.46 (P < 0.001) in group 3. The mean central subfield thickness increased from 263.57 µm at baseline to 274.57 µm at 6 months (P = 0.088) in group 1, from 316.02 µm to 339.56 µm (P = 0.184) in group 2, and from 259.18 µm to 282.21 µm (P = 0.044) in group 3. The authors concluded that cataract surgery may be successfully performed in patients with preexisting DME when treated prophylactically with intravitreal bevacizumab or triamcinolone [26].

In another study of eyes with DR that underwent cataract surgery, the average macular thickness increased by 11% in eyes that did not receive prophylactic intravitreal injections but did not increase in eyes with CSME that received prophylactic bevacizumab [7].

The best approach to cataract surgery in eyes with DR depends upon the preoperative status of the fovea. For eyes without DME, prophylactic pharmacotherapy beyond the customary corticosteroid and nonsteroidal anti-inflammatory drops does not appear necessary. For eyes with DME, preoperative intravitreal injections of an anti-VEGF drug or corticosteroid within 1 week of surgery appear to be prudent.

6.8 **Proliferative Diabetic Retinopathy**

Recent advances in the treatment of DME have been remarkable, but those applicable to the management of PDR have been more modest. Panretinal photocoagulation remains the most commonly used treatment for PDR, but promising new data show excellent disease control after 2 years of intravitreal ranibizumab [79]. For highly compliant patients who wish to avoid the visual side effects of panretinal photocoagulation, intravitreal ranibizumab therapy appears to be an attractive option. Controlled trials for the treatment of PDR with bevacizumab and affibercept have not been performed, but most investigators believe that these drugs will produce results comparable to those achieved with ranibizumab. Results with ranibizumab are limited to 2 years, so continued follow-up of these patients is required to show durability. The phase III anti-VEGF DME registration trials showed that the ETDRS severity scores decrease after 2 years of regular injections [8, 9], suggesting that the VEGF synthesis and treatment burden may decrease over time.

Compared with data published by the Diabetic Retinopathy Vitrectomy Study (DRVS), recent results support better visual and anatomic outcomes with vitrectomy and decreased complication rates [71]. Many surgeons now perform vitreoretinal surgery earlier in the course of disease management instead of waiting for the development of more advanced PDR complications [67]. Surgeons should also consider preoperative anti-VEGF injections to decrease neovascularization and minimize intraoperative bleeding. If the eye has significant vitreoretinal traction, however, surgeons must be willing to proceed quickly to vitrectomy if anti-VEGF therapy precipitates the "crunch" syndrome. Further management options for



Fig. 6.8 This composite figure shows worsening of diabetic retinopathy during pregnancy. The patient was first examined at 5 months gestation (*upper left*) and was found to have moderate non-proliferative diabetic retinopathy without macular edema. Over the next 3 months (*upper right and lower left*), she developed severe nonproliferative diabetic retinopathy with macular edema appearing at 8 months gestation (*lower left*). The macular edema was not treated at this time, and by 3 months after delivery (*lower right*), the retinopathy improved and the macular edema resolved

established PDR are still needed, and an in-depth discussion of the surgical management of PDR is beyond the scope of this book.

6.9 Pregnancy

The development of DME during pregnancy poses a significant management challenge. Spontaneous miscarriages have followed the intravitreal injections of bevacizumab [57], making pregnancy a high-risk condition for the institution of intravitreal anti-VEGF therapy. Diabetic macular edema in pregnancy can be approached in two ways. Firstly, intravitreal dexamethasone inserts are generally effective and safe without known systemic adverse events [14]. No more than 2 injections per eye would be needed before delivery of the baby, after which all treatment options – including anti-VEGF drugs – become available. Secondly, intraocular pharmaco-therapy could be withheld for the balance of the pregnancy and anti-VEGF injections started after delivery (Fig. 6.8). The RESTORE trial showed that delaying anti-VEGF therapy for up to 1 year after laser does not cause permanent long-term vision loss. Newly developed PDR during pregnancy should be treated with PRP.

6.10 Conclusions

Intravitreal anti-VEGF therapy has become the standard of care for most cases of center-involving diabetic macular edema. Eyes that respond poorly to first-line therapy remain a therapeutic challenge, and consensus recommendations for the treatment of these eyes have not yet been developed.

References

- 1. American Academy of Ophthalmology Retina Panel. Preferred Practice Pattern Guidelines. Diabetic retinopathy. San Francisco: American Academy of Ophthalmology; 2008.
- 2. American Academy of Pediatrics, Section on Endocrinology and Section on Ophthalmology. Screening for retinopathy in the pediatric patient with type 1 diabetes mellitus. Pediatrics. 1998;101:313–4.
- 3. American Diabetes Association. Diabetic retinopathy. Diabetes Care. 2002;25(1):S90-3.
- Barteselli G, Kozak I, El-Emam S, Chhablani J, Cortes MA, Freeman WR. 12-month results of the standardized combination therapy for diabetic macular oedema: intravitreal bevacizumab and navigated retinal photocoagulation. Br J Ophthalmol. 2014;98:1036–41.
- Boyer DS, Yoon YH, Belfort Jr R, et al. Three-year, randomized, sham-controlled trial of dexamethasone intravitreal implant in patients with diabetic macular edema. Ophthalmology. 2014;121(10):1904–14.
- 6. Boyer DS. Oral Presentation. Paris: Retina Society. 2015.
- Brito PN, Rosas VM, Coentrão LM, Carneiro ÂV, Rocha-Sousa A, Brandão E, Falcão-Reis F, Falcão MA. Evaluation of visual acuity, macular status, and subfoveal choroidal thickness changes after cataract surgery in eyes with diabetic retinopathy. Retina. 2015;35(2):294–302.
- Brown DM, Nguyen QD, Marcus DM, et al. Long-term outcomes of ranibizumab therapy for diabetic macular edema: the 36-month results from two phase III trials. Ophthalmology. 2013;120:2013–22.
- Brown DM, Schmidt-Erfurth U, Do DV, Holz FG, Boyer DS, Midena E, Heier JS, Terasaki H, Kaiser PK, Marcus DM, Nguyen QD, Jaffe GJ, Slakter JS, Simader C, Soo Y, Schmelter T, Yancopoulos GD, Stahl N, Vitti R, Berliner AJ, Zeitz O, Metzig C, Korobelnik JF. Intravitreal aflibercept for diabetic macular edema: 100-week results from the VISTA and VIVID studies. Ophthalmology. 2015;122(10):2044–52.
- 10. Canadian Ophthalmological Society Diabetic Retinopathy Clinical Practice Guideline Expert Committee. Canadian Ophthalmological Society Evidence-Based Clinical Practice Guidelines for the Management of Diabetic Retinopathy. Can J Ophthalmol. 2012;47(2):91–6.
- Chhablani JK, Kim JS, Cheng L, et al. External limiting membrane as a predictor of visual improvement in diabetic macular edema after pars plana vitrectomy. Graefes Arch Clin Exp Ophthalmol. 2012;250(10):1415–20.
- 12. Chin EK, Ventura BV, See K-Y, et al. Nonmydriatic fundus photography for teleophthalmology diabetic retinopathy screening in rural and urban clinics. Telemed J e-Health. 2014;20:102–8.
- Ciulla TA, Hussain RM, Ciulla LM, Sink B, Harris A. Ranibizumab for diabetic macular edema refractory to multiple prior treatments. Retina. 2015 Nov 18;7 [Epub ahead of print].

- 14. Concillado M, Lund-Andersen H, Mathiesen ER, Larsen M. Dexamethasone intravitreal implant for diabetic macular edema during pregnancy. Am J Ophthalmol. 2016;165:7–15.
- Cunha-Vaz J, Ashton P, Iezzi R, et al. for the FAME Study Group. Sustained delivery fluocinolone acetonide vitreous implants: long-term benefit in patients with chronic diabetic macular edema. Ophthalmology. 2014;121(10):1892–903.
- Diabetes Control and Complications Trial Research Group. Early worsening of diabetic retinopathy in the Diabetes Control and Complications Trial. Arch Ophthalmol. 1998;116:874–86.
- 17. The Diabetes Control and Complications Trial (DCCT)/Epidemiology of Diabetes Interventions and Complications (EDIC) Research Group. Effect of intensive diabetes therapy on the progression of diabetic retinopathy in patients with type 1 diabetes: 18 years of follow-up in the DCCT/EDIC. Diabetes. 2015;64:631–42.
- 18. Diabetic Retinopathy Clinical Research Network. Krzystolik MG, Strauber SF, Aiello LP, Beck RW, Berger BB, Bressler NM, Browning DJ, Chambers RB, Danis RP, Davis MD, Glassman AR, Gonzalez VH, Greenberg PB, Gross JG, Kim JE, Kollman C. Reproducibility of macular thickness and volume using Zeiss optical coherence tomography in patients with diabetic macular edema. Ophthalmology. 2007;114(8):1520–5.
- 19. Diabetic Retinopathy Clinical Research Network Writing Committee. Elman MJ, Bressler NM, Qin H, Beck RW, Ferris III FL, Friedman SM, Glassman AR, Scott IU, Stockdale CR, Sun JK. Expanded 2-year follow-up of ranibizumab plus prompt or deferred laser or triam-cinolone plus prompt laser for diabetic macular edema. Ophthalmology. 2011;118:609–14.
- 20. Diabetic Retinopathy Clinical Research Network Writing Committee. Elman MJ, Qin H, Aiello LP, Beck RW, Bressler NM, Ferris FL III, Glassman AR, Maturi RK, Melia M. Intravitreal ranibizumab for diabetic macular edema with prompt versus deferred laser treatment. Three-year randomized trial results. Ophthalmology. 2012;119(11):2312–8.
- 21. Diabetic Retinopathy Clinical Research Network. Wells JA, Glassman AR, Ayala AR, Jampol LM, Aiello LP, Antoszyk AN, Arnold-Bush B, Baker CW, Bressler NM, Browning DJ, Elman MJ, Ferris FL, Friedman SM, Melia M, Pieramici DJ, Sun JK, Beck RW. Aflibercept, bevacizumab, or ranibizumab for diabetic macular edema. N Engl J Med. 2015;372(13):1193–203.
- 22. Donaghue KC, Fung AT, Hing S, Fairchild J, King J, Chang A, Howard NJ, Silink M. The effect of prepubertal diabetes duration on diabetes. Microvascular complication in early and late adolescence. Diabetes Care. 1997;20(1):77–80.
- Early Treatment Diabetic Retinopathy Study Research Group. Photocoagulation for diabetic macular edema. Early treatment diabetic retinopathy Study Research Group report number 1. Arch Ophthalmol. 1985;103(12):1796–806.
- 24. EUCLID Study Group. Randomised placebo-controlled trial of lisinopril in normotensive patients with insulin-dependent diabetes and normoalbuminuria or microalbuminuria. Lancet. 1997;349(9068):1787–92.
- Flaxel CJ, Edwards AR, Aiello LP, et al. Factors associated with visual acuity outcomes after vitrectomy for diabetic macular edema: diabetic retinopathy clinical research network. Retina. 2010;30(9):1488–95.
- 26. Gallego-Pinazo R, Dolz-Marco R, Berrocal M, Wu L, Maia M, Serrano M, Alezzandrini A, Arévalo JF. Díaz-Llopis M; Pan-American Collaborative Retina Study Group (PACORES). Outcomes of cataract surgery in diabetic patients: results of the Pan American Collaborative Retina Study Group. Arq Bras Oftalmol. 2014;77(6):355–9.
- 27. Gerstein HC, Ambrosius WT, Danis R, Ismail-Beigi F, Cushman W, Calles J, Banerji M, Schubard U, Chew EY, for the ACCORD Study Group. Diabetic retinopathy, its progression, and incident cardiovascular events in the ACCORD trial. Diabetes Care. 2013;36:1266–71.
- Haller JA, Qin H, Apte RS, et al. on behalf of the Diabetic Retinopathy Clinical Research Network (DRCR.net). Vitrectomy outcomes in eyes with diabetic macular edema and vitreomacular traction. Ophthalmology. 2010;117(6):1087–93.
- 29. Haynie J. Obstructive sleep apnea and diabetic macular edema. Presented at Mountain West Optometric Council annual meeting. Las Vegas, NV. 2016.

- 30. Holl RW, Lang GE, Grabert M, Heinze E, Lang GK, Debatin M. Diabetic retinopathy in pediatric patients with type-1 diabetes: effect of diabetes duration, prepubertal and pubertal onset of diabetes, and metabolic control. J Pediatrics. 1998;132(5):790–4.
- International Consortium Investigating Early Vitrectomy in Diabetic Macular Edema Patients (ICV-DME) ClinicalTrials.gov Identifier: NCT02639507.
- Wu J, Xin J, Hong L, You J, Zheng N. New hierarchical approach for microaneurysms detection with matched filter and machine learning. Conf Proc IEEE Eng Med Biol Soc. 2015;2015:4322–5.
- 33. Keech AC, Mitchell P, Summanen PA, O'Day J, Davis TM, Moffitt MS, Taskinen MR, Simes RJ, Tse D, Williamson E, Merrifield A, Laatikainen LT, d'Emden MC, Crimet DC, O'Connell RL, Colman PG; FIELD study investigators. Effect of fenofibrate on the need for laser treatment for diabetic retinopathy (FIELD study): a randomised controlled trial. Lancet 2007;370:1687–1697.
- 34. King P, Peacock I, Donnelly R. The UK Prospective Diabetes Study (UKPDS): clinical and therapeutic implications for Type 2 diabetes. Br J Clin Pharmacol. 1999;48:643–8.
- 35. Kiss S, Berenberg TL. Ultra widefield fundus imaging for diabetic retinopathy. Curr Diab Rep. 2014;14(8):514.
- 36. Klein R, Klein B. The epidemiology of diabetic retinopathy. In: Ryan SJ, editor. Retina II. London: Elsevier; 2013. p. 907–24.
- 37. Klein R, Klein BE, Moss SE, Cruickshanks KJ. The Wisconsin Epidemiologic Study of Diabetic Retinopathy: XVII. The 14-year Incidence and Progression of Diabetic Retinopathy and Associated Risk Factor in Type 1 Diabetes. Ophthalmology. 1998;105(10):1801–15.
- Klein R, Klein BE, Moss SE, Davis MD, DeMets DL. The Wisconsin epidemiologic study of diabetic retinopathy. II. Prevalence and risk of diabetic retinopathy when age at diagnosis is less than 30 years. Arch Ophthalmol. 1984;102(4):520–6.
- 39. Klein R, Zinman B, Gardiner R, Suissa S, Donnelly SM, Sinaiko AR, Kramer MS, Goodyer P, Moss SE, Strand T, Mauer M. Renin-Angiotensin System Study. The relationship of diabetic retinopathy to preclinical diabetic glomerulopathy lesions in type 1 diabetic patients: the Renin-Angiotensin System Study. Diabetes. 2005;54(2):527–33.
- 40. Korobelnik JF, Do DV, Schmidt-Erfurth U, Boyer DS, Holz FG, Heier JS, Midena E, Kaiser PK, Terasaki H, Marcus DM, Nguyen QD, Jaffe GJ, Slakter JS, Simader C, Soo Y, Schmelter T, Yancopoulos GD, Stahl N, Vitti R, Berliner AJ, Zeitz O, Metzig C, Brown DM. Intravitreal aflibercept for diabetic macular edema. Ophthalmology. 2014;121(11):2247–54.
- Kozak I, Oster SF, Cortes MA, Dowell D, Hartmann K, Kim JS, Freeman WR. Clinical evaluation and treatment accuracy in diabetic macular edema using navigated laser photocoagulator NAVILAS. Ophthalmology. 2011;118(6):1119–24.
- 42. Liegl R, Langer J, Seidensticker F, et al. Comparative evaluation of combined navigated laser photocoagulation and intravitreal ranibizumab in the treatment of diabetic macular edema. PLoS One. 2014;26:e113981.
- 43. Lim LS, Ng WY, Mathur R, Wong D, Wong EY, Yeo I, Cheung CM, Lee SY, Wong TY, Papakostas TD, Kim LA. Conversion to aflibercept for diabetic macular edema unresponsive to ranibizumab or bevacizumab. Clin Ophthalmol. 2015:1715–8.
- Lueder GT, Silverstein J. Screening for retinopathy in the pediatric patient with type 1 diabetes mellitus. Pediatrics. 2005;116:270–3.
- Malone JI, Grizzard S, Espinoza LR, Archenbach KE, Van Cader TC. Risk factors for diabetic retinopathy in youth. Pediatrics. 1984;73(6):756–61.
- 46. Matsuda S, Tam T, Singh RP, Kaiser PK, Petkovsek D, Zanella MT, Ehlers JP. Impact of insulin treatment in diabetic macular edema therapy in type 2 diabetes. Can J Diabetes. 2015;39(1):73–7.
- 47. Maturi RK, Bleau L, Saunders J, Mubasher M, Stewart MW. A 12-month, single-masked, randomized controlled study of eyes with persistent diabetic macular edema after multiple anti-VEGF injections to assess the efficacy of the dexamethasone-delayed delivery system as an adjunct to bevacizumab compared with continued bevacizumab monotherapy. Retina. 2015;35(8):1604–14.

- 48. McCarty CA, Taylor KI, McKay R, Keeffe JE; Working group on Evaluation of NHMRC Diabetic Retinopathy Guidelines. Diabetic retinopathy: effects of national guidelines on the referral, examination and treatment practices of ophthalmologists and optometrists. Clin Experiment Ophthalmol. 2001;29:52–8.
- 49. Michaelides M, Kaines A, Hamilton RD, et al. A prospective randomized trial of intravitreal bevacizumab or laser therapy in the management of diabetic macular edema (BOLT Study). Ophthalmology. 2010;117:1078–86.
- 50. Mitchell P, Bandello F, Schmidt-Erfurth U, Lang GE, Massin P, Schlingemann RO, Sutter F, Simader C, Burian G, Gerstner O, Weischelberger A, on behalf of the RESTORE study group. The RESTORE Study. Ranibizumab monotherapy or combined with laser versus laser monotherapy for diabetic macular edema. Ophthalmology. 2011;118(4):615–25.
- 51. Mukamel BD, Bresnick GH, Wang Q, Dickey CF. Barriers to compliance with screening guidelines for diabetic retinopathy. Ophthalmic Epidemiology. 1999;6:61–72.
- Myers FL, Bresnick GH. Pars plana vitrectomy. Vitrectomy in diabetic retinopathy. Trans Sect Ophthalmol Am Acad Ophthalmol Otolaryngol. 1976;81(3 Pt 1):399–401.
- 53. Nguyen QD, Brown DM, Marcus DM, et al. Ranibizumab for diabetic macular edema: results from 2 phase III randomized trials: RISE and RIDE. Ophthalmology. 2012;119(4):789–801.
- Olk RJ. Modified grid argon (blue-green) laser photocoagulation for diffuse diabetic macular edema. Ophthalmology. 1986;93(7):938–50.
- 55. Palmer JJ, Chinanayi F, Gilbert A, et al. Trends and implications for achieving VISION 2020 human resources for eye health targets in 16 countries of sub-Saharan Africa by the year 2020. Hum Resour Health. 2014;12:45.
- Patz A, Finkelstein D, Fine SL, Murphy RP. The role of fluorescein angiography in national collaborative studies. Ophthalmology. 1986;93(11):1466–70.
- 57. Petrou P, Georgalas I, Giavaras G, Anastasiou E, Ntana Z, Petrou C. Early loss of pregnancy after intravitreal bevacizumab injection. Acta Ophthalmol. 2010;88(4):e136.
- Phase II Combination Steroid and Anti-VEGF for Persistent DME. https://clinicaltrials.gov/ ct2/show/NCT01945866?term=protocol+u&rank=2.
- 59. Prünte C, Fajnkuchen F, Mahmood S, Ricci F, Hatz K, Studnička J, Bezlyak V, Parikh S, Stubbings WJ, Wenzel A, Figueira J; and the RETAIN Study Group. Ranibizumab 0.5 mg treat-and-extend regimen for diabetic oedema: the RETAIN study. Br J Ophthalmol. 2015. Epub before print [PMID 26453639 doi:10.1136/bjophthalmol-2015-307249].
- 60. Raizada S, Al Kandari J, Al Diab F, Al Sabah K, Kumar N, Mathew S. Pars plana vitrectomy versus three intravitreal injections of bevacizumab for nontractional diabetic macular edema. A prospective, randomized comparative study. Indian J Ophthalmol. 2015;63(6):504–10.
- Rajalakshmi R, Arulmalar S, Usha M, et al. Validation of smartphone based retinal photography for diabetic retinopathy screening. PLoS One. 2015;10:e0138285.
- 62. Ranibizumab for treating diabetic macular oedema. https://www.nice.org.uk/guidance/ta274. Accessed 16 May 2016.
- 63. Rosenberg JB, Friedman IB, Gurland JE. Compliance with screening guidelines for diabetic retinopathy in a large academic children's hospital in the Bronx. J Diabetes Complications. 2011;25:222–6.
- 64. Ryan Jr EH, Han DP, Ramsay RC, Cantrill HL, Bennett SR, Dev S, Williams DF. Diabetic macular edema associated with glitazone use. Retina. 2006;26(5):562–70.
- 65. Sivaprasad S, Crosby-Nwaobi R, Heng LZ, Peto T, Michaelides M, Hykin P. Injection frequency and response to bevacizumab monotherapy for diabetic macular oedema (BOLT Report 5). Br J Ophthalmol. 2013;97(9):1177–80.
- 66. Sjølie AK, Klein R, Porta M, Orchard T, Fuller J, Parving HH, Bilous R. Chaturvedi N; DIRECT Programme Study Group. Effect of candesartan on progression and regression of retinopathy in type 2 diabetes (DIRECT-Protect 2): a randomised placebo-controlled trial. Lancet. 2008;372(9647):1385–93.
- Stitt AW, Lois N, Medina RJ, Adamson P, Curtis TM. Advances in our understanding of diabetic retinopathy. Clin Sci (Lond). 2013;125(1):1–17.

- 68. Stewart MW, Rosenfeld PJ, Penha FM, Wang F, Yehoshua Z, Bueno-Lopez E, Lopez PF. Pharmacokinetic rationale for dosing every 2 weeks versus 4 weeks with intravitreal ranibizumab, bevacizumab, and aflibercept (vascular endothelial growth factor Trap-eye). Retina. 2012;32(3):434–57.
- 69. Takamura Y, Tomomatsu T, Matsumura T, Arimura S, Gozawa M, Takihara Y, Inatani M. The effect of photocoagulation in ischemic areas to prevent recurrence of diabetic macular edema after intravitreal bevacizumab injection. Invest Ophthalmol Vis Sci. 2014 Jul 15;55(8):4741–6.
- 70. The ACCORD, Study Group ACCORD, Eye Study Group. Effects of medical therapies on retinopathy progression in type 2 diabetes. N Engl J Med. 2010;363:233–44.
- The Diabetic Retinopathy Vitrectomy Study Research Group. Early vitrectomy for severe vitreous hemorrhage in diabetic retinopathy. Two-year results of a randomized trial. Diabetic Retinopathy Vitrectomy Study report 2. Arch Ophthalmol. 1985;103(11):1644–52.
- 72. Totan Y, Güler E, Gürağaç FB. Dexamethasone Intravitreal Implant for Chronic Diabetic Macular Edema Resistant to Intravitreal Bevacizumab Treatment. Curr Eye Res. 2015;22:1–7.
- Treatment for CI-DME in eyes with very good VA Study (Protocol V). https://clinicaltrials. gov/ct2/show/NCT01909791?term=drcr.net+protocol+v&rank=1. Accessed 16 May 2016.
- 74. UK Prospective Diabetes Study Group. Tight blood pressure control and risk of macrovascular and microvascular complications in type 2 diabetes: UKPDS 38. Br Med J. 1998;317: 703–13.
- Verougstraete C, Toussaint D, De Scheppter J, Haentjens M, Dorchy H. First microangiographic abnormalities in childhood diabetes – types of lesions. Graefes Arch Clin Exp Ophthalmol. 1991;229(1):24–32.
- Walton 4th OB, Garoon RB, Weng CY, et al. Evaluation of Automated Teleretinal Screening Program for Diabetic Retinopathy. JAMA Ophthalmol. 2015;17:1–6.
- 77. Witkin SR, Klein R. Ophthalmologic care for persons with diabetes. JAMA. 1984;251: 2534–7.
- Wood EH, Karth PA, Moshfeghi DM, Leng T. Short-term outcomes of aflibercept therapy for diabetic macular edema in patients with incomplete response to ranibizumab and/or bevacizumab. Ophthalmic Surg Lasers Imaging Retina. 2015;46(9):950–4.
- 79. Writing Committee for the Diabetic Retinopathy Clinical Research Network, Gross JG, Glassman AR, Jampol LM, Inusah S, Aiello LP, Antoszyk AN, Baker CW, Berger BB, Bressler NM, Browning D, Elman MJ, Ferris FL 3rd, Friedman SM, Marcus DM, Melia M, Stockdale CR, Sun JK, Beck RW. Panretinal photocoagulation vs intravitreous ranibizumab for proliferative diabetic retinopathy. A randomized clinical trial. JAMA. 2015;314(20):2137–46.
- Yolcu U, Sobaci G. The effect of combined treatment of bevacizumab and triamcinolone for diabetic macular edema refractory to previous intravitreal mono-injections. Int Ophthalmol. 2014 Nov 26. [Epub ahead of print].
- Zhioua I, Semoun O, Lalloum F, Souied EH. Intravitreal dexamethasone implant in patients with ranibizumab persistent diabetic macular edema. Retina. 2015;35:1429–35.

Chapter 7 Vitreolysis: Targeting the Vitreoretinal Interface

7.1 Introduction

The vitreous humor is the most abundant ocular tissue, comprising the majority (4 ml) of the eye's volume. Often referred to as the "vitreous gel," a reference to both its consistency and biomechanical composition, the vitreous extends from the posterior lens capsule back to the retina's internal limiting membrane (ILM). For centuries, ophthalmologists believed that the vitreous could not be safely manipulated, altered, or removed without precipitating catastrophic consequences such as retinal tears, retinal detachments, suprachoroidal hemorrhages, or endophthalmitis. In retrospect, it has become obvious that physicians lacked a sufficient understanding of vitreous anatomy and physiology and did not possess the proper technology with which to safely and effectively handle the vitreous [60].

The gel-like consistency of the vitreous impedes the diffusion of substances such as vascular endothelial growth factor out of the retina and the movement of oxygen from anterior structures into the retina. The high viscosity of the vitreous (up to six times that of water) slows the resorption of hemorrhage that may result from insufficiently treated proliferative diabetic retinopathy (PDR), and it impedes the elimination of intravitreally administered drugs, thereby improving the effectiveness and duration of pharmacologic therapy. Trial designers recognize the importance of an intact vitreous since they routinely exclude eyes that have already undergone vitrectomy from participating in intravitreal drug trials.

An intact vitreous provides a scaffold for fibrovascular growth in various proliferative retinopathies. Vitreomacular traction (VMT) due to partial vitreous contraction and collapse without complete separation of the posterior hyaloid from the ILM upregulates VEGF production, promotes the formation of macular edema, and limits the efficacy of intravitreal pharmacotherapy.

In eyes with diabetic retinopathy (DR), the vitreous is viewed as both an enemy that contributes to the disease and as an ally that potentiates treatment by providing a reservoir for drug placement. Successfully changing the vitreous by inducing a vitreous detachment or removing most of it has become an important therapeutic approach for many retinal and vitreoretinal interface diseases. Pars plana vitrectomy with the removal of VMT has been used for two decades to treat diabetic macular edema (DME) (Chap. 3).

This chapter will discuss pharmacologic approaches to eliminate the adverse effects of VMT on the formation of diabetic retinopathy.

7.2 Vitreous Anatomy

The vitreous comprises about 80% of the ocular volume [82], and its transparent matrix is composed primarily (98–99%) of water [6]. The gel structure is maintained by a complex branching network of collagen fibrils held apart by hyaluronic acid and other macromolecules. Type II collagen is the most common vitreous protein with highest concentrations found at the vitreous base and in the posterior vitreous cortex [41]. The collagen fiber network gives the gel strength and stability, which enables it to absorb shock from blunt trauma and resume its previous shape after being subjected to distorting forces.

Hyaluronic acid, the most common glycosaminoglycan within the vitreous, stabilizes the collagen network [7, 78]. The hyaluronic acid-collagen matrix correctly spaces fibrils to optimize transparency and decrease light scatter, and endows the vitreous with viscoelastic properties [7, 8, 13]. The vitreous is nearly acellular except for a small number of hyalocytes, phagocytic cells found mostly at the vitreous base and posterior pole [4, 8, 78, 94]. Hyalocytes also synthesize collagen fibrils and hyaluronic acid.

The interface between the vitreous and adjacent structures is composed of vitreous cortical fibrils and the basal laminae of the other tissue [30]. Changes at the vitreous base usually result in retinal tears and detachments, but the interface between the posterior hyaloid and macula is more important in patients with DR. The ILM of the retina consists of the footplates or basal laminae of the Müller cells [79] together with types I and IV collagen, proteoglycans, fibronectin, and laminin [36, 47, 52, 53]. It is not clear which of these molecules must be dissolved to facilitate a posterior vitreous detachment, and even "linker molecules" like lectins, integrins, and chondroitin sulfate may ultimately be responsible for the attachment of the vitreous cortex to the ILM. The ILM varies in thickness from 50 μ m at the vitreous base to 300 μ m at the equator to 1890 μ m over the posterior pole before it thins to only 10–20 μ m at the fovea [25, 30, 37]. The firmest vitreous adhesions occur where the ILM is thinnest – at the vitreous base and the fovea.

7.3 Posterior Vitreous Detachment

The vitreous both liquefies and aggregates as it ages. Vitreous liquefaction, or syneresis, begins at the age of 4 years, whereas aggregation of fibrils, or synchysis, begins in midlife and continues into old age (Fig. 7.1) [48, 49]. Changes in



Fig. 7.1 This swept-source optical coherence tomography image shows excellent detail in the vitreous, retina, and choroid. The gel-like consistency of the vitreous appears hyperreflective (*top of image*). A large preretinal bursa (posterior to the "PH") due to liquefaction of the vitreous is present over the macula. *PH* posterior hyaloid, *NFL* nerve fiber layer, *GCL* ganglion cell layer, *IPL* inner plexiform layer, *INL* inner nuclear layer, *OPL* outer plexiform layer, *ONL* outer nuclear layer, *ELM* external limiting membrane, *IS* inner segment/outer segment line, *VM* cone outer segments, *RPE* retinal pigment epithelium, *CH* choriocapillaris, *SL* Sattler's layer, *HL* Haller's layer, *LS* junction between choroid and sclera

hyaluronic acid and its interaction with the collagen fibrils create areas of liquefaction that alternate with areas of increased collagen fiber concentration within the residual gel [45, 59]. Liquefaction in the central vitreous overlying the macula enables incident light to produce free radicals that weaken the residual collagen and decrease the concentrations of glycosaminoglycans and chondroitin sulfate [51]. This promotes posterior hyaloid separation from the ILM (posterior vitreous detachment).

Most posterior vitreous detachments (PVDs) result from a combination of vitreous synchysis and weakening of the adhesions between the ILM and posterior hyaloid. Liquefied premacular vitreous slowly migrates into the potential space between the posterior hyaloid and ILM, and since the adhesions have been weakened, the two surfaces slowly begin to separate, beginning in the peripheral macula (Fig. 7.2). Saccadic eye movements and continued loss of adherent molecules cause further cleavage with progressive enlargement of the preretinal space that marches toward the fovea. A final collapse of the vitreous gel with complete separation of the posterior hyaloid from the fovea and optic disc characterizes a PVD [22, 58, 77].

7.4 Anomalous Posterior Vitreous Detachment

Incomplete separation of the posterior hyaloid because of residual vitreomacular adhesion constitutes an anomalous PVD [84]. If the residual vitreous adhesion causes distortion of the macula by elevating the ILM, this is termed vitreomacular









traction. Excessive gel liquefaction is seen in many systemic conditions including diabetes [80] and together with strong vitreomacular adhesion may cause symptomatic vitreomacular traction or even macular holes. In patients with diabetic retinopathy, this can cause diabetic macular edema (DME).

Diagnosing an anomalous vitreous detachment can sometimes be done with slitlamp biomicroscopy and a magnifying fundus lens (Fig. 7.3), but spectral domain optical coherence tomography (SD-OCT) is an easier and more accurate method of visualizing the partially detached posterior hyaloid [31, 54, 75, 97]. Correctly differentiating between a complete PVD and a completely attached posterior hyaloid can be difficult with OCT, and ultrasound is a more accurate tool to discriminate between these two conditions.

7.5 Association Between Posterior Hyaloid and Diabetic Retinopathy

Several studies have investigated the association between integrity of the vitreous and the development and progression of DR. Retinal vessels are normally excluded from the vitreous because development of neovascularization on the retinal surface – proliferative diabetic retinopathy (PVR) – requires a scaffold of collagenous material [21]. Elevated glucose levels in the vitreous induce the formation of advanced glycation end products [85] and promote cross-linking of collagen and the development of excess vitreous liquefaction, without promoting dehiscence of the posterior cortical vitreous. Intravitreal proteins that normally inhibit the development of neovascularization may lose this ability because they undergo nonenzymatic glycation [38, 81]. In eyes with DR, growth factors that are synthesized in the retina diffuse into the vitreous and encourage the development of neovascularization [33, 34, 61]. Oxygen transport from the ciliary body to the inner retina may be impeded, which worsens retinal ischemia and further increases the synthesis of growth factors.

Diabetes-mediated breakdown of the blood-retinal barrier leads to an extracellular accumulation of serum proteins, which increases the concentrations of fibronectin and other chemokines at the vitreoretinal surface by tenfold [53]. These molecules stimulate the migration and adhesion of proliferating vascular endothelial cells [9] and fibroblasts to the posterior hyaloid. Contraction of these cells, in the presence of an anomalous PVD, exerts tangential traction on the retina, further upregulates VEGF, and exacerbates DME [70]. Traction also decreases interstitial pressure, which, according to Starling's Law, promotes passage of fluid out of the vascular lumens [15].

A complete PVD should provide some protection against the development of neovascularization [65, 66], but since islands of residual cortical vitreous remain attached to the ILM, such protection is generally incomplete. Five studies that included over 2000 eyes have looked at the association of PDR and BDR development according to the status of the posterior hyaloid [2, 46, 69, 90, 91]. A pooled analysis of these data found that eyes with complete PVDs had a significantly lower prevalence of PDR (OR 0.1, 95% CI, 0.05–0.18), and the development of a PVD is frequently followed by resolution of DME [42, 100]. Two of the studies found that eyes with partial PVDs had six times the progression rate as those without a PVD and 15 times the progression rate as those with complete PVDs [2, 46, 69, 90, 91].

Removal of the posterior hyaloid may help resolve DME by relieving traction and releasing trapped preretinal growth factors. A liquefied vitreous may accelerate the removal of vascular endothelial growth factor from the inner retina. When surgeons propose treatments for DR that involve the induction of a PVD, they need to remember that a partial PVD may actually worsen the retinopathy.

7.6 Treatment of Vitreomacular Traction

Spectral domain OCT imaging has demonstrated that vitreomacular adhesion (VMA) occurs in the majority of patients over the age of 50 years. Vitreomacular adhesion that distorts the inner retinal contour is termed vitreomacular traction (Fig. 7.4). Macular changes may be limited to loss of the foveal depression, to macular splitting or schisis, or to traction foveal detachment. Patients usually experience decreased central visual function though some may be asymptomatic. The natural history of this process is highly variable, as some eyes achieve complete vitreoretinal separation, others progress to partial or full-thickness macular holes, and a third group remains stable for years.

Creating a complete PVD to relieve an anomalous partial vitreous detachment has been a topic of considerable interest to retinal surgeons over the last 5 years. Surgical intervention for eyes with symptomatic VMT can be done in two ways. Pars plana vitrectomy removes the core vitreous, after which the posterior hyaloid is detached with aspiration. Despite what appears to be complete removal of the posterior cortical vitreous, islands of cortex frequently remain attached to the ILM. These may sequester growth factors and promote fibrovascular growth. Many surgeons perform a dye-assisted peeling of the ILM to remove all vitreous fragments and minimize the chance of postoperative angiogenesis or traction.

Vitrectomy with ILM removal effectively and predictably removes traction but not without associated risks. Many phakic patients develop cataracts within 2 years that require removal. The risks of retinal tears, retinal detachments, and endophthalmitis are relatively low but still of concern. If the adhesive forces between the posterior cortical vitreous and the ILM are sufficiently strong, vitrectomy can convert VME into a full-thickness macular hole.

Pharmacologic vitreolysis involves the use of drugs that liquefy the vitreous (synchysis) and weaken the adhesion between the posterior hyaloid and the ILM [83]. Intravitreal hyaluronidase was used as early as 1946 [72] and collagenase was used in 1973 [67]. Chondroitinase, dispase, plasmin, and tissue plasminogen

Fig. 7.4 This optical coherence tomography scan shows that the partially detached posterior hyaloid is exerting traction on the inner limiting membrane. The traction causes splitting of the inner retina

Fig. 7.5 The top optical coherence tomography scan shows vitreomacular traction in a symptomatic patient with visual acuity of 20/25. One month after the intravitreal injection of ocriplasmin (middle scan), the contour of the internal limiting membrane has inverted, and the splitting of the inner retina has diminished. One year after the ocriplasmin injection (bottom scan), the macula appears normal and the posterior hyaloid has completely separated



activator (tPA) have all been tested in animals and used sporadically in humans [67, 71, 93, 99]. Results with intravitreal hyaluronidase and dispase were disappointing because of insufficient vitreoretinal separation or partial digestion of the inner retina [32, 44, 50, 55, 56, 68, 93]. Only the fibrinolytic enzymes plasmin and tissue plasminogen activator (tPA) showed some success in PVD induction when injected into human eyes [3, 12, 17, 57, 63, 95]. Successful induction of PVDs has been achieved after intravitreal injections of ocriplasmin (Jetrea®, Thrombogenics, Leuven, Belgium) into eyes with symptomatic vitreomacular traction (Fig. 7.5) or stage 2 macular holes (Fig. 7.6). Eyes that responded best to ocriplasmin had the following characteristics: age >65 years of age, vitreomacular adhesion <1500 μ m, phakic lens status. and the absence of epiretinal membrane. Macular holes of <400 μ m diameter with persistent vitreomacular adhesion also responded well to ocriplasmin.

Many surgeons believe that pharmacologic vitreolysis produces a more complete and less traumatic PVD than does surgery [3]. Vitreolysis trials with ocriplasmin and other pharmacologic agents in patients with DME are underway. Table 7.1 lists the drugs that have been used to induce posterior vitreous detachments in animals and humans. Fig. 7.6 The top scan shows a full-thickness macular hole with persistent traction on an inner retinal flap. One month after the intravitreal injection of ocriplasmin (*middle scan*), the hole has closed, but the photoreceptor layer is still separated from the retinal pigment epithelium (RPE). Nine months after the intravitreal injection of ocriplasmin (bottom scan), the photoreceptors have moved closer to the RPE



 Table 7.1
 The drugs that have been investigated for the induction of posterior vitreous detachment (PVD)

Drugs used to induce posterior vitreous detachments				
Drug	Characteristics	Clinical data		
Streptokinase	 Binds to and activates plasminogen Used to dissolve clots causing myocardial infarctions and pulmonary emboli 	 Intravitreal injections used to dissolve post-vitrectomy fibrin Used to induce PVDs in rabbit eyes Pilot study showed creation of PVDs in humans 		
Hyaluronidase	 Cleaves glycosidic bonds of hyaluronic acid Mimics liquefaction by lowering vitreous viscosity 	 Mixed ability to induce PVDs in rabbits Caused retinal necrosis in rabbits and monkeys 		
Nattokinase	 Activates plasminogen activator and inactivates an inhibitor Induces PVDs in rabbit eyes Causes retinal hemorrhages an changes in electroretinogram 			
Chondroitinase	 Degrades chondroitin sulfate and depolymerases hyaluronic acid Has liquefactant and interfactant properties 	• Detached posterior hyaloid in monkeys after 5 to 15 min		

Drugs used to induce posterior vitreous detachments				
Drug	Characteristics	Clinical data		
Plasmin	 Nonspecific protease from plasminogen activation No activity against type IV collagen 	 In DR pilot study (7 eyes), vitreoretinal interface weakened PVD induced in 10 of 25 DME eyes with improved VA and CRT PVD induced in 4 of 12 eyes; most had improved VA 		
Tissue plasminogen activator (tPA)	Catalyzes conversion of plasminogen to plasmin	 Induces PVDs in rabbits PVDs reported in case studies and retrospective series Prospective, randomized trial failed to show development of PVD 		
Autologous plasmin enzyme (APE)	• Easier and less expensive to isolate than plasmin	 85% of eyes found to have PVD 1 h after injection Compared to controls, decreased DME and improved VA after 1 month 		
Vitreosolve	Nonenzymatic agent comprised of urea	• Phase III trial failed to show difference between vitreosolve and controls		
Ocriplasmin	 Truncated version of plasmin Acts on several molecules at vitreoretinal interface 	 Phase III MIVT trials, 26.5% with vitreomacular traction achieved PVD Fewer than 20% of DME eyes achieved vitreoretinal separation in MIVI-II Phase III trial for DME currently underway 		

Table 7.1	(continued)
-----------	-------------

Several drugs have appeared promising, but only ocriplasmin has been fully developed and approved for the treatment of vitreomacular traction. A phase III trial is currently evaluating the efficacy of ocriplasmin in eyes with diabetic macular edema.

DR diabetic retinopathy, VA visual acuity, CRT central retinal thickness, DME diabetic macular edema

7.6.1 Streptokinase

Streptokinase, an enzyme produced by several *Streptococci* sp., binds to and activates human plasminogen. Streptokinase is a fibrinolytic that is used to dissolve clots in patients with acute myocardial infarctions [86] and pulmonary emboli [63]. No adverse effects have been noted after the intravitreal injection of 1000 IU.

Intravitreal streptokinase has been used to dissolve post-vitrectomy fibrin in patients with advanced PDR [11]. Posterior vitreous detachments have been induced in rabbit eyes after the injection of 1500 IU, and only mild toxic effects on the retina have been noted [19]. Pilot studies have been performed to evaluate the efficacy of streptokinase-plasmin injections into human eyes for the induction of PVDs [14].

Though the drug has demonstrated potential for vitreolysis, further studies are required to establish efficacy.

7.6.2 Hyaluronidase

Hyaluronidase cleaves glycosidic bonds of hyaluronic acid and other glycosaminoglycans. This mimics liquefaction by lowering the viscosity of the vitreous. Commercially available hyaluronidase includes a bovine testicular protein enzyme (Wydase®), which has been used in ophthalmology for over 40 years as a spreading or diffusing agent with local anesthetics.

Hyaluronidase has been used in experimental intravitreal models, but its ability to cause a PVD remains unclear. Intravitreal bovine hyaluronidase has produced mixed results in rabbits [35, 44], but retinal necrosis has been noted in both rabbit and monkey eyes [29, 36]. Hyaluronidase has been suggested as a drug to induce a PVD, but since it does not weaken the vitreoretinal interface, there is little evidence that it would work.

7.6.3 Nattokinase

Nattokinase is a 275-amino acid serine protease produced by *Bacillus subtilis* [64]. Nattokinase has potent fibrinolytic activity as a plasminogen activator and by inactivating a plasminogen activator inhibitor, and it hydrolyzes collagen fibrils [88, 89]. Nattokinase induces a PVD in rabbit eyes 30 min after injection but also causes retinal hemorrhage and ERG changes [92].

7.6.4 Chondroitinase

Chondroitinase degrades chondroitin sulfate and depolymerases hyaluronic acid. A low dose failed to induce PVD in pigs but higher doses detached the posterior hyaloid in monkeys within 5 to 15 min [32]. Chondroitinase remains an attractive molecule for vitreolysis because of both its liquefactant and interfactant properties, but much more studies need to be done to determine its clinical efficacy.

7.6.5 Plasmin

Plasmin is a nonspecific protease resulting from plasminogen activation. Plasmin mediates fibrinolysis and also acts on numerous glycoproteins such as laminin and fibronectin. Plasmin is not believed to alter the intact ILM because it has no activity

on type IV collagen. Intravitreal plasmin can create posterior vitreous detachments (dose dependent) in rabbit and pigs [23, 40, 43], and 0.4 IU can induce complete PVDs in postmortem human eyes [24]. Intravitreal autologous plasmin injections have been used as primary or adjunctive therapy for the treatment of pediatric macular holes [62] and advanced diabetic retinopathy [7–9]. It can be prepared in vitro by hydrolyzation with urokinase.

In a pilot study, seven eyes with advanced diabetic retinopathy (six with traction retinal detachment and one with treatment resistant DME) received intravitreal plasmin. The vitreoretinal interface was weakened in all eyes leading to either spontaneous PVD or easier surgical dissection. All eyes achieved resolution of DME without the need for additional laser [99].

In a prospective interventional case study, 25 eyes with clinically significant DME and vitreomacular traction received 0.2 IU/0.2 ml of autologous plasmin. PVD was achieved in ten (41.3%) eyes – complete in six and partial in four. All ten eyes had vitreous separation from the fovea. Visual acuity improved by one Snellen line in four eyes. The mean foveal thickness improved from 480 to 226 μ m (*P* = 0.05) [20].

A prospective study evaluated the efficacy of intravitreal plasmin in 12 eyes with SD-OCT-diagnosed VMT that were injected over the course of 4 years. Five eyes were injected once and seven eyes each received three injections. Four eyes (33%) developed PVDs, two after single injections and two after multiple injections. Central macular thickness improved significantly (P = 0.016), and seven eyes experienced BCVA improvement of at least one line (P = 0.017). The only complication was an immediate elevation in intraocular pressure in one eye [74].

7.6.6 Tissue Plasminogen Activator (tPA)

Tissue plasminogen activator, a serine protease found on the surface of endothelial cells, catalyzes the conversion of plasminogen to plasmin. tPA was first reported to induce a PVD in 1995 [39], and several subsequent reports have described its mechanism of action – plasma activation that degrades the extracellular matrix proteins [25, 43, 96]. PVDs were noted in all rabbit eyes treated with intravitreal tPA, and retrospective series and case studies have reported the induction of PVD following tPA [28]. A prospective, randomized, case-control trial, however, failed to show surgical benefits after the intravitreal injection of 25 μ g of tPA 15 min prior to vitrectomy [57].

7.6.7 Autologous Plasmin Enzyme (APE)

Autologous plasmin enzyme is easier and less expensive to isolate than is plasmin and has been used as an adjuvant during vitrectomy [99]. During vitrectomy, 85% of eyes were found to have PVDs after they had received intravitreal APE 1 h prior to surgery [73]. There appears to be a direct correlation between the exposure time of the posterior hyaloid to plasmin and the development of vitreoretinal separation in pig eyes [26].

The efficacy of 0.2 ml intravitreal APE was tested in eyes with DME that had failed prior laser photocoagulation. Compared to the fellow control eye, intravitreal APE led to decreased macular edema and improved vision 1 month later [18].

7.6.8 Vitreosolve

Vitreosolve (Vitreoretinal Technologies, Inc.) is a nonenzymatic agent comprised of urea. It has been reported to "unravel" collagen to produce both liquefaction and vitreoretinal separation. Phase III trials failed to detect a clinically significant difference in PVD rates between controls and eyes treated with vitreosolve. As a result, subsequent development of vitreosolve has been discontinued.

7.6.9 Ocriplasmin

Microplasmin is a truncated version of plasmin, with retained protease activity and considerably more stability [74]. At doses of 125 μ g, microplasmin induces PVDs in pig, cat, and human eyes and leaves a smooth surface on the ILM [15, 25, 76]. Studies in rabbit eyes showed no histologic toxicity and only transient decreases in electroretinographic a- and b-wave amplitudes [25, 98].

Ocriplasmin, a recombinant version of microplasmin, has been approved for the enzymatic lysis of persistent vitreomacular adhesions. Ocriplasmin acts on several molecules within the vitreoretinal interface – laminectin, fibronectin, and type IV collagen [10]. Like other serine proteases, ocriplasmin is highly autolytic and behaves according to a second-order pharmacokinetic profile that results in a very short intravitreal half-life [1, 16]. Vitreous gel liquefaction occurs after injection, but the total effect depends upon the baseline status, since ocriplasmin must diffuse through the gel to achieve optimum effect.

The prospective, uncontrolled MIVI trial reported that 25 ug to 125 ug doses of ocriplasmin induced posterior vitreous separation when given to eyes with VMT prior to vitrectomy [14]. A subsequent phase II dose-ranging trial reported that ocriplasmin induced a PVD in 31% of eyes with VMT or macular holes [5].

In the two randomized, phase III registration trials (MIVI-006 and MIVI-007), patients with symptomatic VMT were randomized to receive ocriplasmin or sham. At the 4-week primary temporal endpoint, 26.5% of ocriplasmin patients achieved a PVD (compared to 10.1% of sham patients; P < 0.001), and this did not change with follow-up through 6 months. Of patients with successful VMT resolution, 41.1% achieved BCVA improvements of at least two lines by 6 months. For the subgroup with full-thickness macular holes, the closure rate at 28 days was 40.6%

(compared to 10.6% of sham patients; P < 0.001). Results of the phase III registration trials indicate that single intravitreal injections of ocriplasmin are indicated in patients with symptomatic vitreomacular traction and small diameter macular holes with persistent vitreoretinal traction.

Despite the encouraging findings of the MIVI trials, a majority of patients fail to achieve successful PVD with single injections of ocriplasmin. Since a partial PVD may actually increase the risk of PDR development in patients with DR, a strategy to increase the success rate and control complications in those who fail pharmacologic vitreolysis would be needed in these patients. As discussed earlier in this chapter, PVD appears to decrease the incidence and severity of DME so pharmacologic vitreolysis in patients with DR may be beneficial.

Patients with diabetes were excluded from the phase III MIVI-006 and MIVI-007 registration trials, but they were included in other published trials [14, 87]. No DME studies with ocriplasmin have yet been published, and the closest has been the plasmin study discussed previously. Fewer than 20% of eyes with DME that received intravitreal microplasmin in the MIVI-II trial developed PVDs.

7.7 Conclusions

There is no current indication for vitreolysis in eyes with DME, but phase II clinical trials with ocriplasmin are in the planning stages. The area of vitreoretinal adhesion is frequently greater in eyes with DME than in those studied for VMT in the MIVI-006 and MIVI-007 trials perhaps making successful vitreolysis rates with single intravitreal injections more uncertain. Plasmin injections to induce PVDs in eyes with DME appeared promising but plasmin is difficult to prepare. Since ocriplasmin and plasmin have the same enzymatic profile and ocriplasmin should diffuse better (due to its smaller molecular weight), results with ocriplasmin may be as good as those reported with plasmin. Short-term visual acuity results after enzymatic vitreolysis in eyes with DME appear good, but long-term results are still unknown. The MIVI-006 and MIVI-007 prohibited repeated intravitreal injections, but a small number of eyes from the phase II MIVI-IIT trial that failed to respond to single intravitreal microplasmin injections received additional (up to three) injections. Two of seven eyes that ultimately achieved PVD and had no adverse events, such as lens dislocation, due to the repeated injections were reported [87].

A phase II randomized, double-masked, sham-controlled, multicenter study (CIRCLE Trial) will evaluate the efficacy and safety of up to three intravitreal injections of either 0.125 or 0.0625 mg of ocriplasmin in 230 patients with moderately severe to very severe NPDR. The primary endpoint is the proportion of eyes with total PVD by the 3-month visit. Several secondary endpoints will evaluate ocriplasmin's potential for reducing the risk of progression of NPDR to PDR. Preliminary results are expected in late 2017.

References

- 1. Aerts F, Noppen B, Fonteyn L, et al. Mechanism of inactivation of ocriplasmin in porcine vitreous. Biophys Chem. 2012;165:30–8.
- Akiba J, Arzabe CW, Trempe CL. Posterior vitreous detachment and neovascularization in diabetic retinopathy. Ophthalmology. 1990;97:889–91.
- Azzolini C, D'angelo A, Maestranzi G, et al. Intrasurgical plasmin enzyme in diabetic macular edema. Am J Ophthalmol. 2004;138:560–6.
- Balazs EA, Toth LZ, Eckl EA, Mitchell AP. Studies on the structure of the vitreous body. XII. Cytological and histochemical studies on the cortical tissue layer. Exp Eye Res. 1964;3:57–71.
- Benz MS, Packo KH, Gonzalez V, Pakola S, Bezner D, Haller JA, Schwartz SD. A placebocontrolled trial of microplasmin intravitreous injection to facilitate posterior vitreous detachment before vitrectomy. Ophthalmology. 2010;117(4):791–7.
- 6. Bishop PN. The biochemical structure of the mammalian vitreous. Eye. 1996;10:664–70.
- 7. Bishop PN. Structural macromolecules and supramolecular organization of the vitreous gel. Prog Retin Eye Res. 2000;19:323–44.
- Bishop PN, Holmes DF, Kadler KE, McLeod D, Bos JK. Age-related changes on the surface of vitreous collagen fibrils. Invest Ophthalmol Vis Sci. 2004;45:1041–6.
- Casaroli Marano RP, Preissner KT, Vilaro S. Fibronectin, laminin, vitronectin and their receptors at newly-formed capillaries in proliferative diabetic retinopathy. Exp Eye Res. 1995;60(1):5–17.
- Chen WL, Mo W, Sun K, et al. Microplasmin degrades fibronectin and laminin at vitreoretinal interface and outer retina during enzymatic vitrectomy. Curr Eye Res. 2009;34(12):1057–64.
- 11. Cherfan GM, el Maghraby A, Tabbara KF, Nasr Y, Hassan H. Dissolution of intraocular fibrinous exudate by streptokinase. Ophthalmology. 1991;98(6):870–4.
- 12. Chow DR, Williams GA, Trese MT, et al. Successful closure of traumatic macular holes. Retina. 1999;19:405–9.
- Comper WD, Laurent TC. Physiological function of connective tissue polysaccharides. Physiol Rev. 1978;58:255–315.
- 14. de Smet MD, Gandorfer A, Stalmans P, Veckeneer M, Feron E, Pakola S, Kampik A. Microplasmin intravitreal administration in patients with vitreomacular traction scheduled for vitrectomy: the MIVI I trial. Ophthalmology. 2009;116(7):1349–55.
- de Smet MD, Valmaggia C, Zarranz-Ventura J, Willekens B. Microplasmin: ex vivo characterization of its activity in porcine vitreous. Invest Ophthalmol Vis Sci. 2009;50(2):814–9.
- de Smet MD, Jonckx B, Vanhove M, et al. Pharmacokinetics of ocriplasmin in vitreous. Invest Ophthalmol Vis Sci. 2012;53:8208–13.
- Diaz-llopis M, Udaondo P, Garcia-delpech S, et al. Enzymatic vitrectomy by intravitreal autologous plasmin injection as initial treatment for diffuse diabetic macular edema. Arch Soc Esp Oftalmol. 2008;83:77–84.
- Diaz-Llopis M, Udaondo P, Arevalo F, Salom D, Garcia-Delpech S, Quijada A, Romero FJ. Intravitreal plasmin without associated vitrectomy as a treatment for refractory diabetic macular edema. J Ocul Pharmacol Ther. 2009;25(4):379–84.
- El Baha SM, Abou-Nazel MW, Idriss HF, Abdel-Megeed AS. The role of streptokinase in induction of posterior vitreous detachment: a scanning and transmission electron microscopic study of the retina in rabbits. Retina. 2003;23(5):698–704.
- Elbendary AM, Elwan MM, Azzam HA, Eldeeb DR. Predictability of vitreous detachment following intravitreal plasmin injection in diabetic macular edema associated with vitreomacular traction. Curr Eye Res. 2011;36(6):534–9.
- Faulborn J, Bowald S. Microproliferations in proliferative diabetic retinopathy and their relationship to the vitreous: corresponding light and electron microscopic studies. Graefes Arch Clin Exp Ophthalmol. 1985;223(3):130–8.

- Foos RY, Kreiger AE, Forsythe AB, Zakka KA. Posterior vitreous detachment in diabetic subjects. Ophthalmology. 1980;87:122–8.
- Gandorfer A, Putz E, Welge-Luben U, et al. Ultrastructure of the vitreoretinal interface following plasmin assisted vitrectomy. Br J Ophthalmol. 2001;85:6–10.
- Gandorfer A, Priglinger S, Schebitz K, Hoops J, Ulbig M, Ruckhofer J, Grabner G, Kampik A. Vitreoretinal morphology of plasmin-treated human eyes. Am J Ophthalmol. 2002;133(1):156–9.
- Gandorfer A, Rohleder M, Sethi C, Eckle D, Welge-Lussen U, Kampik A, Luthert P, Charteris D. Posterior vitreous detachment induced by microplasmin. Invest Ophthalmol Vis Sci. 2004;45:641–7.
- Gandorfer A, Kampik A. Enzyme-assisted vitrectomy in enucleated pig eyes. Curr Eye Res. 2005;30(10):821–2.
- Gaucher D, Tadayoni R, Erginay A, Haouchine B, Gaudric A, Massin P. Optical coherence tomography assessment of the vitreoretinal relationship in diabetic macular edema. Am J Ophthalmol. 2005;139:807–13.
- Glacet-Bernard A, Kuhn D, Vine AK, Oubraham H, Coscas G, Soubrane G. Treatment of recent onset central retinal vein occlusion with intravitreal tissue plasminogen activator: a pilot study. Br J Ophthalmol. 2000;84(6):609–13.
- Gottlieb JL, Antoszyk AN, Hatchell DL, Saloupis P. The safety of intravitreal hyaluronidase. A clinical and histologic study. Invest Ophthalmol Vis Sci. 1990;31(11):2345–52.
- 30. Grabner G, Boltz G, Förster O. Macrophage like properties of human hyalocytes. Invest Ophthalmol Vis Sci. 1980;19:333–40.
- Gupta P, Sadun AA, Sebag J. Multifocal retinal contraction in macular pucker analyzed with combined optical coherence tomography/scanning laser ophthalmoscopy. Retina. 2008;28:447–52.
- Hageman GS, Russell SR. Chondroitinase-mediated disinsertion of the primate vitreous body. Invest Ophthalmol Vis Sci. 1994;35(4):1260.
- 33. Hammes HP, Lin J, Bretzel RG, Brownlee M, Breier G. Upregulation of the vascular endothelial growth factor/vascular endothelial growth factor receptor system in experimental background diabetic retinopathy of the rat. Diabetes. 1998;47(3):401–6.
- 34. Hammes HP, Wellensiek B, Kloting I, Sickel E, Bretzel RG, Brownlee M. The relationship of glycaemic level to advanced glycation end-product (AGE) accumulation and retinal pathology in the spontaneous diabetic hamster. Diabetologia. 1998;41(2):165–70.
- Harooni M, McMillan T, Refojo M. Efficacy and safety of enzymatic posterior vitreous detachment by intravitreal injection of hyaluronidase. Retina. 1998;18(1):16–22.
- 36. Heegaard S, Jensen OA, Prause JU. Structure and composition of the inner limiting membrane of the retina. SEM on frozen resin-cracked and enzyme-digested retinas of Macaca mulatta. Graefes Arch Clin Exp Ophthalmol. 1986;224(4):355–60.
- 37. Heegaard S, Jensen OA, Prause JU. Structure of the vitread face of the monkey optic disc (Macaca mulatta). SEM on frozen resin cracked optic nerve heads supplemented by TEM and immunohistochemistry. Graefes Arch Clin Exp Ophthalmol. 1988;226:377–83.
- Hendrikse F, Yeo KT. Role of the vitreous body in diabetic retinopathy. Klin Monbl Augenheilkd. 1993;203(5):319–23.
- Hesse L, Chofflet J, Kroll P. Tissue plasminogen activator as a biochemical adjuvant in vitrectomy for proliferative diabetic vitreoretinopathy. Ger J Ophthalmol. 1995;4:323–7.
- Hesse L, Kroll P. Enzymatically induced posterior vitreous detachment in proliferative diabetic vitreoretinopathy. Klin Monbl Augenheilkd. 1999;214:84–9.
- Hikichi T, Akiba J, Ueno N, Yoshida A, Chakrabarti B. Cross-linking of vitreous collagen and degradation of hyaluronic acid induced by bilirubin-sensitized photochemical reaction. Jpn J Ophthalmol. 1997;41:154–9.
- 42. Hikichi T, Fujio N, Akiba J, Azuma Y, Takahashi M, Yoshida A. Association between the short-term natural history of diabetic macular edema and the vitreomacular relationship in type II diabetes mellitus. Ophthalmology. 1997;104:473–8.

- Hikichi T, Yanagiya N, Kado M, et al. Posterior vitreous detachment induced by injection of plasmin and sulfur hexafluoride in the rabbit vitreous. Retina. 1999;19:55–8.
- 44. Hikichi T, Kado M, Yoshida A. Intravitreal injection of hyaluronidase cannot induce posterior vitreous detachment in the rabbit. Retina. 2000;20(2):195–8.
- 45. Itakura H, Kishi S. Aging changes of vitreomacular interface. Retina. 2011;31:1400-4.
- 46. Jalkh A, Takahashi M, Topilow HW, Trempe CL, McMeel JW. Prognostic value of vitreous findings in diabetic retinopathy. Arch Ophthalmol. 1982;100:432–4.
- Jerdan J, Kao L-Y, Glaser B. The inner limiting membrane: a modified basement membrane? Invest Ophthalmol Vis Sci. 1986;27(Suppl):230.
- Johnson MW. Posterior vitreous detachment: evolution and complications of its early stages. Am J Ophthalmol. 2010;149:371–82.
- Johnson MW. How should we release vitreomacular traction: surgically, pharmacologically, or pneumatically? Am J Ophthalmol. 2013;155:203–5.
- Jorge R, Oyamaguchi EK, Cardillo JA, Gobbi A, Laicine EM, Haddad A. Intravitreal injection of dispase causes retinal hemorrhages in rabbit and human eyes. Curr Eye Res. 2003;26(2):107–12.
- Kamei A, Totani A. Isolation and characterization of minor glycosaminoglycans in the rabbit vitreous body. Biochem Biophys Res Commun. 1982;109:881–7.
- 52. Kohno T, Sorgente N, Goodnight R, Ryan SJ. Alterations in the distribution of fibronectin and laminin in the diabetic human eye. Invest Ophthalmol Vis Sci. 1987;28:515–21.
- 53. Kohno T, Sorgente N, Ishibashi T, Goodnight R, Ryan SJ. Immunofluorescent studies of fibronectin and laminin in the human eye. Invest Ophthalmol Vis Sci. 1987;28:506–14.
- 54. Koizumi H, Spaide RF, Fisher YL, Freund KB, Klancnik Jr JM, Yannuzzi LA. Three dimensional evaluation of vitreomacular traction and epiretinal membrane using spectral-domain optical coherence tomography. Am J Ophthalmol. 2008;145:509–17.
- 55. Kuppermann BD, Thomas EL, de Smet MD, Grillone LR. Pooled efficacy results from two multinational randomized controlled clinical trials of a single intravitreous injection of highly purified ovine hyaluronidase (Vitrase) for the management of vitreous hemorrhage. Am J Ophthalmol. 2005;140(4):573–84.
- 56. Kuppermann BD, Thomas EL, de Smet MD, Grillone LR. Safety results of two phase III trials of an intravitreous injection of highly purified ovine hyaluronidase (Vitrase) for the management of vitreous hemorrhage. Am J Ophthalmol. 2005;140(4):585–97.
- Le Mer Y, Korobelnik JF, Morel C, Ullern M, Berrod JP. TPA-assisted vitrectomy for proliferative diabetic retinopathy: results of a double-masked, multicenter trial. Retina. 1999;19(5):378–82.
- Linder B. Acute posterior vitreous detachment and its retinal complications. Acta Ophthalmol Suppl. 1966;87:5–107.
- Los LI, van der Worp RJ, van Luyn MJ, Hooymans JM. Age-related liquefaction of the human vitreous body: LM and TEM evaluation of the role of proteoglycans and collagen. Invest Ophthalmol Vis Sci. 2003;44:2828–33.
- 60. Machemer R. Pars plana vitrectomy. Introduction. Trans Sect Ophthalmol Am Acad Ophthalmol Otolaryngol. 1976;81(3 Pt 1):350–1.
- 61. Malecaze F, Clamens S, Simorre-Pinatel V, Mathis A, Chollet P, Favard C, Bayard F, Plouet J. Detection of vascular endothelial growth factor messenger RNA and vascular endothelial growth factor-like activity in proliferative diabetic retinopathy. Arch Ophthalmol. 1994;112(11):1476–182.
- 62. Margherio AR, Margherio RR, Hartzer MK, et al. Plasmin enzyme-assisted vitrectomy in traumatic pediatric macular holes. Ophthalmology. 1998;105:1617–20.
- 63. Meneveau N, Schiele F, Vuillemenot A, Valette B, Grollier G, Bernard Y, Bassand JP. Streptokinase vs alteplase in massive pulmonary embolism. A randomized trial assessing right heart haemodynamics and pulmonary vascular obstruction. Eur Heart J. 1997;18(7):1141–8.
- Nakamura T, Yamagata Y, Ichishima E. Nucleotide sequence of the subtilisin NAT gene, aprN, of Bacillus subtilis (natto). Biosci Biotechnol Biochem. 1992;56(11):1869–71.

- 65. Nasrallah FP, Jalkh AE, Van Coppenolle F, Kado M, Trempe CL, McMeel JW, Schepens CL. The role of the vitreous in diabetic macular edema. Ophthalmology. 1988;95: 1335–9.
- Nasrallah FP, van de Velde F, Jalkh AE, Trempe CL, McMeel JW, Schepens CL. Importance of the vitreous in young diabetics with macular edema. Ophthalmology. 1989;96:1511–6.
- 67. O'Neill R, Shea M. The effects of bacterial collagenase in rabbit vitreous. Can J Ophthalmol. 1973;8(2):366–70.
- Oliveira LB, Tatebayashi M, Mahmoud TH, Blackmon SM, Wong F, McCuen 2nd BW. Dispase facilitates posterior vitreous detachment during vitrectomy in young pigs. Retina. 2001;21(4):324–31.
- 69. Ono R, Kakehashi A, Yamagami H, Sugi N, Kinoshita N, Saito T, Tamemoto H, Kuroki M, Ishikawa S, Kawakami M. Prospective assessment of proliferative diabetic retinopathy with observations of posterior vitreous detachment. Int Ophthalmol. 2005;26:15–9.
- Pendergast SD, Hassan TS, Williams GA, Cox MS, Margherio RR, Ferrone PJ, Garretson BR, Trese MT. Vitrectomy for diffuse diabetic macular edema associated with a taut premacular posterior hyaloid. Am J Ophthalmol. 2000;130(2):178–86.
- Pirie A. Effect of hyaluronidase injection on vitreous humor of the rabbit. Br J Ophthalmol. 1948;33:678–84.
- 72. Pirie A. Ox vitreous humour; hyaluronic acid relationships. Br J Ophthalmol. 1949;33(5):271–83.
- 73. Rizzo S, Pellegrini G, Benocci F, Belting C, Baicchi U, Vispi M. Autologous plasmin for pharmacologic vitreolysis prepared 1 hour before surgery. Retina. 2006;26(7):792–6.
- Rodriguez-Hurtado FJ, Garrido Collado MP, Delgado Ceballos VC. Treatment of vitreomacular traction syndrome with autologous plasmin enzyme. Arch Soc Esp Oftalmol. 2015;90:2689–703.
- 75. Rosen R, van Velthoven M, Garcia P, Cucu RG, de Smet MD, Muldoon TO, Podoleanu AG. Ultrahigh-resolution combined coronal optical coherence tomography confocal scanning ophthalmoscope (OCT/SLO): a pilot study. Spektrum Augenheilkd. 2007;21:36–47.
- Sakuma T, Tanaka M, Mizota A, Inoue J, Pakola S. Safety of *in vivo* pharmacologic vitreolysis with recombinant microplasmin in rabbit eyes. Invest Ophthalmol Vis Sci. 2005;46(9):3295–9.
- 77. Sebag J. Aging of the vitreous. Eye. 1987;1:254-62.
- 78. Sebag J. The vitreous structure, function and pathobiology. New York: Springer; 1989.
- 79. Sebag J. Anatomy and pathology of the vitreoretinal interface. Eye. 1992;6:541-52.
- Sebag J. Abnormalities of human vitreous structure in diabetes. Graefes Arch Clin Exp Ophthalmol. 1993;231:257–60.
- 81. Sebag J. Diabetic vitreopathy. Ophthalmology. 1996;103(2):205-6.
- 82. Sebag J. Macromolecular structure of vitreous. Prog Polym Sci. 1998;23:415-46.
- 83. Sebag J. Pharmacologic vitreolysis. Retina. 1998;18:1-3.
- Sebag J. Anomalous posterior vitreous detachment: a unifying concept in vitreo-retinal disease. Graefes Arch Clin Exp Ophthalmol. 2004;242:690–8.
- Sebag J, Buckingham B, Charles MA, Reiser K. Biochemical abnormalities in vitreous of humans with proliferative diabetic retinopathy. Arch Ophthalmol. 1992;110(10):1472–6.
- Sikri N, Bardia AA. History of streptokinase use in acute myocardial infarction. Tex Heart Inst J. 2007;34(3):318–27.
- 87. Stalmans P, de Laey C, de Smet M, et al. Intravitreal injection of microplasmin for treatment of vitreomacular adhesion: results of a prospective, randomized, sham-controlled phase II trial (the MIVI-IIT trial). Retina. 2010;30:1122–7.
- Sumi H, Hamada H, Tsushima H, Mihara H, Muraki H. A novel fibrinolytic enzyme (nattokinase) in the vegetable cheese Natto; a typical and popular soybean food in the Japanese diet. Experientia. 1987;43(10):1110–1.
- Sumi H, Hamada H, Nakanishi K, Hiratani H. Enhancement of the fibrinolytic activity in plasma by oral administration of nattokinase. Acta Haematol. 1990;84(3):139–43.

- Tagawa H, McMeel JW, Furukawa H, Quiroz H, Murakami K, Takahashi M, Trempe CL. Role of the vitreous in diabetic retinopathy. 1. Vitreous changes in diabetic retinopathy and in physiologic aging. Ophthalmology. 1986;93:596–601.
- Takahashi M, Trempe CL, Maguire K, McMeel W. Vitreoretinal relationship in diabetic retinopathy: a biomicroscopic evaluation. Arch Ophthalmol. 1981;99:241–5.
- 92. Takano A, Hirata A, Ogasawara K, Sagara N, Inomata Y, Kawaji T, Tanihara H. Posterior vitreous detachment induced by nattokinase (subtilisin NAT): a novel enzyme for pharmacologic vitreolysis. Invest Ophthalmol Vis Sci. 2006;47(5):2075–9.
- Tezel TH, Del Priore LV, Kaplan HJ. Posterior vitreous detachment with dispase. Retina. 1998;18:7–15.
- 94. Theopold H, Faulborn J. Scanning electron microscopic aspects of the vitreous body: technique of preparation. Albrecht Von Graefes Arch Klin Exp Ophthalmol. 1980;214:33–8.
- 95. Trese MT, Williams GA, Hartzer MK. A new approach to stage 3 macular holes. Ophthalmology. 2000;107:1607–11. discussion 1611.
- Unal M, Peyman GA. The efficacy of plasminogen-urokinase combination in inducing posterior vitreous detachment. Retina. 2000;20(1):69–75.
- Van Velthoven MEJ, Faber DJ, Verbraak FD, van Leeuwen TG, de Smet MD. Recent developments in optical coherence tomography for imaging the retina. Prog Retin Eye Res. 2007;26:57–77.
- Verstraeten TC, Chapman C, Hartzer M, Winkler BS, Trese MT, Williams GA. Pharmacologic induction of posterior vitreous detachment in the rabbit. Arch Ophthalmol. 1993;111(6):849–54.
- 99. Williams JG, Trese M, Williams GA, et al. Autologous plasmin enzyme in the surgical management of diabetic retinopathy. Ophthalmology. 2001;108:1902–5.
- 100. Yamaguchi Y, Otani T, Kishi S. Resolution of diabetic cystoid macular edema associated with spontaneous vitreofoveal separation. Am J Ophthalmol. 2003;135:116–8.

Chapter 8 Investigational Medications

Pharmacologic treatment of diabetic retinopathy (DR) began with the use of intravitreal triamcinolone acetonide and has evolved rapidly over the past 16 years. Dozens of drugs are in the development pipeline and will probably give us several new therapeutic options within the next decade. Development of new pharmacotherapeutic agents is being driven by several factors. Our understanding of the molecular pathways responsible for DR has improved, particularly with the discovery of vascular endothelial growth factor (VEGF) and other cytokines and chemokines. Successful completion of the Human Genome Project has spun off techniques that have enabled the discovery of various genetic abnormalities associated with DR.

The success of intravitreal anti-VEGF and corticosteroid therapy for neovascular age-related macular degeneration (nAMD), diabetic macular edema (DME), and edema due to retinal vein occlusions has transformed the way companies, scientists, and clinicians think about treating these conditions. Development of ocular pharma-cotherapies can be financially lucrative as ranibizumb and affibercept have become the second and third highest reimbursed medications on the Medicare Part B list of reimbursed medications. Clinicians have embraced the high efficacy and favorable safety profiles of intraocular therapy, and they continue to move away from laser photocoagulation. New indications for currently available drugs, such as treatment of diabetic retinopathy in eyes with diabetic macular edema (DME) and treatment of proliferative diabetic retinopathy (PDR), continue to emerge. Patients have accepted intravitreal injections, and compliance with demanding treatment regimens is generally favorable.

Despite the numerous drawbacks associated with long-term intravitreal anti-VEGF use in DR, these drugs are now standard of care for the treatment of most patients. When considering a new therapeutic strategy for DR, one must understand whether the targeted mechanism is independent of VEGF or if it ultimately suppresses VEGF. If the action of a new therapeutic is to primarily impact the VEGF pathway or VEGF levels, then it is unlikely that this treatment will be superior to currently available intravitreal anti-VEGF therapies. This may impact the likelihood of successful drug development since reduction of treatment burden, though attractive to physicians and patients, is often not an acceptable regulatory endpoint in some areas. Therefore, the new therapeutic must be non-inferior to intravitreal anti-VEGF therapy or result in increased efficacy as an adjuvant therapy.

This chapter will discuss drugs that are in preclinical testing, the early stages of human testing (phase I, II, or III trials) for DR, are likely to be used in patients with DR after they have been successfully tested for other conditions such as nAMD and off-label use of drugs that have already been approved for other indications. The drugs discussed in this chapter fall into several categories: biologics, corticosteroids, and antibiotics, among others. These drugs have not been subjected to multicenter, randomized, controlled, masked trials against standard-of-care therapies and at this time are rarely used outside of laboratory studies and clinical trials.

8.1 **Corticosteroid-Related Molecules**

Corticosteroid-related drugs that are being investigated for the treatment of diabetic macular edema are listed in Table 8.1.

Table 8.1 Corticosteroid-related drugs that are being evaluated for the treatment of diabetic macular edema are listed, along with their clinical trial status and important biochemical characteristics and study results

Corticosteroid-related drugs being evaluated for the treatment of diabetic macular edema				
Drug	Clinical phase	Important characteristics		
Danazol	Phase IIb	 Binds androgens and steroid-binding globulins Approved for the treatment of endometriosis In Phase IIa study, oral danazol outperformed placebo in: Decreasing excess retinal thickness (<i>P</i> = 0.05) Improving BCVA by 1 category (47% vs. 14%) 		
Dexamethasone- cyclodextrin Microparticulate Drops	Phase I completed	 Dissolves in tear film to form microparticles Penetrates to retina in rabbits, anterior chamber in humans In phase I trial, 19 eyes treated 3 or 6 times per day for 4 weeks and observed for 4 weeks: LogMAR BCVA improved by -0.15 and -0.07 CMT improved by -113 µm and -24 µm 		
Difluprednate (Durezol®)	Phase I completed	 Refractory DME after vitrectomy in 11 eyes, difluprednate was compared to subtenon's triamcinolone: 1. 4 times daily for 1 month then twice daily for 2 months 2. Mean improvement in CRT at 3 months of -159 μm 3. No change in BCVA at 3 months 		
Corticosteroid-related	d drugs being evaluated f	for the treatment of diabetic macular edema		
--------------------------------------	---	--		
Drug	Clinical phase	Important characteristics		
EGP-437	Completed phase Ib/IIa Extension study in progress	 Iontophoresis drug delivery to the retina Open-label study of 19 patients with DME, RVO, and CME: 1. Drops delivered on days 0, 4, and 9 2. Pseudophakic patients and those with postoperative CME did best 		
Loteprednol etabonate (KP-121)	Phase II trial underway	 Topically administered Mucus-penetrating platform (MPP) QID dosing 		

Table 8.1 (continued)

BCVA best corrected visual acuity, *LogMAR* logarithm of the minimal angle of resolution, *CMT* central macular thickness, *DME* diabetic macular edema, *RVO* retinal vein occlusion, *CME* cystoid macular edema, *QID* 4 times per day

8.1.1 Danazol

Danazol binds to androgens and steroid-binding globulins, stimulating the formation of a cortical actin ring that enhances endothelial barrier function. Danazol has been approved by the US Food and Drug Administration (US FDA) for the oral treatment of endometriosis.

A randomized, 12-week, placebo-controlled study evaluated the safety and efficacy of twice daily oral danazol in patients with DME. Twenty-three patients with DME and central retinal thickness (CRT) >300 μ m were enrolled. The study's primary functional endpoint was change in CRT, and secondary endpoints were changes in macular volume and best corrected visual acuity (BCVA). Compared to placebo treated eyes, those receiving danazol achieved significant decreases in excess CRT (-29% vs. -86%) and macular volume (P = 0.05) and modest improvements in BCVA (improvement by 1 category: 14% vs. 47%).

The FDA mandated that Ampio Pharmaceuticals perform a confirmatory study on patients with DME who are refractory to approved medications. Since patients will already have failed to respond to anti-VEGF medications, placebo will serve as the control group. The company projects that 80 patients (40 in each of the danazol and control groups) will be needed for a 12-month study. The FDA did not require that safety be a specified endpoint because danazol would be used at doses only 10% of those approved for use in patients with endometriosis [7].

8.1.2 Dexamethasone-Cyclodextrin Microparticle Drops

A drug delivery platform based on cyclodextrin microparticles that dissolve in the tear film to form water-soluble drug/cyclodextrin complex microparticles has been developed for ocular pharmacology [44]. Microparticulate 1.5% (wt/vol) dexamethasone-cyclodextrin eye drops can deliver the drug to the retina and vitreous

humor in rabbits [63, 80, 81, 122]. Cyclodextrin-based dexamethasone eye drop solutions penetrate well into the anterior segment of the human eye [75, 110]. Since early pharmacology studies in rabbits and humans suggested that dexamethasone-cyclodextrin microparticle eye drops may reach the human retina, clinical trials of dexamethasone eye drops were initiated for the topical treatment of DME.

Nineteen eyes of 19 patients with DME received dexamethasone-cyclodextrin eye drops three or six times a day for 4 weeks and were then observed for 4 weeks without treatment [127]. At weeks 0 (baseline), 4, and 8, logMAR visual acuity (mean \pm SD) was 0.52 \pm 0.41, 0.37 \pm 0.40 (P = 0.0025 vs. baseline), and 0.45 \pm 0.41, respectively; central macular thickness (µm) was 512 \pm 164, 399 \pm 154 (P = 0.0016 vs. baseline), and 488 \pm 172 (P = 0.0116 vs. week 4), respectively; and intraocular pressure (mm Hg) was 15.2 \pm 3.1, 17.4 \pm 4.2 (P = 0.0015 vs. baseline) and 15.8 \pm 4.0, respectively. At week 4, central macular thickness had decreased more than 10% in 12 of 19 eyes (63%), and the mean change was -20% (-65% to -10%). In 14 of 19 eyes (74%) visual acuity had improved more than 0.1 logMAR at week 4. No eye drop-related adverse effects were noted.

8.1.3 Difluprednate Ophthalmic Emulsion

The efficacy of topical difluprednate ophthalmic emulsion 0.05% (Durezol[®], Sirion Therapeutics Inc., USA) on the treatment of refractory DME after vitrectomy was compared to sub-Tenon's injections of triamcinolone (STTA) [87]. Eleven eyes of ten subjects were treated with STTA (STTA group), and 11 eyes of seven subjects were treated with difluprednate ophthalmic emulsion 0.05% four times daily for the first month and then twice daily for 2 months (eye drop group).

The mean VA (±SD) in the eye drop group was similar at 3 months (0.67 ± 0.29 loMAR) as at baseline (0.67 ± 0.35 logMAR); mean retinal thickness (µm) decreased from baseline (500.6 ± 207.7) to 3 months (341.2 ± 194.8), with a mean minimum retinal thickness during the treatment period (300.6 ± 123.2) that was significantly lower than that at baseline (Mann-Whitney U test: P = 0.003). In the STTA group, mean VA (±SD) was 0.67 ± 0.35 logMAR, and mean retinal thickness was 543.3 ± 132.6 µm at baseline. After 3 months of treatment, mean VA improved to 0.49 ± 0.67 logMAR, and mean retinal thickness during the treatment period (349.9 ± 113.8 µm) was significantly lower than at baseline (Mann-Whitney U test: P = 0.003). The rate of improvement in retinal thickness did not differ between the eye drop group (73%) and STTA group (84%) (Fisher's exact test: P = 1).

8.1.4 EGP-437

Eyegate Pharmaceuticals is using iontophoresis to deliver the experimental drug EGP-437 (reformulated, topically active dexamethasone phosphate) to the retina of patients with DME. A multicenter, open-label, phase Ib/IIa trial has enrolled 19

patients with DME, retinal vein occlusions, and postsurgical cystoid macular edema. Treatments with an electrical impulse of 4.0 mA-min (3.5 mA) were administered on days 0, 4, and 9 with the primary outcome being the reduction in central subfield thickness (CST) on days 4, 9, and 14. The dexamethasone insert was administered to patients who did not respond favorably by day 14.

The interim results showed that some patients, particularly those with postoperative cystoid macular edema (CME) and those who were pseudophakic, responded positively [36]. Edema re-accumulated when the drug was cleared from the tissues. An extension study will include an additional 15 patients who will be dosed on 3 consecutive days.

8.1.5 Loteprednol Etabonate

Kala Pharmaceuticals is developing nanotechonology-based ophthalmic products to treat DME. They are initiating a phase II clinical trial (KPI-121-C-004) to evaluate KP-121 (LE-MPP), a topically administered loteprednol etabonate mucuspenetrating platform (MPP), for the treatment of macular edema due to DR and retinal vein occlusions [82]. This single-masked, randomized trial will investigate the efficacy and safety of 1% LP-MPP and 0.25% LPP administered QID to 20 patients.

8.2 Vascular Endothelial Growth Factor Inhibitors

Vascular endothelial growth factor inhibitor drugs that are being investigated for the treatment of diabetic macular edema are listed in Table 8.2.

Table 8.2 Vascular endothelial growth factor inhibitory drugs that are being evaluated for the treatment of diabetic macular edema are listed, along with their clinical trial status and important biochemical characteristics and study results

Drug	Clinical phase	Important characteristics
Abicipar pegol	Phase II	 Designed ankyrin repeat protein Currently in phase III trial for nAMD In phase I/II trial of 18 patients with DME: 1. Estimated half-life of 13.4 days 2. BCVA improvement of +10 letters at 12 weeks after single injection
Conbercept	Phase III	 Fusion protein with receptor binding sequences attached to Fc fragment of IgG Binding affinity of 0.75 pM for VEGF₁₆₅ Binds VEGF-A, VEGF-B, placental growth factor Approved for treatment of nAMD in China In vitro suppression of glucose-induced endothelial cell migration and proliferation

Vascular endothelial growth factor inhibitory drugs being evaluated for the treatment of diabetic macular edema

diabetic macular ec	iema	
Drug	Clinical phase	Important characteristics
Encapsulated cell technology (NCT-03)	Phase II nAMD trial terminated early due to poor efficacy	 Immortalized retinal pigment epithelial cells in semipermeable implanted chamber produce aflibercept-like fusion protein Ciliary neurotrophic factor producing implant failed trials for dry AMD and retinitis pigmentosa. Failed phase II trial for nAMD Future development is uncertain
Gene therapy (AVA-101)	Completed phase IIa nAMD trial	 Subretinally injected adenovirus DNA for slt-1 (soluble VEGFR1) into retinal pigment epithelial cells In phase IIa trial with 21 nAMD patients: Ranibizumab injected at baseline, AVA-101 injected at day 7 I1 patients received ranibizumab only (control) At 52 weeks, BCVA in AVA (+2.2 letters) vs. ranibizumab (+9.3 letters) Mean center point thickness improved by -27 μm and -85 μm Future development is uncertain
Implantable drug delivery pump (PMP)	Phase I completed	 Miniature pump delivers drug to retina Long-term safety seen in animals Phase I trial of 11 patients with DME treated for 3 months: 1. No cases of endophthalmitis or strabismus
PAN-90806	Phase II trial underway	 Low molecular weight, topical anti-VEGF medication Excellent drug concentrations in the retina 17 h after administration Animal studies show CNVM control comparable to antibodies Phase I/II trial data in 2016 Phase I proliferative diabetic retinopathy trial underway
Ranibizumab sustained release reservoir	Currently in phase II trial for nAMD	 A refillable port delivery system that is implanted through the pars plana 1-year phase I nAMD trial of 20 patients found: 1. Mean of 4.8 reinjections 2. BCVA improvement of +10 letters 3. Four implant-related SAEs
RTH258	Currently in phase III trial for nAMD	 High-affinity, single-chain antibody fragment High injected concentration (6 mg/0.05 ml) produces extended duration of action Phase II nAMD trial showed comparable efficacy to aflibercept. Extended duration of action suggests q3month dosing
Ziv-aflibercept	Off-label use in DME	 Intravenous formulation of aflibercept indicated for treatment of advanced solid tumors Two DME patients had significant improvements in BCVA and macular thickness Ongoing off-label treatment continues

Table 8.2 (continued)

Vascular endothelial growth factor inhibitory drugs being evaluated for the treatment of diabetic macular edema

nAMD neovascular age-related macular degeneration, *DME* diabetic macular edema, *BCVA* best corrected visual acuity, *VEGF* vascular endothelial growth factor, *CNVM* choroidal neovascular growth factor, *SAEs* serious adverse events

8.2.1 Abicipar Pegol

DARPins (designed ankyrin repeat proteins) are small molecular weight (14–20 kDa) molecules with high solubility (>100 mg/L) in saline. They are a flexible design platform that allows for the creation of genetically engineered mimetic proteins that can target any molecule [17]. Several DARPin molecules (at least 15) are being developed for chorioretinal vascular conditions, oncologic indications, and inflammatory diseases. DARPin technology also allows for the design of dual action proteins.

Abicipar pegol (Allergan, Irvine, CA), formerly known as DARPin MP0112, binds all isoforms of VEGF-A. This high-affinity molecule ($K_D = 2 \text{ pM}$ for VEGF₁₆₅) has a surprisingly long intravitreal half-life in rabbits (6 days) that may be attributed to its pegylation. A phase I/II multicenter, open-label, dose-escalation trial evaluated the safety and bioactivity of abicipar in 18 patients with DME [21]. Patients receiving 1 mg injections experienced excellent reductions in macular thickness and mean improvement in VA (+10 letters) 12 weeks after single injections. Pharmacokinetic analyses based on anterior chamber drug concentrations suggest an extended intraocular half-life of 13.4 days. Multicenter, randomized, double-masked, phase III nAMD trials (CEDAR and SEQUOIA) are comparing q8wk and q12wk abicipar with q4wk ranibizumab. Given abicipar's high binding affinity, apparently long intraocular half-life, and encouraging results from the early phase trials, the developers are hoping to establish the efficacy of 3-month dosing. Phase III DME trials are being planned but have not yet begun.

8.2.2 Conbercept

Conbercept (KH902, Chengdu Kanghong Biotech Co., Sichuan, China) is a recombinant, fusion protein that, like aflibercept, acts as a decoy receptor. Conbercept (MW of 143 kDa) contains the second immunoglobulin (Ig)-binding domain from VEGF receptor 1 (VEGFR1), the third and the fourth binding domains from VEGFR2, and the Fc region of human IgG. The difference between aflibercept and conbercept is that aflibercept does not contain the fourth domain of VEGFR2 [53, 141]. Conbercept has a high affinity for VEGF because the fourth Ig domain of VEGFR2 is essential for receptor dimerization and it enhances the association rate of VEGF to the receptor [141]. Like aflibercept, conbercept binds all isoforms of VEGF-A, VEGF-B, and placental growth factor. At concentrations from 100 ng/ml to 100 μ g/ml, conbercept was not cytotoxic to cultured human retinal endothelial cells (HRECs). A 500 ng/ml solution of conbercept significantly suppresses high glucose-induced migration and sprouting of HRECs by downregulating the expression of PI3K and inhibiting the activation of Src, Akt1, and Erk1/2 [25].

Four weeks after intravitreal injection, conbercept-treated rats had better retinal electrophysiological function, less retinal vessel leakage, and lower levels of PIGF, VEGFR2, PI3K, AKT, p-AKT, p-ERK, and p-SRC than PBS or Avastin-treated rats [58]. The distribution of claudin-5 and occludin in the retinal vessels of diabetic rats

treated with conbercept was smoother and more uniform than those of diabetic rats treated by PBS or Avastin. Conbercept has already been approved in China for the treatment of nAMD [79], and a phase III trial evaluating the efficacy of conbercept for the treatment of DME is currently enrolling patients.

8.2.3 Encapsulated Cell Technology

Encapsulated cell technology (ECT) was first reported by Bisceglie (1934) as a way to prevent rejection of foreign cell, tissues, or organisms. ECT involves the use of immortalized cells that have been programmed to overproduce a specified biochemical product. The cells are grown in a cylinder lined by semipermeable membranes that allow ingress of nutrients and egress of the synthesized product. The membrane prevents migration of the modified cells and shields them from the body's immune system. The cylinder is 10 mm long and is surgically implanted through the pars plana and sutured to the sclera.

Clinical studies using ECT to produce ciliary neurotrophic factor (CNF) have been completed in eyes with retinitis pigmentosa and atrophic AMD [70]. The ECT cylinder was well tolerated, but the trials failed to meet their primary therapeutic endpoints. Pharmacokinetic analyses showed that the half-life of CNF production by the cylinder was 54 months. Phase I trials with a cylinder that produced a high-affinity VEGF-binding protein similar to aflibercept have been performed, and a multicenter phase II trial [28] failed to meet its primary efficacy endpoint, thereby calling into question future development and use in patients with DME.

8.2.4 Gene Therapy

Avalanche Biotechnologies has developed a viral delivery system (AVA-101) to enable the eye to produce long-term anti-VEGF therapy. An adenovirus vector inserts the DNA for a naturally occurring slt-1 (soluble VEGF receptor-1) into retinal pigment epithelial cells. Infected cells synthesize and excrete the soluble VEGF inhibitory protein into the outer retina and choriocapillaris.

In a phase IIa trial, 21 patients with nAMD received AVA-101, with 0.5 mg ranibizumab injected both at baseline and 1 month and again as rescue therapy. Patients underwent core vitrectomy and subretinal injection of AVA-101 adjacent to the macula at day 7. Patients were evaluated monthly and were eligible for rescue ranibizumab therapy based on prespecified criteria. Eleven control patients received only 0.5 mg ranibizumab monthly.

At the 52-week endpoint, mean improvement in BCVA was +2.2 letters in the AVA-101 group compared to +9.3 letters in the ranibizumab group [51]. These differences were statistically significant, but they were largely driven by three subjects in the AVA-101 group who each lost at least four lines of vision. Mean center point

thickness improved by $-27 \ \mu\text{m}$ in the AVA-101 group and $-85 \ \mu\text{m}$ in the control group. There were no serious ocular adverse events in the AVA-101 group, and no systemic safety signals were noted. All patients in the AVA-101 group who were phakic at baseline developed cataracts, and three (14%) developed moderate vitreous hemorrhages. Gene therapy was well tolerated by patients, but the technology failed to provide a complete or durable anti-VEGF response.

8.2.5 Implantable Drug Delivery Pump (PMP)

Microelectromechanical system (MEMS) technology is a miniaturized system that is currently used in insulin pumps to deliver drug to the tissues. The Posterior MicroPump Drug Delivery System (PMP, Replenish Inc., Pasadena, CA) uses MEMS technology to deliver drug within the eye. Long-term safety after implantation into animal eyes has been demonstrated [47, 111]. The PMP can reliably deliver 100 programmed doses of an anti-VEGF drug, equivalent to over 8 years of therapy. The PMP was evaluated for 3 months in 11 patients with DME [59]. After episcleral implantation, similar to placement of a glaucoma drainage device, the PMP was well tolerated with no cases of endophthalmitis or strabismus.

8.2.6 PAN-90806

PanOptica, Inc. is developing a topical anti-VEGF medication for the treatment of nAMD and PDR. PAN-90806 is a low molecular weight, VEGF receptor blocker administered in eye drop form. Pharmacokinetic studies show excellent drug concentrations in the central retina and choroid as late as 17 h after administration. Animal studies are reported to show control of leakage and bleeding from choroidal neovascular membranes, comparable to that achievable with intravitreal anti-VEGF antibodies, with minimal systemic exposure to the drug. Preliminary results from each of four monotherapy treatment arms in a phase I/II trial were judged by an independent panel of experts to show promise for the treatment of nAMD [94]. Results from a phase II trial of PAN-90806 maintenance therapy after a single anti-VEGF injection are expected to be presented in 2017. The company is moving forward with a phase I trial for the treatment of PDR.

8.2.7 Ranibizumab Sustained Release Reservoir

A refillable ranibizumab port delivery system is being codeveloped by Genentech and ForSight Vision 4 to reduce the need for repeated intravitreal anti-VEGF injections. The pre-loaded implant is surgically implanted beneath the conjunctiva through a 3.2 mm scleral incision over the pars plana. The reservoir tip can be accessed easily through the conjunctiva and refilled in the office as needed. The device continuously releases ranibizumab into the vitreous between refills.

A phase I trial for patients with nAMD was performed in Riga, Latvia [105]. At baseline, the reservoir was implanted, and eyes were given 500 µg ranibizumab injections, 250 µg into the vitreous and 250 µg into the reservoir for sustained release. Additional injections were given based on optical coherence tomography (OCT) evaluation of disease activity. The primary endpoint was 12 months with an observation period that extended through 36 months. The primary objective of the study was safety assessment with secondary objectives that included functional measurements.

Four of the patients had significant or serious adverse events (endophthalmitis, vitreous hemorrhage (2), and traumatic cataract), but 3 of these 4 had improved vision by the study's endpoint. The average visual acuity gains for the cohort were +10 letters; 10 eyes (50%) gained at least 3 lines and 2 (10%) lost at least 3 lines. The mean number of refills was 4.8 per patient.

The planned phase II trial will feature 750 μ g injections, with hopes of extending the treatment interval to 4 months.

8.2.8 RTH258

RTH258 (formerly known as ESBA 1008) is a single-chain, VEGF-binding antibody fragment currently being developed by Alcon (Ft. Worth, TX) for the treatment of nAMD. It has been touted by its developer as having a longer duration of action than currently available anti-VEGF drugs, thereby requiring fewer injections.

A phase II clinical trial compared RTH258 to aflibercept in patients with nAMD [92]. The trial's primary objective was to compare the efficacy of 6 mg RTH258 against 2 mg aflibercept with the primary endpoint of the study being the mean change in BCVA from baseline to 12 weeks. Secondary endpoints included improvement in central subfield foveal thickness (CSFT) on SD-OCT. Preliminary reports stated that RTH258 produced BCVA gains that were non-inferior to aflibercept with a greater reduction in macular fluid. Patients treated every 3 months experienced a positive effect, suggesting a long duration of action. No new safety signal was seen.

The phase III clinical trial program was initiated in December 2014, with an enrollment goal of 1700 patients in more than 50 countries. These 2-year, double-masked, multicenter trials will randomize patients with untreated nAMD to one of two dosage levels of RTH258 or 2 mg affibercept bimonthly [29]. The primary end-point will be changed in BCVA at 48 weeks with several additional secondary functional and morphologic endpoints. No DME trials have yet been announced.

8.2.9 Ziv-Aflibercept

Ziv-aflibercept (Zaltrap[®], Regeneron, Tarrytown, NY) is the systemic formulation of Eylea[®] that is indicated for the intravenous treatment of advanced colorectal carcinoma. Single use vials contain 4 ml (25 mg/ml) of

characteristics and study results

ziv-aflibercept in a buffered solution of polysorbate 20 (0.1%), sodium chloride (100 mM), sodium citrate (5 mM), sodium phosphate (5 mM), and sucrose (20%), with a pH of 6.2. Small series of patients with nAMD that received single injections of ziv-aflibercept had anatomic and visual acuity improvements at 1 month without evidence of toxicity [26]. Two patients with DME had improved VA (20/800 to 20/100; 20/800 to 20/200) and macular thickness (CST: $-65 \mu m$ and $-352 \mu m$) 1 week after intravitreal injections [84]. Additional studies continue to evaluate the use of ziv-aflibercept for nAMD, DME, and retinal vein occlusions.

8.3 Tumor Necrosis Factor-α Inhibitors

Tumor necrosis factor (TNF)- α is a pro-inflammatory cytokine that is synthesized by T-lymphocytes, macrophages, neutrophils, and mast cells. It plays an important role in mediating the immune response, tumorigenesis, and inhibiting viral replication. TNF- α is upregulated in eyes with uveitis, nAMD, and diabetic retinopathy [130]. Several anti-TNF- α biologicals have been approved for the treatment of systemic inflammatory conditions including rheumatoid arthritis [97]. In animal models, TNF- α has been shown to contribute to the development of DR [46, 54, 66], and TNF- α inhibition limits breakdown of the blood-retinal barrier [67]. It is also possible that high-dose NSAIDs delay the onset of diabetic retinopathy via TNF- α suppression [107].

Tumor necrosis factor inhibitor drugs that are being investigated for the treatment of diabetic macular edema are listed in Table 8.3.

Table 8.3 Tumor necrosis factor inhibitory drugs that are being evaluated for the treatment of diabetic macular edema are listed, along with their clinical trial status and important biochemical

edenna		
Drug	Clinical phase	Important characteristics
Adalimumab	Off-label use	 Was not effective when injected into five eyes with DME When injected into seven eyes with DME, positive results were seen only when combined with bevacizumab
Etanercept	Phase I trial completed	 Prevents TNF-α binding to transmembrane receptor Two injections, 2 weeks apart given to seven eyes with DME yielded no clinical response
Infliximab	Phase IIa trial completed	 15 patients with DME received single 1.0 mg injections and 19 received 2.0 mg injections. 42% developed uveitis In a double-blind, randomized, placebo-controlled study of patients with persistent DME, patients receiving infliximab had improved BCVA and retinal thickness
Pegsunercept	Preclinical	• Injections into rats led to reduction in pericyte loss and capillary degeneration

Tumor necrosis factor inhibitory drugs being evaluated for the treatment of diabetic macular edema

DME diabetic macular edema, TNF tumor necrosis factor, BCVA best corrected visual acuity

8.3.1 Adalimumab

Intravitreal injections of 2.0 mg adalimumab were ineffective in five eyes with DME that had been refractive to anti-VEGF therapy. No adverse side effects were noted in any of these eyes [136]. In a series of seven eyes of five patients with macular edema from various causes, favorable clinical responses were noted when adalimumab was combined with bevacizumab [9].

8.3.2 Etanercept

Etanercept is a soluble TNF- α receptor that acts as a competitive inhibitor to block TNF- α binding to transmembrane receptors. It reduces leukocyte adherence in retinal blood vessels [67], blood-retinal barrier breakdown, and NF- κ B activation in the diabetic retina [66]. Two injections of etanercept (2.5 mg) were performed 2 weeks apart to seven eyes with refractory DME, but no clinical responses were noted at 3 months [131].

8.3.3 Infliximab

Visual acuity changes from baseline to 3 months in 15 patients with DME receiving single injections of 1.0 mg infliximab (1.49 LogMAR to 1.38 LogMAR), and 19 patients receiving 2.0 mg infliximab (0.76 LogMAR to 1.03 LogMAR) were disappointing. Furthermore, 42% of eyes developed severe uveitis with 37% requiring vitrectomy, thereby halting further research with intravitreal infliximab [136].

A double-blind, randomized, placebo-controlled, crossover study in patients with DME that had persisted after two sessions of laser photocoagulation showed that patients receiving intravenous infliximab (5 mg) had significantly improved visual acuity and reduced retinal thickness [118].

8.3.4 Pegsunercept

Intravitreal injection of the TNF- α -specific inhibitor, pegsunercept, led to a significant reduction in pericyte loss and capillary degeneration in diabetic rats [14, 15].

8.4 Nonsteroidal Anti-inflammatories

Nonsteroidal anti-inflammatory drugs that are being investigated for the treatment of diabetic macular edema are listed in Table 8.4.

Nonsteroidal inflammatory drugs being evaluated for the treatment of diabetic macular edema			
Drug	Clinical phase	Important characteristics	
Aspirin	Phase II completed	• Low-dose aspirin provides little to no benefit in prevent- ing diabetic retinopathy	
Diclofenac	Phase IIa completed	 57 patients with treatment naïve DME received single injections of diclofenac or bevacizumab: 1. Diclofenac patients had better improvements in BCVA: -0.08 LogMAR vs. +0.04 LogMAR 2. Bevacizumab improved edema better 	

Table 8.4 Nonsteroidal anti-inflammatory drugs that are being evaluated for the treatment of diabetic macular edema are listed, along with their clinical trial status and important biochemical characteristics and study results

DME diabetic macular edema, *BCVA* best corrected visual acuity, *LogMAR* logarithm of the minimum angle of resolution

8.4.1 Aspirin

In clinical studies, low-dose aspirin has shown only little or no benefit in preventing diabetic retinopathy [93]. However, further work is still needed to determine if high-dose aspirin can prevent the development of diabetic retinopathy.

8.4.2 Diclofenac

In a randomized trial, 57 eyes with treatment naïve DME received single intravitreal injections of either diclofenac (500 µg/0.1 ml) or bevacizumab. The primary outcome was the change in mean BCVA at 12 weeks, and secondary outcomes included changes in macular thickness, macular leakage, and safety. Eyes receiving diclofenac had better mean improvements in BCVA compared to bevacizumab (Δ –0.08 LogMAR vs. Δ +0.04 LogMAR, *P* = 0.033), but bevacizumab improved macular edema slightly better [124].

8.5 Other

Drugs that do not fit into the other listed categories that are being investigated for the treatment of diabetic macular edema are listed in Table 8.5.

8.5.1 Adenosine Kinase Inhibitor

Adenosine is centrally involved in the signaling cascade that regulates antiinflammatory actions, angiogenesis, the oxygen supply/demand ratio, and ischemic pre- and post-conditioning [65]. Under these circumstances, the local levels of

Table 8.5 Nonsteroidal anti-inflammatory drugs that are being evaluated for the treatment of diabetic macular edema are listed, along with their clinical trial status and important biochemical characteristics and study results

maculai edema		
Drug	Clinical phase	Important characteristics
Adenosine kinase inhibitor (ABT-702)	Preclinical	 Adenosine helps regulate anti-inflammatory actions, angiogenesis, and oxygen supply and demand Adenosine is a major source of stored energy (ATP) Intraperitoneal adenosine in rats decreased signs of inflammation in experimental diabetes
Angiopoietin-2 inhibition	Phase II Trial Underway (AVENUE)	 Competes with Ang-1 for Tie2 receptor Bi-specific antibody (VEGF and Ang2) currently being studied
Antioxidants	Phase II Trials Completed	 Calcium dobesilate has been studied in several trials Failed in most trials to prevent the development of macular edema
ASP8232	Phase II trial underway	 Inhibitor of vascular adhesion protein-1 VIDI trial is evaluating ASP8232 monotherapy and in combination with ranibizumab
Darapladib	Phase II trial underway	 Inhibits lipoprotein-associated phospholipase CA2 Protects against atherogenesis and vascular leakage in animal models
Epalrastat	Phase II trial completed	 Inhibits production of protein kinase C Prevented progression of early retinopathy and neuropathy
Fasudil	Phase I trial completed	 Rho-kinase inhibitor used to treat cerebral vasospasm Can suppress leukocyte adhesion and prevent neutrophil-mediated capillary endothelial cell damage In a small prospective study, fasudil + bevaci-zumab improved BCVA and CRT at 4 weeks
iCo-007	Phase II trial completed	• iCo-007 and iCo-007 + ranibizumab were compared to laser (iDEAL study). No difference among groups for proportion of patients with 15-letter BCVA loss
Luminate (ALG-1001)	Phase IIb trial underway	 Integrin receptor antagonist May be effective for VMT and DME. Promotes vitreolysis and interferes with angiogenesis Phase I trial in patients with DME that were refractory to standard care. At 150 days: BCVA improved from 20/200 to 20/125 CMT improved from 519 µm to 387 µm

Drugs not in the previously identified categories being evaluated for the treatment of diabetic macular edema

Table 8.5 (continued)

Drugs not in the previously identified categories being evaluated for the treatment of diabetic macular edema

Drug	Clinical phase	Important characteristics
Mecamylamine	Phase I/II trial completed	 Antagonist of n-acetyl choline receptors 23 patients with DME were treated with BID drops for 12 weeks. At 16 weeks: 1. BCVA improved by +3.1 letters 2. No change in foveal thickness
Microspheres	Preclinical	 Local administration of sustained release particles that can be loaded with several molecules Subconjunctival injections of sustained release celecoxib-loaded microspheres decreased VEGF production and blood-barrier breakdown in rat model of diabetes
Minocycline	Phase I/II trial completed	 Exhibits anti-inflammatory effect against glial activation Six months oral administration in prospective, open-label study resulted in: BCVA improvement of +5.8 letters CST improvement of 8.1%
PF-04523655	Phase II trial completed	 Small interfering ribonucleic acid that inhibits expression of hypoxia-inducible gene RTP801 May work independent of and possibly complimentary to anti-VEGF drugs 184 patients were randomized to one of three doses of PF-0423655 or laser. At 12 months: 1. BCVA in highest dose improved nonsignificantly more than laser (+5.77 vs. +2.39 letters; <i>P</i>=0.08) 2. No evidence that macular fluid changes were dose related 3. Study was terminated based on predetermined futility criteria
Plasma kallikrein inhibitor	Phase II trials planned	 A serine protease that is part of the body's inflammatory response. Increases levels of bradykinin Increased kallikrein activity seen in DME, hereditary angioedema, and cerebral hemorrhage In a phase I study, 14 patients received single injections of three doses. At day 84: 1. BCVA improved by +4 letters 2. CST improved by -40 µm
Ruboxistaurin	Phase III trials completed	 Orally administered protein kinase C inhibitor Improves DR in animal models and retinal hemodynamics in patients with DM Studied in phase III trials but failed to meet primary endpoints

(continued)

Drug	Clinical phase	Important characteristics
Sirolimus	Phase II trials underway	 mTOR inhibitor that modulates HIF-1α- mediated activation of growth factors In phase I trial, single subconjunctival or intravitreal injections were given to 50 eyes. At day 45, median improvements in subconjuncti- val and intravitreal eyes were: BCVA: +4 letters in both groups Decrease in retinal thickness: -23.7 µm and -52 µm
Squalamine	Phase I trials underway	 Small antiangiogenic molecule that interferes with several growth factors including VEGF Has been evaluated in early phase nAMD trials and investigator-initiated DR trials
Tie2 agonist (AKB-9778)	Phase IIa trial completed	 Tie 2 is a transmembrane receptor that stabilizes vasculature and decreases leakage 12-week randomized trial evaluated AKB-9778 monotherapy and in combination with ranibizumab: 1. AKB-9778 monotherapy was not effective 2. Compared to ranibizumab monotherapy, combination therapy: (a) Improved CST (-164 µm vs -110 µm; P = 0.008) (b) Improved BCVA (+6.3 letters vs. +5.7 letters)
Tocilizumab (TCZ)	Phase II trial underway	 Inhibits interleukin-6 READ-4 trial randomizes patients to ranibizumab, TCZ, or combination therapy
Teprotumumab	Phase I trial underway	Insulin-like growth factor inhibitorIntravenous administration for DME

Table 8.5 (continued)

Drugs not in the previously identified categories being evaluated for the treatment of diabetic macular edema

ATP adenosine triphosphate, VEGF vascular endothelial growth factor, Ang2 angiopoietin-2, BCVA best corrected visual acuity, CRT central retinal thickness, VMT vitreomacular traction, DME diabetic macular edema, CMT central macular thickness, BID twice daily, CST central subfield thickness, DR diabetic retinopathy, DM diabetes mellitus, HIF hypoxia-inducible factor, nAMD neovascular age-related macular degeneration

extracellular adenosine are increased due to the increased need for energy supplied by adenosine triphosphate [132]. The increased extracellular adenosine at inflamed sites can protect against cellular damage by activating the A2A adenosine receptor (A2AAR), a Gs-coupled receptor [60].

The selective adenosine kinase inhibitor (AKI), ABT-702, was injected intraperitoneally twice weekly to streptozotocin-induced diabetic mice [35]. Retinal inflammation was evaluated using Western blot, real-time PCR, and immunostaining analyses, and the role of A2AAR signaling in the anti-inflammation regulation of ABT-702 was analyzed in amadori-glycated-albumin (AGA)-treated microglial cells. At 16 weeks, when diabetic mice exhibit significant signs of retinal inflammation including upregulation of oxidative/nitrosative stress, A2AAR, ENT1, Iba1, TNF- α , ICAM1, retinal cell death, and downregulation of AK, the ABT-702-treated group showed decreased signs of inflammation compared to control animals receiving the vehicle. The involvement of adenosine signaling in the anti-inflammation effect of ABT-702 was supported by the tumor necrosis factor (TNF)- α release blocking effect of an A2AAR antagonist in AGA-treated microglial cells. These results suggest a role for adenosine kinase in regulating adenosine receptor signaling in the retina.

8.5.2 Ang2 Inhibition

Compromise of the blood-retinal barrier has been associated with elevated vitreous concentrations of angiopoietin-2 (Ang2) in patients with clinically significant macular edema (CSME) [98]. Ang2 promotes angiogenesis and vascular leakage in the presence of VEGF and pro-inflammatory cytokines but facilitates vascular regression in the absence of VEGF [12]. Intravitreal injection of Ang2 in nondiabetic rats increases retinal vascular permeability, and Ang2 also leads to a loss of VE-cadherin function [102]. Ang2 sensitizes endothelial cells to TNF- α -induced expression of ICAM-1, the critical player in the pathogenesis of inflammation-induced retinopathy [38]. Pharmacologic blockade of Ang2 might also prevent pericyte dropout in DR.

A bispecific – anti-VEGF and anti-Ang2 – antibody is currently in phase II testing for patients with DME (AVENUE Trial, Regeneron).

8.5.3 Antioxidants

Evidence from animal studies speaks both for and against the use of antioxidants to prevent experimental diabetic retinopathy [50, 73]. This use of antioxidants has not been supported by clinical trials [128].

Despite disappointing data from human studies, controversy over the advantages of calcium dobesilate (CaD) in the treatment of DR remains. Several reports suggest that CaD slows the progression of DR [13, 16, 40, 106, 112, 133]. One study [41] suggested that CaD might protect against endothelial cell dysfunction, reduce apoptosis, and retard the local proliferation of vascular endothelial cells, but others [78, 104] failed to show that CaD benefits the capillary resistance in diabetic patients or decreases the progression of DR. In a recent double-blind, multicenter trial [49], CaD could not prevent or reduce the development of macular edema in patients with nonproliferative DR during a 5-year follow-up period. In a trial of obese nondiabetic male smokers [114], CaD (1000 mg/d) did not improve endothelial function.

8.5.4 ASP8232

ASP8232 belongs to a novel class of orally administered vascular adhesion protein-1 inhibitors. It is being evaluated in a phase 2 multicenter, randomized, controlled trial (the VIDI study) for the treatment of DME. The safety and efficacy of ASP8232 + sham are being compared to ASP8232 + ranibizumab and placebo + ranibizumab. The trial has an enrollment target of 84 patients and is expected to reach the primary completion date in 2017 [11].

8.5.5 Darapladib

Darapladib, a specific inhibitor of lipoprotein-associated phospholipase CA2 (Lp-PLA₂), is protective against atherogenesis and vascular leakage in diabetic and hypercholesterolemic animal models. It effectively suppresses BRB breakdown in streptozotocin-diabetic brown Norway rats, comparable to that of intravitreal anti-VEGF therapy [24].

8.5.6 Epalrestat

Neuropathy, retinopathy, and nephropathy, which are all microvascular complications of diabetes, may be mutually and closely related, with diabetic neuropathy acting as a possible trigger for the onset or progression of the other complications. Incubation of rat aortic smooth muscle cells in the presence of a high glucose concentration significantly increases protein kinase C activity and expression of the protein kinase C bII isoform, and these increases are suppressed by epalrestat [86]. In a study of human coronary artery smooth muscle cells [137], epalrestat inhibited an increase in membrane-bound protein kinase C. An aldose reductase inhibitor also reduced hyperglycemia-induced apoptosis in cultured bovine retinal pericytes [88] and inhibited upregulation of genes in the transforming growth factor- β pathway and apoptosis in retinal vessels of diabetic rats [43]. Therefore, increased activity of the polyol pathway may also be closely related to increased activity of protein kinase Cb and transforming growth factor- β in microangiopathies.

Epalrestat, an aldose reductase inhibitor, was found to be effective for the treatment of both diabetic neuropathy and for early retinopathy [55–57]. Progression of diabetic retinopathy and nephropathy was significantly inhibited in a group treated with epalrestat compared with a control group (odds ratio = 0.323, P = 0.014) and was dependent on the severity of diabetic neuropathy at the end of the study (odds ratio = 2.131, P = 0.025) [57].

8.5.7 Fasudil

Fasudil (Asahi Kasei Pharma Corporation, Tokyo, Japan) is a rho-kinase (ROCK) inhibitor that is used to treat cerebral vasospasm after aneurysm rupture and stroke. It has also been used for primary pulmonary hypertension and memory deficits in patients with Alzheimer's disease. Experimental studies have demonstrated that fasudil can suppress leukocyte adhesion and prevent neutrophil-induced retinal capillary endothelial cell damage [10]. Fasudil may directly protect vascular endothelial cells by reversing endothelial nitric oxide synthase activity.

In a small, prospective study, patients with DME received single intravitreal injections of bevacizumab combined with fasudil (0.025 mg). Compared to baseline, eyes had significant improvements in mean BCVA (0.84 logMAR to 0.49 logMAR; P = 0.003) and mean CRT (448 µm to 347 µm; P = 0.001) at 4 weeks [91].

8.5.8 iCo-007

The RAF proto-oncogene serine/threonine-protein kinase, also known as protooncogene c-RAF or c-Raf, is a principal component of the first described mitogenactivated protein kinase (MAPK) pathway: ERK1/2 signaling. c-Raf acts as a MAP3 kinase, thereby initiating the entire kinase cascade. It has been hypothesized that several growth factors including VEGF, basic fibroblastic growth factor (bFGF), insulin-like growth factor (IGF)-1, epopoietin (EPO), hepatocyte growth factor (HGF), and integrin, signal through c-Raf. The antisense oligonucleotide iCo-007 inhibits c-Raf expression and blocks MAP kinase signaling. iCo-007 has a favorable ocular pharmacokinetic profile and an intraocular half-life of 6–8 weeks in rabbits and monkeys after intravitreal injection.

iCo-007 completed a phase 1, open-label, dose-escalation study in 15 patients with diffuse DME with a 6-month follow-up after a single intravitreal injection (doses ranging between 110 µg and 1000 µg). The study included patients with diffuse DME within 300 µm of the foveal center, OCT-measured macular thickness at baseline of >250 µm, and BCVA at baseline of 60 ± 15 ETDRS letters (20/63 to 20/500 Snellen). Enrolled patients were divided into four cohorts (a total of 15 patients, six in the last cohort).

No drug-related adverse effects were seen during the study. Pharmacokinetic results indicated that iCo-007 concentrations were below the detectable level of 2 ng/mL in the blood. At a secondary endpoint at week 24, mean reduction of excess retinal thickness compared to baseline was 40%, with a 69% improvement in BCVA [18].

A multicenter, phase II trial evaluated iCo-007 as monotherapy and in combination with ranibizumab or laser for patients with DME involving the foveal center (the iDEAL Study). When using multiple imputation analysis, there was no statistically significant difference between 350 μ g iCo-007 monotherapy and each of the 700 μ g monotherapy, 350 μ g plus laser, and 350 μ g plus ranibizumab arms. At 8 months, in the 700 μ g monotherapy arm, 64% of patients experienced a 15-letter or greater loss of vision, compared to 33% in the 350 μ g monotherapy arm, 33% in the 350 μ g plus laser arm, and 41% in the 350 μ g plus ranibizumab arm. At 4 months, the corresponding numbers were 29%, 9%, 9%, and 14%, respectively [61].

8.5.9 Luminate (ALG-1001)

Integrin peptide therapy is a novel approach to the treatment of DME. Integrins are cell-surface receptors that participate in cell signal transduction, mediation of attachments between cells, and regulation of the cell cycle. Integrins interact extracellularly with important proteins, such as collagen and fibronectin, and by intracellularly regulating cell survival, proliferation, and trafficking.

Luminate (ALG-1001, Allegro Ophthalmics, San Juan Capistrano, CA) is a firstin-class integrin peptide therapy that targets integrin receptors involved in cell signaling and regulation and in the formation of new blood vessels. Luminate may be useful in the treatment of vitreomacular traction diseases by promoting vitreolysis and in macular vascular diseases by interfering with angiogenesis. ALG-1001 binds to all integrin receptors involved with retinal angiogenesis and has a long-lasting effect.

Early clinical trials showed that luminate inhibits new vessel growth and decreases vascular leakage. A phase I study evaluated the safety and efficacy of luminate in 15 subjects with advanced DME. Patients had BCVA of 20/100 or worse, some had early proliferative DR, and many were already refractory to standard-of-care therapy. After a period of 90 days without any DME therapy, the patients received three intravitreal 2.5 mg injections of luminate at monthly intervals as monotherapy. Follow-up continued for 3 months after the last injection.

No subjects in the study lost BCVA or experienced an increase in CMT. No serious or significant adverse events were seen during follow-up. Mean BCVA improved from 20/200 at baseline to 20/125 at 60 days (last treatment) and remained stable through 150 days. The mean central macular thickness of 519 μ m at baseline decreased to 387 μ m at 150 days [76].

Luminate is presently being evaluated in a phase IIb clinical trial against bevacizumab and focal laser for DME [6]. The enrollment goal for the phase IIb DME trial (150 patients) was met in late 2015, and top-line data may be released by the third quarter of 2016.

8.5.10 Mecamylamine

In a multicenter phase I/II trial, the safety and bioactivity of topical mecamylamine, an antagonist of n-acetyl choline (ACh) receptors, was tested in 23 patients with DME [23]. Mecamylamine (1%) was administered topically twice daily for

12 weeks. Patients underwent safety assessments and measurements of BCVA and foveal thickness at baseline, 1, 4, 8, 12, and 16 weeks.

Mecamylamine drops were well tolerated, and there were no drug-related safety problems. Mean improvement in BCVA at 1, 4, 8, 12, and 16 weeks was +2.8, +1.9, +2.4, +0.8, and +3.1 letters, respectively. There was little change in mean excess foveal thickness, but there was substantial heterogeneity in response as eight patients had improved BCVA, foveal thickness, or both. Nine patients experienced no significant changes and four patients worsened. Five patients had a significant improvement in BCVA, foveal thickness, or both between weeks 12 (last visit while receiving mecamylamine) and 16 (1 month after stopping mecamylamine). The study suggested that administration of topical mecamylamine, a nonspecific nACh receptor blocker, has heterogeneous effects in patients with diabetic macular edema.

8.5.11 Microspheres

Local administration of biodegradable microspheres may be an attractive alternative to multiple injections since they are able to deliver drug in a controlled fashion. Since most of the treatable retinal diseases are multifactorial, microspheres are particularly promising as they can be filled with more than one active substance and complemented with pharmacologically active additives.

Microsphere carriers have been loaded with budesonide and celecoxib for the treatment of diabetic retinopathy. Comparison between nano- and microspheres prepared from poly(lactic acid) (PLA; intrinsic viscosity, 1.1 dL/g) and loaded with budesonide was performed after subconjunctival injection of particles in rats. In this study, microspheres delivered the active substance in a more sustained fashion than nanospheres [72] because the nanospheres were removed more rapidly from the subconjunctival administration site.

Nanoparticles, microspheres, and budesonide in solution (75 mg) were administered to rat eyes, and drug concentrations in different tissues (retina, vitreous, lens, and cornea) were compared at various times after administration. On days 7 and 14, drug levels in the eyes treated with microspheres were higher than those treated with solution and nanoparticles. Sustained release of celecoxib from poly(lactic-coglycolic acid) (PLGA) was evaluated in a diabetic rat model [8]. A posterior subconjunctival injection of 0.05 mL of the celecoxib-microsphere suspension inhibited diabetes-induced VEGF elevations and blood-retinal barrier leakage.

8.5.12 Minocycline

Microglial activation due to induced inflammation within the retina usually precedes the microvascular findings in DR [42, 74, 139]. Tetracycline reduces connective tissue breakdown [109], protein glycation [108] and excessive collagen synthesis [30] and limits microglial-mediated cell death, retinal cell apoptosis, and capillary damage by inhibiting caspase [39, 74, 134]. Minocycline, a commonly used second-generation tetracycline, has been demonstrated in cell culture and animal models to have anti-inflammatory properties that are independent of its antibacterial property [37, 74].

Oral minocycline (100 mg twice daily for 6 months) was investigated in a single-center, prospective, open-label, phase I/II clinical trial of five participants with fovea-involving DME [31]. Minocycline hydrochloride (Ranbaxy Pharmaceutical Inc., Princeton, NJ; National Drug Code 63304 696 50) was reformulated by the NIH Research Pharmacy as capsules for oral administration. Main outcome measurements included changes in BCVA, CST, and central macular volume using SD-OCT. Mean BCVA improved continuously from baseline through 1, 2, 4, and 6 months by +1.0, +4.0, +4.0, and +5.8 letters, respectively, while CST decreased by 2.9%, 5.7%, 13.9, and 8.1% for the same time points. At month 6, the mean area of late leakage on fluorescein angiography decreased by 34.4% in study eyes. Mean changes in fellow eyes demonstrated similar trends. Improvements in outcome measures were not correlated with concurrent changes in systemic factors. The study drug was well tolerated and not associated with significant safety issues.

Two trials with oral doxycycline, however, produced conflicting results [115, 116].

8.5.13 PF-04523655

PF-04523655 is an O-methyl stabilized, 19-nucleotide, double-stranded, small interfering ribonucleic acid (siRNA) that inhibits expression of a hypoxia-inducible gene, RTP801, via RNA interference [121]. RTP801 expression is upregulated in streptozotocin-induced diabetic mice and rats [19] but is suppressed by intravitreal PF-04523655. Blocking the RTP801 hypoxia/stress pathway with PF-04523655 could provide a new treatment option that is independent of, and possibly complementary to, anti-VEGF therapies for the treatment of DME.

In a multicenter, prospective, masked, phase II clinical trial, 184 patients with DME were randomized to receive PF-04523655 (0.4 mg, 1 mg, or 3 mg) or focal/ grid laser photocoagulation [89]. The main outcome measure was the change in BCVA from baseline to month 12.

All doses of PF-04523655 improved BCVA from baseline through month 12. The 3 mg PF-04523655 group showed a trend for greater improvement in BCVA from baseline compared to laser (respectively, +5.77 vs. +2.39 letters; P = 0.08), but the study was terminated early at month 12 based on predetermined futility criteria for efficacy and discontinuation rates. PF-04523655 was generally safe and well tolerated with few treatment-related adverse events.

There was no evidence that the changes in CST, macula volume, fluorescein angiogram leakage area, or diabetic retinopathy were related to the dose of PF-04523655. The lack of correlation between the changes in DME structural

measures and PF-04523655 doses suggests that PF-04523655 may be working through a mechanism that does not involve vascular permeability, such as increased pigment epithelial-derived factor (PEDF) expression combined with suppression of the RTP 801 gene. PF-04523655 showed a dose-related tendency for improvement in BCVA in DME patients, and studies with higher doses have been considered to determine the optimum dose [27].

8.5.14 Plasma Kallikrein Inhibitor

Plasma kallikrein, a serine protease, is an important component of the body's inflammatory response. Kallikrein circulates as an inactive enzyme (pre-kallikrein) that becomes activated at the site of vascular injury and initiates a cascade that increases the levels of bradykinin. This potent, vasoactive protein dilates blood vessels and increases vascular permeability, edema, and inflammation. Plasma kallikrein is regulated by a C1-inhibitor.

Increased plasma kallikrein activity is implicated in many diseases including DME, hereditary angioedema, and cerebral hemorrhage. Several components of the kallikrein-kinin system (KKS), including plasma kallikrein, factor XII, and kininogen, are found in the vitreous of patients with advanced diabetic retinopathy. Preclinical studies in rodents showed that plasma kallikrein activation in the vitreous increases retinal vascular permeability, whereas kallikrein inhibition reduces diabetes- and hypertension-induced retinal leakage.

Small molecule plasma kallikrein inhibitors delivered intravitreally and orally have the potential to treat these and other retinal vascular problems. KalVista Pharmaceuticals is developing plasma kallikrein inhibitors for the treatment of DME and hereditary angioedema. A phase I study at 5 US sites established the safety of an intravitreally injected kallikrein inhibitor (KVD001). Three cohorts (14 total patients) previously treated with anti-VEGF injections received single injections of 1, 3, and 10 μ g. Thirty-six minor adverse events were recorded. At day 84, the mean BCVA and CST improved by +4 letters and -40 μ m, respectively. Systemic drug absorption was low [125].

Two phase II trials are being planned: one combines a plasma kallikrein inhibitor with an anti-VEGF drug; the other employs plasma kallikrein inhibitor monotherapy for eyes with anti-VEGF-resistant DME [69].

8.5.15 Ruboxistaurin

Ruboxistaurin (RBX) is an orally administered, isoform-selective inhibitor of protein kinase C that positively affects animal models of diabetic retinopathy [3] and improves diabetes-induced retinal hemodynamic abnormalities in patients with diabetes [4]. Two randomized controlled studies and a combined analysis of two additional studies have suggested that oral RBX reduces the rate of sustained moderate vision loss in patients with diabetes [100, 101, 120]. These studies did not meet their primary endpoints; consistent improvements in macular edema and reduction in need for laser photocoagulation were not seen.

8.5.16 Sirolimus

The mTOR inhibitors are uniquely suited to address both early and advanced manifestations of DR. The mTOR inhibitors have the potential to delay or prevent the progression of retinal microangiopathies by averting breakdown of the blood-retinal barrier by modulating HIF-1 α -mediated downstream activation of growth factors. As DR progresses and proliferative lesions develop, inhibiting the PI3K/Akt/mTOR pathway may promote neovascular regression by downregulating pro-survival growth factors, modulating the inflammatory cascade, preventing angiogenesis, and promoting apoptosis of nascent vessels [62].

A randomized, open-label, dose-escalating, phase I study evaluated the safety and tolerability of sirolimus (Perceiva, Macusight, Union City CA) for the treatment of DME [33]. Single subconjunctival (SCJ: 220, 440, 880, 1320, or 1760 μ g) or intravitreal (IVT: 44, 110, 176, 264, or 352 μ g) injections were given to 50 eyes of 50 patients with retinal thickness of at least 300 μ m and BCVA of 20/40 to 20/200. The primary endpoints were the frequency and severity of ocular and systemic adverse events at 90 days. Secondary endpoints were changes in BCVA and retinal thickness.

No dose-limiting toxicities were observed, and ocular adverse events were mostly mild and transient. Conjunctival hyperemia, hemorrhage, and edema were common after SCJ injections, and conjunctival hemorrhage was common after IVT injections. Three patients experienced ocular adverse events considered possibly related to study drug: conjunctival edema and reduced visual acuity were reported in one SCJ patient each, and iritis was reported in one IVT patient. No serious ocular adverse events were reported. No non-ocular adverse events were considered related to study drug. Systemic exposure to sirolimus was low with blood concentrations below levels necessary for systemic immunosuppression.

For the SCJ group, a median increase in BCVA started at day 7 (+5.0 letters) and was +3.0, +4.0, and +4.0 letters at days 14, 45, and 90, respectively. At day 45, median decrease in retinal thickness was $-23.7 \mu m$. For the IVT group, the median increase in BCVA was +2.0 letters at day 7; at days 14, 45, and 90, the median increase was maintained (+4.0 letters); the median decrease in retinal thickness was $-52.0 \mu m$ at day 45.

The investigators concluded that locally administered sirolimus was well tolerated with minimal systemic exposure at all doses tested in this small phase I population. These findings support advancing the present sirolimus formulation into phase II studies.

8.5.17 Squalamine

Squalamine (Ohr Pharmaceuticals) is a small antiangiogenic molecule that interferes with the actions of several growth factors including VEGF, platelet-derived growth factor (PDGF), and basic fibroblastic growth factor (bFGF). It enters the cell and sequesters intracellular calmodulin to inactivate the growth factor receptors. It has been evaluated in early phase nAMD trials, and investigator-initiated trials have studied its effect on DR.

8.5.18 Tie2 Agonist (AKB-9778)

Tyrosine kinase with immunoglobulin-like and epidermal growth factor-like domains 2 (Tie2) is a transmembrane receptor located on vascular endothelial cells that has been associated with vascular permeability. Unlike activation of the VEGF receptors, activation of Tie2 stabilizes vasculature and decreases leakage. Angiopoietin 1 stimulates Tie2 phosphorylation, whereas angiopoietin 2 only partially stimulates Tie2 phosphorylation and, therefore, competes with angiopoietin 1 to suppress Tie2 phosphorylation.

A phase IIa, randomized, placebo and sham injection-controlled, double-masked clinical trial assessed the effect of the Tie2 agonist AKB-9778 alone or in combination with ranibizumab in subjects with DME. Patients (n = 144) with decreased VA from DME and CST \geq 325 µm measured by SD-OCT were enrolled at 36 sites. Patients were randomized to receive AKB-9778 monotherapy (subcutaneous AKB-9778 15 mg twice per day (BID) + monthly sham intraocular injections), combination therapy (subcutaneous AKB-9778 15 mg BID + monthly 0.3 mg ranibizumab), or ranibizumab monotherapy (subcutaneous placebo injections BID + monthly 0.3 mg ranibizumab). Best corrected visual acuity and CST were measured at baseline and every 4 weeks. The primary outcome measure was the mean change from baseline CST at week 12. Other predetermined outcome measures included changes in BCVA, diabetic retinopathy severity score (DRSS), and safety assessments.

At week 12, mean change from baseline CST was significantly greater in the combination group $(-164.4 \pm 24.2 \ \mu\text{m})$ compared with the ranibizumab monotherapy group $(-110.4 \pm 17.2 \ \mu\text{m}; P = 0.008)$ but was only $6.2 \pm 13.0 \ \mu\text{m}$ in the AKB-9778 monotherapy group. The mean CST at week 12 and percentage of eyes with resolved edema was $340.0 \pm 11.2 \ \mu\text{m}$ and 29.2%, respectively, in the combination group versus $392.1 \pm 17.1 \ \mu\text{m}$ and 17.0%, respectively, in the ranibizumab monotherapy group. The mean change from baseline BCVA (ETDRS letters) was $+6.3 \pm 1.3$ in the combination group, $+5.7 \pm 1.2$ in the ranibizumab monotherapy group, and $+1.5 \pm 1.2$ in the AKB-9778 monotherapy group. The percentage of study eyes that gained ≥ 10 or ≥ 15 letters in the AKB-9778 monotherapy group, ranibizumab monotherapy group, and 4.3%, 17.0%, and 20.8%. Improvements in DRSS in study

eyes were similar across groups, and the percentage of qualified fellow eyes with a \geq 2-step change was 11.4% in all AKB-9778-treated subjects compared with 4.2% in the ranibizumab monotherapy group. AKB-9778 was well tolerated, with no clear differences in adverse events.

The authors concluded that activation of Tie2 by subcutaneous injections of AKB-9778 combined with suppression of VEGF reduces DME greater than that seen with anti-VEGF monotherapy [22].

8.5.19 Tocilizumab

The READ-4 study compares ranibizumab and tocilizumab (TCZ), an interleukin-6 inhibitor, for the treatment of DME. The study will randomize patients to receive ranibizumab, TCZ, or a combination, with the primary endpoint analysis at month 6. Enrollment for the study is expected to begin in the first quarter of 2016 [103].

8.5.20 Teprotumumab

The safety and efficacy of insulin-like growth factor (IGF-1) inhibitors are being evaluated in patients with DME. The intravenously administrated IGF-1 inhibitor teprotumumab (RV001) is being evaluated in an open-label phase 1 study in three centers in the United States [129].

8.6 Future Therapies

Experimental work has shown that leukostasis-related death of endothelial cells can be prevented by blocking or genetically eliminating either ICAM-1 or CD18 [1, 68] and by administering anti-inflammatory agents. High-dose aspirin reduces expression of CD11a, CD11b, and CD18 [68]; high-dose aspirin, etanercept (soluble TNFR-Fc, tumor necrosis factor-1 linked to a human FC region), and meloxicam (a cyclooxygenase 2 inhibitor) reduce leukostasis and suppress BRB breakdown in diabetic rats [68].

Repopulating diabetes-damaged acellular capillaries with vascular progenitor cells constitutes another therapeutic approach [64, 119]. Unfortunately, diabetic endothelial progenitor cells (EPCs) are deficient in homing ability and have limited engraftment capacity [20], but this can be increased by treating the cells with nitric oxide [117]. Vascular progenitors have been generated from CD34b cord blood cells that have been induced nonvirally into pluripotent stem cells (iPSCs) [95]. iPSCs were trained to be vascular progenitors on fibronectin substratum with high levels of VEGF [96]. When CD31b/CD146b cells were injected into the vitreous of NOD/

Scid mice that had experienced ischemia-reperfusion injury to retina resulting in acellular capillaries, the cells homed to the ablumenal surface of the acellular capillaries, taking a pericyte position. When the cells were delivered intravenously, they were engrafted in a lumenal position, suggesting that they were assuming the role of endothelial cells. The hypoxic adjacent retina makes both stromal-derived factor-1 (SDF-1) and VEGF, which would provide the stimulus for homing of vascular progenitors to the acellular capillaries. Future work will focus on repopulating acellular capillaries with iPSC vascular progenitors in diabetic animals [83].

Neuroprotective factors such as pigment epithelium-derived factor (PEDF) [90, 99, 126, 138], somatostatin (SST) [71], nerve growth factor (NGF) [48], brainderived neurotrophic factor [45], and epopoietin [85, 135, 140] have been used to treat experimental DR. Intraocular gene transfer of PEDF significantly increases neuroretinal cell survival after an ischemia-reperfusion injury [126]. Intravitreal injections of PEDF prevent neuronal derangements and vascular hyperpermeability in early DR [138]. Intravitreally administered somatostatin and somatostatin analogs protect the retina from alpha-amino-3-hydroxyl-5-methyl-4-isoxazole-propionate-induced neurotoxicity [71]. Treating diabetic rats with NGF prevents ganglion cell and Müller cell apoptosis [48]. These promising results suggest that enhancing the expression and function of the neuroprotective factors synthesized by the retina could be a therapeutic strategy to treat DR.

The ability to clinically identify retinal neurodegeneration will be crucial for developing early treatment strategies with drugs that possess neuroprotective effects. At this stage of disease, however, patients are still asymptomatic, so invasive treatments such as intravitreal injections are not appropriate. Emerging experimental evidence indicates that many drugs, including some that are administered topically, can reach the retina in adequate concentrations [34]. In fact, the retinal neuroprotective effects of topically administered brimonidine, NGF, PEDF, insulin, and SST have already been reported in experimental models [5, 32, 39, 52, 77, 113, 123]. Topical administration limits drug actions primarily to the eye and minimizes unwanted systemic effects, which should result in higher patient compliance [2]. Effective topical therapies could revolutionize the care of diabetic patients, but clinical trials are needed to test the safety and effectiveness of neuroprotective agents.

References

- 1. Adamis AP. Is diabetic retinopathy an inflammatory disease? Br J Ophthalmol. 2002;86: 363–5.
- 2. Aiello LP. Targeting intraocular neovascularization and edema one drop at a time. N Engl J Med. 2008;359:967–9.
- Aiello LP, Bursell SE, Clermont A, et al. Vascular endothelial growth factor induced retinal permeability is mediated by protein kinase C in vivo and suppressed by an orally effective betaisoform-selective inhibitor. Diabetes. 1997;46:1473–80.
- Aiello LP, Clermont A, Arora V, et al. Inhibition of PKC beta by oral administration of ruboxistaurin is well tolerated and ameliorates diabetes-induced retinal hemodynamic abnormalities in patients. Invest Ophthalmol Vis Sci. 2006;47:86–92.

- 5. Aizu Y et al. Topical instillation of ciliary neurotrophic factor inhibits retinal degeneration in streptozotocin-induced diabetic rats. Neuroreport. 2003;14:2067–71.
- 6. http://www.allegroeye.com/press-release/allegro-ophthalmics-announces-last-patient-enrolled-in-del-mar-phase-2b-clinical-trial-of-luminate-for-the-treatment-of-diabeticmacular-edema/#sthash.YKXN8VQd.dpuf. Accessed 10 Jan 2016.
- FDA advisory to Ampio Pharmaceuticals. http://campaign.r20.constantcontact.com/ render?ca=6f759dbd-431a-417b-94b1-bfd8be310ff7&c=de0e4120-b64d-11e3-bf1ed4ae527b6fcc&ch=deced200-b64d-11e3-bf7e-d4ae527b6fcc. Accessed 23 Oct 2015.
- Amrite AC, Ayalasomayajula SP, Cheruvu NP, Kompella UB. Single periocular injection of celecoxib-PLGA microparticles inhibits diabetes-induced elevations in retinal PGE2, VEGF, and vascular leakage. Invest Ophthalmol Vis Sci. 2006;47:1149e1160.
- 9. Arevalo JF, Serrano MA, Wu L. Combined inhibition of tumor necrosis factor (TNF) and vascular endothelial growth factor (VEGF) for the treatment of macular edema of various etiologies: a short-term pilot study. Eye (Lond). 2013;27:569–71.
- Arita R, Hata Y, Nakao S, et al. Rho kinase inhibition by fasudil ameliorates diabetes-induced microvascular damage. Diabetes. 2009;58(1):215–26.
- A Study to Evaluate ASP8232 in Reducing Central Retinal Thickness in Subjects With Diabetic Macular Edema (DME) (VIDI) 2015. [Last accessed on 2015 May 30]. Available from: https://www.clinicaltrials.gov/ct2/show/NCT02302079.
- Augustin HG, Koh GY, Thurston G, Alitalo K. Control of vascular morphogenesis and homeostasis through the angiopoietin-Tie system. Nat Rev Mol Cell Biol. 2009;10:165–77.
- Barras JP, Michal M. Effect of calcium dobesilate on blood viscosity in diabetic microangiopathy. A review. Vasa. 1986;15:200–5.
- 14. Behl Y, Krothapalli P, Desta T, DiPiazza A, Roy S, Graves DT. Diabetes-enhanced tumor necrosis factor-alpha production promotes apoptosis and the loss of retinal microvascular cells in type 1 and type 2 models of diabetic retinopathy. Am J Pathol. 2008;172:1411–8.
- Behl Y, Krothapalli P, Desta T, Roy S, Graves DT. FOXO1 plays an important role in enhanced microvascular cell apoptosis and microvascular cell loss in type 1 and type 2 diabetic rats. Diabetes. 2009;58:917–25.
- Binkhorst PG, Van Binsterveld OP. Calcium dobesilate versus placebo in the treatment of diabetic retinopathy: a double-blind cross-over study. Curr Ther Res Clin Exp. 1976;20:283–8.
- Binz HK, Stumpp MT, Forrer P, Armstutz P, Pluckthun A. Designing repeat proteins: wellexpressed, soluble and stable proteins from combinatorial libraries of consensus ankyrin repeat proteins. J Mol Biol. 2003;332(2):489–503.
- Boyer D. iCo-007 for treatment of diffuse diabetic macular edema: phase 1, dose escalation, open label clinical trial. iCo Therapeutics Web site. Available at: http://www.icotherapeutics. com/_resources/presentations/presentation_david_boyer.pdf. Accessed 21 May 2013.
- Brafman A, Mett I, Shafir M, et al. Inhibition of oxygen-induced retinopathy RTP801- deficient mice. Invest Ophthalmol Vis Sci. 2004;45:3796–805.
- Caballero S, Sengupta N, Afzal A, et al. Ischemic vascular damage can be repaired by healthy, but not diabetic, endothelial progenitor cells. Diabetes. 2007;56:960–7.
- Campochiaro PA, Channa R, Berger BB, Heier JS, Brown DM, Fiedler U, Hepp J, Stumpp MT. Treatment of Diabetic Macular Edema With a Designed Ankyrin Repeat Protein That Binds Vascular Endothelial Growth Factor: A Phase 1/2 Study. Am J Ophthalmol. 2013;155(4):697–704.
- 22. Campochiaro PA, Khanani A, Singer M, Patel S, Boyer D, Dugel P, Kherani S, Withers B, Gambino L, Peters K, Brigell M. TIME-2 Study Group. Enhanced Benefit in Diabetic Macular Edema from AKB-9778 Tie2 Activation Combined with Vascular Endothelial Growth Factor Suppression. Ophthalmology. 2016; [Epub ahead of print].
- Campochiaro PA, Shah SM, Hafiz G, Heier JS, Lit ES, Zimmer-Galler I, Channa R, Nguyen QD, Syed B, Do DV, Lu L, Monk J, Cooke JP, Kengatharan MK, Hsu HH. Topical mecamylamine for diabetic macular edema. Am J Ophthalmol. 2010;149(5):839–51.

- 24. Canning P, Kenny B-A, Prise V, Glenn J, Sarker MH, Hudson N, Brandt M, Lopez FJ, Gale D, Luthert PJ, Adamson P, Turowski P, Stitt AW. Lipoprotein-associated phospholipase A2 (Lp-PLA2) as a therapeutic target to prevent retinal vasopermeability during diabetes. Proc Natl Acad Sci. 2016;113(26):7213–8. doi:10.1073/pnas.1514213113.
- 25. Chen X, Li J, Li M, Zeng M, Li T, Xiao W, Li J, Wu Q, Ke X, Luo D, Tang S, Luo Y. KH902 suppresses high glucose-induced migration and sprouting of human retinal endothelial cells by blocking VEGF and PIGF. Diabetes Obesity Metab. 2013;15:224–33.
- Chhablani J, Narayanan R, Mathai A, Yogi R, Stewart M. Short-term safety profile of intravitreal ziv-aflibercept. Retina. 2016;36(6):1126–31.
- 27. ClinicalTrials.gov Identifier: NCT00701181. Accessed 4 July 2016.
- 28. ClinicalTrials.gov Identifier: NCT02228304. Accessed 24 Jan 2016.
- 29. ClinicalTrials.gov Identifiers: NCT02307682, NCT02434328. Accessed 24 Feb 2016.
- Craig RG, Yu Z, Xu L, et al. A chemically modified tetracycline inhibits streptozotocininduced diabetic depression of skin collagen synthesis and steady-state type I procollagen mRNA. Biochim Biophys Acta. 1998;1402(3):250–60.
- 31. Cukras CA, Petrou P, Chew EY, Meyerle CB, Wong WT. Oral minocycline for the treatment of diabetic macular edema (DME): results of a phase I/II clinical study. Invest Ophthalmol Vis Sci. 2012;53(7):3865–74.
- 32. Dong CJ et al. Alpha 2 adrenergic modulation of NMDA receptor function as a major mechanism of RGC protection in experimental glaucoma and retinal excitotoxicity. Invest Ophthalmol Vis Sci. 2008;49:4515–22.
- 33. Dugel PU, Blumenkranz MS, Haller JA, Williams GA, Solley WA, Kleiman DM, Naor J. A randomized, dose-escalation study of subconjunctival and intravitreal injections of sirolimus in patients with diabetic macular edema. Ophthalmology. 2012;119:124–31.
- 34. Eljarrat-Binstock E et al. New techniques for drug delivery to the posterior eye segment. Pharm Res. 2010;27:530–43.
- Elsherbiny NM, Ahmad S, Naime M, Elsherbini AM, Fulzele S, Al-Gayyar MM, Eissa LA, El-Shishtawy MM, Liou GI. ABT-702, an adenosine kinase inhibitor, attenuates inflammation in diabetic retinopathy. Life Sci. 2013;93:78–88.
- http://www.eyegatepharma.com/uncategorized/eyegate-announces-interim-data-from-phase-1b-2a-clinical-trial-of-iontophoretic-egp-437-ophthalmic-solution-in-macular-edemapatients/. Accessed 24 Nov 2015.
- Federici TJ. The non-antibiotic properties of tetracyclines: clinical potential in ophthalmic disease. Pharmacol Res. 2011;64(6):614–23.
- 38. Fiedler U, Reiss Y, Scharpfenecker M, Grunow V, Koidl S, Thurston G, Gale NW, Witzenrath M, Rosseau S, Suttorp N, Sobke A, Herrmann M, Preissner KT, Vajkoczy P, Augustin HG. Angiopoietin-2 sensitizes endothelial cells to TNF-α and has a crucial role in the induction of inflammation. Nat Med. 2006;12(2):235–9.
- 39. Fort PE et al. Differential roles of hyperglycemia and hypoinsulinemia in diabetes induced retinal cell death: evidence for retinal insulin resistance. PLoS One. 2011;6:e26498.
- Freyler H. Microvascular protection with calcium dobesilate (Doxium) in diabetic retinopathy. Ophthalmologica. 1974;168:400–16.
- Garay RP, Hannaert P, Chiavaroli C. Calcium dobesilate in the treatment of diabetic retinopathy. Treat Endocrinol. 2005;4:221–32.
- Gaucher D, Chiappore JA, Paques M, et al. Microglial changes occur without neural cell death in diabetic retinopathy. Vision Res. 2007;47(5):612–23.
- 43. Gerhardinger C, Dagher Z, Sebastiani P, Park YS, Lorenzi M. The transforming growth factor-beta pathway is a common target of drugs that prevent experimental diabetic retinopathy. Diabetes. 2009;58:1659–67.
- 44. Ginka T, inventor; Mpex Pharmaceuticals, Inc., assignee. Bacterial efflux pump inhibitors and methods of treating bacterial infections. US patent 7,893,020 B2. 22 Feb 2011.
- 45. Gong Y et al. Protective effect of adeno-associated virus mediated brain-derived neurotrophic factor expression on retinal ganglion cells in diabetic rats. Cell Mol Neurobiol. 2012;32:467–75.

- 46. Grant MB, Afzal A, Spoerri P, Pan H, Shaw LC, Mames RN. The role of growth factors in the pathogenesis of diabetic retinopathy. Expert Opin Investig Drugs. 2004;13:1275–93.
- 47. Gutiérrez-Hernándes J-C, Caffey S, Abdallah W, et al. Pump for intravitreal drug delivery: a pilot study. Transl Vis Sci Technol. 2014;3:1–13.
- 48. Hammes HP et al. Nerve growth factor prevents both neuroretinal programmed cell death and capillary pathology in experimental diabetes. Mol Med. 1995;1:527–34.
- 49. Haritoglou C, Gerss J, Sauerland C, Kampik A, Ulbig MW. CALDIRET study group. Effect of calcium dobesilate on occurrence of diabetic macular oedema (CALDIRET study): randomized, double-blind, placebo-controlled, multi-centre trial. Lancet. 2009;373:1364–71.
- Haskins K, Bradley K, Powers K, et al. Oxidative stress in type 1 diabetes. Ann N Y Acad Sci. 2003;1005:43–54.
- Heier J. Presented at the 2015 American Academy of Ophthalmology Annual Meeting. Las Vegas, NV. 15 Nov 2015.
- 52. Hernández C et al. Topical administration of somatostatin prevents retinal neurodegeneration in experimental diabetes. 2013;62:2569–78.
- Holash J, Davis S, Papadopoulos N, et al. VEGF-Trap: a VEGF blocker with potent antitumor effects. Proc Natl Acad Sci U S A. 2002;99:11393–8.
- 54. Hotta N. Is there a place for inhibition of transforming growth factor-β and the polyol pathway in therapy for diabetic retinopathy? J Diabetes Investiq. 2010;1:134–6.
- 55. Hotta N, Akanuma Y, Kawamori R, Matsuoka K, Oka Y, Shichiri M, et al. Long-term clinical effects of epalrestat, an aldose reductase inhibitor, on diabetic peripheral neuropathy: the 3-year, multicenter, comparative Aldose Reductase Inhibitor–Diabetes Complications Trial. Diabetes Care. 2006;29:1538–44.
- 56. Hotta N, Kawamori R, Atsumi Y, Baba M, Kishikawa H, Nakamura J, et al. Stratified analyses for selecting appropriate target patients with diabetic peripheral neuropathy for long-term treatment with an aldose reductase inhibitor, epalrestat. Diabet Med. 2008;25:818–25.
- 57. Hotta N, Kawamori R, Fukuda M, Shigeta Y, the Aldose Reductase Inhibitor-Diabetes Complications Trial Study Group. Long-term clinical effects of epalrestat, an aldose reductase inhibitor, on progression of diabetic neuropathy and other microvascular complications: multivariate epidemiological analysis based on patient background factors and severity of diabetic neuropathy. Diabet Med. 2012;29:1529–33.
- Huang J, Li X, Li M, Li S, Xiao W, Chen X, Cai M, Wu Q, Luo D, Tang S, Luo Y. Effects of intravitreal injection of KH902, a vascular endothelial growth factor receptor decoy, on the retina of streptozotocin-induced diabetic rats. Diabetes Obesity Metab. 2012;14:644–53.
- Humayan M, Santos A, Altamirano JC, et al. Implantable micropump for drug delivery in patients with diabetic macular edema. Transl Vis Sci Technol. 2014;3(6):5. eCollection 2014.
- 60. Ibrahim AS, El-Shishtawy MM, Zhang W, Caldwell RB, Liou GI. A2A Adenosine Receptor (A2AAR) as a Therapeutic Target in Diabetic Retinopathy. Am J Pathol. 2011b;178:2136–45.
- https://www.asrs.org/education/clinical-updates/289/ico-therapeutics-announces-top-line-primary-endpoint-data-from-phase-2-ideal-study-in-dme. Accessed 3 July 2016.
- Jacot JL, Sherris D. Potential therapeutic roles for inhibition of the PI3K/Akt/mTOR pathway in the pathophysiology of diabetic retinopathy. J Ophthalmol. 2011; Article ID 589813:1–19.
- Jansook P, Ritthidej GC, Ueda H, Stefansson E, Loftsson T. yCD/HPyCD mixtures as solubilizer: solid-state characterization and sample dexamethasone eye drop suspension. J Pharm Pharm Sci. 2010;13:336–50.
- 64. Jarajapu YP, Caballero S, Verma A, et al. Blockade of NADPH oxidase restores vasoreparative function in diabetic CD34b cells. Invest Ophthalmol Vis Sci. 2011;52:5093–104.
- Johnston-Cox HA, Ravid K. Adenosine and blood platelets. Purinergic Signal. 2011;7:357–65.
- Joussen AM, Doehmen S, Le ML, et al. TNF-alpha mediated apoptosis plays an important role in the development of early diabetic retinopathy and long-term histopathological alterations. Mol Vis. 2009;15:1418–28.

- 67. Joussen AM, Poulaki V, Mitsiades N, et al. Nonsteroidal anti-inflammatory drugs prevent early diabetic retinopathy via TNF-alpha suppression. FASEB J. 2002;16:438–40.
- Joussen AM, Poulaki V, Le ML, et al. A central role for inflammation in the pathogenesis of diabetic retinopathy. FASEB J. 2004;18:1450–2.
- 69. KalVista Pharmaceuticals Web site. Available at: http://www.kalvista.com/news/37/131/ KalVista-Pharmaceuticals-Wins-2-4m-Bio-medical-Catalyst-Grant-to-Further-Develop-Oral-Plasma-Kallikrein-Inhibitors-as-a-Treatment-for-Diabetic-Macular-Edema.html. Accessed 21 May 2013.
- Kauper K, McGovern C, Sherman S, et al. Two-year intraocular delivery of ciliary neurotrophic factor by encapsulated cell technology implants in patients with chronic retinal degenerative diseases. Invest Ophthalmol Vis Sci. 2012;53:7484–91.
- 71. Kiagiadaki F et al. Activation of somatostatin receptor (sst 5) protects the rat retina from AMPA-induced neurotoxicity. Neuropharmacology. 2010;58:297–303.
- Kompella UB, Bandi N, Ayalasomayajula SP. Subconjunctival nano- and microparticles sustain retinal delivery of budesonide, a corticosteroid capable of inhibiting VEGF expression. Invest Ophthalmol Vis Sci. 2003;44:1192–e1201.
- Kowluru RA, Kern TS, Engerman RL, Armstrong D. Abnormalities of retinal metabolism in diabetes or experimental galactosemia. III. Diabetes. 1996;45(9):1233–7.
- 74. Krady JK, Basu A, Allen CM, et al. Minocycline reduces proinflammatory cytokine expression, microglial activation, and caspase-3 activation in a rodent model of diabetic retinopathy. Diabetes. 2005;54(5):1559–65.
- 75. Kristinsson JK, Fridriksdottir H, Thorisdottir S, Sigurdardottir AM, Stefansson E, Loftsson T. Dexamethasone-cyclodextrin-polymer co-complexes in aqueous eye drops: aqueous humor pharmacokinetics in humans. Invest Ophthalmol Vis Sci. 1996;37:1199–203.
- Kuppermann BD. Integrin peptide therapy for the treatment of vascular eye diseases. Retina Today. 2013;2:60–2.
- 77. Lambiase A et al. Experimental and clinical evidence of neuroprotection by nerve growth factor eye drops: implications for glaucoma. Proc Natl Acad Sci U S A. 2008;106:13469–74.
- Larsen HW, Sander E, Hoppe R. The value of calcium dobesilate in the treatment of diabetic retinopathy. A controlled clinical trial. Diabetologia. 1977;13:105–9.
- Li X, Xu G, Wang Y, et al. Safety and efficacy of conbercept in neovascular age-related macular degeneration: results from a 12-month randomized phase II trial study: AURORA study. Ophthalmology. 2014;121(9):1740–7.
- Loftsson T, Hreinsdottir D, Stefansson E. Cyclodextrin microparticles for drug delivery to the posterior segment of the eye: aqueous dexamethasone eye drops. J Pharm Pharmacol. 2007;59:629–35.
- Loftsson T, Sigurdsson HH, Konradsdottir F, Gisladottir S, Jansook P, Stefansson E. Topical drug delivery to the posterior segment of the eye: anatomical and physiological considerations. Pharmazie. 2008;63:171–9.
- Topical loteprednol for the treatment of DME. http://www.marketwatch.com/story/kalapharmaceuticals-initiates-phase-2-clinical-trial-to-evaluate-le-mpp-kpi-121-in-patientswith-retinal-vein-occlusion-and-diabetic-macular-edema-2014-07-31. Accessed 20 Oct 2015.
- 83. Lutty GA. Effects of diabetes on the eye. Invest Ophthalmol Vis Sci. 2013;54:ORSF81-7.
- Mansour AM, Al-Ghadban SI, Yunis MH, El-Sabban ME. Ziv-aflibercept in macular disease. Br J Ophthalmol. 2015;99(8):1055–9.
- McVicar CM et al. Intervention with an erythropoietin derived peptide protects against neuroglial and vascular degeneration during diabetic retinopathy. Diabetes. 2011;60:2995–3005.
- Nakamura J, Kasuya Y, Hamada Y, Nakashima E, Naruse K, Yasuda Y, et al. Glucose-induced hyperproliferation of cultured rat aortic smooth muscle cells through polyol pathway hyperactivity. Diabetologia. 2001;44:480–7.
- Nakano S, Yamamoto T, Kirii E, Abe S, Yamashita H. Steroid eye drop treatment (difluprednate ophthalmic emulsion) is effective in reducing refractory diabetic macular edema. Graefes Arch Clin Exp Ophthalmol. 2010;248(6):805–10.

- Naruse K, Nakamura J, Hamada Y, Nakayama M, Chaya S, Komori T, et al. Aldose reductase inhibition prevents glucose induced apoptosis in cultured bovine retinal microvascular pericytes. Exp Eye Res. 2000;71:309–15.
- 89. Nguyen QD, Schachar RA, Nduaka CI, Sperling M, Basile AS, Klamerus KJ, Chi-Burris K, Yan E, Paggiarino DA, Rosenblatt I, Aitchison R, Erlich SS, DEGAS Clinical Study Group. Dose-ranging evaluation of intravitreal siRNA PF-04523655 for diabetic macular edema (the DEGAS Study). Invest Ophthalmol Vis Sci. 2012;53:7666–74.
- Nguyen TT et al. Flicker light-induced retinal vasodilation in diabetes and diabetic retinopathy. Diabetes Care. 2009;32:2075–80.
- Nourinia R, Ahmadieh H, Shahheidari MH, Zandi S, Nakao S, Hafezi-Moghadam A. Intravitreal fasudil combined with bevacizumab for treatment of refractory diabetic macular edema; a pilot study. J Ophthalmic Vis Res. 2013;8(4):337–40.
- http://ophthalmologytimes.modernmedicine.com/ophthalmologytimes/news/novel-antivegf-agent-may-provide-important-advancement-amd-treatment?page=full. Accessed 28 Jan 2016.
- 93. Ottiger M, Thiel MA, Feige P, Lichtlen P, Urech DM. Efficient intraocular penetration of topical anti-TNF-α single-chain antibody (ESBA105) to anterior and posterior segment without penetration enhancer. Invest Ophthalmol Vis Sci. 2009;50(2):779–86.
- http://www.businesswire.com/news/home/20151112005385/en/PanOptica-Reports-Progress-PAN-90806-Topical-Anti-VEGF-Eyedrop. Accessed 22 Nov 2015.
- Park TS, Huo JS, Peters A, et al. Growth factor-activated stem cell circuits and stromal signals cooperatively accelerate nonintegrated iPSC reprogramming of human myeloid progenitors. PLoS One. 2012;7:e42838.
- Park TS, Zimmerlin L, Zambidis ET. Efficient and simultaneous generation of hematopoietic and vascular progenitors from human induced pluripotent stem cells. Cytometry A. 2013;83:114–26.
- 97. Pascual-Camps I, Hernández-Martínez P, Monje-Fernández L, Dolz-Marco R, Gallego-Pinazo R, Wu L, Arévalo JF, Díaz-Llopis M. Update on intravitreal anti-tumor necrosis factor alpha therapies for ocular disorders. J Ophthalmic Inflamm Infect. 2014;4:26.
- Patel JI, Hykin PG, Gregor ZJ, Boulton M, Cree IA. Angiopoietin concentrations in diabetic retinopathy. Br J Ophthalmol. 2005;89(4):480–3.
- 99. Pemp B et al. Reduced retinal vessel response to flicker stimulation but not exogenous nitric oxide in type 1 diabetes. Invest Ophthalmol Vis Sci. 2009;50:4029–32.
- 100. The PKC-DRS Study Group. The effect of ruboxistaurin on visual loss in patients with moderately severe to very severe nonproliferative diabetic retinopathy: initial results of the Protein Kinase C beta Inhibitor Diabetic Retinopathy Study (PKC-DRS) multicenter, randomized clinical trial. Diabetes. 2005;54:2188–97.
- 101. The PKC-DRS2 Study Group. The effect of ruboxistaurin on visual loss in patients with diabetic retinopathy. Ophthalmology. 2006;113:2221–30.
- 102. Rangasamy S, Srinivasan R, Maestas J, McGuire PG, Das A. A potential role for angiopoietin 2 in the regulation of the blood-retinal barrier in diabetic retinopathy. Invest Ophthalmol Vis Sci. 2011;52(6):3784–91.
- 103. Ranibizumab for Edema of the mAcula in Diabetes: Protocol 4 with Tocilizumab: The READ-4 Study. 2015. [Last accessed on 2015 Aug 15]. https://www.clinicaltrials.gov/ct2/ show/study/NCT02511067.
- 104. Rasch R. Capillary fragility and doxium. A controlled clinical trial. Diabetologia. 1973;9:483–5.
- 105. http://retinatoday.com/2014/08/long-acting-anti-vegf-delivery. Accessed 15 Jan 2016.
- 106. Robeiro ML, Seres AI, Carneiro AM, Stur M, Zourdani A, Caillon P, Cunha-Vaz JG. DX-Retinopathy Study Group. Effect of calcium dobesilate on progression of early diabetic retinopathy: a randomized double-blind study. Graefes Arch Clin Exp Ophthalmol. 2006;244:1591–600.
- 107. Rosberger DF. Diabetic retinopathy. Current concepts and emerging therapy. Endocrinol Metab Clin North Am. 2013;42:721–45.

- Ryan ME, Ramamurthy NS, Golub LM. Tetracyclines inhibit protein glycation in experimental diabetes. Adv Dent Res. 1998;12(2):152–8.
- 109. Ryan ME, Usman A, Ramamurthy NS, Golub LM, Greenwald RA. Excessive matrix metalloproteinase activity in diabetes: inhibition by tetracycline analogues with zinc reactivity. Curr Med Chem. 2001;8(3):305–16.
- 110. Saari KM, Nelimarkka L, Ahola V, Loftsson T, Stefansson E. Comparison of topical 0.7% dexamethasone-cyclodextrin with 0.1% dexamethasone sodium phosphate for postcataract inflammation. Graefes Arch Clin Exp Ophthalmol. 2006;244:620–6.
- 111. Saati S, Lo R, Li PY, Meng E, Varma R, Humayun MS. Mini drug pump for ophthalmic use. Trans Am Ophthalmol Soc. 2009;107:60–70.
- 112. Salama Benarroch I, Nano H, Pérez H, Elizalde F, Bisceglia H, Salama A. Assessment of calcium dobesilate in diabetic retinopathy. A double-blind clinical investigation. Ophthalmologica. 1977;174:47–51.
- 113. Saylor M et al. Experimental and clinical evidence for brimonidine as an optic nerve and retinal neuroprotective agent: an evidence-based review. Arch Ophthalmol. 2009;127:402–6.
- 114. Schram MT, Stam F, de Jongh RT, de Vries G, van Dijk FA, Serné EH, Lampe D, Nanayakkara PW, Tushuizen ME, Scheffer PG, Schalkwijk CG, Kamper AM, Stehouwer CD. The effect of calcium dobesilate on vascular endothelium function, blood pressure, and markers of oxidation in obese male smokers: a placebo-controlled randomized clinical trial. Atherosclerosis. 2003;170:59–72.
- 115. Scott IU, Jackson GR, Quillen DA, et al. Effect of doxycycline vs placebo on retinal function and diabetic retinopathy progression in patients with severe nonproliferative or non-high-risk proliferative diabetic retinopathy: a randomized clinical trial. JAMA Ophthalmol. 2014;132(5):535–43.
- 116. Scott IU, Jackson GR, Quillen DA, Klein R, Liao J, Gardner TW. Effect of doxycycline vs placebo on retinal function and diabetic retinopathy progression in mild to moderate nonproliferative diabetic retinopathy: a randomized proof-of-concept clinical trial. JAMA Ophthalmol. 2014;132(9):1137–42.
- 117. Segal MS, Shah R, Afzal A, et al. Nitric oxide cytoskeletal-induced alterations reverse the endothelial progenitor cell migratory defect associated with diabetes. Diabetes. 2006;55:102–9.
- 118. Sfikakis P, Grigoropoulos V, Emfietzoglou I, et al. Infliximab for diabetic macular edema refractory to laser photocoagulation. Diabetes Care. 2010;33:1523–8.
- Shaw LC, Neu MB, Grant MB. Cell-based therapies for diabetic retinopathy. Curr Diab Rep. 2011;11:265–74.
- 120. Sheetz MJ, Aiello LP, Davis MD, et al. The effect of oral PKC beta inhibitor ruboxistaurin on vision loss in two phase 3 studies. Invest Ophthalmol Vis Sci. 2013;54:1750–7.
- 121. Shoshani T, Faerman A, Mett I, et al. Identification of a novel hypoxia-inducible factor1responsive gene, RTP801, involved in apoptosis. Mol Cell Biol. 2002;22:2283–93.
- 122. Sigurdsson HH, Konraethsdottir F, Loftsson T, Stefansson E. Topical and systemic absorption in delivery of dexamethasone to the anterior and posterior segments of the eye. Acta Ophthalmol Scand. 2007;85:598–602.
- 123. Silva KC et al. Diabetic retinal neurodegeneration is associated with mitochondrial oxidative stress and is improved by an angiotensin receptor blocker in a model combining hypertension and diabetes. Diabetes. 2009;58:1382–90.
- 124. Soheilian M, Karimi S, Ramezani A, Montahai T, Yaseri M, Soheilian R, Peyman GA. Intravitreal diclofenac versus intravitreal bevacizumab in naïve diabetic macular edema: a randomized double-masked clinical trial. Int Ophthalmol. 2015;35(3):421–8.
- 125. Sun J. Presented at the 2016 Annual Macula Society Meeting. Miami Beach, FL. 25 Feb 2016.
- 126. Takita H et al. Retinal neuroprotection against ischemic injury mediated by intraocular gene transfer of pigment epithelium-derived factor. Invest Ophthalmol Vis Sci. 2003;44:4497–504.

- 127. Tanito M, Hara K, Takai Y, Matsuoka Y, Nishimura N, Jansook P, Loftsson T, Stefansson E, Ohira A. Topical dexamethasone-cyclodextrin microparticle eye drops for diabetic macular edema. Invest Ophthalmol Vis Sci. 2011;52(11):7944–7.
- Tarr JM, Maul K, Chopra M, Kohner EM, Chibber R. Pathophysiology of diabetic retinopathy. ISRN Ophthalmol. 2013;Article ID 343560:1–13.
- 129. A phase 1, open-label study of teprotumumab in patients with diabetic macular edema (DME) 2014. Last accessed on 2015 May 30. Available from: https://www.clinicaltrials.gov/ct2/show/NCT02103283.
- 130. Theodossiadis PG, Markomichelakis NN, Sfikakis PP. Tumor necrosis factor antagonists: preliminary evidence for an emerging approach in the treatment of ocular inflammation. Retina. 2007;27:399–413.
- 131. Tsilimbaris MK, Panagiotoglou TD, Charisis SK, Anastasakis A, Krikonis TS, Christodoulakis E. The use of intravitreal etanercept in diabetic macular oedema. Semin Ophthalmol. 2007;22:75–9.
- 132. Vallon V, Mühlbauer B, Osswald H. Adenosine and kidney function. Physiol Rev. 2006;86:901-40.
- 133. Vojnikovic B. Doxium (calcium dobesilate) reduces blood hyperviscosity and lowers elevated intraocular pressure in patients with diabetic retinopathy and glaucoma. Ophthalmic Res. 1991;23:12–20.
- 134. Wang AL, Yu AC, Lau LT, et al. Minocycline inhibits LPS-induced retinal microglia activation. Neurochem Int. 2005;47(1–2):152–8.
- 135. Wang Q et al. Long-term treatment with suberythropoietic Epo is vaso- and neuroprotective in experimental diabetic retinopathy. Cell Physiol Biochem. 2011;27:769–82.
- 136. Wu L, Hernandez-Bogantes E, Roca JA, Arevalo JF, Barraza K, Lasave AF. Intravitreal tumor necrosis factor inhibitors in the treatment of refractory diabetic macular edema: a pilot study from the Pan-American Collaborative Retina Study Group. Retina. 2011;31:298–303.
- 137. Yasunari K, Kohno M, Kano H, Minami M, Yoshikawa J. Aldose reductase inhibitor improves insulin-mediated glucose uptake and prevents migration of human coronary artery smooth muscle cells induced by high glucose. Hypertension. 2000;35:1092–8.
- 138. Yoshida Y et al. Protective role of pigment epithelium-derived factor (PEDF) in early phase of experimental diabetic retinopathy. Diabetes Metab Res Rev. 2009;25:678–86.
- 139. Zeng HY, Green WR, Tso MO. Microglial activation in human diabetic retinopathy. Arch Ophthalmol. 2008;126(2):227–32.
- 140. Zhang J et al. Intravitreal injection of erythropoietin protects both retinal vascular and neuronal cells in early diabetes. Invest Ophthalmol Vis Sci. 2008;49:732–42.
- 141. Zhang M, Zhang J, Yan M, et al. A phase 1 study of KH902, a vascular endothelial growth factor receptor decoy, for exudative age-related macular degeneration. Ophthalmology. 2011;118:672–8.

Chapter 9 Safety Considerations of Pharmacotherapy

9.1 Introduction

Current pharmacologic treatment of diabetic retinopathy (DR) is based on the intraocular administration of several medications. Injections into the vitreous have emerged as the preferred route of drug delivery for several reasons. Firstly, drug concentrations in the vitreous, retina, and choroid are higher after intravitreal injections than after any other route of administration. Secondly, the vitreous acts as a depot that slows drug elimination from the eye and prolongs its duration of action. Thirdly, the small volume of injected drug results in low serum concentrations and may minimize the likelihood of systemic adverse events.

The favorable safety profile attributed to intraocular pharmacotherapy has contributed to the rapid development and widespread adoption of corticosteroid formulations and drugs that bind vascular endothelial growth factor (VEGF). The pivotal phase III registration trials were sufficiently sized to demonstrate clinically efficacy and garner approval by the United States Food and Drug Administration (US FDA), but they were insufficiently powered to identify low-frequency ocular and systemic adverse events. Consequently, systematic reviews and meta-analyses have been performed to better understand the risks of therapy.

The safety profiles of intraocular corticosteroids differ significantly from those of the anti-VEGF drugs. The volume of administered corticosteroids is not sufficient to cause systemic adverse events, but steroid therapy is complicated by high rates of cataract development and glaucoma, and triamcinolone acetonide injections may cause a severe sterile endophthalmitis or pseudoendophthalmitis.

Compared to both the general population and other groups who receive intraocular pharmacotherapy, patients with diabetes may be at greater risk of developing thromboembolic events and impaired wound healing, which might be worsened by anti-VEGF therapy. Physicians, therefore, remain keenly interested in the ocular and systemic safety profiles of drugs used to treat DR. The aim of this chapter is to discuss known risks of drugs used to treat DR.

9.1.1 Intravitreal Injections

Drugs are injected into the mid-vitreous through the pars plana (3–4 mm posterior to the limbus) (Fig. 9.1). Vascular endothelial growth factor inhibitory drugs may be injected through 30- or 32-gauge needles, whereas triamcinolone acetonide can be injected through 27- or 30-gauge needles. Large volumes (1 ml) of triamcinolone suspension frequently occlude 30-gauge needles, but the author has never experienced occlusion when injecting small intravitreal volumes (0.05 ml). Needles may be safely inserted to a depth of 12 mm and should be directed toward the mid-vitreous. The bioerodible, sustained release dexamethasone phosphate insert (Ozurdex®, Allergan, Irvine, CA) is injected through a pre-loaded 22-gauge insertion system, whereas the non-bioerodible, sustained release fluocinolone acetonide insert (Iluvien®, Alimera, Alpharetta, GA) is injected through a pre-loaded 25-gauge insertion system.

Neither preinjection nor postinjection antibiotics appear to alter low postinjection endophthalmitis rates [10, 48], but practice patterns regarding antibiotic use differ widely among surgeons. Most physicians use an eyelid speculum to stabilize the eyelids and improve exposure of the eye, but excellent results have been reported by having the surgeon or an assistant manually retract the eyelids (Fig. 9.2) [68]. Topical povidone-iodine, either 5% or 10%, should be used to sterilize the conjunctiva prior to the injection [17, 66]. True allergy to povidone-iodine is rare and most adverse reactions are due to a toxic keratitis. Patients who claim to be "allergic to iodine" because of reactions to shellfish (allergy to myosin) or intravenous contrast dye (hyperosmotic reaction) can be assured that there is no relationship between these adverse reactions and either iodine or povidone-iodine.

Various forms of anesthesia can be administered – topical proparacaine or tetracaine drops, lidocaine gel, subconjunctival lidocaine injection, or peribulbar lidocaine injection – to decrease the pain experienced by patients. Each form of



Fig. 9.1 This photograph of the author's technique of performing an intravitreal injection shows the inferior insertion position of the needle as the patient looks up

anesthesia has its associated advantages and drawbacks. For example, topical drops provide less predictable anesthesia but allows the procedure to be performed very quickly; subconjunctival lidocaine improves the depth of the anesthesia but prolongs the procedure and causes a subconjunctival hemorrhage in more than 50% of cases. The overall experience with the procedure appears to be independent of the method of anesthesia [11]. Surgeons, therefore, should find a technique with which they are most comfortable but should always consider individualizing anesthesia to meet the needs of each patient. Since the needles on the dexamethasone (22 gauge) and fluocinolone (25 gauge) insertion devices have considerably larger bores than those usually used for anti-VEGF injections (30 gauge), surgeons should consider the routine use of subconjunctival lidocaine when injecting these steroids.

Injections may be safely and efficiently performed in an outpatient clinic or, as is required in some European countries, within an operating suite. Some physicians contend that the incidence of endophthalmitis may be higher when injections are performed in the clinic rather than in the operating room [1], but the results from large pooled series of injections do not support this. Some clinics deal with the increasing number of intravitreal injections by employing specially trained nurses. In one New Zealand clinic, nurse specialists administered a total of 2900 injections over an 18-month period. Two patients (0.07%) developed postinjection endophthalmitis (1 microbial and 1 nonmicrobial), two patients (0.07%) developed vitreous hemorrhages, and five patients (0.17%) had elevated intraocular pressures (IOPs) [60]. The anticipated growth of intravitreal pharmacotherapy will likely result in an increase in the use of physician extenders.

9.2 Ocular Complications

Tables 9.1 and 9.2 list the complication rates of ocular adverse events due to anti-VEGF drugs and corticosteroids, respectively.

Fig. 9.2 This photograph of the author's technique for performing an intravitreal injection shows the manual retraction of the eyelids without the use of an eyelid speculum or surgical assistant



Ocular adverse events due	to anti-VEGF drugs
Complication	Incidence rate
Increased intraocular pressure	3–12%
Cataracts	<1% to 3.7%
Endophthalmitis	From: 0.02% per injection to 1%/patient over 3 years
Sterile inflammation	0-1.49%/injection
Retinal detachment	0.9%/injection to 3% per eye *most may be exacerbation of traction detachment
Vitreous hemorrhage	0.4%
Worsening traction detachment	5.2%

 Table 9.1
 The incidences of ocular adverse events that have been described in patients receiving treatment with vascular endothelial growth factor inhibitors

 Table 9.2
 The incidences of ocular adverse events that have been described in patients receiving treatment with intravitreal corticosteroids

Ocular adverse events due to cortice	osteroids	
Complication	Incidence rate	
Cataracts	37–80%	
Glaucoma	32–40%	
Endophthalmitis	0.03-0.09%	
	per Injection	
Migration of dexamethasone insert	into anterior chamber	

ingration of dexamethasone insert into anterior chamber

9.3 Elevated Intraocular Pressure (IOP) and Glaucoma

Population studies have defined the normal human IOP (encompassing 95% of the population) as 9–22 mmHg. Intraocular pressure is maintained through a complex balance of aqueous production, ocular rigidity, tissue deturgescence, and aqueous outflow. Most studies do not implicate diabetes as a risk factor for glaucoma, but intravitreal pharmacotherapy increases the risk of glaucoma in this population. In patients with diabetic macular edema (DME) that receive pharmacotherapy, elevated IOP usually results from increased resistance to aqueous outflow. Since eyes of diabetic patients frequently have complex vitreoretinal pathologies, IOP elevations should prompt evaluations for causes of secondary open-angle (corticosteroid-induced, anti-VEGF-induced, hemolytic, ghost-cell) and secondary angle-closure (neovascular) glaucoma.

The evaluation and treatment of glaucoma encompasses an entire subspecialty within ophthalmology that is beyond the scope of this book. Physicians who manage DME, however, need to be continually aware of the IOP and health of the optic nerve and nerve fiber layer. Intraocular pressure should be measured at each visit,
and the status of the optic nerve and nerve fiber layer should be assessed periodically with disc photographs and/or optical coherence tomography (OCT) measurements. Any IOP elevation should prompt the physician to consider testing (assessment of optic disc anatomy with biomicroscopy and stereoscopic disc photographs, OCT analysis of the nerve fiber layer, threshold visual fields, and pachymetry). If treatment of prolonged IOP elevation is outside the scope of practice of the treating physician, then referral to a glaucoma subspecialist or qualified comprehensive ophthalmologist should be considered.

Intraocular pressure often rises immediately after intravitreal injections but usually normalizes quickly. One study demonstrated that >95% of eyes had an IOP <35 mmHg immediately after injection and only 33% had an IOP elevation of 10 mmHg [8]. Sustained elevations in intraocular pressure occur in approximately 6% of eyes receiving anti-VEGF injections [30] and after a median of five injections. Preexisting glaucoma is a risk factor for sustained IOP elevation, probably because of an already compromised aqueous outflow. Patients receiving at least 29 injections have a 5.75 times risk of developing sustained IOP elevation compared to those receiving 12 or fewer injections [33]. Reasons for a higher risk with more injections may include prolonged VEGF blockade, development of an inflammatory trabeculitis, impaired aqueous outflow due to protein aggregates in the trabecular meshwork, large macromolecules obstructing the meshwork because of bulk deposition, or silicone droplets from the needles and syringes.

An exploratory post hoc analysis of DRCR.net Protocol I sought to determine the incidence of elevated IOP in eyes without preexisting glaucoma. During the 3-year trial, sustained IOP elevation was seen in 6 patients in the laser/sham group and 22 patients receiving ranibizumab (Lucentis®, Genentech, S. San Francisco, CA/Roche, Basel, Switzerland). The risk of developing a sustained IOP of >22 mmHg with an elevation of ≥ 6 mmHg above baseline was 10% in patients receiving ranibizumab, but only 3% in patients randomized to laser/sham (hazard ratio 2.9 [95% CI, 1.0–7.9]). Through 1 year in the DRCR.net Protocol T trial, patients treated with aflibercept (Eylea®, Regeneron, Tarrytown, NY), bevacizumab (Avastin®, Genentech, S. San Francisco, CA/Roche, Basel, Switzerland), and ranibizumab had the following incidences of elevated intraocular pressures: 12%, 9%, and 9% [20].

In a large retrospective study of 760 eyes, both transient (7%) and sustained (5.8%) elevations of intraocular pressure (defined as increases ≥ 6 mmHg, $\geq 20\%$ above baseline, or ≥ 24 mmHg on two or more consecutive visits) were seen after treatment. The probability of an IOP rise increased with the number of injections.

Development of glaucoma is a greater concern in patients receiving intraocular corticosteroids. Patients susceptible to steroid-induced glaucoma are believed to have upregulation of corticosteroid receptors in trabecular meshwork cells [81], which leads to increased expression of the extracellular protein fibronectin, glycosaminoglycans, and elastin [37, 67]. Steroids also downregulate phagocytic activity, which allows obstructive material to accumulate in the meshwork [58, 59]. Increased pressure in patients receiving triamcinolone acetonide may be due to obstruction of aqueous outflow due to steroid deposits in the angle. Approximately 50% of patients receiving intravitreal triamcinolone will experience an elevation in pressure within 2 to 4 weeks [64]. In the DRCR.net trial comparing macular laser photocoagulation with 1 mg and 4 mg intravitreal triamcinolone acetonide, the incidences of IOP elevation in the 1 mg and 4 mg triamcinolone cohorts were 20% and 40%, respectively [19]. Incisional glaucoma surgery was eventually required in 1.6% of patients receiving 4 mg triamcinolone. In DRCR.net Protocol I trial, 42% of patients receiving 4 mg triamcinolone experienced elevated IOP [22].

Intraocular pressure increases with the dexamethasone insert can generally be managed with pressure-lowering medications and are usually transient [9, 16, 31]. In a report of 186 eyes that received the dexamethasone insert through 6 months, eight eyes (4.3%) had intraocular pressure increases to at least 30 mmHg and one eye had an IOP of 50 mmHg [51]. The elevated pressure was successfully treated in all eyes with medications. In the phase III MEAD trials, patients receiving the 0.7 mg dexamethasone insert had a 32% incidence of IOP >25 mmHg and a 6% incidence of IOP >35 mmHg [12]. Less than 1% of eyes required incisional glaucoma surgery.

In the 36-month FAME trial, 38.4% of patients receiving the 0.2 mg fluocinolone insert (Iluvien) developed increased intraocular pressure, and 18.4% had IOPs over 30 mmHg [14]. Approximately 4% of patients required incisional glaucoma surgery to control the pressure. In a series of 17 eyes that were treated with the fluocinolone insert after having previously responded poorly to anti-VEGF therapy, only three had elevated IOP by 12 months and all were successfully treated with topical therapy [46].

9.4 Cataracts

Cataracts due to intravitreal injections of anti-VEGF drugs are very unusual. Only 1 injection-related cataract occurred after 6000 injections in DRCR.net Protocol T [20], presumably from needle trauma to the crystalline lens capsule (Fig. 9.3).

Corticosteroids induce posterior subcapsular cataracts (Fig. 9.4) because upregulated fibroblast growth factor-1, epidermal growth factor, transforming growth factor- β , lens epithelium-derived growth factor, platelet-derived growth factor, and bone morphogenic proteins induce proliferation and migration of lens epithelial cells. Local activation of glucocorticoid receptors in the lens may also downregulate apoptosis [35, 36].

Prospective studies evaluating the treatment of DME with triamcinolone acetonide report that the incidence of cataracts appears to be dose-related [19] and 37–54% of eyes undergo cataract surgery within 2 years [19, 28]. In the Protocol I trial, 15% of patients randomized to triamcinolone required cataract surgery within the first year compared to 3.7% of patients receiving ranibizumab and 6% of patients randomized to sham/laser [22]. The incidence of cataract surgery in the triamcinolone group increased to 55% at 2 years [23].

In the cohort of phakic eyes receiving the 0.7 mg dexamethasone insert in the phase III MEAD trials, progression of cataracts was experienced by 67.9% of eyes

Fig. 9.3 This slit-lamp photograph shows a perforation of the posterior lens capsule following the intravitreal injection of a vascular endothelial growth factor inhibitory drug. Localized posterior subcapsular lens opacifications have developed



Fig. 9.4 This slit-lamp photograph shows a posterior subcapsular cataract that formed after treatment with intravitreal corticosteroids



over 3 years [12]. In another dexamethasone insert study, progression of lens opacities was seen in 26 of 112 (23.2%) phakic patients within 6 months and 7 (6.3%) underwent cataract extraction [51]. This rate was higher than had been reported in other studies [9, 16, 27, 31].

9.5 Endophthalmitis

Individual cases and clusters of patients with noninfectious inflammation that followed injections of each anti-VEGF drug have been reported. In a study of 61 eyes with AMD, DME, or RVO, increased aqueous flare was detected in eyes that



Fig. 9.5 This external photograph shows a large hypopyon in an eye with streptococcal endophthalmitis

received bevacizumab but not ranibizumab or aflibercept (P = 0.006) [11]. Rates of inflammation after bevacizumab range from 0.14% to 0.32% and have been attributed to bevacizumab's large size and Fc fragment [24, 26]. A review of postinjection inflammation reported rates of 0.09–1.49% with bevacizumab, 0–0.83% with ranibizumab, and 0–0.05% with aflibercept [3].

A large meta-analysis of intravitreal injections found that *Staphylococcus epidermidis* was the most common cause of infectious endophthalmitis [47], but a single-center report of over 65,000 injections reported that *Streptococci* was the most common organism (Fig. 9.5) [52]. Since most *Streptococci* originate from the oropharynx, some surgeons have begun wearing masks while injecting or minimizing conversation to minimize aerosol spread of organisms.

The single-injection rates of endophthalmitis in the phase II and III DME trials were low, but the cumulative per patient endophthalmitis rates in the RESOLVE [45] and DRCR.net Protocol I [23] trials were approximately 1%. On the other hand, no cases of infectious endophthalmitis were reported in either the phase III RESTORE trial, which followed RESOLVE [62], or the DRCR.net Protocol T trial. Two cases of noninfectious inflammation occurred in each drug group (out of approximately 2000 injections with each drug) in Protocol T [20].

In some countries surgeons compound several doses of bevacizumab from the same vial to use in different patients during a single day. This may be done inside or outside of an operating suite. In one study, surgeons aliquoted bevacizumab from six vials and stored the syringes for up to 1 week at 4° C [18]. Subsequent analysis demonstrated no evidence of bacterial contamination and no cases of endophthalmitis occurred after 973 injections. The authors concluded that this technique is safe if "standard precautions" are adhered to. A retrospective report described a technique in which up to ten doses of bevacizumab were prepared by the surgeon from each vial [53], and no cases of endophthalmitis occurred after 1184 injections. Though this compounding procedure may be performed commonly in some areas, it does not comply with United States Pharmacopeia Chapter 797 guidelines and should be viewed with skepticism [29].

Bilateral same-day injections are performed by some surgeons and patients favor this technique since it decreases the number of office visits. In a series of 342 bilat-

eral injections, one patient developed infectious endophthalmitis in only one eye. The authors claim that this technique is safe if the surgeon changes povidone-iodine, eyelid speculums, and all needles and syringes between injecting the two eyes [2].

Because bevacizumab must be compounded or fractionated from 4 ml multiuse vials into 20 to 30 single-dose 1 ml syringes, it may cause clusters of infectious endophthalmitis because of compounding errors. Several outbreaks of compounding-related endophthalmitis due to highly virulent organisms have occurred [29]. Visual outcomes usually depend upon the organism, but severe vision loss is common. Compounding is best performed by a trained technician working under a high-quality, laminar flow hood. By following the guidelines in Chapter 797 of the United States Pharmacopeia, these technicians are able to safely and inexpensively compound bevacizumab for intraocular use.

The overall incidence of endophthalmitis following intravitreal triamcinolone injections has been estimated at 0.8% for noninfectious causes and 0.6% from infectious agents [63]. The DRCR.net Protocol B trial reported only one case of endophthalmitis in 3159 triamcinolone injections (0.03% per injection) [19] though smaller series have reported higher incidence rates primarily because of the smaller numbers of injections. The cause of sterile or noninfectious endophthalmitis is unknown, but some investigators believe that it may be due to preservatives. Pseudoendophthalmitis occurs when the crystalline triamcinolone suspension diffuses rapidly throughout an eye that has previously undergone vitrectomy; fortunately this is a self-limited condition.

The incidence of endophthalmitis following insertion of the dexamethasone insert appears to be low. In the MEAD trial, 2928 DEX injections were performed, with only two cases of endophthalmitis [12]. One of these occurred after cataract surgery and was felt to be unrelated to the DEX.

9.6 Counterfeit Drugs

Bevacizumab ranks seventh in world drug sales (by dollars) with most of this stemming from oncologic use [56]. Because bevacizumab is manufactured as a clear liquid, it has become an easy target for counterfeiters. Counterfeit bevacizumab naturally lacks biological efficacy and the substitute solution may be contaminated with microbes or pro-inflammatory substances. Though intravenous injections of counterfeit bevacizumab may not cause observable adverse events, intraocular injections have caused both infectious endophthalmitis and severe sterile inflammation with retinal necrosis.

Series of patients receiving intravitreal injections of counterfeit bevacizumab have been reported in China [75], Mexico [25], and India [69]. Counterfeit bevacizumab has entered the supply chain in the United States though no injuries were reported [76].

Bevacizumab should always be purchased from a reputable supplier, and physicians and pharmacists must carefully scrutinize bottles for authenticity to be sure that they are not using counterfeit products.

9.7 Other Complications

Retinal detachments (RDs) are frequently cited as complications of intravitreal injections, but they occur only rarely. A systematic review put the RD rate at 3.9% per eye and 0.9% per injection [34], but most of these detachments were believed due to vitreoretinal traction from the underlying disorder rather than due to the injection procedure. Detachment rates from the phase III DME registration trials, from which eyes with significant fibrovascular proliferation were excluded, are generally much lower.

Intraocular hemorrhages (retinal, vitreous, or hyphema) have been reported after intraocular injections, but like RDs, these are mostly due to the underlying diabetic retinopathy. The incidence of vitreous hemorrhage in RISE and RIDE was only 0.4% in the treatment arms compared to 2.8% in the sham arm [54], suggesting that most hemorrhages were due to progression of the diabetic retinopathy. Subconjunctival hemorrhages are quite common (10%), particularly in patients taking aspirin (17%) [41]. Most studies do not report an increased incidence of intraocular hemorrhages in patients taking systemic anticoagulants [42, 44], and special precautions are not required when injections are performed in these patients.

Considerable in vitro and in vivo research has been done to determine whether anti-VEGF drugs are cytotoxic. Neither bevacizumab nor ranibizumab is toxic to the corneal endothelium when injected intravitreally into patients with DME [32]. Repeated intravitreal injections of ranibizumab for the treatment of nAMD do not appear to adversely affect retinal nerve fiber layer thickness [21], and intravitreal bevacizumab injections into rabbit eyes do not cause retinal apoptosis [71]. Consistent cytotoxicity patterns have not been observed for any of the drugs.

Rapidly worsening traction retinal detachments, the "crunch syndrome," have been reported after injections of bevacizumab into eyes with preexisting fibrovascular proliferation due to proliferative diabetic retinopathy. These may occur because rapidly falling VEGF concentrations lead to increasing connective tissue growth factor levels and a switch from angiogenesis to fibrosis [40]. In one report, the incidence of traction retinal detachment after bevacizumab was 11 out of 211 injections (5.2%), with 82% of these occurring within 5 days of the injections [4].

Early reports suggested that anti-VEGF injections might decrease retrobulbar blood flow, constrict retinal arterioles, and worsen macular ischemia [55]. Macular perfusion studies, however, have not demonstrated a significant decrease in macular perfusion [49], and current observations suggest that anti-VEGF injections may actually improve retinal capillary blood flow.

Migration of the dexamethasone insert into the anterior chamber has been reported in eyes without an intact lens-iris diaphragm. Though 28-gauge bimatoprost SR inserts (constructed with the same design as the dexamethasone inserts) are designed to settle safely into the anterior chamber angle, the larger dexamethasone insert fits poorly and causes significant corneal edema. Dexamethasone inserts may be injected into eyes with in-the-bag intraocular lenses and YAG capsulotomies, but they should be avoided in aphakic eyes and those with sutured posterior chamber intraocular lenses.

9.8 Systemic Complications

Intravenous injections of VEGF inhibitors lower serum VEGF concentrations [74] and increase the risks of VEGF-related complications. Systemic arterial hypertension and proteinuria are the most commonly seen adverse effects, and delayed wound healing, gastrointestinal bleeding, and Antiplatelet Trialists' Collaborative (APTC) defined events such as myocardial infarction and stroke are less frequently seen (Fig. 9.6) [57, 61]. Patients with advanced solid carcinomas have double the risk of stroke when receiving intravenous bevacizumab.

The Pan-American Collaborative Retina Study Group (PACORES) found that 7 of 1173 patients (0.6%) developed transient systemic arterial hypertension between 7 h and 2 weeks after intravitreal bevacizumab injections [77]. Strokes and myocardial infarctions occurred in 1.2% and 0.4% of patients, respectively.

The phase II and III DME trials were designed to demonstrate the clinical superiority of anti-VEGF therapy over laser photocoagulation/sham injections, but they were insufficiently powered to detect infrequently occurring APTC events. Metaanalyses of large treatment populations have generally determined that intravitreal anti-VEGF therapy is not associated with an increased risk of systemic complications [78], but many of these studies have not focused on higher-risk populations such as patients with diabetes mellitus [13, 50, 65, 70, 72, 73]. Data from several studies



Fig. 9.6 Systemic adverse events related to vascular endothelial growth factor binding have been reported after intravenous and intravitreal drug injections. Systemic arterial hypertension, proteinuria, and Antiplatelet Trialists' Collaborative defined events (myocardial infarction and cerebrovascular event) are of particular concern in diabetic patients receiving intravitreal therapy

have shown that serum concentrations of VEGF fall after intravitreal injections of bevacizumab and aflibercept (but not ranibizumab), thus providing a plausible mechanism for the development of some VEGF-related systemic adverse events [5, 15, 75, 79, 80]. Associations between low VEGF concentrations and higher risks of APTC events, however, have never been established.

In the phase II RESOLVE trial, the incidences of hypertension and arteriothrombotic events were similar in the ranibizumab and sham groups [45]. The 1-year results in the RISE and RIDE trials showed a dose-dependent, increased incidence of stroke with ranibizumab. When also considering that the 0.3 mg and 0.5 mg ranibizumab doses were similarly effective for DME, the US FDA approved the 0.3 mg dose for DME. At 3 years in the RISE and RIDE trials, the incidences of APTC-related events in the sham/laser, 0.3 mg ranibizumab, and 0.5 mg ranibizumab groups were 7%, 11%, and 10%, and the overall incidences of death were 3%, 4%, and 6%. The causes of death were typical of a cohort with advanced diabetes mellitus. These data must be considered carefully since most patients in the sham/laser group crossed over to monthly 0.5 mg ranibizumab at 24 months and had 1 year of drug exposure by the study's conclusion. The use of the lower ranibizumab dose may be of particular importance in a diabetic population that is at risk of APTC events since over 50% of patients may require bilateral treatment.

Increased risk of death and cerebrovascular events was not seen in patients receiving 0.5 mg ranibizumab in the DRCR.net Protocol I trial [23]. Patients in this trial were not required to receive monthly injections, so total exposure to ranibizumab was considerably lower than in RISE and RIDE. In the phase III VISTA and VIVID trials, APTC events occurred in 3.3% of patients receiving aflibercept and 2.8% of patients treated with laser/sham [39].

A Cochrane meta-analysis evaluating the safety of anti-VEGF drug use for DME did not find an increased incidence in non-ocular serious adverse events [73]. The relative risks of anti-VEGF-treated patients experiencing serious systemic adverse events (2985 patients from 15 trials, relative risk (RR) 0.98, 98% confidence interval (CI) 0.83 to 1.17), arterial thrombotic events (3034 participants from 14 studies, RR 0.89, 95% CI 0.63 to 1.25), and overall mortality (3562 participants, RR 0.88, 95% CI 0.52 to 1.47) were all insignificant.

Primary analysis of the DRCR.net Protocol T trial detected no significant differences in adverse events between patients treated with aflibercept, bevacizumab, and ranibizumab, but a post hoc analysis found a higher incidence of cardiovascular events in patients treated with ranibizumab (17%), as compared to aflibercept (9%) and bevacizumab (9%) (P = 0.01) [43].

A meta-analysis of phase II and III trials in which affibercept had been administered for several conditions found that the risk of death in affibercept-treated patients with DME was 1.70 per 100 patient-years at risk compared to 0.55 per 100 patientyears at risk for patients in the sham/laser groups [38].

In the first year of the VISTA and VIVID trials, the incidences of ocular and nonocular adverse events and serious adverse events including APTC-defined events and deaths were similar among all groups [39]. Serious non-ocular adverse events were uncommon (hypertension, 9.7%; cerebrovascular accidents, 1.1%; and myocardial infarction, 1.1%). The incidences of noninfectious intraocular inflammation were <0.5% in each trial, and there were no incidences of infectious endophthalmitis. The incidences of congestive heart failure and anemia were higher in the aflibercept groups, but the incidences of myocardial infarction and osteoarthritis were higher in the laser groups. The total numbers of deaths due to vascular and non-vascular causes were similar among all groups. Through 148 weeks in VIVID and VISTA, 6.8% eyes in the aflibercept arms and 4.5% of eyes in the laser arms experienced significant ocular adverse events, and 47.6% and 48.7% of patients, respectively, experienced serious systemic adverse events.

A meta-analysis was performed to examine the systemic safety profile of anti-VEGF agents in patients with several chorioretinal vascular conditions [6]. Higher risks for death, CVA, and impaired wound healing were found only in patients with DME. The authors performed a subsequent systematic review and meta-analysis to evaluate the systemic safety of monthly (the most intensive dosing regimen) intravitreal anti-VEGF injections administered to patients with DME over the course of at least 2 years [7]. Four studies (two using monthly aflibercept and two using monthly ranibizumab) with a total of 1328 patients met the authors' entry criteria and were included in the analysis. In the combined analysis, patients receiving monthly injections had an increased risk for death (odds ratio of 2.98; 95% confidence interval: 1.94–5.22; P = 0.04) compared to those receiving sham injections or laser. However, no significantly increased risk of myocardial infarctions or arteriothrombotic events could be identified.

Determining the risk of developing APTC events with anti-VEGF therapy has been difficult since patients with recent (generally <6 months) myocardial infarctions or strokes have been excluded from phase III trials. Physicians, therefore, have very little guidance regarding the safety of anti-VEGF therapy in high-risk patients that have experienced recent thromboembolic events. When considering treatment for a patient with neovascular age-related macular degeneration, for which there are no good alternatives to anti-VEGF drugs, physicians will generally provide appropriate informed consent that includes disclosure of possible systemic risks and then will usually proceed with injections. Fortunately for patients with DME, intravitreal corticosteroids are an effective alternative to anti-VEGF drugs, without known systemic risks.

9.9 Conclusion

The first intravitreal injection was performed over 100 years ago, but only with the use of anti-VEGF drugs during the last 11 years has the performance of intravitreal therapy become a common procedure. Fortunately, anti-VEGF treatment of diabetic macular edema is effective and safe. Most treatment-associated adverse events are related to the injection itself and are minor. Patients with diabetes are at higher risk for APTC events, but there is no conclusive evidence that anti-VEGF injections increase the occurrence rates, and neither is there evidence of different rates among the drugs. Intraocular corticosteroids do not cause systemic adverse events, but rates of cataract progression and glaucoma development must always be considered.

References

- 1. Abell RG, Kerr NM, Allen P, et al. Intravitreal injections: is there benefit for a theatre setting? Br J Ophthalmol. 2012;96(12):1474–8.
- Abu-Yaghi NE, Shokry AN, Abu-Sbeit RH. Bilateral same-session intravitreal injections of anti-vascular endothelial growth factors. Int J Ophthalmol. 2014;7(6):1017–21.
- Agrawal S, Joshi M, Christoforidis JB. Vitreous inflammation associated with intravitreal anti-VEGF pharmacotherapy. Mediators Inflamm. 2013;2013:943409.
- Arevalo JF, Maia M, Flynn Jr HW, et al. Tractional retinal detachment following intravitreal bevacizumab (Avastin) in patients with severe proliferative diabetic retinopathy. Br J Ophthalmol. 2008;92(2):213–6.
- Avery RL, Castellarin AA, Steinle NC, et al. Systemic pharmacokinetics following intravitreal injections of ranibizumab, bevacizumab or affibercept in patients with neovascular AMD. Br J Ophthalmol. 2014;98(12):1636–41.
- Avery RL, Francom S, Lai P, Melson C, Cha SB, Tuomi L. Meta-analysis examining the systemic safety profile of intravitreal ranibizumab injections in AMD, RVO and DME [ARVO Abstract]. Invest Ophthalmol Vis Sci. 2013;54:1535.
- Avery RL, Gordon GM. Systemic safety of prolonged monthly anti-vascular endothelial growth factor therapy for diabetic macular edema. A systematic review and meta-analysis. JAMA Ophthalmol. 2016;134(1):21–9.
- Bakri SJ, Pulido JS, McCannel CA, et al. Immediate intraocular pressure changes following intravitreal injections of triamcinolone, pegaptanib, and bevacizumab. Eye (Lond). 2009;23(1):181–5.
- Bansal P, Gupta V, Gupta A, Dogra MR, Ram J. Efficacy of Ozurdex implant in recalcitrant diabetic macular edema-a single-center experience. Int Ophthalmol. 2016;36(2):207–16.
- Bhavsar AR, Stockdale CR, Ferris 3rd FL, et al. Update on risk of endophthalmitis after intravitreal drug injections and potential impact of elimination of topical antibiotics. Arch Ophthalmol. 2012;130(6):809–10.
- Blaha GR, Tilton EP, Barouch FC, Marx JL. Randomized trial of anesthetic methods for intravitreal injections. Retina. 2011;31(3):535–9.
- 12. Boyer DS, Yoon YH, Belfort Jr R, et al. Three-year, randomized, sham-controlled trial of dexamethasone intravitreal implant in patients with diabetic macular edema. Ophthalmology. 2014;121(10):1904–14.
- Braithwaite T, Nanji AA, Lindsley K, Greenberg PB. Anti–vascular endothelial growth factor for macular oedema secondary to central retinal vein occlusion. Cochrane Database Syst Rev. 2014;5:CD007325.
- Campochiaro PA, Brown DM, Pearson A, et al. FAME Study Group. Sustained delivery fluocinolone acetonide vitreous inserts provide benefit for at least 3 years in patients with diabetic macular edema. Ophthalmology. 2012;119(10):2125–32.
- Chakravarthy U, Harding SP, Rogers CA, et al. IVAN Study Investigators. Ranibizumab versus bevacizumab to treat neovascular age-related macular degeneration: one-year findings from the IVAN randomized trial. Ophthalmology. 2012;119(7):1399–411.
- 16. Chhablani J, Bansal P, Veritti D, et al. Dexamethasone implant in diabetic macular edema in real-life situations. Eye (Lond). 2015;30(3):426–30.
- 17. Ciulla TA, Starr MB, Masket S. Bacterial endophthalmitis prophylaxis for cataract surgery: an evidence-based update. Ophthalmology. 2002;109(1):13–24.
- Das T, Volety S, Ahsan SM, Thakur AK, Sharma S, Padhi TR, Basu S, Rao CM. Safety, sterility and stability of direct-from-vial multiple dosing intravitreal injection of bevacizumab. Clin Experiment Ophthalmol. 2015;43(5):466–73.
- 19. Diabetic Retinopathy Clinical Research Network. A randomized trial comparing intravitreal triamcinolone acetonide and focal/grid photocoagulation for diabetic macular edema. Ophthalmology. 2008;115(9):1447–9.

- 20. Diabetic Retinopathy Clinical Research Network. Wells JA, Glassman AR, Ayala AR, Jampol LM, Aiello LP, Antoszyk AN, Arnold-Bush B, Baker CW, Bressler NM, Browning DJ, Elman MJ, Ferris FL, Friedman SM, Melia M, Pieramici DJ, Sun JK, Beck RW. Aflibercept, bevacizumab, or ranibizumab for diabetic macular edema. N Engl J Med. 2015;372(13):1193–203.
- El-Ashry MF, Lascaratos G, Dhillon B. Evaluation of the effect of intravitreal ranibizumab injections in patients with neovascular age related macular degeneration on retinal nerve fiber layer thickness using optical coherence tomography. Clin Ophthalmol. 2015;9:1269–74.
- 22. Elman MJ, Aiello LP, Beck RW, et al. Randomized trial evaluating ranibizumab plus prompt or deferred laser or triamcinolone plus prompt laser for diabetic macular edema. Ophthalmology. 2010;117(6):1064–77.
- 23. Elman MJ, Qin H, Aiello LP, et al. Diabetic Retinopathy Clinical Research Network. Intravitreal ranibizumab for diabetic macular edema with prompt versus deferred laser treatment: three-year randomized trial results. Ophthalmology. 2012;119(11):2312–8.
- 24. Fung AE, Rosenfeld PJ, Reichel E. The International Intravitreal Bevacizumab Safety Survey: using the internet to assess drug safety worldwide. Br J Ophthalmol. 2006;90(11):1344–9.
- Garcia-Aguirre G, Vanzinni-Zago V, Quiroz-Mercado H. Growth of Scytalidium sp. in a counterfeit bevacizumab bottle. Indian J Ophthalmol. 2013;61(9):523–5.
- Georgopoulos M, Polak K, Prager F, et al. Characteristics of severe intraocular inflammation following intravitreal injection of bevacizumab (Avastin). Br J Ophthalmol. 2009;93(4):457–62.
- Gillies MC, Lim LL, Campain A, et al. A randomized clinical trial of intravitreal bevacizumab versus intravitreal dexamethasone for diabetic macular edema The BEVORDEX Study. Ophthalmol. 2014;121(12):2473–81.
- Gillies MC, Sutter FK, Simpson JM, et al. Intravitreal triamcinolone for refractory diabetic macular edema: two-year results of a double-masked, placebo-controlled, randomized clinical trial. Ophthalmology. 2006;113(9):1533–8.
- Gonzalez S, Rosenfeld PJ, Stewart MW, Brown J, Murphy SP. Avastin doesn't blind people, people blind people. Am J Ophthalmol. 2012;152(2):196–203.
- 30. Good TJ, Kimura AE, Mandava N, et al. Sustained elevation of intraocular pressure after intravitreal injections of anti-VEGF agents. Br J Ophthalmol. 2011;95(8):1111–4.
- 31. Guigou S, Pommier S, Meyer F, et al. Efficacy and safety of intravitreal dexamethasone implant in patients with diabetic macular edema. Ophthalmologica. 2015;233:169–75.
- 32. Guzel H, Bakbak B, Koylu MT, Gonul S, Ozturk B, Gedik S. The effect and safety of intravitreal injection of ranibizumab and bevacizumab on the corneal endothelium in the treatment of diabetic macular edema. Cutan Ocul Toxicol. 2016;24:1–4.
- Hoang QV, Mendonca LS, Della Torre KE, et al. Effect on intraocular pressure in patients receiving unilateral intravitreal anti-vascular endothelial growth factor injections. Ophthalmology. 2012;119(2):321–6.
- Jager RD, Aiello LP, Patel SC, et al. Risks of intravitreous injection: a comprehensive review. Retina. 2004;24(5):676–98.
- 35. James ER. The etiology of steroid cataract. J Ocul Pharmacol Ther. 2007;23(5):403-20.
- Jobling AI, Augusteyn RC. What causes steroid cataracts? A review of steroid-induced posterior subcapsular cataracts. Clin Exp Optom. 2002;85(2):61–75.
- Johnson DH, Bradley JM, Acott TS. The effect of dexamethasone on glycosaminoglycans on human trabecular meshwork in perfusion organ culture. Invest Ophthalmol Vis Sci. 1990;31(12):2568–71.
- 38. Kitchens J. Systematic review of safety across the phase 2 and 3 clinical trials of intravitreal affibercept injection in neovascular age-related macular degeneration, macular edema following retinal vein occlusion, and diabetic macular edema. Poster presented at: 2015 Meeting of the Association for Research in Vision and Ophthalmology; May 6, 2015; Denver, CO.
- Korobelnik JF, Do DV, Schmidt-Erfurth U, Boyer DS, Holz FG, Heier JS, Midena E, Kaiser PK, Terasaki H, Marcus DM, Nguyen QD, Jaffe GJ, Slakter JS, Simader C, Soo Y, Schmelter T,

Yancopoulos GD, Stahl N, Vitti R, Berliner AJ, Zeitz O, Metzig C, Brown DM. Intravitreal aflibercept for diabetic macular edema. Ophthalmology. 2014;121(11):2247–54.

- 40. Kuiper EJ, Van Nieuwenhoven FA, de Smet MD, et al. The angio-fibrotic switch of VEGF and CTGF in proliferative diabetic retinopathy. PLoS One. 2008;3(7):e2675.
- Ladas ID, Karagiannis DA, Rouvas AA, et al. Safety of repeat intravitreal injections of bevacizumab versus ranibizumab: our experience after 2,000 injections. Retina. 2009;29(3):313–8.
- 42. Loukopoulos V, Meier C, Gerding H. Hemorrhagic complications after intravitreal injections of ranibizumab in patients under coumarin-type anticoagulation. Klin Monbl Augenheilkd. 2010;227(4):289–91.
- Martin DF, Maguire MG. Treatment of choice for diabetic macular edema. N Engl J Med. 2015;372(13):1260–1.
- 44. Mason 3rd JO, Frederick PA, Neimkin MG, et al. Incidence of hemorrhagic complications after intravitreal bevacizumab (avastin) or ranibizumab (lucentis) injections on systemically anticoagulated patients. Retina. 2010;30(9):1386–9.
- 45. Massin P, Bandello F, Garweg JG, et al. Safety and efficacy of ranibizumab in diabetic macular edema (RESOLVE Study). Diabetes Care. 2010;33:2399–405.
- 46. Massin P, Erginay A, Dupas B, Couturier A, Tadayoni R. Efficacy and safety of sustaineddelivery fluocinolone acetonide intravitreal implant in patients with chronic diabetic macular edema insufficiently responsive to available therapies: a real-life study. Clin Ophthalmol. 2016;10:1257–64.
- 47. McCannel CA. Meta-analysis of endophthalmitis after intravitreal injection of anti-vascular endothelial growth factor agents: causative organisms and possible prevention strategies. Retina. 2011;31(4):654–61.
- Meyer CH, Mennel S, Eter N. Incidence of endophthalmitis after intravitreal Avastin injection with and without postoperative topical antibiotic application. Ophthalmology. 2007;104(11):952–7.
- 49. Michaelides M, Fraser-Bell S, Hamilton R, et al. Macular perfusion determined by fundus fluorescein angiography at the 4-month time point in a prospective randomized trial of intravitreal bevacizumab or laser therapy in the management of diabetic macular edema (Bolt Study): Report 1. Retina. 2010;30(5):781–6.
- Mitry D, Bunce C, Charteris D. Anti–vascular endothelial growth factor for macular oedema secondary to branch retinal vein occlusion. Cochrane Database Syst Rev. 2013;1:CD009510.
- 51. Moon BG, Lee JY, Yu HG, Song JH, Park Y-H, Kim HW, Ji Y-S, Chang W, Lee JE, Oh J, Chung I. Efficacy and safety of a dexamethasone implant in patients with diabetic macular edema at tertiary centers in Korea. J Ophthalmol. 2016;Article ID 9810270.
- 52. Moshfeghi AA, Rosenfeld PJ, Flynn Jr HW, Schwartz SG, Davis JL, Murray TG, Smiddy WE, Berrocal AM, Dubovy SR, Lee WH, Albini TA, Lalwani GA, Kovach JL, Puliafito CA. Endophthalmitis after intravitreal vascular [corrected] endothelial growth factor antagonists: a six-year experience at a university referral center. Retina. 2011;31(4):662–8.
- Ng DS, Kwok AK, Chan CW, Li WW. Intravitreal bevacizumab: safety of multiple doses from a single vial for consecutive patients. Hong Kong Med J. 2012;18(6):488–95.
- 54. Nguyen QD, Brown DM, Marcus DM, et al. Ranibizumab for diabetic macular edema: results from 2 phase III randomized trials: RISE and RIDE. Ophthalmology. 2012;119(4):789–801.
- 55. Papadopoulou DN, Mendrinos E, Mangioris G, et al. Intravitreal ranibizumab may induce retinal arteriolar vasoconstriction in patients with neovascular age-related macular degeneration. Ophthalmology. 2009;116(9):1755–61.
- 56. http://www.pmlive.com/top_pharma_list/Top_50_pharmaceutical_products_by_global_sales. Accessed 24 Feb 2016.
- Ranpura V, Hapani S, Chuang J, Wu S. Risk of cardiac ischemia and arterial thromboembolic events with the angiogenesis inhibitor bevacizumab in cancer patients: a meta-analysis of randomized controlled trials. Acta Oncol. 2010;49(3):287–97.

- 58. Rohen JW, Linner E, Witmer R. Electron microscopic studies on the trabecular meshwork in two cases of corticosteroid-glaucoma. Exp Eye Res. 1973;17(1):19–31.
- 59. Roll P, Benedikt O. Electron microscopic studies of the trabecular meshwork in corticosteroid glaucoma (in German). Klin Monbl Augenheilkd. 1979;174(3):421–8.
- 60. Samalia P, Garland D, Squirrell D. Nurse specialists for the administration of anti-vascular endothelial growth factor intravitreal injections. N Z Med J. 2016;129(1438):32–8.
- 61. Schutz FA, Je Y, Azzi GR, Nguyen PL, Choueiri TK. Bevacizumab increases the risk of arterial ischemia: a large study in cancer patients with a focus on different subgroup outcomes. Ann Oncol. 2011;22(6):1404–12.
- 62. Schmidt-Erfurth U, Lang GE, Holz FG, et al. RESTORE extension study group. Three-year outcomes of individualized ranibizumab treatment in patients with diabetic macular edema: the RESTORE extension study. Ophthalmology. 2014;121:1045–53.
- Scott IU, Flynn Jr HW. Reducing the risk of endophthalmitis following intravitreal injections. Retina. 2007;27(1):10–2.
- 64. Singh IP, Ahmad SI, Yeh D, et al. Early rapid rise in intraocular pressure after intravitreal triamcinolone acetonide injection. Am J Ophthalmol. 2004;138(2):286–7.
- Solomon SD, Lindsley K, Vedula SS, Krzystolik MG, Hawkins BS. Anti–vascular endothelial growth factor for neovascular age-related macular degeneration. Cochrane Database Syst Rev. 2014;8:CD005139.
- Speaker MG, Menikoff JA. Prophylaxis of endophthalmitis with topical povidone-iodine. Ophthalmology. 1991;98(12):1769–75.
- Steely HT, Bowder SL, Julian MB, et al. The effects of dexamethasone on fibronectin expression in cultured human trabecular meshwork cells. Invest Ophthalmol Vis Sci. 1992;33(7):2242–50.
- 68. Stewart MW. Unassisted intravitreal injection without an eyelid speculum or gloves. Retina. 2014;34(4):e11–3.
- 69. Stewart MW, Narayanan R, Gupta V, Rosenfeld PJ, Martin DF, Chakravarthy U. Counterfeit Avastin in India: punish the criminals, not the patients. Am J Ophthalmol. 2016. Epub 2016 June 07. PMID:27287822. doi:10.1016/j.ajo.2016.05.023.
- Thulliez M, Angoulvant D, Le Lez ML, et al. Cardiovascular events and bleeding risk associated with intravitreal antivascular endothelial growth factor monoclonal antibodies: systematic review and meta-analysis. JAMA Ophthalmol. 2014;132(11):1317–26.
- Türkcü FM, Alp MN, Türkcü G, Kulaçoğlu S, Kural G. Short term apoptotic activity of intravitreal bevacizumab on rabbit retina. Int J Ophthalmol. 2013;6(6):785–9.
- 72. Ueta T, Noda Y, Toyama T, Yamaguchi T, Amano S. Systemic vascular safety of ranibizumab for age-related macular degeneration: systematic review and meta-analysis of randomized trials. Ophthalmology. 2014;121(11):2193–2203.e1–7.
- 73. Virgili G, Parravano M, Menchini F, et al. Anti-vascular endothelial growth factor for diabetic macular edema. Cochrane Database Syst Rev. 2014;10:CD007419.
- 74. Wang X, Sawada T, Sawada O, Saishin Y, Liu P, Ohji M. Serum and plasma vascular endothelial growth factor concentrations before and after intravitreal injection of aflibercept or ranibizumab for age-related macular degeneration. Am J Ophthalmol. 2014;158(4):738–744.e1.
- 75. Wang F, Yu S, Liu K, et al. Acute intraocular inflammation caused by endotoxin after intravitreal injection of counterfeit bevacizumab in Shanghai, China. Ophthalmol. 2013;120(2):355–61.
- 76. http://www.wsj.com/articles/SB10001424052702303879604577410430607090226. Accessed 5 Feb 2016.
- 77. Wu L, Martinez-Castellanos MA, Quiroz-Mercado H, et al. Twelve-month safety of intravitreal injections of bevacizumab (Avastin): results of the Pan-American Collaborative Retina Study Group (PACORES). Graefes Arch Clin Exp Ophthalmol. 2008;246(1):81–7.
- 78. Yanagida Y, Ueta T. Systemic safety of ranibizumab for diabetic macular edema: meta-analysis of randomized trials. Retina. 2014;34(4):629–35.

- 79. Yoshida I, Shiba T, Taniguchi H, et al. Evaluation of plasma vascular endothelial growth factor levels after intravitreal injection of ranibizumab and affibercept for exudative age-related macular degeneration. Graefes Arch Clin Exp Ophthalmol. 2014;252(9):1483–9.
- Zehetner C, Kralinger MT, Modi YS, et al. Systemic levels of vascular endothelial growth factor before and after intravitreal injection of aflibercept or ranibizumab in patients with agerelated macular degeneration: a randomised, prospective trial. Acta Ophthalmol. 2015;93(2):e154–9.
- Zhang X, Clark AF, Yorio T. FK 506-binding protein 51 regulates nuclear transport of the glucocorticoid receptor beta and glucocorticoid responsiveness. Invest Ophthalmol Vis Sci. 2008;49(3):1037–47.

Chapter 10 Socioeconomic Cost of Diabetic Retinopathy and Therapy

10.1 Introduction

Diabetes mellitus (DM) significantly impacts quality of life [8, 16, 48] with affected patients unable to manage their diabetes as well when they are visually impaired [6]. Vision impairment compromises quality of life by limiting physical activity, promoting social isolation, and causing dependence on others for the performance of many of life's necessary functions [43]. Patients with diabetic macular edema (DME) also have higher rates of healthcare utilization (doctors' visits, hospitalizations, diagnostic testing, treatments, and medications) compared to diabetic patients without diabetic retinopathy [20, 26, 44]. A US health quality study reported that diabetes is associated with a utility of 0.53, whereas blindness has a utility of only 0.38, with only major stroke (0.31) and end-stage renal disease (0.35) ranking lower [19]. By identifying patients with diabetic retinopathy (DR), physicians are frequently able to stabilize and improve vision, thereby improving the quality of patients' lives. Improvement in visual acuity often enables patients to resume driving and obtain gainful employment.

For decades, the standard treatment for most patients with vision-threatening DR was laser photocoagulation, with pars plana vitrectomy reserved for advanced vitreoretinal pathologies. As discussed in other chapters in this text, laser reduces the risk of vision loss by approximately 50% in most patients with DME and proliferative diabetic retinopathy, but does not usually improve visual acuity. Maintaining good visual acuity also depends upon early diagnosis through effective screening programs, regularly scheduled examinations, and prompt examination and treatment of symptomatic and at-risk patients.

Intravitreal pharmacotherapy with corticosteroids and drugs that inhibit the actions of vascular endothelial growth factor (VEGF) is capable of significantly improving vision in a majority of treated patients. Physicians in several countries have studied the cost-effectiveness of diabetic retinopathy treatments, and their conclusions vary according to the drug costs in each country and the assumptions used to create each of the respective economic models. All generally agree, however, that

treatment of DR with laser and pharmacotherapy is cost-effective. The cost of drugs in each country remains a major determinant of cost-effectiveness, and future models will have to contend with the major price differences between compounded bevacizumab and commercially provided ranibizumab and aflibercept. In some countries, this comparison is moot because intraocular injection of compounded bevacizumab is prohibited by regulatory authorities.

The costs associated with intravitreal pharmacotherapy can be considerable and they can impact private insurance companies and national health systems. This chapter examines the costs of therapy and the subsequent effects on insurance and healthcare policy.

10.2 National Costs

In the United States, diabetes accounts for 10% of healthcare spending and is the leading cause of new blindness in working aged adults [5]. Approximately 75,000 new cases of DME are diagnosed yearly and medical costs for these patients are 29% higher than for diabetic patients without DME. Inpatient care comprises nearly half of these costs [44].

The pharmaceutical costs associated with the treatment of DME in the United States are considerable. Insurance payments for intravitreal drugs administered to Medicare patients are paid through Part B. In 2010, when intravitreal pharmacotherapy was being used predominantly for the treatment of neovascular age-related degeneration, VEGF inhibitors accounted for \$2 billion or one-sixth of the US Medicare Part B drug budget [49]. By 2013, Part B expenditures for ranibizumab and aflibercept totaled \$2.5 billion [50]; during calendar year 2014, ranibizumab (\$1.3 billion) and aflibercept (\$1.3 billion) were the second and third most expensive drugs (behind rituximab) (Table 10.1). The majority of these payments were for the treatment of neovascular age-related macular degeneration, but an increasing proportion was for DME.

Generic and brand		
names	Major indications	Total spending
Rituximab (Rituxan)	Non-Hodgkin lymphoma, chronic lymphocytic leukemia	\$1.4 Billion
Ranibizumab (Lucentis)	Neovascular AMD, DME, edema due to RVOs	\$1.3 Billion
Aflibercept (Eylea)	Neovascular AMD, DME, edema due to RVOs	\$1.3 Billion
Infliximab (Remicade)	Rheumatoid arthritis, psoriasis, Crohn's disease	\$1.2 Billion
Pegfilgrastim (Neulasta)	Myelosuppressive chemotherapy adjunct, myelosuppression	\$1.2 Billion

Table 10.1 The 2014 Medicare Part B expenditures for the top five drugs

Note that ranibizumab and affibercept rank second and third

AMD age-related macular degeneration, DME diabetic macular edema, RVO retinal vein occlusions

The costs of healthcare in Canada have risen considerably faster over the past few decades than the gross domestic product, and the expenditure on vision care has risen even faster, from 1.8% of total health expenditures in 1975 to 2.2% in 2007 [4]. The annual cost of vision loss in Canada due to age-related macular degeneration (AMD), diabetic retinopathy (DR), cataract, glaucoma, and refractive error was estimated at \$15.80 billion (2007), with \$8.6 billion resulting from direct healthcare expenditures, \$4.4 billion from productivity loss (lower employment, higher absenteeism, and premature death), \$1.8 billion from dead weight losses (welfare payments and lost taxation), \$0.7 billion for care of people with vision loss, \$305 million due to aids and home modifications, and \$11.7 billion due to lost wellbeing. This equates to an annual financial cost of \$19,370 per affected person, or \$33,704 if the value of well-being is included [7]. The prevalence of vision loss is expected to increase from 2.5% of the population in 2007 to 4.0% in 2032.

The number of people over the age of 11 years with diabetes mellitus in England was estimated at 2,334,550 in 2010 with 166,325 (7.12%) believed to have DME. The total cost of health and social care attributed to these patients was $\pounds 116,296,038$ [28].

The cost to the German statutory health insurance system of caring for the ocular complications of diabetes was about $\varepsilon 2.23$ billion in 2002. German patients with DME use almost twice the medical resources as those with only mild retinopathy [15].

10.3 Cost-Effectiveness of Treating Diabetic Macular Edema

The concept of "cost-effectiveness" only has meaning when a particular intervention is compared to a standard [24]. Policy makers in the United States have generally not established formal standards for cost-effectiveness though they probably exist in an unpublished form [34]. Some other nations base standards of effectiveness on the quality-adjusted life year (QALY), a measure of health-related quality of life in which every health state is assigned a value between 1 (perfect health) and 0 (death) [12]. For ophthalmic conditions, most models are based on health states determined by visual acuities. A cost of \$50,000/QALY is generally accepted as a willing-to-pay cost-effectiveness standard [17, 53], but some authors have suggested an upper limit of \$100,000/QALY [25] (Fig. 10.1). The United Kingdom's National Institute for Health and Care Excellence (NICE) generally approves interventions with cost-utility ratios of less than £20,000/QALY (approximately \$32,000/ QALY) but usually requires strong reasons to approve interventions costing more than \$48,000/QALY. Some researchers have suggested that the United States should be willing to spend \$100,000/QALY or more [52]. The World Health Organization recommends that interventions that cost less than the per capita gross domestic product per individual of a country per disability-adjusted life year are very costeffective [51]. An intervention that costs less than 3 times the gross domestic product per disability-adjusted life year is considered cost-effective. According to these standards, ranibizumab remains cost-effective in the United States if 3.5 or fewer



Fig. 10.1 The recommendations of cost-effectiveness standards made by several groups and government agencies. Some analysts have proposed that standards of over \$100,000/QALY are still cost-effective. *QALY* quality adjusted life year, *NICE* United Kingdom's National Institute for Health and Care Excellence, *GDP* gross domestic product

injections are required each year from years 3 through 14. Data from The Diabetic Retinopathy Clinical Research network Protocol I suggests that fewer injections are likely to be given after the first year of DME treatment, thereby suggesting that ranibizumab is cost-effective over extended periods of time [11]. Though cost-effective analyses are common in US medical literature, legislation in the United States has banned the use of QALY for making insurance coverage decisions [35].

Laser photocoagulation and vitrectomy surgery are usually considered durable therapies for DR. Using data from the Early Treatment Diabetic Retinopathy Study [13], laser photocoagulation was estimated to produce a gain of 0.236 QALY and was considered to be highly cost-effective compared to no treatment [43]. Though repeat sessions of laser and return visits to the operating room are not uncommon for patients with advanced diabetic retinopathy, the total number of these interventions is generally far fewer than the number of required intravitreal anti-VEGF injections. Randomized, prospective trials suggest that patients require at least 15 anti-VEGF injections over the course of 3 years, and RISE and RIDE featured monthly injections for 36 months. In most vitreoretinal practices, physicians individualize therapy with pro re nata (PRN) or treat-and-extend (T&E) regimens. Compared to monthly injections, both strategies decrease the number of injections and T&E also decreases the number of clinic visits. Both regimens decrease the total direct costs for treating DME.

Treatment costs vary considerably depending upon the choice of drugs [42]. In the United States, compounded bevacizumab costs approximately \$60 per dose, ranibizumab (0.3 mg) costs \$1170 per dose, and affibercept costs \$1850 per dose [49, 50]. Triamcinolone (Kenalog®) costs only \$49 per dose, the dexamethasone

insert costs \$1371 per dose, and the fluocinolone insert costs \$8800 per dose. The total lifetime cost for one patient treated with anti-VEGF therapy has been estimated at \$133,126. The cost to achieve one line of visual acuity improvement per year ranges from \$60 to \$561 [46]. When analyzed according to the incremental cost-effective ratio (ICER) per cost of QALY gained, each of the anti-VEGF drugs decreases costs compared to no treatment [18].

No trials have directly compared all of the available treatment options and regimens, and few have detailed the costs of therapy. The cost-effectiveness of therapy varies according to the authors – their assumptions and biases. The United Kingdom's NICE evaluated an industry-sponsored study of ranibizumab for the treatment of DME and concluded that the cost-effectiveness of anti-VEGF therapy is not convincingly superior to laser photocoagulation [32]; two other studies, however, reached the opposite conclusion though they did not compare all treatment options or calculate lifetime costs [10, 29]. In the final step of NICE's assessment of the cost-effectiveness of ranibizumab for DME, NICE used a preplanned subgroup threshold analysis to find that the ICER for ranibizumab treatment fell below the willing-to-pay threshold only for eyes with central retinal thickness greater than 400 μ m at baseline. By approving ranibizumab use only for this subgroup, NICE effectively halved the potential payment budget for the treatment of DME [33].

Data from the RESTORE trial were used to create one of the first cost-effective analyses for the treatment of DME [29]. Using a Markov model, the authors determined that ranibizumab monotherapy resulted in a 0.17 QALY gain at an incremental cost of £4191 relative to laser monotherapy; this yields an ICER of £24,028. A probabilistic analysis reported a 64% probability that ranibizumab was cost-effective at a threshold of £30,000 per QALY. Combined ranibizumab and macular laser photocoagulation results in a 0.13 QALY gain over laser monotherapy at an incremental cost of £4695 with an ICER of £36,106. They concluded that ranibizumab monotherapy is cost-effective, but the cost-effectiveness of ranibizumab combined with laser photocoagulation is less certain because of higher costs and lower effectiveness. If a 40-year time line were used with an average patient age of 47 years – as in the study by Sharma – then the RESTORE model would predict a 0.26 QALY and an ICER of £10,412 for ranibizumab compared to laser.

A Markov model was used to determine the cost-effectiveness of different treatments (laser photocoagulation, intravitreal injections of triamcinolone acetonide, intravitreal injections of anti-VEGF drugs, or a combination of both) for DME [36]. The model determined that all treatments except laser monotherapy substantially reduced costs, and all treatments except triamcinolone increased quality-adjusted life years (QALYs). Combined laser photocoagulation and anti-VEGF therapy produced the greatest benefit by gaining 0.56 QALYs at a cost of \$6975 for an incremental cost-effectiveness ratio per QALY compared to laser plus triamcinolone. Anti-VEGF monotherapy was similarly cost-effective to laser and anti-VEGF combination therapy. The authors concluded that anti-VEGF monotherapy or in combination with laser photocoagulation was the most effective treatment strategy for DME, and both compare favorably with cost-effective interventions for other conditions. According to the authors, the decision to add laser photocoagulation to anti-VEGF therapy should probably be made according to each patient's preference. In most cases, adding laser to an anti-VEGF regimen does not incrementally improve VA beyond that achievable with anti-VEGF monotherapy. Laser minimally extends the durability of anti-VEGF therapy but introduces additional risk of permanent macular complications. Because of its low unit dose cost, the cost-effectiveness of bevacizumab exceeds that of ranibizumab. Absolute visual acuity gains with ranibizumab, however, are generally greater in eyes with moderate to severe vision loss, and this needs to be considered when choosing therapy.

A computer simulation cost-effectiveness analysis was based on 2-year data from DRCR.net Protocol I [10]. The authors calculated several ICER values in terms of dollars per letter: \$393 (sham + laser vs. triamcinolone + laser); \$5943 (ranibizumab + prompt laser vs. sham + laser); and \$20 (ranibizumab + prompt laser vs. ranibizumab + deferred laser). In pseudophakic patients, the ICER value for ranibizumab + deferred laser (compared to triamcinolone + laser) was \$14,690. In phakic patients, ranibizumab + deferred laser produced an additional 6 letters of visual acuity at a cost of \$19,216 compared to triamcinolone + laser. The authors concluded that intravitreal triamcinolone acetonide appears to be the most cost-effective treatment for DME in pseudophakic eyes.

Using results from several of the earlier clinical trials, a cost-benefit analysis compared several treatments [46]. Not surprisingly, the use of lower cost drugs such as bevacizumab and triamcinolone proved to be most cost-effective [45].

A 14-year cost-utility analysis for the treatment of DME with ranibizumab was performed with data from the RISE and RIDE trials [2]. This analysis not only assessed the costs of ranibizumab therapy, but it contrasted these to savings gained from lowered vision-related healthcare costs. The 14-year gain attributed to ranibizumab therapy was 0.9981 QALYs, which produced an 11.6% improvement in quality of life. The direct ophthalmologic cost to treat one eye was \$30,116 and for both eves was \$56,336. The decrease in direct nonophthalmologic medical costs from lower incidences of depression, injury, skilled nursing home admissions, and other costs associated with vision were \$51,758. This resulted in a net medical cost of \$4578. The mean cost to society for bilateral ranibizumab therapy was -\$30,807, consisting of \$31,406 in savings due to decreased caregiver costs and \$3978 in decreased wage losses. The third-party insurer's cost-utility ratio was \$4587/QALY, and the societal cost perspective for bilateral therapy was -\$30,807/QALY. This demonstrated that ranibizumab therapy not only produced a positive QALY of 0.9981 but also a positive financial gain of \$30,807. The authors stated that ranibizumab produces a much greater gain than the 6–9% attributed to β-blocker treatment of systemic arterial hypertension, the 4-6% gain attributed to statin treatment of hyperlipidemia [1], and the 1% gain attributed to the use of bisphosphonates for osteoporosis [3].

A Markov model using RESTORE data was developed to determine the costeffectiveness of DME treatment with ranibizumab monotherapy, ranibizumab in combination with laser photocoagulation, and laser monotherapy for the province of Quebec [14]. The ICERs for ranibizumab monotherapy and combination therapy compared to laser monotherapy were CA\$24,494 and CA\$36,414, respectively. The yearly incremental costs for ranibizumab monotherapy and combination therapy without legal blindness were CA\$15,822 and CA\$20,616. The authors concluded that both ranibizumab treatment regimens were cost-effective and both increased the time without blindness.

The cost-effectiveness of ranibizumab + laser was compared to laser monotherapy with 5-, 7-, and 10-year time horizons in Mexico [41]. The ICERs for combination therapy vs. laser were \$5019, \$2375, and \$622 at each time point, respectively. The authors concluded that ranibizumab was a cost-effective option at 98.7% vs. monotherapy before reaching the GDP per capita.

Using a US healthcare model with a 25-year horizon, ranibizumab (compared to laser) was found to have an ICER of \$89,903 per QALY gained (2013 US dollars) [47]. By comparison, bevacizumab was estimated to cost \$11,138/QALY. This suggests that ranibizumab would be cost-effective at a willing-to-pay threshold of \$100,000/QALY (but not at \$50,000/QALY) if patients received fewer than 0.45 injections annually after year 2, whereas bevacizumab is cost-effective at nearly all willing-to-pay levels.

A Markov model was used to evaluate the cost-effectiveness of the fluocinolone acetonide insert with a 15-year time horizon [31]. The ICER was \$38,763, assuming that 40% of patients are treated unilaterally. The insert is cost saving when 100% of patients receive unilateral treatment.

10.4 Comparative Cost-Effectiveness Models

Several studies have compared the cost-effectiveness of the anti-VEGF drugs in different countries, but these have been plagued by competing interests, since many studies were written by employees of the drug manufacturing companies. Markov models are complex, and selecting the necessary inputs requires considerable expertise by the authors. Most of these studies have been published in peer-reviewed journals, but it is not clear that reviewers, many of whom were probably physicians, had the necessary expertise to adequately evaluate all inputs and assumptions [37]. The reader, therefore, is advised to closely scrutinize the methodology used in each of these studies to determine if the assumptions and comparisons made by the authors are appropriate.

A network meta-analysis was used by many of the national studies to determine the efficacies of ranibizumab, aflibercept, and laser photocoagulation [39]. Data were used from the RESTORE (PRN regimen) [30] and RETAIN (treat and extend regimen) [38] trials for ranibizumab and the VIVID and VISTA (monthly ×5 followed by bimonthly) [22] for aflibercept. Importantly, PRN and T&E regimens, which at that time had not been published, were not modeled for aflibercept.

Using a Markov model previously reviewed by NICE and based on data from the RESTORE study, the costs of treating DME in the United Kingdom were estimated [40]. A 3-year treatment period with a lifetime horizon was assumed. The lifetime costs for treating patients with DME were £20,019 (PRN), £22,930 (T&E), and £25,859 (aflibercept q8weeks). The authors claimed that ranibizumab produced an

incremental gain of 0.05 QALYs over affibercept and produced cost savings of £5841 (PRN) and £2930 (treat-and-extend) compared to affibercept.

A Markov model was used to compare the cost-effectiveness of PRN and T&E ranibizumab against bimonthly affibercept (after 5 monthly injections) in Greece [23]. The authors concluded that ranibizumab offered cost savings of ϵ 2,824 and ϵ 22 for each regimen over affibercept and produced greater gains in QALYs (+0.05) and more time without visual impairment. The net monetary benefits were ϵ 3,984 and ϵ 1,278. They stated that ranibizumab should be the dominant treatment option for the treatment of DME in Greece.

A Markov model was used to compare the cost-effectiveness of intravitreal aflibercept vs. ranibizumab in Turkey [9]. Results from VIVID and VISTA and RESTORE and REVEAL were used to construct the model. The total annual costs of aflibercept and ranibizumab therapy were 15,315 and 14,791 Turkish Lira, respectively, and QALYs were 7343 and 7295. Aflibercept was deemed to be cost-effective compared to ranibizumab at a cost-effectiveness threshold of 26,415 TL. The expected number of years with one blind eye was 0.416 with aflibercept and 0.647 with ranibizumab. The authors concluded that aflibercept is a cost-effective option for the treatment of DME in Turkey compared to ranibizumab.

A Markov model was used to compare the cost-effectiveness of ranibizumab and aflibercept in the Czech Republic [21]. The authors concluded that there is a 62% probability that ranibizumab brings more quality than aflibercept at a lower cost.

The DRCR.net performed a cost-effectiveness analysis based on its data from the Protocol T trial [54]. Whereas Protocol T found that for patients with VA of 20/40 or better, bevacizumab, ranibizumab, and aflibercept were equally effective, for patients with VA of 20/50 or worse, affibercept produced better VA gains at 1 year, though the statistical significance compared to ranibizumab had vanished at 2 years. The network determined that during year 1 the ICERs of aflibercept and ranibizumab compared with bevacizumab were \$1,110,000 and \$1,730,000 per OALY, respectively. Extrapolating to 10 years, they were \$349,000 and \$603,000 per QALY, respectively. Aflibercept's ICER compared to ranibizumab was \$648,000 per QALY at 1 year and \$203,000 per QALY at 10 years. The 10-year ICERs of aflibercept and ranibizumab compared to bevacizumab at 10 years were \$287,000 and \$817,000, respectively. These numbers indicate that though affibercept and ranibizumab are more effective than bevacizumab for the treatment of DME, their relative costs exceed the commonly accepted \$100,000/OALY threshold. Interestingly, even if single doses of bevacizumab were drawn from each 4 ml vial and the rest of the vial was discarded, bevacizumab is still more cost-effective than aflibercept and ranibizumab.

The cost-effectiveness of ranibizumab for the treatment of PDR was recently performed based on the DRCR.net Protocol S data [27]. The authors found that the 2-year cost was \$13,053 with a cost per QALY of \$7988 when pan-retinal photoco-agulation was the primary treatment, compared to a 2-year cost of \$30,328 and a cost per QALY of \$19,150 when ranibizumab was the primary treatment. When extrapolated to lifetime therapy, ranibizumab had a cost per QALY of \$138,852 to \$164,360.

10.5 Future Considerations

The treatment of DME has been shown repeatedly to be cost-effective, but the high aggregate costs due to the large and growing number of affected individuals have already influenced healthcare policy. The treatment landscape is complicated because of bevacizumab, since some countries forbid its use because of its off-label status despite its demonstrably low cost and high cost-effectiveness. Safety issues pertaining to compounding-related sterility and the penetration of counterfeit drugs into the market has created a patchwork of drug availability throughout the world. Physicians in some countries are able to safely use bevacizumab as first-line therapy, whereas bevacizumab cannot be used in other countries, and approved anti-VEGF therapy is limited by inequalities in cost and income.

Safe, universally available, cost-effective therapy is needed throughout the world, but easy solutions are not available. Pars plana vitrectomy has emerged as a low-cost alternative to pharmacotherapy, but despite 25 years of experience, effectiveness has not been consistently demonstrated, and level I evidence supporting its efficacy is lacking.

Drugs are being developed to target new pathogenic pathways, but there is no evidence that these will be less expensive than approved anti-VEGF drugs. Extended duration therapy with longer acting drugs or sustained release devices, if they prove to be effective, will likely decrease the number of clinic visits but these therapies will likely be generally priced higher because of their durations of action.

References

- Brown GC, Brown MM, Kertes P. Value-based medicine, cost-utility analysis. The value of commonly-used pharmaceuticals. Evidence Based Ophthalmol. 2009;10:107–22.
- Brown GC, Brown MM, Turpcu A, Rajput Y. The cost-effectiveness for the treatment of diabetic macular edema. Ophthalmology. 2015;112:1416–25.
- 3. Bureau of Labor Statistics. Employment situation summary table A. Household data seasonally adjusted, August 2012. Available at: http://www.bls.gov/news.release/empsit.a.htm. Accessed 22 Sept 22, 2012.
- Canadian Institute for Health Information. National Health Expenditures data base. Available at http://secure.cihi.ca/cihiweb/downloads/Casemix_ICDimpact_AppA_CMG_02_03.xls. Accessed 2008.
- Centers for Disease Control and Prevention. National diabetes fact sheet: National estimates and general information on diabetes and prediabetes in the United States, 2011. Atlanta: Centers for Disease Control and Prevention; 2011. Accessed at www.cdc.gov/diabetes/pubs/pdf/ndfs_2011. pdf on 8 Dec 2011.
- Chen E, Looman M, Laouri M, Gallagher M, Van Nuys K, Lakdawalla D, et al. Burden of illness of diabetic macular edema: literature review. Curr Med Res Opin. 2010;26:1587–97.
- Cruess AF, Gordon KD, Bellan L, Mitchell S, Pezzullo ML. The cost of vision loss in Canada.
 Results. Can J Ophthalmol. 2011;46(4):315–8.
- Davidov E, Breitscheidel L, Clouth J, Reips M, Happich M. Diabetic retinopathy and healthrelated quality of life. Graefes Arch Clin Exp Ophthalmol. 2009;247:267–72.

- Deger C, Ozdemir O, Eldem B, Unlu N, Alp MN, Saatci AO, Ozmert E, Altintas AK, Sermet F, Erdal E, Sar C, Asan S, Sumer F, Parali E, Ozel O. The cost-effectiveness (CE) of intravitreal aflibercept (IVT-AFL) in the treatment of diabetic macular edema (DME) in Turkey. Value Health. 2015;18:A606.
- Dewan V, Lambert D, Edler J, Kymes S, Apte RS. Cost-effectiveness analysis of ranibizumab plus prompt or deferred laser or triamcinolone plus prompt laser for diabetic macular edema. Ophthalmology. 2012;119:1679–84.
- 11. Diabetic Retinopathy Clinical Research Network, Elman MJ, Aiello LP, Beck RW, et al. Randomized trial evaluating ranibizumab plus prompt or deferred laser or triamcinolone plus prompt laser for diabetic macular edema. Ophthalmology. 2010;117:1064–77.
- 12. Drummond MF, Sculpher MJ, O'Brien BJ, Stoddart GL. Methods for the Economic Evaluation of Health Care Programmes. 3rd ed. Oxford: Oxford University Press; 2005.
- Early Treatment Diabetic Retinopathy Study Research Group. Photocoagulation for diabetic macular edema. Early Treatment Diabetic Retinopathy Study report number 1. Arch Ophthalmol. 1985;103:1796–806.
- 14. Haig J, Barbeau M, Ferreira A. Cost-effectiveness of ranibizumab in the treatment of visual impairment due to diabetic macular edema. J Med Econ. 2016;19(7):663–71.
- Happich M, Reitberger U, Breitscheidel L, Ulbig M, Watkins J. The economic burden of diabetic retinopathy in Germany in 2002. Graefes Arch Clin Exp Ophthalmol. 2008;246:151–9.
- Hariprasad SM, Mieler WF, Grassi M, Green JL, Jager RD, Miller L. Vision related quality of life in patients with diabetic macular oedema. Br J Ophthalmol. 2008;92:89–92.
- 17. Heudebert GR, Centor RM, Klapow JC, et al. What is heartburn worth? A cost-utility analysis of management strategies. J Gen Intern Med. 2000;15:175–82.
- Hodgson N, Wu F, Zhu J, Wang W, Ferreyra H, Zhang K, Wang J. Economic and quality of life benefits of anti-VEGF therapy. Mol Pharmaceutics. 2016;13:2877–80.
- 19. Huang ES, Brown SE, Ewigman BG, Foley EC, Meltzer DO. Patient perceptions of quality of life with diabetes-related complications and treatments. Diabetes Care. 2007;30:2478–83.
- Javitt JC, Aiello LP, Bassi LJ, Chiang YP, Canner JK. Detecting and treating retinopathy in patients with type I diabetes mellitus. Savings associated with improved implementation of current guidelines. American Academy of Ophthalmology. Ophthalmology. 1991;98: 1565–73.
- 21. Klimes J, Regnier SA, Mahon R, Budek T, Dostal F, Skalicky D, Depta J. Cost effectiveness analysis of ranibizumab compared to aflibercept and laser intervention in treatment of diabetic macular edema (DME) in the Czech Republic. Value Health. 2015;18:A419.
- 22. Korobelnik JF, Do DV, Schmidt-Erfurth U, Boyer DS, Holz FG, Heier JS, Midena E, Kaiser PK, Terasaki H, Marcus DM, Nguyen QD, Jaffe GJ, Slakter JS, Simader C, Soo Y, Schmelter T, Yancopoulos GD, Stahl N, Vitti R, Berliner AJ, Zeitz O, Metzig C, Brown DM. Intravitreal aflibercept for diabetic macular edema. Ophthalmology. 2014;121:2247–54.
- 23. Kourlaba G, Relakis J, Mahon R, Kalogeropoulou M, Pantelopoulou G, Kousidou O, Maniadakis N. Cost-utility of ranibizumab versus aflibercept for treating Greek patients with visual impairment due to diabetic macular edema. Cost Eff Resour Alloc. 2016;14:7.
- 24. Kymes SM. An introduction to decision analysis in the economic evaluation of the prevention and treatment of vision-related diseases. Ophthalmic Epidemiol. 2008;15:76–83.
- Laupacis A, Feeny D, Detsky AS, Tugwell PX. How attractive does a new technology have to be to warrant adoption and utilization? Tentative guidelines for using clinical and economic evaluations. CMAJ. 1992;146:473–81.
- 26. Lee LJ, Yu AP, Cahill KE, Oglesby AK, Tang J, Qiu Y, et al. Direct and indirect costs among employees with diabetic retinopathy in the United States. Curr Med Res Opin. 2008;24:1549–59.
- 27. Lin J, Chang JS, Smiddy WE. Cost evaluation of panretinal photocoagulation versus intravitreal ranibizumab for proliferative diabetic retinopathy. Ophthalmology. 2016;123:1912–8.
- Minassian DC, Owens DR, Reidy A. Prevalence of diabetic macular oedema and related health and social care resource use in England. Br J Ophthalmol. 2012;96:345–9.

- 29. Mitchell P, Annemans L, Gallagher M, Hasan R, Thomas S, Gairy K, et al. Cost-effectiveness of ranibizumab in treatment of diabetic macular oedema (DME) causing visual impairment: evidence from the RESTORE trial. Br J Ophthalmol. 2012;96:688–93.
- 30. Mitchell P, Bandello F, Schmidt-Erfurth U, Lang GE, Massin P, Schlingemann RO, Sutter F, Simader C, Burian G, Gerstner O, Weichselberger A, RESTORE Study Group. The RESTORE study: ranibizumab monotherapy or combined with laser versus laser monotherapy for diabetic macular edema. Ophthalmology. 2011;118:615–25.
- Moore PT, Kendall R, Dip PG, Zachary C, Cutino A, Green K. Economic evaluation of a fluocinolone acetonide intravitreal implant for patients with DME based on the FAME study. Am J Manag Care. 2015;21:S63–72.
- 32. National Institute for Health and Clinical Excellence. National Institute for Health and Clinical Excellence Final Appraisal Determination: Ranibizumab for the Treatment of Macular Oedema. July 2011. Accessed at www.nice.org.uk/nicemedia/live/13125/55324/55324.pdf on 30 June 2012.
- National Institute for Health and Clinical Excellence. Ranibizumab for the treatment of diabetic macular oedema (rapid review of TA237). http://guidance.nice.org.uk/TA/Wave23/41. Accessed 21 Oct 2012.
- Neumann PJ. Lessons for health technology assessment: it is not only about the evidence. Value Health. 2009;12(suppl):S45–8.
- Neumann PJ, Weinstein MC. Legislating against use of cost-effectiveness information. N Engl J Med. 2010;363:1495–7.
- 36. Pershing S, Enns EA, Matesic B, Owens DK, Goldhaber-Flebert JD. Cost-effectiveness of treatment of diabetic macular edema. Ann Intern Med. 2014;160:18–29.
- Philips Z, Ginnelly L, Sculpher M, et al. Review of guidelines for good practice in decisionanalytic modelling in health technology assessment. Health Technol Assess. 2004;8:1–158.
- 38. Prünte C, Fajnkuchen F, Mahmood S, Ricci F, Hatz K, Studnička J, Bezlyak V, Parikh S, Stubbings WJ, Wenzel A, Figueira J; and the RETAIN Study Group. Ranibizumab 0.5 mg treat-and-extend regimen for diabetic oedema: the RETAIN study. Br J Ophthalmol. 2016;100:787–95.
- Regnier S, Malcolm W, Allen F, Wright J, Bezlyak V. Efficacy of anti-VEGF and laser photocoagulation in the treatment of visual impairment due to diabetic macular edema: a systematic review and network meta-analysis. PLoS One. 2014;9:e102309.
- 40. Regnier SA, Malcolm W, Haig J, Xue W. Cost-effectiveness of ranibizumab versus aflibercept in the treatment of visual impairment due to diabetic macular edema: a UK healthcare perspective. ClinicoEconomics Outcomes Res. 2015;7:235–47.
- Ruiz Miranda CI, Ubiarco LV. Cost-effectiveness of ranibizumab on patients with diffuse diabetic macular edema within the public Mexican health care system. Value Health. 2015;17:A607.
- 42. Schauwvlieghe AME, Dijkman G, Hooymans JM, et al. Comparing the effectiveness and costs of bevacizumab to ranibizumab in patients with diabetic macular edema: a randomized clinical trial (the BRDME study). BMC Ophthalmol. 2015;15:71.
- 43. Sharma S, Brown GC, Brown MM, et al. The cost-effectiveness of grid laser photocoagulation for the treatment of diabetic macular edema: results of a patient based cost-utility analysis. Curr Opin Ophthalmol. 2000;11:175–9.
- 44. Shea AM, Curtis LH, Hammill BG, Kowalski JW, Ravelo A, Lee PP, et al. Resource use and costs associated with diabetic macular edema in elderly persons. Arch Ophthalmol. 2008;126:1748–54.
- 45. Smiddy WE. Economic considerations of macular edema therapies. Ophthalmology. 2011;118(9):1827–33.
- Smiddy WE. Clinical applications of cost analysis of diabetic macular edema treatments. Ophthalmology. 2012;119:2558–62.
- 47. Stein JD, Newman-Casey PA, Kim DD, Nwanyanwu KH, Johnson MW, Hutton DW. Costeffectiveness of various interventions for newly diagnosed diabetic macular edema. Ophthalmology. 2013;120:1835–42.

- 48. Tranos PG, Topouzis F, Stangos NT, Dimitrakos S, Economidis P, Harris M, et al. Effect of laser photocoagulation treatment for diabetic macular oedema on patient's vision-related quality of life. Curr Eye Res. 2004;29:41–9.
- 49. US Department of Health and Human Services, Office of Inspector General. Medicare payments for drugs used to treat wet age-related macular degeneration. Washington, DC: US Dept. of Health and Human Services; 2012. Publication OEI-03-10-00360.
- 50. US Government Accountability Office. Medicare Part B: expenditures for new drugs concentrated among a few drugs, and most were costly for beneficiaries. Washington, DC: US Government Accountability Office; 2015:QAO-16-12.
- 51. The World Health Organization. The World Health Report 2002 reducing risks, promoting healthy life. Geneva: The World Health Organization; 2002. p. 47e96. Available at: http:// www.who.int/whr/2002/en/index.html. Accessed 18 Mar 2015.
- 52. Ubel PA, Hirth RA, Chernew ME, Fendrick AM. What is the price of life and why doesn't it increase at the rate of inflation? Arch Intern Med. 2003;163:1637–41.
- Wailoo A, Roberts J, Brazier J, McCabe C. Efficiency, equity, and NICE clinical guidelines. BMJ. 2004;328:536–7.
- Wells JA, Glassman AR, Ayala AR, et al.; Diabetic Retinopathy Clinical Research Network. Aflibercept, bevacizumab, or ranibizumab for diabetic macular edema. N Engl J Med. 2015;372(13):1193–203.